TABLE OF CONTENTS

SUMMARY	i
ACKNOWLEDGEMENTS	iii
TABLE OF CONTENTS	v
LIST OF DIAGRAMS	х
LIST OF TABLES	xi
LIST OF ANNEXURES	xii
DECLARATION OF OWN WORK	xiii

CHAPTER 1: GENERAL INTRODUCTION

1.1	1 BACKGROUND TO THE STUDY						
1.2	RATIC	RATIONALE FOR THE STUDY 10					
1.3	STATE	STATEMENT OF THE RESEARCH PROBLEM 14					
1.4	RESE	ARCH OBJECTIVES	15				
1.5	RESE	RESEARCH QUESTIONS 15					
1.6	RESEARCH DESIGN AND METHODOLOGY		16				
	1.6.1	Research purpose	16				
	1.6.2	The approach	17				
	1.6.3	Unit of analysis	17				
	1.6.4	Data sources (Units of observation)	18				
	1.6.5	Data collection	18				
		1.6.5.1 Review of scholarly literature	18				
		1.6.5.2 Reading of official documents	19				
		1.6.5.3 Semi-structured questionnaire	19				
		1.6.5.4 Focus group discussion	20				
		1.6.5.5 Interviews	20				
		1.6.5.6 Sampling techniques	21				
1.7	ETHIC	AL CONSIDERATIONS	22				
1.8	STUD	Y TITLE AND RELATED CONCEPTS CLARIFIED	22				
	1.8.1	Key human factors	22				
	1.8.2	E-readiness	24				
	1.8.3	E-learning	25				
	1.8.4	Information and Communication Technology	26				
	1.8.5	Academic employee	27				

	1.8.6	Online learning facilitator	28
	1.8.7	Higher education institutions (HEIs)	29
	1.8.8	Student vs learner	30
	1.8.9	Assessment	31
	1.8.10	Human resource performance appraisal	31
	1.8.11	Human resource management	31
1.9	LAYOL	JT OF CHAPTERS	32

CHAPTER 2: HUMAN RESOURCE PERFORMANCE APPRAISAL AND EMPLOYEE MOTIVATION

2.1	INTRC	DUCTION	35
2.2	HUMAN RESOURCE PERFORMANCE APPRAISAL FOR		
	E-REA	DINESS ASSESSMENT	36
	2.2.1 Human Resource Performance Management		36
	2.2.2	Human Resource Performance Appraisal	44
2.3	.3 THE ROLE OF EMPLOYEE MOTIVATION TOWARDS INCREASED PERFORMANCE		
			48
	2.3.1	Financial and non-financial incentives and rewards	49
	2.3.2	Goal-setting	51
	2.3.3	Intrinsic motivation and extrinsic motivation	56
	2.3.4	Self-determination theory	57
	2.3.5	Motivation and technology adoption	61
2.4	CONC	LUSION	66

CHAPTER 3: EMPLOYEE TRAINING AND DEVELOPMENT

3.1	INTRODUCTION		
3.2	2 EMPLOYEE TRAINING AND DEVELOPMENT		69
3.2.1 Training and development		Training and development	70
	3.2.2	Employee development and employee performance	74
	3.2.3	The need for employee development for e-learning	76
	3.2.4	The constructivist epistemology	81
3.3	CARE	ER MANAGEMENT	84
	3.3.1	The role of career management in employee development and	
		enhanced performance	85

	3.3.2	Career development	88
3.4	THE A	CADEMIC AS ADULT LEARNER	93
3.5	THE L	INE MANAGER'S ROLE IN EMPLOYEE DEVELOPMENT	96
3.6	CONC	LUSION	97
CH/	APTER 4	4: KEY HUMAN FACTORS IN E-READINESS	
4.1	INTRO	DUCTION	100
4.2	KEY H	UMAN FACTORS	100
	4.2.1	Pace and style of technology adoption	103
	4.2.2	Learning styles	110

	4.2.3	Personal work profile patterns	115
4.3	CONC	LUSION	118

CHAPTER 5: E-READINESS: CRITICAL REQUIREMENT FOR EFFECTIVE E-LEARNING IN THE HIGHER EDUCATION ENVIRONMENT

5.1	INTRC	INTRODUCTION 1	
5.2 THE HIGHE		IGHER EDUCATION ENVIRONMENT OF THE 21 ST CENTURY	120
	5.2.1	The types of learners and their expectations	121
		5.2.1.1 Generation Y learners	122
		5.2.1.2 Underprepared learners	124
	5.2.2	Changing demographics of learners	127
	5.2.3	Lifelong learning	128
	5.2.4	Increased competition among HEIs	129
5.3	E-REA	DINESS OF ONLINE LEARNING FACILITATORS	130
	5.3.1	Role and responsibilities of an online facilitator	131
	5.3.2	E-readiness indicators	132
	5.3.3	Resistance to e-learning	142
5.4	CONC	LUSION	148

CHAPTER 6: PROFILING THE NORTH-WEST UNIVERSITY

INTRC	DUCTION	151
HUMA	N RESOURCE PERFORMANCE APPRAISAL AT THE NWU	151
EMPL	OYEE DEVELOPMENT AT THE NWU	155
6.3.1	Employee training and development programmes at the NWU	156
	INTRO HUMA EMPLO 6.3.1	INTRODUCTION HUMAN RESOURCE PERFORMANCE APPRAISAL AT THE NWU EMPLOYEE DEVELOPMENT AT THE NWU 6.3.1 Employee training and development programmes at the NWU

	6.3.2	Career management programmes at the NWU	162
6.4	6.4 THE NWU LEARNER		166
	6.4.1	Generation Y learners	166
	6.4.2 Task group and pilot project: technology in teaching and learning		169
	6.4.3	Underprepared learners	170
6.5	CONC	LUSION	173

CHAPTER 7: EMPIRICAL RESEARCH: METHODOLOGY AND FINDINGS

7.1	INTRC	INTRODUCTION 17				
7.2	CHALL	CHALLENGES EXPERIENCED DURING DATA COLLECTION 1				
7.3	RESEARCH DESIGN					
	7.3.1	Sampling	181			
	7.3.2	Validity	183			
	7.3.3	7.3.3 Reliability				
	7.3.4	7.3.4 Review of scholarly literature				
	7.3.5	Empirical study	185			
		7.3.5.1 Quantitative data collection	186			
		7.3.5.1.1 Quantitative data collection by means of a questionnaire	187			
		7.3.5.2 Qualitative data collection	189			
7.3.5.2.1 Focus group discussion						
7.3.5.2.2 Individual interviews						
7.4	EMPIRICAL RESEARCH FINDINGS		192			
	7.4.1	7.4.1 Results obtained through the self-administered questionnaire 1				
		7.4.1.1 Section A: biographical information 1				
		7.4.1.1.1 Designation1				
	7.4.1.1.2 School/faculty					
		7.4.1.1.3 Years of teaching experience	195			
		7.4.1.1.4 Age	196			
		7.4.1.1.5 Highest qualification	197			
		7.4.1.2 Section B: work style preferences of academics	197			
		7.4.1.2.1 Style and pace of technology adoption	198			
		7.4.1.2.2 Preferred learning style	200			
		7.4.1.2.3 Personal work profile patterns	202			
		7.4.1.3 Section C: academic employees' motivation in the				

		workplace	206
	7.4.1.3.1	New challenging goals	206
	7.4.1.3.2	Goal preference	207
	7.4.1.3.3	Motivators in the workplace	210
	7.4.1.4	Section D: the use of and perceptions of an e-learning	
		platform	213
	7.4.1.4.1	The use of eFundi	213
	7.4.1.4.2	Use of e-learning platform tools	215
	7.4.1.4.3	Level of comfortableness with e-learning	216
	7.4.1.4.4	Reasons for not using e-learning	217
	7.4.1.4.5	The use of cutting edge technology	218
	7.4.1.4.6	The use of social networks	220
	7.4.1.5	Section E: training and development for e-readiness	221
	7.4.1.5.1	The need for training	222
	7.4.1.5.2	Perceptions when first taught	223
	7.4.1.5.3	Skills included in e-learning training	225
	7.4.1.6	Open-ended questions	227
	7.4.1.6.1	The use of and perceptions pertaining to an	
		e-learning platform	227
	7.4.1.6.2	Training and development for e-readiness	240
7.4.2	Focus gr	oup	261
	7.4.2.1	Question 1: How do you perceive the current role of	
		e-learning in the higher education environment?	262
	7.4.2.1.1	Summary of feedback on question 1	262
	7.4.2.1.2	Researcher's interpretation on question 1	265
	7.4.2.2	Question 2: How do you perceive the future role of	
		e-learning in higher education environment?	270
	7.4.2.2.1	Summary of feedback on question 2	271
	7.4.2.2.2	Researcher's interpretation on question 2	271
	7.4.2.3	Question 3: What should be done to ensure that	
		e-learning is able to fulfil this role?	272
	7.4.2.3.1	Summary of feedback on question 3	272
	7.4.2.3.2	Researcher's interpretation on question 3	273
	7.4.2.4	Question 4: What are your perceptions pertaining to the	

	expectations of students in terms of e-learning?	274
	7.4.2.4.1 Summary of feedback on question 4	275
	7.4.2.4.2 Researcher's interpretation on question 4	275
	7.4.2.5 Question 5: How should the role of academic employees	
	change to accommodate students' expectations?	276
	7.4.2.5.1 Summary of feedback on question 5	276
	7.4.2.5.2 Researcher's interpretation on question 5	277
7.5	SUMMARY AND DEDUCTIONS FROM EMPIRICAL RESULTS	281
7.6	FRAMEWORK FOR THE E-READINESS ASSESSMENT OF ACADEMICS	287
7.7	CONCLUSION	299

CHAPTER 8: REFLECTIONS AND RECOMMENDATIONS

8.1	INTRODUCTION	302
8.2	SUMMARY OF CHAPTERS	302
8.3	PRIMARY FINDINGS OF THE STUDY	307
8.4	CONCLUSION	322
8.5	CONTRIBUTION OF THE STUDY	323
8.6	RECOMMENDATIONS	323
8.7	FURTHER RESEARCH	327

328

BIBLIOGRAPHY

LIST OF DIAGRAMS

Diagram 2.1:	Self-determination theory and the adoption to technology	63
Diagram 4.1:	S-Curve of technology adoption	104
Diagram 4.2:	Adopter categories defined by Rogers	106
Diagram 7.1:	Designation of academics	194
Diagram 7.2:	Faculties of academics	195
Diagram 7.3:	Years of teaching experience	196
Diagram 7.4:	Age of academics	196
Diagram 7.5:	Highest qualifications of academics	197
Diagram 7.6:	Style and pace of technology adoption	199
Diagram 7.7:	Learning styles	201
Diagram 7.8:	Relation between learning styles and technology adoption	202

Diagram 7.9:	Personal work profile patterns	203
Diagram 7.10:	Comparison between pace and style of technology	
	adoption, preferred learning style, and personal work	
	profile patterns	204
Diagram 7.11:	Most likely reaction when new challenging goals are	
	assigned	207
Diagram 7.12:	Goal preference	208
Diagram 7.13:	Motivators (more than one chosen)	210
Diagram 7.14:	Motivators: single most important motivator	211
Diagram 7.15:	The use of eFundi for modules	214
Diagram 7.16:	The use of eFundi tools	215
Diagram 7.17:	Reasons for not using e-learning	218
Diagram 7.18:	The use of cutting edge technology	219
Diagram 7.19:	The use of social networks	220
Diagram 7.20:	Included in e-learning training	222
Diagram 7.21:	Comparison between the DISC profiles, learning styles,	
	pace and style of technology adoption, and	
	initial feelings/perceptions after e-learning training	224
Diagram 7.22:	Included in e-learning training	225
Diagram 7.23:	Future interactive use of eFundi	232
Diagram 7.24:	E-learning training as part of personal career development	
	plan	240
Diagram 7.25:	Assistance expected from School Directors/Deans	243
Diagram 7.26:	E-readiness assessment preference	247
Diagram 7.27	Teaching experience and e-readiness preference	249
Diagram 7.28	Age and e-readiness assessment preference	251

LIST OF TABLES

Table 5.1:	Learner enrolments at public HEIs, by age	124
Table 5.2:	The e-readiness construct	139
Table 6.1:	Undergraduate learner enrolments at the NWU	167
Table 7.1:	Matches between personal profile patterns, learning styles and	
	pace and style of technology adoption	283
Table 7.2:	Matches between personal profile patterns, learning styles, pace	

and style of technology adoption, and e-learning perceptions	
when trained and first being used	284
E-readiness levels	284
Matches between personal profile patterns, learning styles,	
pace and style of technology adoption, e-learning perceptions	
when trained and first being used and type of motivation	285
The technology adoption – personal work profile pattern match	292
The technology adoption – preferred learning style match	293
Personal work profile pattern – preferred learning style match	294
E-readiness construct	317
The technology adoption – personal work profile pattern match	319
The technology adoption – preferred learning style match	320
Personal work profile pattern – preferred learning style match	321
	and style of technology adoption, and e-learning perceptions when trained and first being used E-readiness levels Matches between personal profile patterns, learning styles, pace and style of technology adoption, e-learning perceptions when trained and first being used and type of motivation The technology adoption – personal work profile pattern match The technology adoption – preferred learning style match Personal work profile pattern – preferred learning style match E-readiness construct The technology adoption – personal work profile pattern match The technology adoption – personal work profile pattern match Personal work profile pattern – preferred learning style match Phetechnology adoption – personal work profile pattern match The technology adoption – perferred learning style match

LIST OF ANNEXURES

Annexure A:	Letter of request: School Directors	390
Annexure B:	Letter of request: academic employees	391
Annexure C:	Questionnaire	392
Annexure D:	Focus group discussion consent form	404
Annexure E:	Focus group power point presentation	405
Annexure F:	Focus group discussion questions	411
Annexure G:	Form – Personal development plan for academic staff	412
Annexure H:	Form – Performance agreement form for academic staff	414
Annexure I:	Form – Self-evaluation for academic staff	418
Annexure J:	Form – Feedback for mini-contact session	423
Annexure K:	Guidelines for panel evaluations of contact sessions	425
Annexure L:	Form – Assessment of contact opportunity for the Institutional	
	teaching excellence award (ITEA)	429
Annexure M:	Guidelines for the Institutional teaching excellence award	
	(ITEA) portfolio	432
Annexure N:	Score card for the evaluation of the Institutional teaching	
	excellence award (ITEA) portfolio	435

Student number: 3108-324-2

I declare that **KEY HUMAN FACTORS IN THE E-READINESS OF ACADEMIC EMPLOYEES AT HIGHER EDUCATION INSTITUTIONS** is my own work and that all the sources that I have used or quoted have been indicated and acknowledged by means of complete references.

SIGNATURE (L Vermeulen)

DATE



CHAPTER 1

GENERAL INTRODUCTION

1.1 BACKGROUND TO THE STUDY

Societies globally have adopted new strategies and techniques to keep abreast with technological development and to use such to achieve their competitive advantage (De Moraes, Melo, Oliviera & Cabral 2010:225; Moolman & Blignaut 2008:168–169; Adams 2005:12–13; cf Thompson & Strickland 2003:172). Governments, businesses and educational institutions are inter alia making use of technology to enhance the quality of products and services they offer to clients (Bates 2010; cf Archibugi & Pietrobelli 2003:862; 865-868; IBM 2006:2-5). Policies and strategies are subsequently formulated to regulate the use of technology (UN 2006; UN 2010; Dada 2006:2). South Africa is no exception since various legislation and a number of policies have been formulated and promulgated in recognition of technology as a key enhancer in modernisation and globalisation (DPSA 2001:4; SA 2004:i). Technology is used as tool to enhance service delivery (e-government), increase public participation (e-governance), provide for better accessibility and convenience of services to inhabitants/customers (DPSA 2001:4) and can be applied to enhance the teaching and learning experience of learners and facilitators (e-learning) (SA 2004a:6;8;14). The White Paper on e-Education (2004b:8) emphasises the responsibility of education systems to provide in public expectations pertaining to quality education for economic growth and social development. As in other areas of social and economic development, technology has the possibility to increase the quality of education and training (SA 2004:8; Pandor 2007). The South African government is therefore supporting the use of technology to enhance and support teaching and learning in the 21st century (SA 2004:8; SA Government 2010; USAID 2005:6).

The technological revolution of the 21st century presents higher education institutions (HEIs) with one of the biggest adaptation challenges in its history (Amirault & Visser 2009:62; Smyre 2006; Tadmore 2006:287; Carnesale 2000:3–4;7). Higher education institutions should either adjust to this innovative reality or

they will be in danger of losing their standing as principal educational institutions (Amirault & Visser 2009:62; Bates 2010). Several decades ago HEIs perhaps still had the luxury of examining, debating, and arguing the advantages and disadvantages of advanced technology in teaching and learning, but today, in the 21st century, the inevitable progress of technology into a meaningful, ever-present and sustained reality (Bennet 2002:2; cf Albright & Nworie 2008:15) has largely cancelled out such debate and arguments in support of and not in favour of the notion (Amirault & Visser 2009:66).

The majority of HEIs in Europe, the United Kingdom, Australia and New Zealand realised the need to include technology in teaching and learning to provide for the need of learners globally (Elgort 2005:182; EC 2005:7; Rossiter 2006:iii; Kanuka 2006:1; Kwache 2007:395; Wong & Fitzsimmons 2008:1121; Harasim 2000:42). In many instances the use of e-learning at South African HEIs have also become increasingly important (Le Roux 2009; Njenga & Fourie 2010:199; Mutula 2003:1), not only for creating new teaching and learning opportunities by means of greater knowledge sharing than traditional classroom teaching and learning and control over time, place and pace of study, but also in providing access to resources beyond the borders of those traditionally available in university libraries, interactive learning, simulations, podcasts, and communication tools (JISC 2009:8; Stone 2008:526; SA 2004b:i).

Learners at HEIs are exposed to e-learning through inter alia ICT subjects; the use of a Learning Management System (LMS) such as inter alia *WebCT/Blackboard, Sakai* and *Moodle*; multimedia; or through other methods of digital learning such as mobile learning. For the purpose of this study the e-learning focus will predominantly be on web-based teaching and learning. Typically, with web-based teaching and learning a LMS will be used as platform for learning. The LMS could be an established system such as those mentioned in this paragraph or a newly designed and developed LMS, customised for a particular HEI's needs.

The Potchefstroom campus of the North-West University (NWU), which serves as case study for this study, uses a LMS called *eFundi* as e-platform. *eFundi* is supported by the LMS *Sakai*, but customised to serve the specific needs of the

NWU. Web-based teaching and learning is, however, not restricted to the use of an LMS¹ and other e-learning tools and media (such as podcasts, blogs, wikis, etc) can be used in conjunction with the e-platform and can be uploaded to the platform or accessed by means of hyperlinks from the platform (Sakai 2009).

The NWU also introduced e-learning as teaching and learning delivery mode and it is subsequently specified and supported in various University policies and other official documents:

- Through its *mission statement* the NWU, aims to "Develop, educate and empower through innovative and high quality teaching-learning, produce well-rounded graduates who are able to think laterally and critically in their service to the country and its people" (NWU 2010f), which can be accomplished through the use of *inter alia* e-learning.
- The *Institutional plan, 2010–2012*, of the NWU supports the achievement of the aforementioned mission statement (NWU 2009b:4;6;7–8) as it states that the University aims to be the leading HEI in terms of moving from <u>E</u>-learning to <u>I</u>-learning (*Innovative learning*) as a future prospect (NWU 2009b:4). The NWU thus places a significant emphasis on innovation in teaching and learning, which implies that the NWU should excel at e-learning if it wishes to take the lead in innovation in the higher education environment (NWU 2009b:4; NWU 2010d). The NWU should therefore strengthen the quality of teaching and learning by *inter alia* enhanced e-learning (NWU 2009b:5).
- The NWU *Teaching and learning policy* (2007a:2), states that academic programmes at the University are "delivered by means of a blended mode, which can include a combination of face-to-face contact between lecturer and student, distance learning and/or e-learning".
- The *Teaching and learning framework* (2009a:11), which makes provision for the use of e-learning in academic programmes, e-learning training for academic employees, hereafter referred to as academics, and a helpdesk for e-learning support to both employees and students.

¹ The concept *e-platform* is used in the study when referring to a Learning Management System (LMS).

 The NWU Quality policy (2007b:1) aims to establish conditions for assuring and improving the quality of the NWU's core business, including inter alia teaching, learning.

According to the Joint Information Systems Committee (JISC) (2009:6-8) of the Higher Education Funding Council of England e-learning is no longer a distance learning or remote learning tool, or a specialist area, but it has become part of the broader aim of enhancing teaching and learning through the use of suitable technology (cf Birkbeck 2005:1). As seen by Oliver (2004) in Johannes (2007:87) elearning should be viewed as a mainstream activity and not something separate from or in addition to other modes of learning. The focus of e-learning has thus become part of a broader debate on the enhancement of learning by more effective and comprehensive use of digital technologies (JISC 2009:8) and is therefore not intended to replace the teacher, but to use technology to increase the focus on pedagogic skills (JISC 2009:5). This notion is supported by the White Paper on e-Education (2004b:14), which provides for e-learning to go beyond the mere exchange of information to provide a variety of learning activities that meet educational objectives. Teaching and learning with technology entails more than providing a technological edge to education and should employ ICT skills to access, analyse, evaluate, integrate, and present information; to create knowledge and new information by adapting to and functioning in a knowledge society by using suitable technology and communicate information; by constructing information; and by mastering communication and collaboration skills (SA 2004b:14).

The online learning facilitator, which decides which technologies to use and how it should be used, plays a significant role in this process (JISC 2009:8; Achimugu Oluwagbemi & Oluwaranti 2010:27). Online learning facilitators need to be aware of the various tools and media that are available and how to select and use those that will best serve the learning purpose (JISC 2009:8; Takalani 2008:1–2). E-learning as blended approach to teaching and learning, with the skilful and suitable incorporation of technology with face-to-face classroom teaching (Cardwell & Madigan 2004:26–27; University of Calgary sa:1; Dziuban, Moskal & Hartman 2005:4; cf Graham, 2004:3), is a significant element of 21st century teaching and learning practice (JISC 2009:9). When planning and designing learning, online learning facilitators must

ensure that they make use of the appropriate learning strategies and technology that will have the biggest impact on the way students learn. (JISC 2009:6–8.)

According to the *E-learning facilitation toolkit* of the United Kingdom National College for School Leadership (UKNCSL) online learning facilitators plan, design, evaluate and develop online communities (UKNCSL 2006). Online learning facilitators encourage and enable groups of learners to share learning in a friendly environment (Conceição-Runlee & Daley 2005; cf Sutton 2004). Online learning facilitators should therefore possess the ability to let learners feel comfortable about participating in online discussions and sharing information (UKNCSL 2006; Sutton 2004; Hrastinski 2008). Salmon (2003:4) explains that the primary role of an online learning facilitator is to promote human interaction and communication through knowledge construction and modelling, as well as transferring of skills and knowledge (cf Thanasingam & Soong 2007:1003–1004; AFLF 2003:2). The task of acting as an online learning facilitator therefore not only requires a combination of fresh insights and technical skill, it also calls for understanding and appreciating the management of online learning and group working, thus focusing on the human aspects and variables of learning as well (Salmon 2003:4).

Although students are familiar and comfortable with technology, JISC (2009:7) further explains that they lack the "critical and evaluative skills" required to construe information from online resources. This shortcoming highlights the need to support students in developing digital literacy skills (JISC 2010; cf Eshet-Alkalai 2004); thus further emphasising the importance of the online learning facilitator's skill to the effectiveness of learning (JISC 2009:7; Achimugu et al 2010:27), emphasising the need for e-readiness of academics in order to become skilled online learning facilitators. It is evident that more than technical skill is required of online learning facilitators, which indicates that academics should also be e-ready in terms of how to best use e-learning tools and how to use teaching and learning strategies online to optimise learning.

From the preceding paragraphs it is clear that the use of technology in teaching and learning brought about changed work environments and changed job requirements for academics at HEIs (cf OECD 2004). The changed job requirement compels job

incumbents, in this case academics, to adapt to new job requirements. All job incumbents did not necessarily change with the new job requirements associated with their jobs and therefore did not necessarily embrace the use of technology in teaching and learning for several reasons. Research (Proctor & Doukakis 2003:268; Robbins 2003:559–560) shows that employees usually resist change in the workplace due to various reasons such as inter alia, fear of the unknown, habits (comfort zones), inadequate information and communication, threats to status, fear of failure and lack of perceived benefits. It is, however, important that academics adapt to changing job requirements in order for the university to remain relevant in times of change (Amirault & Visser 2009:62) and to be able to optimally service the 21st century student, predominantly belonging to Generation Y (HEMIS 2010; Halse & Mallinson 2008:1). The changed job requirements therefore call for the e-readiness of academics to successfully take on the use of e-learning as a learning enhancing tool.

From a human resource management point of view it can be argued that it is important to enhance the e-readiness of academics through training and development to ensure employees are well-skilled and capable to fulfil the new job requirements. This view is supported by the NWU *Teaching and learning framework* (NWU 2009a:16), which focuses on teaching and learning development initiatives, including the use of technology in teaching and learning. It is also supported by the Performance management policy (NWU 2005a:2;9), emphasising the need for personal development plans (PDPs). Further, from a line management viewpoint it is necessary to enhance the e-readiness of academics as well-skilled and competent employees are more likely to perform optimally. Lastly, from an institutional management viewpoint it is necessary to enhance the e-readiness of academics to ensure quality service delivery and world class education to students. This view is supported by the mission of the University, stating that the NWU aims to offer "innovative and high quality teaching and learning" (NWU 2010f), as well as the NWU's Institutional plan, 2010-2012, stating that the University wants to continuously enhance the quality of teaching-learning and advance student preparedness to increase access (NWU 2009b:7).

It is thus evident that University policies are in support of providing students with the best possible education and learning experiences. However, an academic that is not e-ready will not be able to use e-learning optimally as learning tool. This lack of e-readiness can most probably be attributed to a number of factors such as challenging technology, a lack of sufficient computer literacy, inadequate technology-related training, an inability to make the paradigm shift from traditional classroom teaching and learning to e-learning, a lack of understanding the use and purpose of e-learning, an increased workload, fear of the unknown and a threat to the comfort zone, to name a few (cf Bozarth 2006:2–4; cf Kottolli 2008:1). It can be assumed that institutional managers would want to effectively manage the changed job requirement to ultimately enhance productivity and organisational performance. It is therefore necessary to explore the barriers to e-readiness with the aim of providing solutions in this regard.

In order to enhance the e-readiness of academics, it is necessary to assess their level of e-readiness against particular criteria. The study therefore argues for the inclusion of the assessment of the e-readiness of academics during human resource performance appraisals. The study also explores the reasons for a lack of e-readiness and determines what is regarded as an acceptable level of e-readiness. Furthermore, the relation between key human factors and e-readiness is explained. For the purpose of this study key human factors such as personal work profile patterns (the DISC factors of Thomas International), preferred learning styles (as outlined by Honey & Mumford 1986) and employees' pace and style of technology adoption (as outlined by Zemsky & Massey 2004) are considered. The impact of these human factors on the e-readiness of an academic is determined in the study.

It is further argued that these human factors impact on an employee's e-profile and should thus be assessed to determine an employee's level of e-readiness. A personal development plan will subsequently be drafted for each employee. For this assessment and development purpose a theoretical framework is developed where employees can be plotted according to their personal work profile patterns, preferred learning styles and pace and style of technology adoption. The framework will thus make provision to identify uniquely structured training and development interventions that should be introduced through the employee development process. It will also

make provision for which motivational strategies should be applied to enhance ereadiness and performance of academics. It will further be determined how career management programmes can enhance the e-readiness of academics.

In addition to the requirement of e-readiness, academics need to realise that the same teaching and learning strategies applied for traditional classroom learning cannot be used for online learning. The use and implementation of e-learning as a changed job requirement calls for academics at HEIs to acquire not only new skills to teach online, but also how to use online resources and tools to enhance the learning experience and facilitate communication for learning to take place (JISC 2009:7). Communication in the online environment is different from face-to-face communication, as it can be both synchronous and asynchronous and is not time dependant (Huang 2002:28; Miller 2005:1). The importance of communication and support from online learning facilitators as a significant factor to their online learning experience is constantly highlighted by students (Sutton 2004; Alexander 2001:242).

Furthermore, the profile of the current day student has changed from that of a traditional student. The average tertiary student is between 18 and 25 years old with the vast majority between 18 and 22 years (HEMIS 2010) and can be categorised as Generation Y. Generation Y refers to people born between 1980 and 2000 (Naidoo 2005). This age category refers to Generation Y in the United States of America; in the South African context these students are mostly regarded as born from 1990 onwards (Steyn, Badenhorst & Kamper 2010:177;185; cf Msimang 2008).

Generation Y was born in an era of technological and sociological change (Kezi 2009:1). This generation prefers learning to be fun, relaxed and interactive and therefore a traditional teaching and learning approach does not appeal to them (Gleeson 2003:4; Price 2009:3; cf Naidoo 2005). According to Halse and Mallinson (2008:1) students belonging to Generation Y portray particular characteristics which impact on their interaction with others and their environment, how they connect and learn, as well as the assortment of technologies they use to do so. They are characterised by a high level of technical literacy (Gen Y Report 2010:24; Song, Singleton, Hill & Koh 2004:59). Some of the technologies often used by this generation include live virtual classrooms, podcasts, blogs, social networks and

collaborative editing (Halse & Mallison 2008:1). Considering these characteristics and technologies the generation prefer can guide facilitators to adapt or customise teaching and learning approaches to enhance learning (Halse & Mallinson 2008:1; cf Song et al 2004:59).

In an interview with the Section Head: Information Technology Support at the NWU, Potchefstroom campus, it was revealed that the majority of online learning facilitators make use of only the basic applications of e-leaning, namely the communications tools (messages and announcements) and information tools (providing resources, and a schedule), but do not use technologies such as podcasts, blogs, forums, wikis and *social networks* which are the typical technologies that appeal to this generation of students and therefore the e-learning platform is not used innovatively to enhance the learning experience (Le Roux 2009). The Section Head: Information Technology Support also indicated that in many instances academics make use of the e-learning platform because it is expected of them by their line managers, but that it is not used in a manner that facilitate learning to take place (Le Roux 2009). E-learning interventions are therefore usually not interactively and creatively applied, which can easily cause disinterest and boredom with the Generation Y learners. This situation further stresses the need for an adjustment of teaching and learning strategies to use e-learning as a learning enhancing tool, which can be handled through training and development.

The vast majority of students at the Potchefstroom campus of the North-West University (NWU), used as a case study for the empirical research component of the study, belong to Generation Y (NWU 2010e). It can be argued that full time residential students from this generation will embrace a blended approach of learning where technology is incorporated in the learning experience. It is therefore evident that academics at HEIs cannot ad infinitum continue to use traditional classroom teaching and learning strategies and need to incorporate e-learning in their teaching and learning strategies.

Furthermore, Rautenbach (2007:16) indicates that teaching and learning has changed. Outcomes based education (OBE) principles call for a changed role of both the learner and the facilitator where learners must have more independence and

must control their own learning events (Rautenbach 2007:16; Butler 2004:11; Gunderman, Williamson, Frank, Heitkamp & Kipfer 2003:16). The facilitator is no longer a traditional teacher (someone who provides knowledge), but a facilitator of learning (someone who enables learners) (Rautenbach 2007:16). Ramsden (2003:xii) is of the opinion that the first step to become a good facilitator is to understand learners' experiences of learning. Milliken and Barnes (2002:225) indicate that this implies that teaching and learning strategies may have to be adapted to focus on the enhancement of learners' learning. The application of new technology can be brought into play to improve both the teaching and learning experience (Milliken & Barnes 2002:226; SA 2004:8).

According to research done by various experts on teaching and learning (Ramsden 2003:106; Elgort 2005:184; Milliken & Barnes 2002:225; Smith 2003) people form their personal theories regarding teaching and learning early on in their lives and rarely change them when they become academics themselves. More authors on teaching and learning (Thomas & Pederson 2003:319; Smith 2000; Efaw 2005), support this notion by stating that academics usually teach in the manner in which they are taught and their principles, approaches, attitudes, and practices are likely to be connected to prior experiences. Robertson (2004) in Elgort (2005:184) further states that academics use technology only if these tools are aligned with their own philosophy of teaching and learning. Thus, from the research done by these authors it can be deduced that an academic's preference for and use of a teaching and learning and learning.

1.2 RATIONALE FOR THE STUDY

The preceding section indicated the global technological change and advancement and the subsequent growth and importance of e-learning. The academic as online learning facilitator plays an important role in this regard which necessitates e-ready academics to use and apply technology tools and media optimally to enhance learning. It is thus necessary to address the lack of e-readiness amongst academics for inter alia the following reasons:



- The inevitable progress of technology urged universities to adapt to these innovations in a competitive higher education environment and to ensure the use of technology in teaching and learning (Bennet 2002:2; cf Albright & Nworie 2008:15; Amirault & Visser 2009:66).
- The NWU's mission promotes the development, education and empowerment of university graduates through "innovative and high quality teaching-learning" that enables them to take responsibility and service their country through creative and critical thinking and solutions (NWU 2010f). E-learning can support in the development of higher order thinking skills (SA 2004:14) and expose learners to the global community and abundant resources (JISC 2009:8; Stone 2008:526; SA 2004:i).
- Traditional classroom teaching and learning does not appeal to the current university learner belonging to Generation Y (Halse & Mallinson 2008:2). Since the vast majority of current learners of HEIs can be categorised as Generation Y (cf HEMIS 2010), academics should be skilled and equipped to train them. This also entails that academics need to adjust teaching and learning strategies to accommodate learners' needs and preferences (Van der Watt 2009:11; Milliken & Barnes 2002:226).
- The two types of learners, those who perfectly fit the Generation Y profile, and those who are either not e-ready or do not have access to e-learning, is a reality experienced in the higher education environment. The income inequality in South Africa affects affordability and access to various social resources including ICT services (Oyedemi 2009:153), and therefore all students did not necessarily have constant access to technology throughout their childhood. Thus, some learners will need guidance and direction, not only in terms of subject-related problems, but also in terms of technology-related problems and the use of it to optimally learn. In many instances adult learners are not e-ready and will need guidance from an online learning facilitator.
- An online learning facilitator needs to constantly act as motivator to learners. Elearning does not entail using the e-platform as a distribution mechanism but to use it interactively in which communication, facilitation and online socialisation are essential (Sutton 2004; Alexander 2001:242; cf Salmon 2002:20–23).

 An online learning facilitator has to "act the part" and demand respect in terms of not only subject matter, but also computer literacy and technical competence.

Presently, neither the literature of human resource performance appraisal nor the literature of e-readiness or e-learning makes provision for particularly assessing the e-readiness of academics. Furthermore, the e-readiness of academics is currently not included in their performance appraisals at the NWU as it is not yet compulsory for employees to use in their teaching and learning (De Wit 2010). However, this may soon change as it becomes increasingly clear that this is the method favoured by the 21st century students and that the university needs to adapt to the technological developments if it does not want to lose its students (De Wit 2010).

Previous research on e-readiness indicates that various e-readiness assessments have been done in the past decade. Specific tools for the assessment of e-readiness have been developed. The purpose of such assessments is to provide benchmarks and to guide the e-readiness development process (Bridges.org 2005a:1). These assessment tools were primarily used to assess e-readiness of countries (especially developing countries), governments, companies (eg banking; property) and to a limited extent HEIs (EIU 2007:1–3; Bridges.org 2005b:1–10; Mutula 2006; Machado 2007:73–74; McConnell International 2001:1–23; Maugis, Choucri, Madnick, Siegel, Gillett, Haghseta, Zhu & Best 2005:313–342; Choucri, Maugis, Madnick, Siegel, Gillet, O'Donnel, Best, Zhu & Haghseta 2003; Ifinedo 2005; SchoolNet Africa 2007; ShoolNet Africa 2008; Bridges.org 2005a:1–4; Brendan 2006:276).

The Economist Intelligence Unit (EIU) has assessed the world's leading economies since 2000 on their capacity to absorb ICT and use it for economic and societal benefit. The quality of a country's ICT infrastructure and the capability of its consumers, businesses and governments to use ICT to their benefit are measured (EIU 2009:1). All the current e-readiness assessment tools focus primarily on facilities, connectivity, accessibility, hardware, software, the digital divide, technological status, policies and regulations (Machado 2007:73–75). The person using the technology is only mentioned as role-player, but very few assessment tools exist to assess the e-readiness of the users of technology and currently no

assessment tool is available to assess the e-readiness of online learning facilitators at HEIs.

Machado (2007:72–82) refers to an e-readiness assessment tool that has been developed specifically for HEIs as result of a focus group study. The particular assessment tool focuses on e-readiness factors such as the ability of HEIs in terms of accessibility, connectivity, technological status, policies, and the capacity of organisational stakeholders. The tool tested the "feelings" of participants towards implementation of e-learning of which the outcomes were positive. During this assessment, the need for abilities at administrative level, instructor level and learner level was identified. However, it does not assess the readiness of the e-user per se.

In 2009 the EIU has for the first time included "consumer and business adoption" as one of the categories in their e-readiness rankings. This refers to the use of the internet by consumers, use of online public services by citizens and the use of online public services by businesses (EIU 2009:4). However, it still does not make provision for the users of technology in HEIs. Due to a lack in contributions in literature on an e-readiness assessment tool for assessing the e-readiness of academics, a framework will be developed for this purpose.

Previous research on academics as online learning facilitators include studies on the role of the online learning facilitator, profiles of the online learning facilitator, skills and competencies needed for online learning facilitators, behavioural requirements and characteristics/personality traits of online learning facilitators (Adendorff 2004; Van Ryneveld 2005; Johannes 2007:13–14). This study will take cognisance of results obtained from the abovementioned studies, but will place the focus on the e-readiness of academics as online learning facilitators. Further, key human factors impacting on the e-readiness of academics regarding personal work profile patterns, preferred learning style and style and pace of technology adoption influence an employee's e-readiness of an online learning facilitator it is first and foremost necessary to determine what is required from an academic to be e-ready. In this regard an e-readiness construct has been developed.

The issue, whether the academic's job specification (indicating the competencies to perform the job successfully) needs to be adjusted, will be raised. It will thus implicate that the matter of an e-ready academic should already be addressed in the recruitment and selection phases. One can furthermore pose the question whether e-learning should become a critical evaluation area and be part of the performance appraisal of an academic. After a performance evaluation it could amongst others be determined whether an incumbent needs further training and development and if a personal development plan should consequently be drafted.

1.3. STATEMENT OF THE RESEARCH PROBLEM

The global technological developments and advancement has led to the increased use of technology in teaching and learning. All academics have however, not made the paradigm shift from traditional classroom teaching and learning to teaching and learning with technology and are thus resisting the changed job requirement (Le Roux 2009; Focus group 2010d). From a public human resource performance appraisal perspective it is expected from academics to be skilled and ready to include technology in teaching and learning to optimally enhance student learning and thus training and development for this purpose should be made available (UNESCO 2008:1; 5; 9–11; Andrew & Durand 2001:9).

Training and development needs are usually identified during a performance appraisal (Kuvaas 2006:504) and it is therefore necessary to include e-readiness assessment during the performance appraisals of academics. Certain key human factors, impacting on the e-readiness of an academic, such as personal work profile pattern, preferred learning style and the pace and style of technology adoption, should therefore be assessed during performance appraisals. The research problem therefore is how to determine the role of these key human factors in the e-readiness of academics. Once the role of the key human factors in the e-readiness of academics have been established, a framework that makes provision for the ereadiness assessment of academics and for the structuring of unique training and development interventions to enhance their levels of e-readiness can be developed.

1.4 RESEARCH OBJECTIVES

In addressing the research problem, particular objectives were set to be obtained.

The primary objective of the study aims to:

 determine the role of key human factors in the e-readiness of academics, with specific focus on personal work profile patterns, preferred learning style and pace and style of technology adoption

The secondary objectives of the study aim to:

- determine the role of human resource performance appraisal in assessing the e-readiness of academics
- determine the role of motivation in employee performance and its relevance to enhancing the e-readiness of academics
- determine the role of employee training and development in enhancing the ereadiness of academics
- determine the reasons for resistance to e-learning and a lack of e-readiness of academics
- determine the indicators for e-readiness of academics by creating an ereadiness construct
- develop a framework assessing the e-readiness of academics during performance appraisals to determine their level of e-readiness and subsequent training and development needs that will be specified in a personal development plan

1.5 RESEARCH QUESTIONS

In addressing the research problem, particular questions were set to be obtained.

The primary question of the study aims to discern:

 how can the role of key human factors in the e-readiness of academics, with specific focus on personal work profile patterns, preferred learning style and pace and style of technology adoption be determined? (chapter 4, 7 and 8) The secondary questions of the study aim to discern:

- what is the role of human resource performance appraisal in assessing the ereadiness of academics? (chapter 2)
- what is the role of motivation in employee performance and its relevance to enhancing the e-readiness of academics? (chapter 2)
- what is the role of employee training and development in enhancing the ereadiness of academics? (chapter 3)
- what are the reasons for resistance to e-learning and a lack of e-readiness of academics? (chapter 5)
- how can the indicators for e-readiness of academics be determined by the construction of an e-readiness construct? (chapter 5)
- can a framework be developed for the e-readiness assessment of academics during performance appraisals to determine their level of e-readiness and subsequent training and development needs that will be specified in a personal development plan? (chapter 7)

1.6 THE RESEARCH DESIGN AND METHODOLOGY

The research design of this study is outlined in the following paragraphs.

1.6.1 Research purpose

The purpose of this research is to develop a framework for assessing the ereadiness of academics during their performance appraisals, as well as determining and assessing the role of key human factors in their e-readiness in order to draft an e-profile which will provide guidance in terms of structuring unique training and development approaches for each e-profile. The performance appraisals will be followed by a personal development plan of each employee, including these uniquely structured training and development interventions, determined by the e-profile of the academic. The most suitable motivating factors for each employee will also be considered during the drafting of the personal development plan. The study contributes to the scholarly discourse and knowledge in the field of public human resource management, as sub-field of Public Administration, by developing a theoretical framework (that currently does not exist), consisting of three matrixes for this purpose.

1.6.2 The approach

The study follows an explanatory approach. Explanatory research aims to provide explanations of phenomena (Durrheim 2009:44). The study focuses on the assessment of the e-readiness of academics and seeks to induce an explanatory value by analysing the reasons for a lack of e-readiness and suggesting particular methods to address the lack of e-readiness.

1.6.3 Unit of analysis

An e-readiness construct is used as the unit of analysis. A construct is defined as "attributes that have been conceptualised and defined in language, and which have been theoretically elaborated in terms of how they are related to other constructs" (Durrheim & Painter 2009:142). For example, the attributes of people that a researcher wishes to assess is seen as a methodological arranged set of ideas, or in other words as constructs (Durrheim & Painter 2009:142). Babbie (2008:135) describes constructs as "theoretical creations that are based on observations but that cannot be observed directly or indirectly".

An e-readiness construct therefore consists of a systematically arranged set of indicators that serve as standards to determine the e-readiness of academics. This arranged set of ideas can be compared to other constructs, in this case key human factors of academics. The e-readiness construct can be regarded as valid as it produced the occurrence of a trend between the e-readiness indicators of the construct and the key human factors of academics. The e-readiness construct assisted the researcher to understand the contributing factors to the e-readiness of academics, as well as factors that cause resistance to e-learning. Thus recommendations on how to enhance the e-readiness of academics could be made.

1.6.4 Data sources (units of observation)

The primary unit of observation in this study is permanent academics at the Potchefstroom campus of the NWU. Additional literature sources that will be consulted include scholarly work (books, scientific journal articles, theses, dissertations and conference papers), official documents (legislation and policies relating to e-learning and human resource performance appraisal) as well as internet sources.

1.6.5 Data collection

The above mentioned data sources will be utilised by means of a variety of methods and techniques, namely a review of scholarship, the interpretive reading of official documents, and the application of qualitative and quantitative data collection techniques by means of semi-structured questionnaires, a focus group discussion and interviews.

1.6.5.1 Review of scholarly literature

In order to contribute to the scholarly knowledge in a field, it is necessary to be knowledgeable on the current state of knowledge and its limitations (Snieder & Larner 2009:133). The review of scholarly literature for this study enabled the researcher to identify a gap in the literature on both human resource performance appraisal and e-learning with regard to e-readiness of academics at HEIs.

The review of scholarly literature informed the data collection of the most significant variables of the study: public human resource performance appraisal and the consequent employee development process; career management and career development; the key human factors included in this study, namely personal profile patterns, preferred learning style and technology adoption pace and style; e-learning and e-readiness. The literature review also assisted in data collection pertaining to the most likely motivating factors of academics and their preferences of and reactions to goal-setting.

The review of scholarly literature includes the critical reading of books, scholarly articles, conference papers, dissertations and thesis. Relevant information will be applied to this study.

1.6.5.2 Reading of official documents

Legislation, policies, official government reports and databases pertaining to the field of human resource management, in particular human resource performance appraisal and employee development, education and e-learning were reviewed. Official documents of the NWU were consulted and referred to in the study, particularly with reference to policy on human resource performance appraisal and teaching and learning.

1.6.5.3 Semi-structured questionnaire

In order to obtain data from the primary unit of observation (academics at the Potchefstroom campus of the NWU), a semi-structured questionnaire, specifically a self-administered questionnaire² (see Annexure C), was designed and distributed via electronic mail to 300 permanent academics at the Potchefstroom campus of the NWU. A self-administered questionnaire can only be used if the population is sufficiently literate (Babbie & Mouton 2004:258).

The questionnaire comprised of closed and open-ended questions. Closed questions were used to determine respondents' work style patterns, preferred learning styles, pace and style of technology adoption, motivating factors, goal-setting preferences, preferences pertaining to the use of e-learning and opinions on e-learning training and development interventions. The feedback provided for comparative data analysis on these aspects and certain conclusions and trends could be drawn from the quantitative data.

Open-ended questions were also directed at the e-learning training and development of academics, as well as their views on the inclusion of e-readiness assessment in

² A self-administered questionnaire can be described as a questionnaire that respondents complete themselves (Babbie 2008:286).

performance appraisals. Responses to the open-ended questions provided insight into respondents' perceptions on e-learning, e-readiness assessment and e-learning training and development, and could be related to respondents' e-learning profiles.

The questionnaire was not designed to do a professional psychological analysis of personal work profile patterns, preferred learning style and pace and style of technology adoption of academics. Particular questions have, however, been asked to give an indication of an employee's personal profile pattern, learning style preference and pace and style of technology adoption.

1.6.5.4 Focus group discussion

The purpose of conducting the focus group³ discussion was to determine academics' perceptions about e-learning and e-readiness in the higher education environment, as well as their perceptions about students' expectations in this regard and the role that academics should play in terms of learning through e-learning. This was done through posing open-ended questions to participants (Annexure F), followed by their discussions.

The focus group participants were selected from a pool of permanent academics at the Potchefstroom campus of the NWU (the unit of observation). A total of 10 participants consisting of lecturers, senior lecturers, associate professors and professors from two faculties (the Faculties of Arts and Natural Sciences), participated in the focus group discussion.

1.6.5.5 Interviews

Semi-structured interviews, with the use of open-ended questions, were conducted with various experts to obtain specialist information on particular variables of the study. The following specialists were interviewed for data collection:

³ A focus group is a group of individuals who have a similar type of experience in common, but is usually not necessarily part of the same social group (Kelly 2009:304). Therefore, with a focus group the researcher gets access to inter-subjective experience, namely a group of people's mutual experience (Kelly 2009:304).

- The Director: Human Resource Management, Institutional Office, NWU.
- The Section Head: Information Technology Support, Academic Support Services, NWU, Potchefstroom Campus.
- The Head: Writing Laboratory, Centre for Academic and Professional Language Practice, NWU, Potchefstroom Campus.
- The Head: Student Learning and Reading Development, Centre for Academic and Professional Language Practice, NWU, Potchefstroom Campus.
- The Manager: Reading Laboratory, Centre for Academic and Professional Language Practice, NWU, Potchefstroom Campus.
- A Subject Specialist: Computer Science and Information Systems, NWU, Potchefstroom Campus.
- Information Technology Laboratory Manager, NWU, Potchefstroom Campus.

1.6.5.6 Sampling techniques

Academics at the Potchefstroom campus of the NWU were used as unit of observation. The researcher ensured that a sample representative of the academics of the Potchefstroom campus of the NWU was chosen. A second consideration with sampling is the size of the sample (Durrheim 2009:49). The sample size must be large enough to make deductions and assumptions about the population (Durrheim 2009:39). In some instances, as was the case with this study, the sample size was determined by practical considerations (Durrheim 2009:49), that is the availability of academics.

Whereas random probability sampling from the identified unit of observation was used for the questionnaire, purposeful non-probability sampling, convenience sampling and maximum variation sampling was used for the focus group discussion. For the interviews respondents, specialising in particular fields relevant to the primary variables of the study (human resource management; e-learning; ICT and student learning), have been selected.

1.7 ETHICAL CONSIDERATIONS

The following principles of ethical conduct were adhered to in the study:

- *Voluntary participation* respondents of the questionnaire, participants in the focus group discussion and interviewees participated voluntary and could withdraw from the study at any time.
- *Informed consent* participants were made aware of the research purpose and have given their consent to participate in the research.
- Privacy the confidentiality and anonymity of respondents to the questionnaire and participants in the focus group discussion were protected at all times; interviewees consented to be quoted.
- *Plagiarism* the researcher is aware of policies in this regard and did not make use of another researcher's work and submitted it as her own. The researcher has submitted her own work.

1.8 STUDY TITLE AND RELATED CONCEPTS CLARIFIED

In this section phrases and concepts from the study title is used to clarify their meaning. Concepts that relate to the study title concepts are also outlined and clarified.

1.8.1 Key human factors

The study focuses on *key human factors* that impact on the e-readiness of an academic. A human factor in this respect refers to a person attribute, observable in a personal work profile pattern (work behavioural style), a preferred individual learning style and an employee's unique style and pace of technology adoption. The term *key* is used as these human factors stand central to an academic's e-readiness.

Shah and Irani (2010:1) explain that it is necessary to understand and appreciate the factors that affect how employees behave in the work environment to manage them effectively. Employees have different personalities, intelligence, abilities, values,

backgrounds and attitudes which influence their behaviour (Armstrong 2006b:240–244). Xu and Tuttle (2004:22) explain that interpersonal aspects are more important to an employee's success in the workplace than technical skill. An interpersonal aspect that is likely to differ between employees is work style; for instance, some employees approach problem-solving in a cautious, systematic manner, whereas others favour innovative solutions (Xu & Tuttle 2004:22). Employees' work styles therefore influence their behaviour in the workplace.

Pearsall (2001:157) describes *behaviour* as "the way in which one acts or conducts oneself, especially towards others" and *style* as "a manner of doing something" or "in a manner characteristic of" something (Pearsall 2001:1847). Thus a *work behaviour style* refers to the characteristic manner in which an employee acts and conducts him- or herself, especially towards other employees, including managers, in the workplace. For the purpose of this study the DISC-profiles of Thomas International is used as indication of **personal work profile preference** of academics as Thomas International is widely accepted as leading expert in the field of work behavioural style.

Another human factor considered in this study is preferred learning style. Learning styles refer to various approaches or techniques of learning and involve educating methods, which seemingly allow learners to learn effectively (Guild 2001; Felder & Henrique 1995:21). *Learning styles* influence the manner in which individuals attach their own meaning to the subject matter or skill being taught (Roy 2006:22). Salmon (2003:110) argues that online teaching and learning must be structured in such a manner that it makes provision for all learning styles as it will enhance an employee's ability to learn (cf Swinton 2006:1). Thus, the study argues that preferred *learning style* should be considered during the e-readiness assessment of academics, as well as with the drafting of their personal development plans. Specific focus is placed on the *learning styles* as outlined by Honey and Mumford (1982) who is regarded as experts in the field of learning style.

The third human factor considered in this study is **style and pace of technology adoption**. For this purpose the various adopter categories as described by Zemsky and Massey (2004) and the technology adoption cycle as illustrated by Rogers

(1962) in his book *Diffusion of innovations*, is considered, namely *Innovator, early adopter, early majority, late majority and laggards* (cf The Computer Language Company 2010). Rogers, as well as Zemsky and Massey, are widely acknowledged as experts in the field of technology adoption.

Alexandrou (2011) describes *technology adoption* as the manner in which people respond to product and service innovations that require them to change their past behaviour. An employee's style and pace of technology adoption therefore refer to the characteristic individual manner in which an employee will respond or adapt to a new technology or innovation (*style*) and how fast or slow the employee will adopt (*pace*). The study argues that style and pace of technology should also be considered with the e-readiness assessment of an academic and the subsequent development process.

1.8.2 E-readiness

The majority of definitions for *e-readiness* refer to readiness of countries, governments and businesses for the use of electronic media (Dada 2006:1; Rautenbach 2007:iv). According to Dada (2006:1) *e-readiness* refers to the extent to which a group of people is prepared and geared to take part in the online world or the potential of an organisation to get involved in e-learning activities.

According to Guglielmino and Guglielmino (2003) in Moolman (2007:65), e-readiness of people can be assessed by evaluating amongst others an individual's technical experience and competency with computers (Schreurs, Sammour & Ehlers 2008:267; Lee-Post 2009:66). In addition to these competencies, a person should have the capability to direct his/her own learning, by means of relevant knowledge, attitudes, skills and habits (Guglielmino & Guglielmino 2003 in Moolman 2007:65). Further, an employee's e-readiness levels may or may not be supported by the readiness of the organisation, therefore institutional management should guide and support e-learning processes for it to be considered a supporting factor to the e-readiness of an employee (Moolman 2007:65). E-readiness assessment is intended to direct development efforts by providing benchmarks for comparisons and measuring progress (Budhiraja & Sachdeva 2010:3).

E-readiness, in the context of e-learning, also entails that the users of technology should possess the necessary capabilities and skills to use e-learning strategies (Moolman & Blignaut 2008:169; Moolman 2007:65). *E-readiness* is not limited to physical readiness (infrastructure, hardware, software), but also consists of the non-physical readiness of an organisation such as psychological readiness (Rautenbach 2007:iv; Hewitt 2003:5). It is thus evident that human factors are related to a person's e-readiness. The focus of this study is on particular human factors that impact on an academic's e-readiness.

For the purpose of this study *e-readiness* refers to the willingness and preparedness of academics to use e-learning in teaching and learning, implying that they have the necessary technical skill, the skill to use and adapt teaching and learning strategies that best suit e-learning, the psychological readiness to use technology in teaching and learning, and the ability to use e-learning to optimally enhance learning. This study explains the role of key human factors such as personal work profile pattern, learning style and pace and style of technology adoption on e-readiness of academics. As the study focuses on the need for academics to be e-ready in order to effectively make use of e-learning, the concept *e-learning* is also clarified.

1.8.3 E-learning

The "e" in *e-learning*, referring to "electronic", adds a technological edge to the "learning", but the focus remains on *learning* as the critical element and e-learning can therefore be seen as enhanced learning (JISC 2009:8). *E-learning* is no longer only linked to distance or remote learning, but has become an element in a conscious decision to use the best and most suitable ways to enhance effective learning (JISC 2009:9; Armstrong 2006b:583). *E-learning* enhances *learning* by broadening and complementing classroom teaching and learning, rather than replacing it (Armstrong 2006b:583). The focus of *e-learning* is not on technology, but on *learning*, supported by technology (Armstrong 2006b:583).

As learning is central to the concept of *e-leaning*, it is necessary to clarify what is meant when referred to learning. According to Pearsall (2001:1048) *learning* is "the

acquisition of knowledge or skills through experience, practice, study or by being taught". The aim thus is to acquire knowledge and skills. Connor (2007) indicates that learning, from the most basic to complex, is an increase in knowledge, memorising information, obtaining knowledge for practical use, finding meaning from what is taught, and a progression of understanding. Learning is the act, process, or experience of acquiring knowledge or skills (Connor 2007).

E-learning refers to learning that can be accessed by any electronic means (Alessi & Trollip 2001:377; Moolman & Blignaut 2008:169; Takalani 2008:3; cf NZCER 2004:21), therefore, acquiring knowledge and skills through electronic means. The European Commission (2001) defines *e-learning* as "the use of new multimedia technologies and the Internet to improve the quality of learning by facilitating access to resource and services as well as remote exchanges and collaborations." *E-learning* can further be defined as an innovative approach to provide learners with well-designed, learner-centred, interactive, and facilitated learning environments with "anyplace, anytime" access (Kahn 2005:3). It is thus evident that e-learning is focused on *learning*.

In this study the *e-learning* focus is in particular on web-based learning. Whereas *e-learning* make use of a variety of ICT tools, web-based (online) learning refers to learning specifically through the use of the internet/web-based applications (SA 2004:16; Alessi & Trollip 2001:5–7). However, other e-learning tools such as podcasts, blog and wikis can also be connected to or uploaded on an e-learning platform.

1.8.4 Information and communication technology (ICT)

As e-learning encompass the use of ICT tools, it is necessary to define *Information* and communication technology (ICT). According to Webster (2011), technology is a method with which a task can be achieved, mainly through the use of technical or practical processes, means, or knowledge. Technology is also defined as the branch of knowledge dealing with applied sciences (Pearsall 2001:1903). *Technology* further refers to the particular aspects of a specific field of undertaking, such as the use of technology in teaching and learning (Webster 2011). The use of technology in teaching and learning involves the use of electronic media such as *inter alia* computers, the Internet, CD-ROMs and e-platforms (Alessi & Trollip 2001:377).

The computer hardware (personal computers, scanners, digital cameras) and software (database programmes and multimedia programmes) that allow us to access, retrieve, store, organise, manipulate and present information by electronic means can be described as *information technology* (IT) (Pearsall 2001:937). *Communications technology* refers to telecommunications equipment that can be used to access, send and seek information, including phones, faxes, modems and computers (SA 2004:16).

Information and communication technology (ICT) then indicates a combination of IT (hardware and software programmes) and communication technology through which the processing, management and exchange of data, information and knowledge is being made able (SA 2004:16; cf DPSA 2004:7). ICTs are therefore used with a LMS in teaching and learning.

1.8.5 Academic employee

An *academic employee* refers to a person that is appointed to teach or to do research at a public HEI or any other employee selected as such by the council of that HEI (SA 1997:7). Currently universities as HEIs take better responsibility for appointing and managing their academics and is therefore required to develop performance measures, training methods and incentives to ensure academics' professional competencies and teaching expertise (Cano-Hurtado, Carot-Sierra, Fernandés-Prada & Fargueta sa:1). The assessment of teaching is in particular significant to universities, as assuring the excellence in teaching, ensure both the professional competency of their academics and the quality of their teaching and learning (Cano-Hurtado et al sa:1).

In this study the focus is on one of the job requirements of an academic: teaching with technology and in particular the e-readiness to use technology successfully in teaching and learning. Before an *academic* is involved in, and skilled in e-learning, the employee cannot be referred to as an *online learning facilitator*. Once the
academic becomes involved in, and skilled in e-learning, the employee will be referred to as an *online learning facilitator*. In other words, an *academic* should be e-ready to act as *online learning facilitator* with e-learning.

1.8.6 Online learning facilitator

When the concept *online learning facilitator* is defined, the various components of the concept need to be clarified:

Online implicates that the teaching and learning process is connected to and accessible via a computer or computer network (Pearsall 2001:1296). For the purpose of this study online will refer to the teaching and learning process that is web-based, including other e-learning tools and media that is connected to web-based learning or uploaded to an e-learning platform.

As indicated by Pearsall (2001:1048) *learning* is "the acquisition of knowledge or skills through experience, practice, study or by being taught". *Online learning* involves the use of a computer or electronic device (eg a mobile phone) in some way (to a bigger or lesser extent) to teach or provide learning material (Stockley 2003; AFLF 2003:2). For the purpose of this study *online learning* implies that an online learning facilitator makes use of electronic means in teaching and learning to contribute to and optimise student learning.

Facilitation means making an action or process easier (Pearsall 2001:656), in the case of this study, learning. *Facilitation* is a pedagogical term that refers to student-centred approaches in teaching and learning, as opposed to teacher-driven approaches (ANTA 2003:2; AFLF 2003:2). Online facilitation thus refers to managing the communication of learners online with a view to accomplish learning. This is also often referred to as *online moderation*.

According to Kaner, Lind and Toldi (2007:xv) a *facilitator* can be described as a person who enables groups to work more effectively, to collaborate and achieve synergy. A facilitator furthermore also adds structure and process to group interaction and encourages full participation amongst participants (Kaner et al

2007:xv). In education, facilitators are usually subject experts, but also draw on the knowledge of learners and fill in any gaps (Answers Corporation 2007). In the higher education environment and in the context of teaching and *learning* a facilitator refers to an academic who tutors/teaches/facilitates learners. An *online learning facilitator* then refers to an academic who effectively uses digital means of learning to teach, tutor or guide learners. In chapter 4 the roles and responsibilities of an academic as online learning facilitator will be elaborated upon.

1.8.7 Higher education institutions (HEIs)

A *Higher Education Institution* (HEI) refers to an institution that provides higher education on a full-time, part-time or on a distance basis (SA 1997:8). The institution should be established or considered to be established, and be declared as a public HEI under the Higher Education Act 101 of 1997. The study focuses predominantly on the university as HEI and the NWU, Potchefstroom campus, is used as case study.

HEIs provide education on tertiary level through which learning, an act, process, or experience of acquiring knowledge or skills (Connor 2007), takes place, thus the result of education. According to Knowles, Holton III and Swanson (2005:10) *education* is an action taken on and/or initiated to result in changes in the knowledge, skills and attitudes of individuals, groups or communities. The educator, who is regarded as the change agent, provides incentives, support and reinforcement for learning and plans activities to encourage change (Knowles et al 2005:10; Contento 2011:387; cf Reeve 2009). *Learning* give emphasis to the person in which the change occurs or is likely to occur, and is the "act or process by which behavioural change, knowledge, skills and attitudes are acquired" (Knowles et al 2005:10). A shift has taken place from teaching to learning and at the same time online education became pivotal in the fundamental institutional change of the higher education sector (Matthias, Schneckenberg & Wildt 2001:5).

1.8.8 Student vs learner

HEIs provide education to *students*. According to Warlick (2010) a *student* follows instructions and is compelled to do what is expected by the teacher, whereas a *learner* has a vested interest in learning and is therefore motivated by it and appreciate the value of learning. A *student* is usually compliant, group-disciplined, objective-oriented (Khosrow-Pour 2000:926), whereas a *learner* is determined, self-disciplined, group- and goal-oriented, resourceful, and is "learning to achieve rather than achieving learning" (Warlick 2010). A *student* is someone caught up in the power structures of a very specific social institution, whereas a *learner* is not necessarily involved in formal education (cf Anon 2008). A *learner* can be viewed as a state of mind, a *student* can be viewed as a social position (Anon 2008). Rossiter (2006:40) explains that the change in terminology from *student* to *learner* is due to the fact that the concept *student* is associated with "dependency or reliance on an institution (or a discipline, profession or teacher)", whereas the concept *learner* implies added commitment on the part of the individual to the gaining of knowledge, skills and understanding in the learning process.

According to Floyd (2009) learning is concerning making the paradigm shift from subject-oriented education to person-oriented learning. With learner-centred education both students and teachers together identify the *what* and the *how* questions to learning. "Continuing education leads us to *know* something different; lifelong learning assumes that we will *be* someone different as a result of the process" (Floyd 2009). The concept of lifelong learning should be embraced to ensure continuous development and considering that organisations have to function in the knowledge economy of the 21st century (Amirault & Visser 2009:66; World Bank 2003:xiii; cf Markkula 2006:2–3).

Therefore, whereas academics teach *students*, who *learn* during the teaching and learning experience, it is ideal that *students* should adapt a *learner* attitude and embrace the concept of lifelong learning. For the purpose of this study, however, academics themselves should become *learners* in the process of enhancing their e-readiness. In the study the concept *learner* is preferred and used.

V-V-List of research project topics and materials

1.8.9 Assessment

According to Pearsall (2001:100–101) to *assess* is to "evaluate or estimate the nature, ability or quality" of something or someone and *an assessment* is the "action of assessing someone or something". In this study the "someone" to be assessed refers to the academic and the "something" to be assessed refers to the e-readiness of the academic. *Assessment* of an employee's skill and capacity to fulfil job requirements (*performance assessment*) is done through a performance appraisal (cf Weiner, Graham, Schinka & Naglieri 2003:321). Therefore the concept human resource performance appraisal is defined in the next section.

1.8.10 Human resource performance appraisal

Human resource performance appraisal is a structured method through which an employee's job-related activities and outcomes are measured and evaluated to determine how the employee is presently performing on the job, why the employee is performing at the particular level and how the employee's performance can be enhanced to benefit all stakeholders, namely, the employee, the organisation and society at large (Schuler 1981:211; Aswathappa 2005:227). Performance appraisal thus refers to the assessment of employees' performance (evaluative part), as well as their potential for further development (developmental part) (Rademan & De Vos 2001:54; cf Grobler, Wärnich, Carrell, Elbert & Hatfield 2002:260;266). This study focuses on both components of performance appraisal: academics will be assessed to determine their level of e-readiness (evaluative part) and the assessment will be followed up with a development plan and development interventions (developmental part).

1.8.11 Human resource management

Human resource management (HRM) refers to the organisational function, facilitating the most effective use of people with the aim to obtain organisational goals and objectives (Ivancevich 2004:4). Human resource management can also be explained as the process of developing, applying and evaluating policies, procedures, methods and programmes relating to the individual in the organisation

(Kleiman 2011). Armstrong (2006b:3) defines human resource management as "a strategic and coherent approach to the management of an organisation's most valuable assets – the people working there who individually and collectively contribute to achievement of its objectives".

It is therefore evident that the primary aim of HRM is to ensure that an institution obtain success through its human resources (employees). According to Erasmus, Swanepoel, Schenk, Van der Westhuizen & Wessels (2005:4) public HRM is part of management, including all the functions, practices, strategies, principles, operations, decisions, activities, processes, procedures, etcetera which relate to the people in the institution.

For the purpose of this study, HRM will refer to the typical HR functions, policies, strategies, structures, procedures and interventions that can be executed and implemented at higher education institutions with the aim to enhance both institutional and individual productivity, with the eventual objective of improving overall effectiveness and providing a quality service to the student. It will be determined therefore which role HRM can play in improving the skills and capacity of academics in order to ensure organisational success and effectiveness, particularly pertaining to e-learning and e-readiness.

1.9 LAYOUT OF CHAPTERS

The next chapters of the study will discuss the following:

In **chapter 2** a literature review of public human resource performance appraisal and its role and place in addressing the e-readiness of academics at HEIs is provided. Current human resource performance appraisal practices and policies of the NWU are also reviewed with a view to determine the possibility and shortcomings for including the e-readiness assessment of academics in the performance appraisal process. Employee performance is directly or indirectly related to motivation, therefore the role and importance of employee motivation is explained and brought in relation to the e-readiness of academics. The significance of motivating factors such as incentives, goal-setting, intrinsic and extrinsic motivation and self-determination

are outlined. Specific attention is also given to the role of self-determination in the motivation towards technology adoption.

Chapter 3 discusses employee development. As human resource performance appraisal entails a developmental component, attention is given to the role of employee development in increasing the e-readiness of academics. Further the rationale for including e-learning training in employee development, career management programmes and talent management programmes receives attention. Attention is also given to current employee development and career management practices at the NWU with a view to make provision or the e-readiness assessment of academics. The chapter also reviews the uniqueness of the academic as an adult learner and the role of the line manager in employee development.

In **chapter 4** the key human factors that impacts on academics' e-readiness, are outlined and explained, namely personal work profile patterns, preferred learning style and pace and style of technology adoption. The interrelatedness of these human factors and how they contribute to an employee's e-profile is also determined and explained.

Chapter 5 reviews the 21st century higher education environment and the typical HEI student. The chapter also determines the role and purpose of e-learning in this environment and in relation to the 21st century student. Further it is explains what e-readiness encompasses, Enabling factors to e-readiness are outlined and discussed and indicators of the e-readiness of academics are established and included in an e-readiness construct. Since a lack of e-readiness usually causes resistance to e-learning, various barriers to e-learning are discussed and its relation to the e-readiness of academics and its relation to the e-readiness of academics of a discussed and its relation to the e-readiness of academics are discussed and its relation to the e-readiness of academics is indicated.

Chapter 6 profiles the NWU, Potcehfstroom campus, in terms of its policies and practices pertaining to performance management, employee and career development, as well as the 21st century undergraduate student.

In **chapter 7** research results obtained from the questionnaire, focus group discussion and interviews are discussed and interpreted. A framework which can be

used to assess the e-readiness of academics during performance appraisals is developed. The impact of key human factors in the e-readiness assessment of academics is considered and included in the development and implementation of the framework.

Finally, in **chapter 8** a summary of the study is provided, particular conclusions are drawn and recommendations are made regarding the e-readiness assessment of academics during performance appraisals with a view to optimise student learning through the use of e-learning.

The next chapter (chapter 2) discusses and explains the role of human resource performance appraisal, employee development and motivation in the e-readiness of academics.

CHAPTER 2

HUMAN RESOURCE PERFORMANCE APPRAISAL AND EMPLOYEE MOTIVATION

2.1 INTRODUCTION

From the discussion in chapter 1 it is eminent that the 21st century technological advancements and the current student profile (Generation Y), compels HEIs to adapt to technological development to remain competitive and to use generation-relevant educational tools. The use of e-learning to enhance learning requires academics to be equipped with the necessary skills to act as online learning facilitators. Thus, the e-readiness of academics is imperative in terms of not only technical and social skills, but also to use effective online teaching and learning strategies and to ensure that learning is optimised. However, when academics experience a lack of e-readiness it will influence the effectiveness with which they fulfil this job requirement – their online teaching duties. Thus, the need for training and development interventions is eminent.

For academics to be competent in discharging their role as facilitator in the learning environment, in particular using e-learning as a learning tool, their competency to use e-learning needs to be assessed. This study therefore argues that, due to the changed job requirements, and highlighting the importance of the role of the online learning facilitator in e-learning, academics need to be assessed on their e-readiness during performance appraisals. The chapter reviews the literature on human resource performance appraisal to determine its role and place in the e-readiness assessment of academics.

A significant factor that managers should take cognisance of, in terms of employee performance and development, particularly with changing job requirements, is employee motivation. When employees resist change, such as the incorporation of technology in teaching and learning, it might be associated with a lack of motivation. The chapter therefore reviews the scholarly literature on the importance and

35

relevance of employee motivation in the performance of academics pertaining to elearning.

2.2 HUMAN RESOURCE PERFORMANCE APPRAISAL FOR E-READINESS ASSESSMENT

Human resource performance appraisal is a HRM practice (Kundu, Malhan & Kumar 2007:75). As alluded to in section 1.8.11 of the previous chapter: HRM is "a strategic and coherent approach to the management of an organisation's most valuable assets – the people working there who individually and collectively contribute to achievement of its objectives" (Armstrong 2006a:1). It is evident that the primary aim of HRM is to ensure that an organisation obtains success through its human resources. This implies that the human resources in an organisation should perform to an acceptable level in order to ensure fulfilment of organisational objectives. The performance of employees should therefore be assessed on a regular basis to determine whether it is on an acceptable level or whether training interventions are needed to develop employees towards an acceptable level of performance. This is done through human resource performance appraisal. Before the concept of *human resource performance appraisal* is defined, it is necessary to first determine where performance appraisal fits into the broader performance management structure of an organisation.

2.2.1 Human resource performance management

According to Amos (2009:8) *performance management* can be described as a strategy that is set in the context of an organisation's HR policies, culture, approach and infrastructure and which relates to all activities of the organisation. Amos (2009:9) further explains that the performance management process entails a continuous cycle of planning, acting, monitoring and reviewing, and again planning to complete the cycle. This cycle links logically with the performance appraisal process (Amos 2009:9). The organisational milieu will dictate the nature of the performance strategy and can differ from organisation to organisation (Amos 2009:8; cf Thomson & Mabey 2001:189–190).

Performance management can also be defined as a "strategic and integrated approach to delivering sustained success to organisations by improving the performance of the people who work for them and by developing the capabilities of teams and individual contributors" (Armstrong 2006a:142). Armstrong (2006b:495) further defines performance management as an organised process by which organisational performance is enhanced through increasing the performance of individual employees and teams. A pre-decided structure of premeditated goals, standards and competency requirements provides for better results and it focuses employees on doing the correct things through goal clarification and is own and steered by line management (Armstrong 2006b:495). It is thus evident that goals are being set for performance management and that employees are assessed against certain predetermined standards.

Further, in their Strategic human resource planning: guideline and toolkit, 2008, the Department of Public Service and Administration (DPSA) explains that a performance management system stems from an organisation's vision and objectives (DPSA 2008:7; DPSA 2002:2-3; cf Thomson & Mabey 2001:189-190; McNamara sa). The performance targets of individuals, against which they will be assessed during a performance appraisal, are therefore set within the framework of organisational objectives and an organisational strategy (Amos 2009:9). Therefore, individual objectives and organisational objectives should be aligned (Armstrong 2006b:496; PSC 2010:x;5; Nickols 2007:12). The mission of the NWU refers to the aim of providing world class education to students and to "educate and empower (them) through innovative and high guality teaching-learning" (NWU 2010f). It is thus evident that the University's mission statement supports innovative methods of teaching and learning, of which e-learning is one. As indicated in the previous chapter, with e-learning the focus should be place on enhanced learning and the use of technology in teaching and learning should thus be applied to optimise learning. When academics are skilled and capable to do so, they will contribute to the university's vision and objectives through their individual performance.

According to the *Toolkit for the management of poor performance in the public service*, drafted by the Public Service Commission (PSC), performance management is intended to be a process that assists organisations in instituting a climate that is

favourable to motivating employees towards the development and achievement of high standards of performance (PSC 2007:5), thus emphasising the importance of motivation in employee performance. In this study it is argued that the key human factors of academics will determine how they should ideally be motivated to adapt to the changed job demand and to increase their performance. Section 2.3 elaborates on employee motivation.

Various authors (Landes 2009:28; Deci & Ryan 2000:227; Cardno 1995:118; Resnick 2007; Momberg 2004:36; Rademan & De Vos 2001:54; PSC 2010:x;5; Moller, Ryan & Deci 2006:105–106;110; Armstrong 2006b: 495–529; Armstrong & Baron 1998:568 in Amos 2009:9; Nickols 2007:12–13; Li & Butler 2004:38) indicate that particular aspects are typical of the performance management process and should be practiced and implemented to ensure the success of performance management. They outline the following aspects relating to performance management and performance appraisal: A meaningful rationale and benefits; well-defined processes; aptitude, skills and knowledge; clear goals, expectations and responsibilities; support structures and guidance systems for academics; patience and reinforcement, feedback, career management; objective assessment and legal protection. The importance of practicing and implementing of these aspects are explained in the following paragraphs:

A meaningful rationale and benefits

For employees to perform, they need a rationale as for why certain activities/responsibilities are expected of them (Landes 2006:28; Moller et al 2006:105–106;110; Deci & Ryan 2000:227). The importance of providing employees with a rationale is cardinal to employee motivation and ultimately performance, particularly when confronted with a changed job requirement, as is the case with academics at HEIs with the inclusion of technology in teaching and learning.

The rationale should include information on who benefits from the new job demand (Landes 2006:28). First and foremost the academic needs to understand how the new job demand will benefit teaching and learning. Further, the organisation (the HEI) and society (the learners and the future work places) will also benefit from

employee development for the new job requirement and the eventual implementation of the new job requirement, e-learning, and should thus also be included in the rationale provided to the employee (Cardno 1995:118; cf Maurer Pierce & Shore 2002:432; Adam 2010). The importance for providing a rationale to employees is thus discussed in sections 2.3.1 and 2.3.2 as part of goal-setting for performance.

• Well-defined performance management processes

Landes (2006:28) explains that a well-defined performance management process simplifies the tasks that employees are expected to perform, particularly those tasks they do not perform on a regular basis. Well-defined processes give direction for the sound implementation of the tasks employees are assigned to perform (Resnick 2007). Therefore, these processes are vital, whether they are developed with support from management, or whether employees create them on their own (Landes 2006:28; Resnick 2007). These well-defined processes can also assist academics, to clearly understand the role and purpose of e-learning in teaching and learning, what is expected of them and how it should be accomplished. One can argue that a well-designed process is likely to contribute positively towards employee motivation.

• Aptitude, skills and knowledge

Whereas aptitude is an intrinsic quality (Moolman & Blignaut 2008:171), skills and knowledge are obtained (Landes 2006:28). When it comes to *aptitude*, it's important that there is a job-person fit (Landes 2006:28; Johannes 2007:135; Momberg 2004:36). One can argue that an employee with an aptitude for technology or an understanding of the constructivist teaching and learning approach, as outlined in chapter 3, will easily fit the position of online learning facilitator, but academics who do not possess this aptitude may not be e-ready to effectively fulfil online teaching tasks.

As for *skills and knowledge*, it is necessary to conduct an assessment in order to determine whether employees' skills and knowledge need to be improved through training and development (Landes 2006:28). Subsequently, suitable development

plans can be set up for each employee (Rademan & De Vos 2001:54). Development plans should support both organisational and individual needs (PSC 2010:x;5; Landes 2006:289). The drafting of development plans to enhance a skill, more specifically a skill to effectively use technology in teaching and learning, is the very purpose of assessing the e-readiness of academics at HEIs.

• Clear goals, expectations and responsibilities

The likelihood of achieving desired outcomes will be increased when goals and expectations are well-defined (Li & Butler 2004:38; Resnick 2007). Armstrong and Baron (1998:568) in Amos (2009:9) agree that employees will do their best to achieve objectives if they know what is expected of them. One can argue that this calls for setting clear and unambiguous goals. Well-defined and clear goals and expectations will also enhance communication between managers and subordinates (Landes 2006:29).

Nickols (2007:12) asserts that the alignment of individual goals with organisational goals should be ensured (Amstrong 2006b:498; PSC 2010:x;5; Gilley, Eggland & Gilley 2002:60). Section 2.3.2 outlines the importance of goal-setting in motivating employees with a view to enhance performance.

In order to minimise resistance to change, it is advisable that management provides an understandable explanation as to which route the organisation wishes to take (Moller et al 2006:105–106;110; Landes 2006:29). Employees should be informed as to what is expected of them and what their individual responsibilities are (Armstrong 2006b:499; Landes 2009:29). This will increase the probability for employees to meet management's expectations (cf Gagné, Koestner & Zuckerman 2000:1843).

• Support structures and guidance systems for academics

Employees are more likely to perform if they get the necessary support to achieve goals (Landes 2006:29). Support mechanisms in the workplace might include resources such as funding, people, communication tools, the right working

V=V List of research project topics and materials

environment, management endorsement, organisational processes, systems, and clearing barriers, to name a few (McNamara sa; Armstrong & Baron 1998:568 in Amos 2009:9). It is thus important that HEIs determine whether employees are being provided with the necessary support to fulfil their e-learning duties, whether it is training and development, equipment, motivation, recognition, or any other form of support.

According to Landes (2006:29) managers need to communicate with employees to give direction (Gitman & McDaniel 2008:187; Hansson 2009) by firstly providing information; secondly, providing a communication system to regularly let them know if they are on target or not; and thirdly, by creating a fixed control mechanism or process to take corrective action if necessary, and to hold employees accountable for their actions. However, if the only tool for holding people accountable is punishment and criticism, a manager will not get the best performance out of employees (Landes 2006:29). Therefore, the performance appraisal process should be used as a mechanism to encourage employees to perform better and to give guidance as to how performance can be improved (Landes 2006:29; Armstrong 2006b:496). It can be argued that putting these support structures and guidance systems for academics in place, will positively impact on employee motivation and ultimately on individual as well as organisational performance.

• Patience and reinforcement

Patience with an employee's performance curve and reinforcement of performance goals usually pay-off in the long run (Landes 2006:29; Vroom 1964:13). Research has shown that old habits die hard and employees find it difficult to move out of their comfort zones (Proctor & Doukakis 2003:268; Robbins 2003:559–560), thus it requires patience of line managers to steer employees towards performance with changed job requirements. As indicated in previous sections, resistance to change will be eminent during periods of change, such as the changed job requirements for academics, which may negatively impact on employees' performance. Therefore, regular reinforcement, repetition and communication of new job requirements are required for employees to become comfortable with change (Landes 2006:28–29).

Skinner's (1969) reinforcement theory refers to the relationship between the desired behaviour (eg performance) and a motivational tool (eg merit pay) and it is based on the philosophy and techniques of organisational behaviour modification (Stajkovic & Luthans 1997 in Perry, Mesch & Paarlberg 2006:505). "Organizational behaviour modification is a framework within which employee behaviours are identified, measured, and analyzed in terms of their functional consequences (i.e., existing reinforcements) and an intervention is developed using principles of reinforcement" (Luthans & Kreitner 1975; Stajkovic & Luthans 1997 in Perry et al 2006:505). Thus, a financial incentive can be used to motivate an employee towards performance by using the incentive as motivational tool. Section 2.3.1 discusses financial and non-financial incentives and rewards and also addresses reinforcement for employee motivation towards increased performance.

• Feedback

During the performance appraisal process employees are given feedback regarding their performance (Armstrong 2006b:521; Dresser & Associates 2008:9). Feedback typically leads to a decrease in errors and waste, improved efficiency, enhanced service delivery and quality, as well as increased employee motivation, commitment and a sense of ownership (Nickols 2007:12).

Feedback, to academics when appraised on a new job demand such as e-learning is of significant value. It can be argued that the feedback to the employee should be coupled with the discussion of a development plan and a motivational strategy. Further, providing a rationale for the consequent development and the end-result is imperative as it can support in increasing the employee's goal commitment (Li & Butler 2004:38; Locke & Latham 1990). The importance of feedback is also emphasised in section 2.3.2 as part of the goal-setting process.

• Career management

As the human resource performance appraisal process includes opportunity for training and development, it provides for discussion on career progression

42

opportunities (Nickols 2007:13; PSC 2000:iv;6; PSC 2010:5; Gilley et al 2002:60–62). As indicated, the developmental part of performance appraisal is relevant to this study, particularly to academics that are not e-ready. Career development can positively contribute in this regard and the e-readiness development of an academic can be included in a career management programme (De Wit 2010). The role of career management in this respect is discussed in chapter 3.

• Objective assessment

Nickols (2007:13) stresses the importance of uniform processes and criteria in the performance appraisal process to ensure objectivity as it is in any circumstance necessary to appraise an employee objectively (Mote 2011; Dresser & Associates 2008:7). Objectivity forms the basis for fair, valid and legally defensible rewards (financial and non-financial) pertaining to performance (Nickols 2007:13).

Employees that experience a lack of e-readiness, who will have to obtain a new skill, may easily be overwhelmed and resistant to the outcomes of the performance appraisal. Therefore, with the performance appraisal of an academic's e-readiness it is imperative to keep the appraisal, as well as the development plan objective with a view on the organisational objectives. The proposed framework for the e-readiness assessment of academics will further contribute to the objectivity of the performance appraisal (see chapter 7).

Legal protection

Performance appraisals offer legal protection against employee lawsuits for discrimination and unfair dismissal (Nickols 2007:13). The following core elements pertaining to performance management are emphasised by Landes (2006:29): Clear goals, expectations and responsibilities, support from management, a meaningful rationale, a well-defined performance process, guidance, feedback and development. In this regard a HEI can be safeguarded if need be, since proof can be shown of a performance appraisal that found an employee to not be e-ready and therefore justifies the required training and/or development expected of the

43

employee. It can be argued that the practice and implementation of the aspects outlined above will ensure a sound performance management process at HEIs.

2.2.2 Human resource performance appraisal

The performance appraisal targets of individuals are set within the performance management framework of organisational objectives and an organisational strategy (Amos 2009:9). The significance of human resource performance appraisal in the e-readiness assessment of academics necessitates a clear understanding of this process. According to Schuler (1981:211) *human resource performance appraisal* is "a formal structured system of measuring and evaluating an employee's job-related behaviours and outcomes to discover how and why the employee is presently performing on the job and how the employee can perform more effectively in future so that the employee, the organisation and society all benefit" (cf Aswathappa 2005:227).

In accordance with Amos' (2009:9) statement in the previous section, Thomson and Mabey (2001:189–190) also contend that that a performance appraisal takes place to determine whether an employee's performance coincides of what is expected in terms of the organisational framework, objectives and strategies. Employee evaluation during a performance appraisal process allows for changes and improvements in employee performance which in return may lead to a development plan (Thomson & Mabey 2001:189–190). Training, development and rewards are identified during a performance appraisal (Kuvaas 2006:504). Performance appraisal thus encompasses the assessment of employees' performance (evaluative part), as well as their potential for further development (Rademan & De Vos 2001:54; cf Grobler et al 2002:260; 266).

The concept and practice of development, indicated in the previous paragraph, is in particular relevant to this study, since the aim is to determine the level of e-readiness of an academic through performance appraisal to determine the need for and the extent of development required to improve the level of e-readiness. As alluded to in the previous section, also relevant to this study is that the academic, the organisation (the HEI) and society (the learners and the future work places) should all benefit

from employee development (Cardno 1995:118; Maurer et al 2002:432; Adam 2010). It can be argued that ultimately the increased e-readiness of an academic will not only enhance the skills and productivity of the employee, but will also benefit the learner which will receive an improved service, more opportunity to enhanced learning, and anytime, anywhere type of access to study material, tutoring and resources through e-learning. Further it enables the institution at large to provide quality world class teaching and learning through incorporating technology

Performance appraisal can be used for a wide range of purposes (Ivancevich 2004:257–258; Rees & Porter 2003:280; Steyn & Van Niekerk 2002:277; Redman & Wilkinson 2001:57; Amos 2009:1): To clarify and define performance expectations; to provide career counselling; succession planning; to improve individual, team and organisational performance; to identify training and development needs; to facilitate communications and involvement; recognition for effective work; to allocate financial rewards; to determine promotion; to motivate and control employees; to achieve cultural change; human resource and employment planning; legal compliance; and HRM research.

Of the abovementioned purposes of performance appraisal, most relevant to this study, in terms of the e-readiness of academics and the introduction of a new job requirement such as e-learning, are the following:

• The identification of training and development needs

It remain the responsibility of line managers to ensure that their subordinates are well-trained and have the competence to provide quality teaching and learning to students (Bartridge 2004b:1). Line managers at HEIs should, through the assessment of e-readiness during the performance appraisal, determine the training and development needs pertaining to e-learning. Line managers are further also responsible to see to it that employees' performance goals are aligned to organisational goals. Not only should training needs be identified, but time and opportunity for training should also be granted. Training and development of academics towards increased e-readiness are discussed in chapter 3.

• The facilitation of communications and involvement

According to Rademan and De Vos (2001:54), performance appraisal is regarded as one of the most significant tools of communication between managers and subordinates since it can either enhance or diminish the effects of other HRM activities such as training and development, talent management and retention. Landes (2006:29) supports the notion of Rademan and De Vos, indicating that the manner in which performance appraisals are practiced and communicated, will either positively or negatively impact on other HRM activities in the organisation. These authors' notions also hold truth for the performance appraisals of academics at HEIs as poor communication of the purpose and expectations of performance appraisals between line managers and academics are likely to stem academics negative towards their jobs. This negativity may very well result in training and development interventions, not having the desired results and even losing these employees to other organisations. It is thus of the utmost importance that performance expectations in terms of e-learning are clearly and regularly communicated to academics.

As indicated in the preceding section, clear goals and expectations will also enhance communication between managers and subordinates (Landes 2006:29). Cook and Jaggers (2005:9) support this notion by indicating that managers who communicate well with employees gain insight into employees' feelings. When employees are well-informed, they are likely to increase their performance since they realise the significance of their contribution to the organisation's performance and objectives (Cook & Jaggers 2005:9; Moller et al 2006:105–106; 110).

Further, research has found that employees are more positively inclined to the performance appraisal system when it provides opportunity for involvement and satisfy their needs (Latham & Locke 1979:68; Nykodym 1996:2; 4). When objectives and plans are talked about candidly, employees respond more favourably to the system (Gagné et al 2000:1843; Nykodym 1996:2;4; Landes 2007:29). It is imperative for managers and employees to mutually agree on the intended purpose, process and functions pertaining to performance appraisal.

46

• Motivation and control of employees

As motivation is regarded by researchers in this field (Saari & Judge 2004:396–397; Gilbert, De Winne & Sels 2010:5) as one of the most significant roles of line managers, it can be rightfully expected that this role takes precedence in the endeavour of changed job requirements, and in particular, if e-readiness assessment is introduced to the performance appraisals of academics. It is further important that academics understand their role in e-learning and be able to motivate themselves in terms of personal growth and development (Nickols 2007:12). It is also vital that employees interpret the worth and integrity of their jobs to successfully perform their e-learning duties. Section 2.3 elaborates on employee motivation.

• Financial or other rewards as incentive

Research (Corporate Leadership Council 2003:3) on best practice in terms of employee performance, employee retention, career management and talent management reveals that rewards and incentives play a significant role in enhancing these human resource management practices (UNDP 2006:4). The study does not focus specifically on rewards and incentives, but does make mention of it as it can be linked to employees' performance and motivation. In the discussion on motivation in section 2.3.1, reference will be made to the role of rewards and incentives.

Whereas the theoretical requirements and the purposes of performance appraisal, as outlined in scholarly literature, have been discussed above, chapter 6 discusses how these performance appraisal requirements and purposes are realised at the NWU. Chapter 6 therefore outlines the current performance management policies and practices followed by the NWU and also seek to determine whether these current policies and practices make provision for the assessment of the e-readiness of academics.

From the descriptions and objectives of performance management and performance appraisal in the preceding sections, two prominent aspects relevant to this study came forth: employee motivation and employee development. It is evident that

47

employee development plays a significant role in the eventual performance of an employee. It is also clear that the human resource performance appraisal process does not consist of an evaluative part only, but also a developmental part. When academics are not e-ready to fulfil their roles as online learning facilitators, e-leaning training and development become imperative. Chapter 3 therefore determines the role of training and development in increasing the e-readiness of academics.

As alluded to several times in the preceding sections, a significant aspect relating to employee performance is motivation. The next section explores the role of employee motivation in employee performance as well as the role of motivation for employees to embrace the new job demand of e-learning and use it optimally for enhanced learning.

2.3 THE ROLE OF EMPLOYEE MOTIVATION TOWARDS INCREASED PERFORMANCE

Research has shown that motivation has a significant impact, positively or negatively, on an employee's performance (Shadare & Hammed 2009:7-8; Oyedele 2010; Shadare & Hammed 2009:8; cf Cooke & Meyer 2007:1–2; Bruce & Pepitone 1999:118). Therefore, in an organisation's endeavour to increase employees' performance it is necessary to understand what specifically motivates employees to optimally perform. Motivation can be described as "the psychological feature that arouses an organism to action toward a desired goal; the reason for the action; that which gives purpose" and can include providing an incentive (Princeton University 2009). Robbins (2009:144) defines motivation as "the processes that accounts for an individual's intensity, direction and persistence of effort toward attaining a goal. Motivation can also be described as the reason or reasons for engaging in a particular behaviour, especially human behaviour (Pearsall 2001:1206; Shadare & Hammed 2009:8) or "the general desire or willingness of someone to do something" (Pearsall 2001:1206). For the purpose of this study the focus will be on employee motivation, namely the reason/s for employee engagement in particular activities, determined by their behaviour towards these activities (specifically the use of technology in teaching and learning).

Increasing the e-readiness of academics through employee development to become skilled online learning facilitators will fail if no reason or motivation exists for employees to become involved (Jordan 2006:1; Roper 2007; cf Chyung 2007:3). This section will therefore focus on a number of aspects identified in literature that proved to be successful in terms of employee motivation towards enhanced performance. For the purpose of this study it should also be kept in mind that line managers not only have to motivate employees in terms of performance, but they will also have to motivate academics to embrace a new job demand for which they are perhaps not ready or have reservations about. This makes it a challenging task for line managers. The literature study will therefore indicate which motivational strategies are most effective towards enhancing employee performance, particularly for employees who resist change. The following aspects will receive attention: financial and non-financial incentives and rewards; goal-setting; intrinsic motivation and extrinsic motivation and self-determination theory. The relation between motivation and technology adoption is also discussed by means of self-determination theory.

2.3.1 Financial and non-financial incentives and rewards

Callaghan, Brownlee, Brtek & Tosi (2003:2518–2519) indicate that incentives have both a direct and indirect effect on task performance (cf Stolovitch, Clark & Condly 2002:2). Incentives are, however, usually linked to goals, either implicitly or explicitly (Callaghan et al 2003:2518). When piece-rate incentives (a rate which an employee is paid for each item produced) (Encarta 2009) are used in combination with assigned goals, it leads to higher self-set goals (Jackson 2007:25; Stolovitch et al 2002:2–3). According to expectancy theory employees are motivated to perform when they believe their effort will lead to obtaining an appreciated reward/incentive (Shadare & Hammed 2009:10; Armstrong 2006b:259). Appropriate allocation of rewards is therefore likely to increase employee motivation and performance and will therefore also add to the successful achievement of organisational goals (Zhu 2007:90). Research by Callaghan et al (2003:2518–2519) also indicates that there is a relationship between incentives and higher self-efficacy as well as higher self-set goals^{4.}

A number of incentives can serve as motivators to performance, namely amongst other financial compensation, participation in decision-making, recognition. promotion, job enrichment, behaviour modification, organisational development, goal-setting and self-determination. These motivators can play a role in academics' e-learning performance and can be applied in training and development interventions to increase their e-readiness levels.

Money assists in attracting quality employees and retaining them (Mason & Watts 2009:1; Lee 2007:1-2). Furthermore, money is found to be most effective as motivator when bonuses offered to employees, are made dependent on the achievement of particular goals (Latham & Locke 1979:68; cf Madden 1997:411). Marshall and Harrison (2005:4) confirm this notion and indicate that financial incentives lead to better work performance. The use of financial incentives (merit pay, pay for performance, bonus) in the performance context, typically is based on the principles of reinforcement theory (Perry et al 2006:505). Skinner's (1969) reinforcement theory refers to the relationship between the desired behaviour (eg performance) and a motivational tool (eg merit pay) as explained in section 2.2.1.

Maslow's hierarchy of needs can be taken into consideration with employee motivation, where higher order needs and lower order needs are included in his hierarchy (Maslow 1946:3). Higher order esteem needs refer to the desire to stability, having a high valuation of oneself, self-respect, self-esteem, based on true capability, accomplishment and respect from others (Maslow 1946:33). Maslow (1946:33) places esteem needs in two categories: firstly, the need for strength, success, competence, confidence, autonomy and freedom; and secondly, the desire for status, prestige, recognition, attention, importance and appreciation. Fulfilment of self-esteem results in self-confidence, value, strength, ability, and competence for being constructive (Maslow 1946:34). The other higher order need on Maslow's hierarchy, self-actualisation, refers to the desire for self-fulfilment, the tendency to

⁴ Self-efficacy and goal-setting is discussed in the next section List of research project topics and materials become actualised in what one latently is and potentially can be, and to grow to be all one is able of becoming (Maslow 1946:34). Based on Maslow's needs hierarchy, it can be argued that academics, functioning within an intellectual, high performance environment, will favour rewards and incentives such as recognition, promotion and flexi time as they would arguably be driven by Maslow's higher order needs. Higher order needs are also associated with intrinsic motivation (Robbins 2009:145), implying that employees would not need external motivation such as financial incentives to perform.

McDonald, Harrison, Checkland, Campbell and Roland (2007:334) is of the opinion that professional performance is usually intrinsically motivated as professionals perform because they find an activity inherently enjoyable and rewarding; professional performance is usually not due to an external reward. Further, research found that "extremely imposed incentives" can weaken internal motivation (McDonald et al 2007:334). This is of significance as the conventional view of intrinsic motivation is that it is regarded as a significant attribute of excellence in professional practice (McDonald et al 2007:334). The determining factor in this study, however, will be whether an employee's position on the technology adoption cycle, preferred learning style and personal work profile pattern can determine the most effective manner in which to motivate the employee. This notion is explored in chapter 6 and the relation between incentives/rewards, learning styles and pace and adoption to technology will be illustrated in chapter 4. As indicated earlier in this section, a relationship can be observed between incentives and higher self-efficacy as well as higher self-set goals (Callaghan et al 2003:2518-2519). The next section discusses the theory on the influence of goal-setting on employee performance.

2.3.2 Goal-setting

The fathers of the goal-setting theory, Latham and Locke (2006:332), are of the opinion that "there is a linear relationship between the degree of goal difficulty to which a person is committed to attaining and that person's subsequent job performance", provided that the employee has the sufficient knowledge and ability (Locke & Latham 2002:706). Goal-setting therefore has the potential to enhance employee performance. Goal-setting, is not only an effective motivator, but may be

the key method by which the other incentives influence motivation (Latham & Locke 1979:68). Job enrichment will have for instance no effect on productivity unless employees in enriched jobs set higher and more specific goals (Latham & Locke 1979:68). Goal commitment, however, is conditional for goal-setting to result in enhanced performance (Locke, Latham & Erez 1988:23; Locke & Latham 1990:240; Locke & Latham 2002:707; Selden & Brewer 2000:535). It is further pointed out by Locke and Latham (2002:707) that goal commitment is most important when goals are difficult as difficult goals will call for high effort from employees. In this regard line managers should play a role in terms of motivation to encourage employee commitment to their goals by focusing them on the positive outcome and rationale of the development (Landes 2006:28; Moller et al 2006:105–106;110; Deci & Ryan 2000:227).

When employees are provided with a rationale/credible reason for a challenging goal it can assist in increasing goal commitment (Li & Butler 2004:38; Locke & Latham 1990:241). Support for this notion has also been indicated in section 2.2.1 and earlier in this section. A rationale or credible/clear explanation will not only enhance employee motivation, but it will ultimately also increase performance of employees which in return will eventually lead to organisational performance. The likelihood that well-defined goals and expectations lead to achievement of desired outcomes (Li & Butler 2004:38; Resnick 2007) was mentioned in section 2.2.1. This notion is supported by Armstrong & Baron (1998:568) in Amos (2009:9) when they agree that employees will do their best to achieve objectives if they know what is expected of them. One can argue that this calls for setting clear and unambiguous goals. It has also been stated that well-defined and clear goals and expectations will also enhance communication between managers and subordinates (Landes, 2006:29).

Literature reviews (Latham & Locke 2006:232; Li & Butler 2004:37; Callaghan et al 2003:2517; Perry et al 2006:509; Locke & Latham 1990; Wegge & Haslam 2005:400) indicate that *difficult* and *specific* goals result in higher performance as opposed to *easy* and *general* goals or no goals. Difficult and specific goals lead to increased rewards, greater satisfaction, increased personal effectiveness and a stronger commitment to the organisation (Li & Butler 2004:37; Callaghan et al 2003: 2517; Perry et al 2006:509; Locke & Latham 1990; Wegge & Haslam 2005:400;

Latham & Locke 2006:333). It is evident that employees are more likely to commit to goals and perform when they are *difficult* and *specific*. Managers should therefore ensure and assist employees in setting difficult and specific goals. Many employees, in particular those who lacks e-readiness, will in all probability already perceive a goal pertaining to e-learning as *difficult*. They can, however, be motivated towards goal accomplishment by making the difficult goal (obtaining the skill to effectively make use of technology in teaching and learning) *specific*. It can be argued that the specificity of a goal will also relate to a rationale and explanation as to why the obtainment of the goal is important.

It is evident that, based on goal-setting theory, difficult and specific goals, linked to a rationale provided pertaining to the expected outcome, will lead to enhanced performance. However, Perry et al (2006:509) propose that the associations between goal-setting and performance processes are moderated by many contextual aspects, such as task arrangement and an employee's profile (cf Latham & Locke 2006:332). This statement is in line with what is agued in this study: an academic's motivation towards goal-setting and goal commitment (to increase e-readiness), is related to the employee's e-profile, which is influenced by particular human factors. These human factors are discussed in the next chapter.

It is interesting to note that research on goal commitment has found that the same level of commitment and performance are reached when goals are assigned to employees by managers, as when employees take part in the setting of their goals or set their own goals (Latham 2007:112; Curtis 1994:41; cf Elston & Ginis 2004:500; Feltz, Short & Sullivan 2008:99–100; cf Lycette & Herniman 2008:27; Curtis 1994:41). Furthermore, assigned goals strongly influence personal goals of employees (Latham 2007:112). The only exception, where assigned goals do not contribute to better performance but actually lead to poorer performance, is when the assigned goals are given with brief, abrupt instructions and with the absence of a rationale (Curtis 1994:41; Locke & Latham 1990:241).

The implication of *assigned goals* to academics that is expected to act as online learning facilitators due to changed job requirements, should therefore according to literature, not be a restricting factor to motivation. Therefore, according to theory,

new challenging goals, such as e-learning, which is perceived to be *difficult* and are *assigned* to employees, should in fact enhance employee performance. However, in practice, resistance to new ideas, in various levels, is still experienced with some academics, as will be discussed in chapter 6 of this study. Though, it can be argued that with commitment from both the employee and the manager goal-setting can lead to increased performance. It is also necessary that a rationale is provided by the manager.

In the context of HEIs this implies that it might be necessary to pay more attention to the explanation/rationale behind the use of technology in teaching and learning. Once the employee has been plotted on the technology adoption cycle, it will be easier to understand to which extent it will be necessary to explain the rationale behind the changed job demands. People adopt to change, including technological change at different rates and some (eg *innovators* who are visionary employees) will most likely not need a rationale, but would jump at the opportunity to experiment with a new challenge. It could be accepted that for the majority of employees, however, an explanation need to be provided.

Locke and Latham (1990:241) further point out that *assigned* goals indirectly affect task performance through two psychological processes regarding self-efficacy and self-set goals. *Self-efficacy* refers to people's perception of how well they can perform a particular task and therefore high self-efficacy beliefs result in individuals who apply more effort and increasing determination and perseverance towards task accomplishment (Locke & Latham 1990:241; cf Garavalia & Gredler 2002:221). Self-efficacy does not only refer to the degree to which an employee's effort will pay off, but also to the employee's ability, adaptability, creativity and capacity to perform in the situational context in which he/she is (Locke & Latham 1990:241; cf Garavalia & Gredler 2002:221). Further it is important to note that employees with high self-efficacy set high goals (Latham & Locke 2006:332).

Callaghan et al (2003:2515–2517) explain that *self-set goals* are aims, set up by individuals for their own task accomplishment. A positive relation has been found between self-efficacy and self-set goals (Callaghan et al 2003:2515–2517). When people have high self-efficacy, they will set higher self-set than people with lower

self-efficacy (Locke & Latham 2002:706). People with high self-efficacy are also more committed to assigned goals than people with low self-efficacy (Locke & Latham 2002:706; Locke & Latham 1990:241). Assigned goals, furthermore, also favourably affects self-efficacy and self-set goals (Locke & Latham 1990:241). Therefore, higher assigned goals lead to higher self-set goals and higher self-efficacy. It is further apparent that there is a direct connection between self-efficacy and task performance – individuals with high self-efficacy beliefs perform better, irrespective of the self-set goal level (Callaghan et al 2003:2515–2517). Self-set goals are also linked directly to task performance (Callaghan et al 2003:2515–2517; Madden 1997:412).

Individuals who set effective goals and combine that with suitable learning strategies, as well as sufficiently assess the requirements of learning tasks, tend to perform at higher levels than those who do not (Garavalia & Gredler 2008:221). Challenging goals, irrespective whether it is assigned or self-set goals, are more likely to lead to performance (Locke & Latham 1990:242). One could therefore argue that a new challenging goal, such as acting as online learning facilitator could very likely lead to improved performance if communicated correctly to the employee and/or when the employee is made part of the goal-setting process. When an employee portrays a significant lack of e-readiness, it could be sensible to commence with moderate goals and gradually increase the difficulty level of goals to provide the employee the opportunity to mature into the role of online learning facilitator.

For a long-term effect on performance through goal-setting it is necessary to provide feedback (Wenschlag 2006:3; Locke & Latham 1990:241). Goals will have a minimal or no effect on performance without feedback, as feedback reveals progress in relation to goals (Locke & Latham 2002:708). Feedback does not usually affect the performance levels of employees who are already performing on or above standard (Selden & Brewer 2000:535). However, it considerably impact on the performance levels of under-achievers (Selden & Brewer 2000:535).

Employee performance is, however, also affected by other factors than goal-setting. The next section will focus on the role of intrinsic motivation and self-determination theory in performance.

2.3.3 Intrinsic motivation and extrinsic motivation

According to self-determination theory there are two overarching categories of motivation: intrinsic motivation, as outlined above; and extrinsic motivation (Vansteenkiste, Neyrinck, Niemiec, Soenens, De Witte & Van den Broeck 2007:251; Armstrong 2006b:254). *Intrinsic motivation* originates from within an individual and that influences people to behave in a particular way or to move in a particular direction (Armstrong 2006b:254; Shamir 1996:151). When an individual performs an activity for its own sake and enjoys performing it, intrinsic motivation is awakened (Gagné & Deci 2005:331; Vansteenkiste et al 2007:253). Intrinsic motivation refers to a condition in which an individual feels interest, pleasure and enthusiasm by taking on task-related activities (Gagné & Deci 2005:331; Vansteenkiste et al 2007:253; Ryan & Deci 2000:56). The individual is self-motivated towards task participation and when there is a natural desire to actualise, develop and grow (Vansteenkiste et al 2007:253; Bainbridge sa:1; Gagné & Deci 2005:331; cf Powell 2008:2).

Extrinsic motivation refers to an employee who does an activity to obtain an outcome that is independent from the activity itself, such as receiving a reward or to avoid punishment (Ryan & Deci 2000:55; Moller et al 2006:104; Vansteenkiste et al 2007:251). According to Ryan and Deci (2000:55) extrinsic motivation has traditionally been considered as a pallid and impoverished form of motivation as opposed to intrinsic motivation, even though extrinsic motivation is also powerful. Self-determination theory proposes that there are varied kinds of extrinsic motivation, of which some are indeed impoverished forms of motivation, but others lively, agentic states. Employees can for example perform extrinsically motivated activities with dislike, resistance, and lack of interest or, in which they are externally forced into action. On the other hand, employees can show an attitude of willingness that depicts an inner acceptance and readiness of the importance of a task, which reveals that the extrinsic goal is self-approved and thus adopted with a sense of willpower. (Ryan & Deci 2000:55.)

Professional performance is usually intrinsically motivated and professionals perform because they find an activity inherently enjoyable and rewarding (Ryan & Deci 2000:56). Professional performance is usually not due to an external reward (McDonald et al 2007:334). For a person that is intrinsically motivated the emphasis is on self-administered rewards (Callaghan et al 200:2519). Further, research found that "extremely imposed incentives" can weaken internal motivation (McDonald et al 2007:334; Deci & Ryan 2000:234) as the focus of intrinsic motivation is on the quality of working life (Armstrong 2006b:254). This is of significance as the conventional view of intrinsic motivation is that it is regarded as a significant attribute of excellence in professional practice (McDonald et al 2007:334). It is thus evident that employees' intrinsic motivation can be negatively influenced by external motivators. This calls for great caution from managers to prevent making this mistake with employees that are intrinsically motivated.

Callaghan et al (2003:2519–2520) explains that for intrinsic motivation to be awakened, it is imperative that a task provides challenges (cf Vansteenkiste et al 2007:253). Intrinsically motivated individuals will seek to master a task and have a need for competence which is fulfilled when the individual meets the challenge and achieves mastery (Callaghan et al 2003:2519–2520; Moller et al 2006:105). Intrinsic motivation relies on self-determination and a direct connection between outcomes such as creativity and innovation and the presence of intrinsic motivation has been revealed (Callaghan et al 2003: 2519–2520; Moller et al 2006:105; cf Vansteenkiste et al 2007:253). As intrinsic motivation is directly connected to self-determination, it is necessary to discuss self-determination and the role it can play in the motivation of academics pertaining to increasing their e-readiness levels.

2.3.4 Self-determination theory

Research of over 30 years on the topic found that intrinsic motivation leads to better perseverance, performance, and fulfilment in a diversity of tasks in various fields (eg education, behavioural health, organisational), as opposed to extrinsic motivation (Baard, Deci & Ryan 2004:2047; Black & Deci 2000; Deci, Connell & Ryan 1989; Williams, Grow, Freedman, Ryan & Deci 1996). According to self-determination theory people have basic psychological needs for *autonomy, competence*, and *relatedness* (Deci & Ryan 2000:227; Ryan & Deci 2000:7). Research has found that when these needs are satisfied, people are more likely to persist and exhibits better

qualitative performance on activities (Roca & Gagné 2008:1588). The three psychological needs can be described as follows (Roca & Gagné 2008:1588; cf Guay, Ratelle, Senécal, Larose & Deschênes 2006:237; Pugno 2008:1330–1331; Ntoumanis 2005:450; Liu, Wang, Tan, Koh & Ee 2009:139; Lynch 2010:1):

- *Autonomy* people's need to self-organise their actions, to pursue activities without restraint and feel the power of using their own will.
- Competence similar to the concept of self-efficacy people have a tendency to be effective when they perform an activity and in their general communications with and in their surroundings.
- Relatedness the need of an individual to feel allied to and supported by significant people in their lives, such as a manager, parents, teachers or teammates.

According to Gagné et al (2000:1843) self-determination should be viewed in the context that human beings have a basic need for autonomy (cf Chirkov, Ryan, Kim & Kaplan 2003:98), therefore autonomy is emphasised in this study. Autonomy support can be described as an interpersonal approach where the significant person/s in an employee's life offer alternatives/options, consider the person's viewpoints, provide suitable information and rationales for activities, and encourage the employee to be innovative (Guay et al 2006:238; Markland, Ryan, Tobun & Rollnick 2005:815-816). For this reason, controls such as rewards, deadlines and competition weaken task interest and the quality of performance when an employee is intrinsically motivated (Ryan & Deci 2000:59; Baard et al 2004:2048; Lam & Gurland 2008:1109). When people are coerced into doing something, they usually become less attracted to the task and will perform it only as long as there is some form of supervision (Gagné et al 2000:1843). This notion is supported by various authors (Roca & Gagné 2008:1588; Pugno 2008:1332; Vansteenkiste et al 2007:252-253), indicating that cognitive evaluation theory, which is a sub-theory of self-determination theory, has revealed that external controls (conditional rewards; deadlines; observation; evaluation; coercion) will most likely reduce employee's intrinsic motivation.

Autonomy support from management has been proven to relate to increased trust in an organisation, satisfaction, engagement and decreased stress (Deci & Ryan 2000:227; Markland et al 2005:816). This notion, once again, emphasise the importance of an employee being involved in the goal-setting process and not coerced. It also stresses the importance of providing a rationale of why a particular task is expected (as indicated in sections 2.2.1 and 2.3.2). Therefore, line managers should take cognisance of the importance of acknowledging negative feelings from employees pertaining to particular tasks and focus on providing a rationale for such a task. If necessary, incentives should be considered to serve as motivation. One can argue that employees with negative feelings towards tasks will need more motivation from superiors than employees who are positively inclined towards their tasks. This may very well be the case with academics that are confronted with a changed job demand such as e-learning. A rationale should be provided for both the use and purpose of e-learning and the subsequent development towards enhanced ereadiness. It can be argued that the rationale should be provided in such a manner and with such motivational skill that the employee wants to embrace the new job demand. This calls for skilled line managers, competent to differentiate between the various motivators for different employees and able to motivate them accordingly. This implies that it may be necessary for line managers to undergo training pertaining to employee motivation as all people do not have the natural skill to motivate.

The most constructive and innovative contribution of self-determination theory is that it, in addition to the abovementioned positive aspects, suggests that extrinsic motivation does not need to be an perpetually controlled form of motivation (Roca & Gagné 2008:1588). Self-determination theory proposes that extrinsic motivation can be internalised and can therefore become autonomously regulated. Internalisation is described as "taking in values, goals, and structures as one's own, so that a behaviour becomes internally regulated as opposed to regulated by external factors like rewards and punishments" (Roca & Gagné 2008:1588). Self-determination theory denotes that the level of internalisation will determine the extent to which extrinsic motivation is autonomous (Niemiec, Lynch, Vansteenkiste, Bernstein, Deci & Ryan 2006:762; Roca & Gagné 2008:1588; cf Chrirkov et al 2003:99).

59

One of the ways in which employees can be assisted to accept change is to introduce participative management programmes which allows for participative decision-making and effective communication. The factors in self-determination theory, posited to assist in this regard, are the following (Gagné et al 2000:1845; Ntoumanis 2005:444):

- Participation in decision-making is parallel to giving some control and choice in how to reach organisational goals.
- Communication and empathy are parallel to providing a rationale and acknowledge feelings.

Managers should consider the abovementioned when assigning new goals pertaining to changed job demands to employees. When employees participate in the decision-making/goal-setting, they might be more incurred to accept the changing work circumstances. It will also be necessary to provide more opportunity and support to the employee to adapt to the changed situation. It can be argued that employees will take ownership if they have internalised the organisation's values, goals and structures as their own, which will positively affect their behaviour, including behaviour towards a new job demand. Again, the importance of a rationale/explanation is emphasised.

In the abovementioned paragraphs, autonomy, as central concept to selfdetermination, was emphasised. One can argue that a lack of autonomy support may be a significant reason for resistance to the use of technology in teaching and learning. Some academics may experience the new job demand as being forced on them and may feel that they did not have the option of choice. As indicated above, taking part in decision-making and having a choice regarding how to perform a task (gradually introducing it or receiving training as a prerequisite) may enhance the motivation of an academic to perform the task.

Self-determination theory is also related to technology adoption of people. As the motivation of academics to become e-ready is significantly important to this study,



the next section will discuss the role of the Technology Adoption Model (TAM) and self-determination theory in this regard.

2.3.5 Motivation and technology adoption

People as users of technology determine the continuation of the use and purpose of a specific technology by their acceptance thereof, or lack of acceptance (Roca & Gagné 2008:1586). A number of theoretical models can be used to explain and predict technology users' attitudes, acceptance and adoption of technology, of which the most well-known and most-used is the Technology Acceptance Model (TAM) (Roca & Gagné 2008:1586; Malhotra & Galetta 1999:1). According to the TAM the most important determinants of a person's acceptance of the use of technology are *perceived usefulness* – "the degree to which a person believes that using a particular system would enhance his/her job performance" (Davis 1989:320 in Roca & Gagné 2008:1586; cf Adams, Nelson & Todd 1992:227–228; Shen, Lin & Huang 2006:272); and *perceived ease of use* – "the degree to which a person believes that using a particular system would be free of physical and mental effort" (Davis 1989:320 in Roca & Gagné 2008:1587; cf Adams et al 1992:227–228; Shen et al 2006:272).

It is therefore evident that academics will more easily accept the changed job demand (the use of technology in teaching and learning) if they are convinced that it will enhance their job performance, career or the performance of a student (*perceived usefulness*). In other words a rationale/explanation for the importance/relevance of the use of technology should be provided as indicated in (as indicated in sections 2.2.1, 2.3.2 and 2.3.1). It is also evident that academics will be more lenient to using technology in teaching and learning if they believe it will not take too much of a high toll on them (addition to workload; inconvenience; mastering a new skill). This indicates that resistance to the use of e-learning may be experienced by academics that portray a lack of e-readiness or are ignorant towards the role and purpose of e-learning (*perceived ease of use*).

Perceived enjoyment – "the extent to which the activity of using a computer system is perceived to be personally enjoyable in its own right aside from the instrumental

value of the technology" (Sun & Zhang 2006:620; Davis et al 1992 in Roca and Gagné 2008:1587; Dickinger, Arami & Meyer 2008:5) was later included in the TAM as an intrinsic motivation and *perceived usefulness* was classified as an extrinsic motivation. Enjoyment and technology self-efficacy furthermore play significant functions in determining a technology user's behavioural intent and real usage (Roca & Gagné 2008:1587). Since *perceived usefulness* and *perceived enjoyment* are motivational factors, it can be accepted that antecedent variables (organisational factors such as supervisors support, top management support and work environment) are likely to have an influence on the type of motivation that e-learning users are to be expected to adopt (Roca & Gagné 2008:1587). One can aim to determine how the different types of motivation will influence intentions to continue using e-learning; in other words, one can ask the question whether the abovementioned antecedents influence technology users' acceptance of technology (Roca & Gagné 2008:1587; Sun & Zhang 2006:620–621).

It can be argued that the *perceived usefulness* of e-learning will play a role in the extent to which academics will resist it. It is most likely that academics will be more inclined to accepting the new job demand if they perceive it to be useful to learners and/or academics. As for *ease of use*, the more technologically skilled academics, the easier they will find he use of the e-learning platform. The position of an employee on the technology adoption cycle (indicating pace and style to technology adoption) will also determine how open an employee will be to realise the perceived usefulness of an e-learning platform. The connection between an employee's position on the technology adoption cycle and the perceived usefulness of technology will be indicated in the next chapter.

As for *perceived playfulness*, it can be assumed that employees who like to explore new ideas and technologies will find the new job demand of using technology in teaching and learning exciting, irrespective of whether the person is skilled in using an e-learning platform. This type of employee will probably learn by "playing around" in the e-learning platform. The connection between perceived playfulness and an employees' position on the technology adoption cycle will also be explained in the next chapter.

62

Self-determination theory, discussed in the previous section, also relates to technology adoption. Perceived autonomy, perceived competence and perceived relatedness (people's psychological needs, according to self-determination theory) can be seen as determinants of perceived usefulness, perceived playfulness and perceived ease of use. Diagram 2.1 below indicates this connection:



Diagram 2.1: Self-determination theory and the adoption to technology

Source: Roca & Gagné (2008:1585–1604)

With the abovementioned figure Roca and Gagné (2008:1590–1593) illustrate the following:

 Perceived autonomy support has a positive effect on perceived usefulness and perceived playfulness

When academics perceive their managers and organisations to be supportive and receive explanations as to the importance of the goals and the necessary resources to carry out their e-learning tasks, they tend to more easily notice the perceived usefulness of the e-learning platform. Autonomy support may also make employees more inclined to "try out" and "play around" within the elearning platform.
Perceived competence has a positive effect on perceived usefulness, perceived playfulness and perceived ease of use of the e-learning platform

When academics feel competent to operate within an e-learning platform, they will experience it easier to use as opposed to an academic that is not e-ready. Competence will also create the perception of usefulness since the employee realises what the intended purpose of the e-learning platform is. The academic may furthermore be more inclined to "play around" within the system, since their perceived competence make them feeling comfortable and confident in doing so.

• Perceived relatedness has a positive effect on perceived usefulness and perceived playfulness of the e-learning platform

Once an academic can relate to the principle and purpose of e-learning and receives the necessary support, it will be easier to realise the usability thereof. Also, if academics can relate to the e-learning platform and its tools they will probably be more inclined to "play around" within it.

• Perceived usefulness has a positive effect on behavioural intention to use the e-learning platform

When academics perceive the e-learning platform to be useful, it will positively affect their behaviour towards using the e-learning platform.

• Perceived ease of use has a positive effect on perceived usefulness and on behavioural intention to use the e-learning platform

Once academics feel comfortable/confident using the e-learning platform, they will be more likely to realise its usefulness and will furthermore be positively inclined to use the platform and continue using it.

 Perceived playfulness has a positive effect on perceived usefulness, perceived ease of use and on behavioural intention to use the e-learning platform

When academics perceive the e-learning platform to be exciting and innovative, they will be more likely to recognise its usefulness and will find it easier to use. They will further be more inclined to continue using it.

In a study done by Roca and Gagné (2008:1605), with employees of four international agencies of the United Nations, results indicate the following:

- Users of technology are more willing and motivated to continue using technology when they experience feelings of autonomy and competence. (Autonomy and competence have an influence on their intrinsic and extrinsic motivation, perceived usefulness and perceived playfulness, which in turn have an effect on their intention to continue using the technology.)
- When employees feel related to and supported (perceived relatedness) by coworkers they will use the technology system simply for the enjoyment they get from it.

The abovementioned study was one of the first to reveal that positive outcomes (increased learning, enhanced performance, and well-being) of employees are more associated with an autonomy-supportive motivating management style than with a controlling management style (Roca & Gagné 2008:1605; cf Niemiec et al 2006:773; cf Perryer & Jordan 2005:381). The same study revealed that employees' perceptions of the extent to which their organisations and managers support autonomy, will predict the perceived usefulness and perceived playfulness, which subsequently indicate their intentions to use technology (Roca & Gagné 2008:1605). It can be argued that when senior managers and line managers support employee autonomy, it will increase both the intrinsic and extrinsic motivation of technology users' to use the technology. This will be due to the support in terms of training, resources, and motivation, which enable employees to perceive the e-learning platform to be constructive and helpful for the achievement of significant goals and subsequently they take pleasure in using it.

Further, when employees perceive their organisations to be autonomy supportive, it will also have an indirect effect on continuance intention, since technology users will then be inclined to stick with their behaviour of using the e-learning platform (Roca & Gagné 2008:1597). Self-determination studies support this notion through results that have revealed that when circumstances support the autonomy contentment, employees are more likely to uphold their engagement in different types of behaviours (Roca & Gagné 2008:1597; cf Kuvaas 2009:40).

Self-determination theory, in which autonomy is central, makes provision for intrinsic and extrinsic motivation. As with goal-setting, self-determination theory also places a focus on providing employees with a rationale and support for expectations and refraining from coercion. It is apparent that motivation, based on self-determination theory can be applied to not only enhance performance, but to also increase technology adoption of employees. It is clear that if there is a direct connection between autonomy and support, employees are likely to be motivated to try out new technologies/innovations, which makes autonomy support a significant factor for line managers of academics who lacks e-readiness. It is evidently clear that the principles of self-determination theory can be used to support the understanding and development of e-learning principles and use as employees are motivated to perform. Thus, the importance of motivation in employee performance should not be underestimated or neglected.

2.4 CONCLUSION

The new job demand of e-learning necessitated the e-readiness of academics to act as online learning facilitators. As many academics portray a lack of e-readiness and is therefore not ready to take on their online duties, training and development opportunities should be used to skill employees in this regard. However, before training and development can take place, employees' e-readiness needs to be assessed to determine their level of e-readiness and the subsequent personal development plan from which training and development interventions will flow. The study argues for the e-readiness assessment of academics during performance appraisals. The study further argues that key human factors such as personal work profile patterns, preferred learning style and pace and style of technology adoption should be included in the e-readiness assessments during performance appraisals. The next chapter discusses these key human factors.

Goals for performance should be set. Managers have a responsibility in this regard to ensure that employees' goals are aligned to the organisation's goals and objectives. Managers should also identify training and development needs during performance appraisal and provide opportunity for employees to attend training and development sessions. Aspects such as clear goals, expectations and responsibilities, support from management, a meaningful rationale, a well-defined performance process, guidance, feedback and development as core elements of performance management should be adhered to. These aspects also relate to employee motivation.

A direct relation exists between employee performance and motivation. The significance of motivating factors such as incentives, goal-setting, intrinsic and extrinsic motivation and self-determination has been outlined. It has been indicated that a lack of motivation can contribute to resistance to a new job requirement; therefore necessitating line managers apply motivational strategies. A lack of sufficient employee motivation can contribute to resistance to a new job demand and a decline in performance.

Goal-setting proved to be one of the best motivational strategies as it leads to enhanced performance, provided that employees remain committed to their goals. Both self-set and assigned goals can lead to increased performance and both are positively connected to self-efficacy, which in return leads to increased performance. Further, difficult and specific goals are more likely to lead to performance than easy goals. Goal-setting should be done in conjunction of line managers, providing clear expectations and a rationale for the changed job and how it fits into the organisational strategy and objectives.

As the human resource performance appraisal process comprise of not only the assessment part, but also a developmental part, the next chapter will give attention to employee development. Specific attention is given to the role of employee

development in increasing the level of e-readiness of academics. The role of career management and career management programmes in e-learning training are also discussed.

CHAPTER 3

EMPLOYEE TRAINING AND DEVELOPMENT

3.1 INTRODUCTION

During the human resource performance appraisals stage, as discussed in chapter 2, managers are able to identify employees' training and development needs. These training and development needs require that managers facilitate the interventions through which employees are able to receive such training and development to mitigate identified performance gaps. This chapter outlines and discusses the training and development interventions that precede performance appraisals of academics.

The training and development of academics in the use of e-learning is viewed as a necessity to meet changed job requirements such as the incorporation of technology in teaching and learning. Therefore, this chapter argues that the results of an assessment of the e-readiness of academics should be followed by a personal development plan that makes provision for uniquely structured training and development interventions for each academic, depending on the results of an assessment. This chapter discusses how training and development can be applied to enhance the e-readiness of academics and how it links to career management, career development and talent management to optimally achieve increased levels of e-readiness and subsequent performance with regard to the use of technology in teaching and learning. It further discusses the academic as adult learner and the unique characteristics of adult learners that should be kept in mind with e-learning training and development of academics. The chapter also pays attention to the pivotal role of the line manager in employee training and development.

3.2 EMPLOYEE TRAINING AND DEVELOPMENT

This section explains the meaning and purpose of employee training and development. It further discusses the link between employee development and

employee performance, as well as its role to develop academics for the use of technology in teaching and learning, thus increasing their levels of e-readiness.

3.2.1 Training and development

Training and development within organisations is meant to ensure that employees perform their tasks can effectively and efficiently perform their newly assigned job requirement (Bartridge 2004a:2; Ogrean, Herciu & Belascu 2009:115). A distinction should however be made between training and development. Training refers to a planned action that is intended to convey information and/or directions to enhance an employee's performance or to assist an employee in attaining a necessary level of knowledge or skill (Armstrong 2006b:535). Training is work-based learning with which employees systematically develop knowledge and skills that are, required to perform a specific task or job effectively (Bartridge 2004a:2; Tatum 2010; Bailey, Hughes & Moore 2004:3). Training can also be defined as the act of teaching someone a skill or kind of behaviour (Pearsall et al 2001:1966). Therefore training has the attainment of skills as an end result.

De Cenzo and Robbins (1999:227) in Babaita (2010:281) describe training as a learning experience that has as objective to improve the ability of an individual to perform an assigned task through a relative permanent change. Thus learning should take place during training. Kim (1993) in Armstrong (2006b:549) defines *learning* as the process through which an employee's ability to take action and move forward is enlarged. Reynolds et al (2002) in Armstrong (2006b:549) makes a distinction between *learning* and *training*: "Learning is the process by which a person acquires new knowledge, skills and capabilities whereas training is one of several responses an organization can take to promote learning". This paragraph shows that training encourages learning to take place during a training intervention.

According to Armstrong (2006b:535) development is defined as the growth and expansion of an employee's capability and potential, obtained through a *learning* and educational experience. Through the *learning* experience, *development* takes place. Learning is a constant process that not only develops current abilities "but also leads" to the development of the skills, knowledge and attitudes that prepare people for List of research project topics and materials

enlarged or higher-level responsibilities in the future" (Armstrong 2006b:560). Bacal and Associates (2008) argue that the *development* process encompasses not only *training* but is more inclusive than *training*: When an experienced employee guides or assists a new employee to perform a certain task, it is regarded as employee *development* (cf Comsats 2010: 9; cf McCrimmon 2007); when an employee is continuously coached by a manager, it can also be regarded as employee *development*. *Development* therefore is a broader term that includes training as one of its techniques for encouraging learning in the workplace (Bacal & Associates 2008).

With the *development* process, learning takes place continuously by means of different interventions, of which training is one. The development process entails that more than one training intervention may be necessary and that training should not only be focused on obtaining a technical skill. If needs be, academics should be coached and mentored as to best practices on online teaching and learning to support their *development. Training* for a skill to teach online should thus be part of a bigger development programme.

It is thus evident that both *training* and *development* have *learning* as end-result. With *training* the focus is on a planned action to obtain *learning* in the form of a particular work-related skill or competency (Bartridge 2004a:2). The term *competency-based training* can also be used in this regard. *Competency-based training* refers to training to perform a new job demand successfully with the new skills and competencies acquired (Bartridge 2004a:2; Tatum 2010; Bailey et al 2004:3). Academics will therefore receive *competency-based training* to obtain a skill to successfully perform the new job demand of e-learning. Thus, a planned and structured *training* intervention can address the technical skill of e-learning and improve the e-readiness of an academic. Therefore the ability to teach online, in this respect, refers to a competence that academics should obtain through training.

Competence can be defined as a combination of attentiveness, skills, knowledge, attitude, behaviour and approach that makes it possible for an employee to perform a particular job satisfactory to meet the required performance standards (DPSA 2008:6; Vathanopas & Thai-ngam 2007:49; Tatum 2010; Bartridge 2004a:2;

UNESCAP, 2009:5; cf Waterhouse 2008; cf Shellabear 2002:1-2; Wright, Dunford, & Snell 2001:712). To be able to be competent in performing a particular task an employee needs to have the ability to meet, or exceed, the job's requirements (Gallagher 2003). This implies that the employee should produce the necessary outputs of the job at an expected level of quality (Gallagher 2003). Competency is essentially about performance, based on both the employee's behavioural and technical ability (Amstrong 2006b:159). Rankin (2002) in Armstrong (2006b:159) explains that competencies symbolise the language of performance. "They can articulate both the expected outcomes from an individual's efforts and the manner in which these activities are carried out. Because everyone in the organization can learn to speak this language, competencies provide a common, universally understood means of describing expected performance in many different contexts" (Rankin 2002 in Armstrong 2006b:159). This statement also holds truth in the higher education environment where a specific competency, such as the ability of an academic to act as an online learning facilitator, will increase the performance of the academic.

Competency-based training therefore needs to focus on providing and enhancing the knowledge and skills of employees to perform for a particular task (Bartridge 2004a:2; cf Shellabear 2002:1–2) and is used to enhance an employee's present job performance, to prepare employees for changing job requirements and must complement the introduction of new tools or technology in the work environment (Vathanopas & Thai-ngam 2007:47). These competencies are derived from job requirements and which implies that line managers should play a pivotal role in the development and implementation of *competency-based training* interventions (Bartridge 2004a:1).

As argued by Shellabear (2002:4) competency-based training in many organisations has a propensity to focus on providing the job-related skills and knowledge (cf Rycus & Hughes 2000:1). Therefore the focus is placed on job requirements and the employer's expectations, rather than on the learning process itself (Shellabear 2002:4). The focus on job-related skills and knowledge has resulted in competency-based training being criticised for failing to attain competences required to support a *learning organisation* (Shellabear 2002:4). Garvin (1993) in Armstrong (2006b:543)

explains that *learning* is: A critical element in a learning organisation if an organisation wants to continue to exist; that learning at operational, policy and strategic levels have to be conscious, continuous and integrated; and that management is responsible to provide a climate, conducive to all employees to constantly learn.

A *learning organisation* can also be described as an organisation that makes learning possible to all its employees and constantly transforms itself (Pedler et al 1991 in Armstrong 2006b:543; cf Smith 2001). It can therefore be argued that an organisation that transforms and renews itself, through innovations such as the increased use of e-learning, is continuously learning and adapting. Continuous learning in an organisation implies that the individual and organisational capacity will be increased over time. Senge (1990:3) argues that learning organisations are: "...organizations where people continually expand their capacity to create the results they truly desire, where new and expansive patterns of thinking are nurtured, where collective aspiration is set free, and where people are continually learning to see the whole together". In the context of HEIs both definitions carries weight as the introduction of e-learning as new job requirement compels HEIs to continue to learn, transform and develop new thinking patterns.

It can be argued that in the global knowledge economy organisations cannot afford to not be learning continuously and should embrace the concept of lifelong learning. Given the expectations of the knowledge economy, lifelong learning is important for an employee to manage and be comfortable with change in the workplace (McLeish 2002:5). Basic computer skills and a willingness to retrain and adapt to rapid technological changes are essential requirements in the 21st century's work environment (McLeish 2002:5). The traditional "one time" learning paradigm, for example the attainment of a first degree before entering the work environment, is no longer sufficient for a knowledge workforce (Amirault & Visser 2009:71; Ramsden 2008:8; cf Markkula 2006:15). The World Bank Report on Lifelong Learning in the Global Knowledge Economy (2003:xiii) explains that "the emergence of the global knowledge economy⁵ has put a premium on learning throughout the world". HEIs

⁵ The knowledge economy includes bringing together powerful computers and well-educated people in order to meet the growing demand for knowledge-based goods and services (Brinkley 2006:3).

constantly need to adapt to environmental changes and demands and therefore need to continue learning and transforming themselves. Some of these changes and demands relevant to this study, as well as the role of HEIs in the knowledge economy are outlined in chapter 5.

This section discussed the concepts of training and development with particular reference to the role of both concepts in learning. It has been determined that training, specifically competency-based training, can be provided to academics to learn the technical skill of e-learning and that a continuous development process should take place for academics to constantly grow in the new job demand and become excellent in online teaching and learning. The development process also entails that academics should continuously adjust to new circumstances and learn how to successfully deal with the new job requirements if they want to remain relevant in terms of teaching and learning in the 21st century. Teaching and learning in the 21st century is elaborated upon in chapter 5.

It is clear that more than one training and development intervention will be necessary for academics to remain updated and skilled in terms of online teaching and learning. The next section will pay attention as to how the training and development interventions to enhance the e-readiness of academics can be supported by employee development.

3.2.2 Employee development and employee performance

As discussed in the previous section, development is the growth and expansion of an employee's capability and potential, obtained through a learning experience. This learning experience can take place through various interventions. Therefore, *employee development* encompasses "an integrated set of planned programmes and/or courses, offered over a period of time, usually in various phases and is aimed at assuring that all employees possess the competence required to optimally perform their duties (Jacobs & Washington 2003:344; cf Ogrean et al 2009:117). Employee development is thus not an activity or event but a process consisting of interrelated programmes/courses/training interventions. Rademan and De Vos (2001:54) point out that employee development should logically flow from an employee's performance appraisal from which an employee's development needs are identified (cf Grobler et al 2002:260;266) and can be viewed as a joint, continuous effort from both the employee and the organisation to improve the employee's knowledge, skills, and abilities (Wellins, Smith & Erker 2010:2; University of Minnesota sa:1; Brown 2010). Hence employee development is a key aspect of human resource performance appraisal (Lee & Bruvold 2003:983; Heathfield 2010a), as indicated in preceding sections. Although the line manager has certain responsibilities pertaining to employee development, one should keep in mind that the employee also has a responsibility to take ownership of his or her development for growth (Trinka 2009:1). A number of initiatives to be considered by employees in their own development are: seeking a variety of assignments, tackling tough problems, asking for feedback, identifying goals for new skills and abilities and looking for ways to meet those goals (Gilley et al 2002:61; 68; Trinka 2009:1; University of Minnesota sa:1).

For the purpose of this study *employee development* will include competency-based training to obtain knowledge and a particular skill (the ability to use technology in teaching), to enhance an academic's present job performance, prepare the employee for changing job demands (such as the increased use of technology in teaching and learning) in order to become a skilled and competent online learning facilitator, and to introduce new tools or technology in the work environment. It can be argued that employee development should further encompass career development and coaching and mentoring with a purpose of not only learning a new skill, but to also enhance the motivation of the employee for the new job demand, expand the employee's knowledge on teaching and learning strategies and to accomplish organisational goals. Employee development should flow from the performance appraisal of an employee and should be implemented in various phases as part of a development programme.

Lee and Bruvold (2003:98) go on to explain that an investment in employee development encompasses providing employees with new knowledge and skills and enabling employees to expect and be ready for new job requirements (Gilley et al 2002:60; cf Hansen & Hansen 2010). The more an organisation empowers an employee, the more the employee is able to contribute to the achievement of organisational goals (Sieber 2008:4). Employee development is not only an

important aspect of human resource performance appraisal, but also an important component of strategic human resource management. Employee development is important in strategic human resource management in that it has as aim to give priority to developing strategies and actions that will improve the skills and competencies of employees (Carpenter, Bauer & Erdogan 2009; Lee & Bruvold 2003:983). Therefore, employee development should be in line with the strategic goals of the organisation (Mayo 2000:529; Šiugždinienė 2008:33; cf Kaplan & Norton 1998, in Moolman 2007:50).

Employee development can be one of the best investments that an organisation can make, particularly when it is integrated to the organisation's operational strategy (Oberstein & Alleman 2003 in Moolman 2007:50). It is furthermore evident from the literature that employee development is regarded by various authors (cf Oberstein & Alleman 2003 in Moolman 2007:50; McNamara 2010; Papasolomou-Doukakis 2002:62) as an integral contributing part to the success of an organisation. Not only will the employee benefit from the development intervention/s, but the organisation will also benefit in terms of increased productivity and output. Likewise, one can argue that the development of an academic's online teaching skills will benefit not only the employee (increased and more comfortable use of the e-learning platform) and the organisation (increased productivity and output), it will also benefit the learner (the client) with increased service delivery and an enhanced learning experience.

3.2.3 The need for employee development for e-learning

This section provides reasons for the need of employee development for e-learning. As indicated in the previous chapter, HEIs have mostly realised the value of teaching and learning with technology (Elgort 2005:182; EC 2005:7; cf Rossiter 2006:iii; Njenga & Fourie 2010:199; Kanuka 2006:1; Kim & Bonk 2006:22), but still face the challenge of motivating all academics into actively participating in e-learning (Le Roux 2009; Focus Group 2006). In many instances where academics are willing to make use of e-learning, it is not necessarily used as an interactive tool and not to its fullest extent, which limits the benefits of e-learning (Le Roux 2009). One of the ways in which the interactive and optimum use of technology in teaching and learning can

be enhanced is by providing employees with the skills, of not only how to use an elearning platform, but also how to use it for maximum benefit and results as a learning tool, thus the need for employee development.

In research done by Elgort (2005:184) on e-learning adoption at Australian universities, she points out that it should be taken into account that academics sometimes adapt to e-learning due to a perceived need, such as learner pressure to make use of e-learning. If academics do not realise the need for e-learning themselves, the role of people responsible for employee development is to act as change agents and to assist them in this regard through creating awareness in academics about a wider range of strengths, weaknesses, potentials, and strategies of e-learning and make it possible for them to construct better e-learning environments (Elgort 2005:184).

In a report on research conducted for the University of Bremen, Atwell (2004:61) reveals that employee development and training is essential for e-learning to be successful and sustainable. Twigg (2001:5) points out that as long as traditional approaches of teaching and learning are used online, and learners are treated as if they are all the same, e-learning will make no significant difference. In an article published in the Journal of Technology and Teacher Education, Bradshaw (2002) asserts that employee development can be significantly helpful and appears to be a useful strategy for change in education when new programmes or new job demands such as, the incorporation of technology in teaching and learning are introduced. In his research Bradshaw (2002) further points out that it is widely accepted that educators must have increased knowledge and expand their skills and attitudes to teach learners about technology or to successfully integrate technology into their teaching and learning (cf JISC 2009:8). However, employee development strategies are not drafted and implemented in a short period of time, but must rather expand over time (Jacobs & Washington 2003:344), deal with requirements and fears of academics, and positively influence learner learning to obtain a lasting effects (Bradshaw 2002).

According to Elgort (2005:184) another reason why e-learning has not reached its full potential as an educational innovation is that the e-learning innovation is

approached form two aspects: the aspect of technology and the aspect of pedagogy (or teaching and learning). It appears that the adoption of e-learning technologies, in particular LMSs, is in more advanced adoption stages than the teaching and learning innovation (Elgort 2005:184). Rogers (1995:225-226) in Elgort (2005:184), also points out that past experiences can result in misadoption of technology. The use of traditional teaching and learning approaches in the context of e-learning is a good example of a past experience that cause the use of e-learning to be thwarted (Elgort 2005:184). This is why the roots of the problems with e-learning adoption in HEIs can predominantly be related to teaching and learning processes, rather than to the use of technology per se (Elgort 2005:184). Further, outcomes-based education (OBE) principles call for a changed role of both the learner and the teacher where learners must have more independence and must control their own learning events (Rautenbach 2007:16). The teacher is no longer a traditional teacher (someone who provides knowledge), but a facilitator of learning (someone who enables learners) (Rautenbach 2007:16). In this regard, employee development programmes can act as change agents in assisting academics to make the mind shift about their beliefs in terms of teaching and teaching practice (Pebble, Hargrave, Leach, Naidoo, Suddaby & Zepke 2005:48).

As mentioned in chapter 1, research done by various experts in the field of teaching and learning (Ramsden 2003:106; Milliken & Barnes 2002:225; Smith 2003; Matei, Bernau, Heyworth, Pohl & Wright 2007:8) revealed that people shape their personal theories regarding teaching and learning early on in their lives and carry it over from generation to generation. Elgort (2005:184) supports this research by stating that these teaching and learning theories are mainly implicit and therefore people do not simply alter them when they become academics themselves. More authors on teaching and learning (Thomas & Pederson 2003:319) add that it is commonly accepted in the educational milieu that a person usually teaches in the manner in which a person is taught and that teachers'/academics' principles, approaches, attitudes, and practices is likely to be connected to prior experiences. Teachers use good teachers' practices and approaches to mould and develop their own images as teachers (Elgort 2005:184; Smith 2001). Furthermore, academics use ICT tools only if these tools are aligned with their own philosophy of teaching and learning (Robertson 2004 in Elgort 2005:184). Based on her research done on e-learning adoption at Australian universities, Elgort (2005:184) explains that academics also usually continue to teach in the same way they always have, whether their teaching is in the context of e-learning or not. She further found that if an academic therefore believes in the approach of conveying information, he/she will use e-learning to facilitate this manner of learning, and any tools on the e-learning platform that are not aligned with this approach will be either ignored or misrepresented (Elgort 2005:184). The Manager: IT Support at the Potchefstroom campus of the NWU indicated that conveying or distributing of information is one of the most common ways in which academics "practice" e-learning (Le Roux 2009). However, to use technology effectively in teaching and learning, a paradigm shift from traditional teaching and learning is required, in particular with regard to the roles of teacher and learner (Hase & Kenyon 2001:1).

The words of Eble (1988:9) come to mind in this regard: "Learning and teaching are constantly interchanging activities. One learns by teaching; one cannot teach except by constantly learning". This statement presupposes continuous learning and implies that through the constant process of learning, academics will realise, in order to provide the best possible quality teaching and learning to 21st century learners, they need to adapt their teaching and learning strategies. Traditional teaching and learning methods are unlikely to appeal to Generation Y learners or enhance their learning. Further, the use an e-learning platform as a distribution mechanism (distributing notes, assignments, power point presentations, announcements, etc) and not as an interactive tool (Le Roux 2009), will not be viable in the long term. More creative ways of applying technology in teaching and learning should be introduced. This notion is supported by Kim and Bonk (2006:22) which state that "bored learners are dropping out of online classes while pleading for richer and more engaging online learning experiences". This situation calls for well-skilled online learning facilitators who are willing to adapt their teaching and learning approaches to accommodate today's learners. Therefore, for e-learning to realise its potential as educational innovation, academics need to have the ability and skill to construct environments that are useful to facilitate learning (Elgort 2005:184; cf Blake 2009). In this regard the constructivist epistemological approach may be relevant. The constructivist approach is discussed later in this section.

Rossiter (2006:23), an expert in the field of education, identifies another factor that also impacts the effectiveness of e-learning as an educational innovation – the use of e-learning is presently, for the most part, based on traditional views of teaching and learning (cf Kim & Bonk 2006). In these traditional views the dominant pedagogy is a content-centred or teacher-centred approach which is based on the delivery of information to learners (Jonassen 1991:28; Hanley 1994:3; Kinchin 2004:302). Thus, transmission of information takes place without a high level of learner contribution and interactivity (Le Roux 2009). In this respect, technologies that facilitate one-way distribution of information are used (Le Roux 2009). This includes technologies such as download technologies (video or audio) on the internet or intranets, or, alternatively stand-alone electronic technologies such as DVD and CD-ROM (Rossitter 2006:23).

It is thus evident that not only the technical skill of using technology in teaching and learning should be focused on during training, but also on how to use if effectively as an educational tool. Therefore it is necessary to emphasise and indicate the educational use of technology in teaching and learning during employee development and academics should be encouraged to revisit their existing paradigms and theories pertaining to teaching and learning. The fact that most academics are subject experts and not necessarily educational experts (Kanuka 2006:6; Fester 2006:6–7) also supports the notion that employee development for elearning should include educational principles and theories. Further, the e-readiness of academics is essential in the successful use of e-learning to optimally enhance learner learning. As the primary purpose of e-learning is the best suitable use of technology to maximally increase learner learning (JISC 2009:8), it is vital that academics are skilled in using the most appropriate teaching and learning strategies to accomplish this. The acceptance of a constructivist epistemology in teaching and learning may bring about the necessary change and is thus elaborated on in the next section.



3.2.4 The constructivist epistemology

Constructivism is essentially an epistemology that has affected the way that educators envisaged learning from the early 1990s (Jonassen 2006:43). Dewey (1916) in Huang (2002:29) is of the opinion that "... knowledge is dynamic and is built around the process of discovery". The learning environment is not static, but interaction takes place between learners and their environment (Huang 2002:29; Cubucku 2008:155). Learners understand and construct a reality based on their experiences and interactions with their environment (Gergen sa). Therefore, knowledge is based on lively experience and learners can construct new knowledge, founded in past knowledge (Huang 2002:28–29). It is necessary to determine which experiences from the environment, and which surroundings, are likely to encourage experiences that lead to growth (Huang 2002:29).

According to Cubucku (2008:155), a constructivist epistemological approach to teaching and learning requires academics to alter their regular thinking approach. Should traditional teaching and learning aim to realise a constructivist worldview, different ways of thinking and doing will be required (Cubukcu 2008:155). Gulati (2008:184) points out that constructivism is not a teaching method or a teaching model, but it is a philosophy that can contribute to critical assessment and problematising of existing and growing educational practices. Jonassen (2006:43) supports the premise that constructivism is not a theory of learning, or a model for designing instruction.

Educationalists and theorists have recognised the constructivist approach as essential for developing learner-centred strategies (Gulati 2008:183). Furthermore, emerging online learning literature often refers to learning as a social constructivist experience (Gulati 2008:184), indicating the preference of the constructivist approach in the online learning environment. When learners participate in archetypal e-learning activities such as structured online discussions, collaborative online activities, online assessment, and interactive course material, the constructivism in online pedagogy is supported as they are contributing to and constructing their own knowledge (Mason 1998 in Gulati 2008:184). When considering the profile of the majority of undergraduate learners, belonging to Generation Y, as explained in

chapter 4, it is evident that a constructivist approach will be most useful with this innovative, technologically driven generation.

Similarly, it can be assumed that this is applicable to academics who have to go through a learning experience as well. One can argue that, in addition to following the constructivist approach in their teaching, academics can also embrace this approach when confronted with obtaining skills pertaining to online teaching and learning. The constructivist approach lends itself ideally to the training and development of academics to become online learning facilitators. It is necessary though that academics change their thinking and approach pertaining to teaching and learning. However, in many instances academics are still embracing the traditional, objectivist epistemological approach.

David H Jonassen, a professor in Educational Psychology and widely respected as an expert on constructivist and objectivist epistemology, describes the suppositions of an objectivist approach to learning (Jonassen 1991:28) by stating that objectivists (supporters of the objectivist approach) believe:

- in the existence of consistent, reliable and dependable knowledge about the world
- that learners should gain this knowledge and educators should transmit it
- that learners gain the same understanding from what is transmitted by educators
- that learning consists of understanding that objective reality
- that the role of education is to help learners learn about the real world
- that the goal of designers or educators is to interpret events for them
- that learners are told about the world and are expected to imitate its content and structure in their thinking

According to Hanley (1994:3) the objectivist approach is driven by "talk-and-chalk" and strongly depends on textbooks for the structure of the course (cf Kinchin 2004:302). Objectivists hold the idea that there is a fixed world of knowledge that the

learner must come to know and educators serve as channels through which their thoughts and meanings are transferred to the passive learner (Hanley 1994:3).

Whereas the objectivist approach gives emphasis to observable, external behaviours, steers clear of reference to meaning, representation and thought, constructivism follows a more cognitive approach (Gergen sa). In the constructivism approach educators are coordinators, facilitators, resource advisors, tutors or coaches (Gergen sa). The role of the academic in the constructivist classroom provides a useful vantage point from which to grasp how the theory impacts on practice. According to Murphy (sa) the role of the educator in the constructivist classroom has two important elements. Firstly, an educator should introduce new ideas or cultural tools where necessary and provide the support and guidance for learners to make sense of these for themselves. Secondly, the educator must identify the ways in which the instructional activities are being interpreted to inform further action. Teaching from this perspective is also a learning process for the teacher. (Murphy sa.)

Tu, Shih and Tsai (2008:1143) indicate that people's epistemological beliefs mirror their perceptions about the nature of knowledge and knowing. These perceptions are found to be related to their common learning habits, or their approaches to processing learning tasks (Tu et al 2008:1143). In recent times, educators drew attention to the role of web users' epistemological beliefs in web-based cognitive activities (Hofer 2004 in Tu et al 2008:1143). Users' epistemological beliefs direct their cognitive as well as meta-cognitive activities in web environments. Users with constructivist-oriented epistemological beliefs are inclined to have greater preferences to engage in meta-cognitive thinking in online environments, as opposed to those who do not have constructivist-oriented epistemological beliefs. Furthermore, research (Braten & Stromso 2006 in Tu et al 2008:1143) showed that learners who held less sophisticated epistemological beliefs were less likely to engage in web-based discussions and online communication activities. Therefore, people's epistemological beliefs should be regarded as a key factor when observing their activities on the web. (Tu et al 2008:1143.)

Considering the role of constructivist-oriented epistemological beliefs in an individual's online activities, online engagement, and online meta-cognitive thinking, it is evident why this approach is sensible to follow in the online teaching and learning environment. It can be argued that the changed job requirement, namely the incorporation of teaching and learning with technology, calls for an adaptation strategy to use different pedagogical approaches with the emphasis on effective learning. It is evidently clear that in training academics can become more skilled and knowledgeable regarding teaching and learning strategies. That cannot be addressed in a single training session and will thus have to be part of a broader development programme. In preceding sections it has been argued that the training for the online teaching skill should ideally be part of a formal employee development programme. Employee development programmes usually contribute positively to organisational performance, in view of the fact that it results in a better skilled workforce that can accomplish more than would have been the case without the development opportunities (Wellins, Smith & Erker 2010:2; University of Minnesota sa:1). Research further indicates that employee development is one of the top three employee retention items in organisations (Smith 2010; Bonadio 2010; Mercer 2010; Abraham 2007; cf Šiugždinienė 2008:32). It can therefore be argued that HEIs will most likely benefit maximally from ensuring that academics are well-trained with regard to e-learning - not only will it enhance goal accomplishment, but it is also likely to improve employee retention. In this regard it is necessary to determine whether the Potchefstroom campus of the NWU development programmes for academics are in place. The employee training and development programmes at the North-West University are therefore discussed in chapter 6.

Career management and career management programmes can also enhance employee development (Erasmus et al 2005:303). Therefore, the next section discusses the role of career management in employee training and development and its contribution to enhanced employee performance.

3.3 CAREER MANAGEMENT

This section discusses whether career management programmes, including career development, is a viable option to enhance the e-readiness levels of academics. In

chapter 6 attention is also given to current career management practices and policies at the NWU.

3.3.1 The role of career management in employee development and enhanced performance

Career management is a process in which planning is done for an employee's progression and development in an organisation (PSC 2000:iv), in which the employee's goals, performance potential and preferences are aligned with organisational goals and objectives (PSC 2000:iv; Clark 2010; Gilley et al 2002:60). According to Erasmus et al (2005:297) career management is a continuous process in which employees collect information about themselves and the workplace and come to a particular realisation of their capacity, abilities, interests, values and ideal lifestyle, and, other jobs and organisations. The career management process serves as basis for employees to plan their careers, set career goals and develop and implement a strategy that is designed to realise goals (cf PSC 2000: 5–6; cf Cline & Kisamore 2008:4).

The most common elements in career management are career planning, career pathing and career development (PSC 2000:iv; PALAMA 2010:130). This implies that an employee's career has to be planned, in conjunction with the employer, to plan and create a career path. Career development should be included in the career plan to enhance the employee's progress and growth on the career path. The responsibility for career management lies with the employee (PSC 2000:6–7; Cline & Kisamore 2008:4), with support from the manager through the following responsibilities (PSC 2000:6–7):

- awareness of the employee's career ambitions
- determining the appropriate career path for the employee's ambitions to be attained
- informing the employee of training and development opportunities that can assist in meeting the career aspirations of the employee and at the same time the operational objectives of the organisation

 providing the employee with the opportunity to make use of these training and development opportunities

Although the primary responsibility of career management lies with the employee, it is clear that the line managers also have a responsibility to support the employee in this regard. Sturges, Conway, Guest and Liefooghe (2005:821) point out that individual career management behaviour is influenced by organisational career management assistance (cf Verbruggen, Sels & Forrier 2007:1). Thus, the more assistance an employee receives from the organisation, the more it will be perceived as fulfilment of the psychological contract⁶ (Sturges et al 2005:821; Armstrong 2006b:226). In turn, achievement of the psychological contract is connected to organisational commitment of employees and is related to behaviours at work (cf Armstrong 2006b:226; 228), including amongst others job performance (Sturges et al 2005:821). It therefore point to the following: "first, it shows that both individual and organizational career management behaviors are linked to psychological contract fulfillment; second, career management help is associated with affective commitment and job performance; third, psychological contract fulfillment plays a key role in mediating the relationship between career management help and such attitudes and and fourth, organizational commitment may mediate behaviors; between psychological contract fulfillment and individual career management behavior aimed at furthering the career outside the organization" (Sturges et al 2005:821). It is evident that both the employee and the organisation can benefit if line managers take cognisance of the effect of a psychological contract. The support of the organisation (mostly by the line manager) with career management will clearly support a sense of belonging and ownership and will eventually result in enhanced performance. The correct use of the psychological contract can thus effectively enhance employee motivation.

Managers can also assist in goal-setting (Landes 2006:29; Li & Butler 2004:38). As career management programmes attempt to involve and encourage employees to set their own goals within the parameters of organisational goals, recognise their

⁶ The psychological contract refers to an unarticulated combination of beliefs held by an employees and their employer about what they expect of each other (Armstrong 2006b:225).

strengths and weaknesses, and improve their performance (Erasmus et al 2005:302–303; PSC 2000:iv;5), it also supports them with the identification and facilitation of training needs and opportunities, which is mostly achieved by a process of feedback and discussion as part of the performance management system of an organisation (Gilley et al 2002:57–58; PSC 2000:5).

It is therefore apparent that the career management programme will entail that goals will be set with a view to grow, develop and to obtain a certain level of performance. It is thus important that the employee is actively involved in the goal-setting process. The line manager however, can also be involved in the goal-setting process to ensure compliance with organisational goals, to give guidance and to provide information pertaining to training and development opportunities towards goal accomplishment (Gilley et al 2002:57-58). In section 2.3.2 the goal-setting process, its significance and how it influences employee performance have been explained. In this section attention is also given to the manner in which the goal-setting process should take place which includes the manager's responsibility of providing support, a rationale and motivation to employees and continuously reminding them of the role and place of their individual goals, skills and contributions in the organisational setup. If development and mentorship programmes are in place and managers provide guidance in terms of career management and career development, it can also be valuable to assist and guide employees on correct career pathing (PSC 2010:6).

Greenhaus, Callanan and Godshalk (2010:7) explain that fast changing technology has produced new career paths for employees with an appropriate combination of skills, while less adaptable employees often found themselves out of synch with their employer's future strategies. Technology development has thus raised the skill needs for many jobs (Greenhaus et al 2010:7). The use of technology in teaching and learning has likewise provided new career opportunities and possibilities to academics and should be considered in their career planning. Thus, not only the obtainment of the skill to teach online, but also the improvement of their levels of ereadiness can be addressed through career development within their career management plans. It can be argued that employees who want to make valuable contributions in their workplace and are motivated to support their organisation's

vision, will most likely be enthusiastic about this process. All employees will, however, not be equally enthusiastic about the new job demand to be included in their career development plans and sections 2.3.3 and 2.3.3.1 therefore discusses the role of intrinsic motivation in this regard.

Based on the discussion above, it can be argued that career management programmes will positively contribute to the development of an employee as argued by Sturgess et al (2005:821) earlier in the section. The enhancement of an employee's e-readiness can be logically incorporated in and driven through a career management programme, in particular through career development. The next section will focus on career development as essential component of career management.

3.3.2 Career development

Career development is a process through which progression and growth through a sequence of jobs and phases is realised. Each of these phases is characterised by a new set of matters, ideas or tasks, and is involving continually more advanced or diverse activities and resulting in wider or improved skills, greater responsibility and prestige (BNet 2010). Career development can also be defined as "an organized, planned effort comprised of structural activities or processes that result in a mutual career plotting effort between employees and the organisation" (Gilley et al 2002:59). The previous section recorded that career management used to be regarded as the responsibility of the employer, whereas is now it is more seen as the responsibility of the employee. As essential element of career management, the same principle is relevant to career development (BNet 2010; Gilley et al 2002: 61). McLeish (2002:3) further indicates that employees need to become proactive in their learning and development as it will advance them on their career paths. According to Gilley et al (2002:60–61) the following can be accomplished through career development:

- Employees can search and discover future career paths.
- Employees' skills, interests and needs for growth and development can be aligned to the needs of the organisation.

- Managers can improve output and competence, employees' attitudes toward work, and job satisfaction.
- Efficient provision and allocation of employees and greater loyalty among employees can be promoted.

It appears that career development of academics will not benefit only the employee, but also the university as employer. The challenge will in all likelihood be to convince and motivate academics who lack the necessary level of e-readiness to undergo a development programme to become e-ready and to take up online teaching responsibilities. As long as e-learning is not part of the formal job requirements of an academic and reflected in the job description, employees will resist the use of e-learning. The manner in which employees will react to the introduction of e-readiness assessment and the consequent development will to a great extent be determined by key human factors, which are discussed in chapter 4. In conjunction with key human factors career anchors usually are leading elements that direct career choices and may also influence an academic's e-readiness as they are indicative of personal work preferences (cf Gilley et al 2002:61–62). Career anchors include:

- managerial competence (development towards more responsibility)
- technical/functional competence (proper person-job fit)
- security and stability (sustaining motivators such as income, benefits, and recognition)
- pure challenge (innovation, assignments, problem-solving)
- autonomy and independence (freedom in decision-making and carrying out responsibilities)
- lifestyle incorporation (work-life balance)
- service/dedication (making a difference)
- entrepreneurship (ownership and responsibility for one's work) (Schien 2007:27; Gilley et al 2002:61–62).

Some individuals flourish by being creative and innovative, others prefer stability and steadiness; challenges and continuous simulation may be important to one person, while creating a work/life balance is vital to another (Mind Tools 2010; Schien

2007:27-33; Bandyopadhyay 2007:34-35; ILO 2007:2). These differences in individuals' personal preferences support the notion that employees will approach changes in the workplace differently, based on what they value and how they are motivated (also see section 2.3). For the same reason employees will approach learning and technology adoption differently (also discussed in chapter 4).

McLeish (2002:2) indicates that career development is no longer merely about the attainment of the skills and knowledge employees need to make progress in an organisation, it is about attaining flexibility and constantly assessing and developing these skills in order to stay employable and content over the long term. In the 21st century organisations employees are expected to do more with less (Fryer 2010; Navran 2010). Stability is no longer the focus, but instant results, along with a demand for high skill sets, are focused upon (Cline & Kisamore 2008:5-6; Khosrow-Pour 2006:256; Auer & Cazes 2000:379). For employees to achieve a high level of flexibility they need to have a strong sense of who they are and what they want from their jobs; people are differently motivated and ambitions vary to a great extent (cf Cobb 2010).

The shift towards more flexibility and versatile skills required of employees underlines the need for continuous career development. Although employees no longer can rely on most organisations for long-term employment, they are increasingly expecting employers to provide career support and enhance their employability (Auer & Cazes 2000:379). The need for flexibility also poses a challenge to line managers with regard to academics who do not view e-learning as part of their own career development. In such a case it will be necessary for the manager to provide a rationale as to the importance and relevance of e-learning, as discussed in sections 2.2.1 and 2.3.2. In particular employees who are not e-ready will have to be motivated. Considering the profile of the majority of learners, and their needs, it is necessary to sensitise academics to the need for e-learning and emphasise the need to become comfortable with e-learning (Alvaro 2010:4-7; Motah 2007:483-484; Le Roux 2009).

Another positive outcome of employee development is retention (Smith 2010; Bonadio 2010; Mercer 2010; Abraham 2007; cf Šiugždinienė 2008:32). Cook and VELList of research project topics and materials

Jaggers (2005:19) indicate that talented individuals may become frustrated when their learning curve starts to plateau. The best employees usually seek opportunities to learn and grow in their careers, knowledge and skills; therefore a career-oriented, valued employee should experience opportunities for growth within the organisation (Cook & Jaggers 2005:19). Such career-oriented and talented employees may be categorised as *innovators* and *early adopters*, and may be portraying the *activist* or *pragmatist* learning style on the technology adoption cycle and will be elaborated upon in chapter 4. It may be ideal to include the employee development of these employees in a career management or talent management programme.

Talent management refers to "a strategic integrated approach to managing human capital throughout the career cycle: attracting, retaining, developing and transitioning the organisation's human assets" (TalentAlign 2007). For talent management to be successful it should be integrated with existing HR processes such as performance management, mentoring career development, succession planning, retention, reward and recognition (Ingham 2006:21; McCauley & Wakefield 2006:4; Wellins et al 2009:2). Talent management processes should preferably be integrated with other developmental practices in the organisation to support and increase the practice and importance of succession planning (Vermeulen 2007:272;277). Vermeulen, 2008:409) further describes that talent management has to do with the human resources of an organisation and how these human resources should be retained, developed and motivated to optimise institutional performance. The integrated approach will ensure that all developmental aspects are addressed. The integrated approach also implies that these developmental human resource management practices are integrated with the University's Human Resource Plan and that monitoring, evaluation and reporting should be done against both the Human resource plan and the Institutional plan as the Human resource plan derives from the Institutional plan (cf DPSA 2008:7).

It is thus evident that the enhancement of performance, and therefore the improvement of e-readiness of academics, can successfully be combined with a talent management strategy. An organisation will determine the focus, extend, capacity and restrictions of its approach to talent management, based on its business strategy (Ingham 2006:21). Talent management is therefore a strategic

process and should be driven by senior management (McCauley & Wakefield 2006:4; Lewis & Heckman 2006:7; Albertsson 2003:3; Guthridge, Komm & Lawson 2006:1).

As indicated in a previous section, for talent management to be effective, it should be integrated with other human resource management practices, especially development practices such as succession planning (Vermeulen 2007:272;277). Wolfe (1996:3) defines succession planning as "the systematic steps or design that allows for one to follow another in time or place". In other words, an organisation should have long-term and leadership plans in place to ensure the organisation will continue as usual when a talented employee leaves (Maurer & Weeks 2010:159). Succession planning is therefore the use of a purposeful process to ensure that employees with the necessary capabilities are developed and prepared to replace senior management as necessary (Prenhall 2007). It can be argued that none of these human resource management practices – talent management, career management, succession planning – are most effective when it stands on its own, but with an integrated approach can optimally ensure employee development.

Part of the development of an employee through a talent management strategy for succession planning is the practice of mentoring and coaching (Cook & Jaggers 2005:19). The *Public service mentorship programme* (2006:5) defines mentoring as "a process of deploying experienced individuals to provide guidance and advice that will help to develop the careers of protégés allocated to them". A mentor acts as counsellor, providing guidance on career paths, development opportunities and an indication of what it entails to become a leader in the organisation (PALAMA 2010:111; Thomas, Willis & Davis 2007:179–180; NASA 2007:3).

Leask (2005) explains that mentoring empowers employees by mentors spending time with them and talking about issues that are important to employees. The mentor (usually a manager) should assist and provide guidance in growing and emphasising their strong points and encouraging their development (PALAMA 2010:111; Thomas et al 2007:179). It requires sensitive and attentive listening from the mentor and introspection and responsible decision-making from the employee (PALAMA

2010:111). When mentees start to reflect on their experience they can turn it into knowledge that can be used for growth and achievement (Leask 2005).

According to Tobin (1998) in Turk (2011:40) "the coach is more of a tutor, observing work and actions, providing comments on execution and teaching skills that may be lacking". Coaching is described as a process that makes it possible for employees to learn and develop and thus enhance their performance (Turk 2011:39). It is essential that the coach has opportunities to observe the employee's work and that the employee respects the coach and is open to feedback in order for the relationship to be successful (Tobin 1998 in PALAMA 2010:112). When managers act as coaches, they should spend time with employees and assist them in mastering their work and developing their knowledge and skills (PALAMA 2010:112–113). Managers should also respect individual capabilities, as employees grow and development at various rates, and provide employees the opportunity for self-development (cf Hicks & McCracken 2009:72–73). A coach's most important task is employee development (Turk 2011:40).

It can be deduced that career management programmes and talent management programmes can play a positive role in employee performance. The integrated approach of a talent management programme, including human resource management development practices such as career management, succession planning and mentoring and coaching, is perceived to have a positive effect on developing the e-readiness of academics. The holistic approach of a talent management programme can ensure that all relevant aspects of an academic's job requirements are addressed.

It is further necessary to take note of the fact that academics are adults and that their training and development should be approached keeping cognisance of this fact. The next section therefore focuses on academics as adult learners.

3.4 THE ACADEMIC AS ADULT LEARNER

When confronted with learning a new skill such as teaching and learning online the academic becomes the learner. With employee development, career management

and talent management therefore, it is important for managers to keep in mind that they are working with adult learners. Knowles et al (2005:38) describe adult learners as those learners who are unlikely to be intellectually motivated or stimulated by the strict, rigid and inflexible requirements of "authoritative, conventionalised institutions of learning". The most significant differences between adult learning and conventional education lie within the learning process itself – adults bring their experience to the learning environment which results in a two-way learning process where both the adult learner and the teacher learn from each other (Knowles et al 2005:39; Fogarty & Pete 2004:12; Levine, 2001:1; Blake 2009).

Adult learners portray certain characteristic and preferences (Knowles et al 2005:3;36–37; Fogarty & Pete 2004:7–13; Blake 2009; Weldon 2010; Levine 2001:1–2). It is necessary that senior managers, line managers and the Human Resource Department of the NWU is informed about the preferences and characteristics of adult learners before they embark on employee development for academics to increase their levels of e-readiness. The theory of learning for adults, called *Androgogy*, is based on the following six principles (Knowles et al 2005:36–37):

- Learners need to know: Adult learners have a need to answer questions about the value of learning and the manner in which it will be accomplished.
- The self-concept of learning: Adult learners want to have the ability to take control of the skills and purposes of learning and self-direct their learning.
- Prior experience of the learner: Past experience of an adult resulted in individuals that have valuable resources, prejudices and a self-identity.
- Readiness to learn: In general adults are ready to learn when their life circumstances necessitate them to learn.
- Orientation to learning: When real-life problems are presented and a problemsolving approach is followed, adults usually learn best.
- Motivation to learn: Adult learners usually have high motivation to learn when the knowledge or skill they obtain can help them to solve significant problems in their life.

The fact that adults are ready to learn when their circumstances necessitate them to learn is encouraging in terms of formalising the use of e-learning for academics. This will necessitate them to undergo a development programme. Course designers should take care however, that employees do not feel coerced, as it will increase resistance and negativity. It is thus necessary to provide a rationale for the new job demand and development interventions. The abovementioned principles support the notion in sections 2.2.1 and 2.3.2 which indicates that it is necessary to provide a rationale for learning a new skill and that employees will be more inclined to learn a new skill when they are aware of the practical value that it will have for their immediate environments.

In addition to the abovementioned principles, Brookfield (1995) in Huang (2002:29) indicates the following four adult learning processes (cf Rabak & Cleveland-Innes 2006:118):

- Adult learners exercise self-directed learning by amongst others setting up their learning goals, searching for appropriate resources, deciding on their learning styles and evaluating their progress (Huang 2002:29; Blake 2009; Weldon 2010). In sections 2.2.1 and 2.3.2 the importance of an employee being involved in the goal-setting process is emphasised, as well as the motivational benefits of self-set goals. In chapter 4 the various learning styles are discussed.
- Critical reflection is part of adult's learning as they think contextually and critically (Huang 2002:29; cf Blake 2009). In particular, an adult learner who falls in the category of a *reflector*, as discussed in Honey and Mumford's learning styles in chapter 4 will portray this kind of learning to a great extent.
- When learning is founded on adult learners' experiences, these experiences are rich resources (Huang 2002:29; Levine 2001:1; Weldon 2010).
- When adults become skilled at learning, they have the capability of lifelong learning (Huang 2002:29).

Brookfield (1995) in Huang (2002:29) as well as Knowles et al (2005:3;36–37) identified the same characteristics that adult learners will portray when confronted with learning a new skill or to obtain new knowledge. It can be assumed that

academics as adult learners will in most probability portray the same or very similar characteristics. Adult learners want to see the value in what they are learning and want to take control of their learning experience. This places a responsibility on line managers and ASS to ensure that an academic's training and development to become an online learning facilitator reflect these values. Proctor and Doukakis (2003: 272) emphasise that it is also necessary to exercise employee development in such a manner that employees experience it as a result of management's commitment to meet their needs. "The successful application of the concept is translated into positive employee attitudes towards their work including organisational commitment, work motivation and job satisfaction" (Proctor & Doukakis 2003:272). This view is supported in sections 2.3.2, 2.3.3.1 and 2.3.4 emphasising the need for a non-coercive approach to maximise employee motivation. It remains evident that the line manager plays a significant role in employee development, particularly when a new job demand is introduces such as the use of technology in teaching and learning. The next section subsequently elaborates on the role of the manager.

3.5 THE LINE MANAGER'S ROLE IN EMPLOYEE DEVELOPMENT

According to Gibb (2003:281), and as alluded to in preceding sections, line managers are increasingly involved in employee development, which reflects the broader changing relations between line managers and the Human Resource Management Department (cf Šiugždinienė 2008:32). From the preceding sections it is evident that a manager has various roles to play in employee development such as assisting employees to identify their development needs, providing positive reinforcement and re-directive feedback, assisting employees in goal-setting and the drafting of a development plan, allocating time for coaching and development activities, providing feedback, providing guidance, ensuring opportunities, and offering organisational insight, information, and advice, and helping employees understand how they fit in the organisational picture. One of the most significant roles of the manager, however, will be to continuously provide information, motivation and support to constantly facilitate the employee's development (PSC 2010:6; Gilley et al 2002:57–58; 61; 68; Trinka 2009:1; Coach4Growth 2007; Heathfield 2010b). One of the biggest misconceptions amongst employees and

managers alike is that performance appraisal is an annual event (PSC 2007:5), thus implying that the development of an employee will also only be followed up once a year during a performance appraisal. It is, however, evident that an ongoing high level of involvement, guidance and feedback of the manager is required on a day-to-day basis.

Further, a line manager should respect an employee's learning curve and realise that it takes time to learn new skills and apply them well (Trinka 2009:1; Ambler 2010; cf Coates 2010). An academic's learning style and position on the technology adoption cycle will guide the manager in this regard, as the position where an employee is plotted on the technology adoption cycle will be directly related to the employee's level of e-readiness. The employee's learning style will direct the manager as to how to approach the employee's training and development as discussed in chapter 4.

According to Mayo (2000:529) an important outcome of employee development should be to give employees the ability to continue adding value to the organisation in the future (cf Jacobs & Washington 2003:343). A mixture of two factors play a role in this regard: an employee's ability to adapt and grow with the changing situation; and an employee's potential to contribute at a higher, broader or deeper level in future (Mayo 2000:529). Everyone is capable of growing and has dormant abilities (Mayo 2000:529). Managers should understand this potential and utilise or create opportunities for development and encourage employees towards further development and growth (PSC 2000:5;7; cf PSC 2010:x;5;6; Gilley et al 2002:68). It is therefore advisable that the development plan is coupled with the most appropriate motivational strategy for a particular employee. It is apparent that line managers should realise their role in an employee's development towards new skills and competencies such as e-learning, make provision for opportunities in this regard and provide the necessary support.

3.6 CONCLUSION

This chapter highlighted the necessity for a development process to follow the performance appraisal of academics with a view to enhance their levels of e-readiness. The role and importance of training and development in this regard was

discussed, as well as how it can be connected to career management, career development and talent management to optimally achieve increased levels of e-readiness and subsequent performance with regard to the use of technology in teaching and learning.

The study further argues that the training and development interventions for academics where it concerns e-learning, should not focus on the technical skill of e-learning only, but follow a broader teaching and learning approach, also including the philosophy of e-learning and the use and adaptation of teaching and learning strategies to obtain optimal learning. It is also argued that these training and development opportunities can be incorporated in a career management programme or a talent management programme.

It was further determined that career management programmes and talent management programmes can play a positive role in employee performance. The integrated approach of a talent management programme including human resource management development practices such as career management, succession planning and mentoring and coaching, is perceived to have a positive effect on developing the e-readiness of academics. The holistic approach of a talent management programme will ensure that all relevant aspects of an academic's job requirements are addressed.

The chapter also discussed the uniqueness of the academic as adult learner and attention was given to the pivotal role of the line manager in employee training and development. The development and training of academics for the use of e-learning due to the changed job demand of increasingly incorporating technology in teaching and learning which necessitated the e-readiness of academics, needs to receive attention.

This chapter outlined a number of career anchors, reflecting that some individuals flourish by being creative and innovative, others prefer stability and steadiness; challenges and continuous simulation may be important to one person, while creating a work/life balance is vital to another. Through the typical traits of the key human factors reviewed in chapter 4, the career anchors outlined in this chapter are

brought into relation with the key human factors identified in chapter 4. In chapter 4 it is determined whether these key human factors influence the e-readiness of academics.
CHAPTER 4

KEY HUMAN FACTORS IN E-READINESS

4.1 INTRODUCTION

Previous chapters alluded to various aspects relating to the purpose of the study – to determine how to assess the e-readiness of academics at HEIs. The study argues for the e-readiness assessment of academics to be included in their performance appraisals. Chapter 2 therefore discussed the process of human resource performance appraisal. As the process of performance management consists of not only an assessment component, but also a developmental component, chapter 3 discussed the role of employee training and development in enhancing the ereadiness levels of academics. The role of career management programmes and talent management programmes, integrated with other human resource management practices such as career development, succession planning and mentoring and coaching, has also been recorded.

This chapter advocates the consideration of key human factors to be included in the e-readiness assessment of academics during performance appraisals as they are believed to impact on the e-readiness of academics. These key human factors are discussed to determine possible e-learning profiles among academics, contributing either positively or negatively to their levels of e-readiness. Chapter 7, discussing the empirical research results of the study, will reveal the e-profiles of academics at the Potchefstroom campus of the NWU, based on their personal work profile patterns, preferred learning style and pace and style of technology adoption.

4.2 **KEY HUMAN FACTORS**

The study focuses on a particular job requirement of an academic, namely the elearning responsibility. The job requirement is brought about by global technological advancement and became an integral component of 21st century higher education (Moolman & Blignaut 2008:168;169; Amirault & Visser 2009:62). However, all academics did not necessarily embrace the changed job requirement (Le Roux Hereian List of research project topics and materials

2009); therefore to a great extent causing ineffective application and use of elearning. A perceived lack of e-readiness⁷ appears to be the biggest contributing factor to the effective use of e-learning as optimal learning tool.

Given the increasing importance of e-learning (Elgort 2005:182; Njenga & Fourie 2010:199; Kanuka 2006:1; Kim & Bonk 2006:23), an assessment of e-readiness of academics need to be enhanced to ensure they discharge their academic role as expected. Organisations obtain goals and objectives through their human resources (Ivancevich 2004:4); thus it necessitates dealing with the factor that appears to prevent academics from optimally using e-learning as pedagogical tool, namely their e-readiness. As indicated in chapter 1, the e-readiness of academics is not only determined by technical readiness, but also involves psychological readiness (Rautenbach 2007:iv; Hewitt 2003:5), therefore indicating that human factors play a role in e-readiness.

As the online learning facilitator plays a significant role in e-learning (Conceição-Runlee & Daley 2005; Thanasingam & Soong 2007:1003–1004; AFLF 2003:2), which necessitates academics (who have to act as online learning facilitators due to the new job requirement), to be e-ready. A variety of responsibilities is required from an online learning facilitator (and thus from the academic who will fulfil this role), including amongst others acting as subject expert, instructor, social director, facilitator, technical advisor and programme manager (Hootstein 2002; cf ANTA 2003:3; Choden 2008; cf ELF 2006:10–14).

The abovementioned responsibilities of an online learning facilitator unmistakably underline the human aspect in e-learning as crucial. However, the majority of research, pertaining to the use of technology in teaching and learning, focuses on the technology itself and not on the person using the technology (Johannes 2007:44–45). Johannes (2007:63) asks the question: "If we need new roles, new pedagogical approaches, new knowledge and skills and new online environments for the e-learning practice, should we not also look at new attributes for the person performing the job?" This question makes sense as the success of e-learning

⁷ Reasons for the perceived lack of e-readiness are discussed in chapter 5.

depends to a great extent on the competence and communication of the online learning facilitator (JISC 2009:8; Achimugu et al 2010:27; Cardwell & Madigan, 2004:26–27; Dziuban et al 2005:4). Human factors are thus significantly important.

Further it has been mentioned in previous chapters that employees will react differently to the changed job requirement and some will resist it (Proctor & Doukakis 2003:268), for the reason that employees have different personalities, intelligence, abilities, values, backgrounds and attitudes which influence their behaviour (Armstrong 2006b:240–244). It is therefore necessary to understand and appreciate the factors that affect how employees behave in the work environment to manage them effectively (Armstrong 2006b:239). Particularly when employees are confronted with a new job requirement, it calls for managers to be sensitive, communicate well, provide the necessary support and motivate employees (Landes 2006:29; Gitman & McDaniel 2008:187; Hansson 2009).

Further, in chapter 3 a number of career anchors have been indicated. These career anchors may influence an academic's e-readiness as they are indicative of personal work preferences (cf Gilley et al 2002:61–62). Individuals' different career anchors (different preferences in the workplace), illustrates that some individuals flourish by being creative, innovative and entrepreneurial; others prefer stability and steadiness; some individuals may prefer challenges and continuous simulation; while others may want a work/life balance (Mind Tools 2010; Friedman 2007; ILO 2007:2). These differences in individuals' personal preferences support the notion that employees will approach changes in the workplace differently, based on what they value and how they are motivated. For the same reason employees will approach learning and technology adoption differently, as also illustrated by the key human factors discussed in this chapter.

The chapter therefore emphasises the importance of considering key human factors when introducing a new technology. The DISC-factors of Thomas International (sa) are used as guideline for personal work profile patterns and will be brought into relation to the likely e-readiness profile of an academic. Further, the learning styles of Honey and Mumford (1982) are discussed with a view to determine the impact of employees' learning style preference on their technology adoption. The technology

adoption cycle of Rogers (1962) will be used to discuss the pace and style of technology adoption, as the technology adoption categories of the cycle reflects academics' levels of e-readiness. The interrelatedness of these human factors also receives attention.

4.2.1 Pace and style of technology adoption

In chapter 1 *technology adoption* is described as the manner in which people respond to product and service innovations that require them to change their past behaviour (Alexandrou 2011). An employee's style and pace of technology adoption therefore refer to the characteristic individual manner in which an employee will respond or adapt to a new technology or innovation (*style*) and how fast or slow the employee will adopt (*pace*). The study argues that style and pace of technology should also be assessed with the e-readiness assessment of an academic and the subsequent development process.

The *technology adoption cycle* is described by The Computer Language Company (2010) as "a model used to describe the adoption of new technologies, typically including the stages of innovators, early adopters, early majority, late majority, and technology laggards". Rogers (1962) introduced the theory of diffusion of innovation in his book, *Diffusion of innovations*, 1962, and defines *diffusion* as "the process by which an innovation is communicated through certain channels over time among the members of a social system" (Rogers 1995:35). He argues that innovations would spread through a society in an S-curve since the *early adopters* will first select the innovation (which may be a technology), where after the majority will follow until a technology or innovation has reached its diffusion point in a community (Rogers 1995:22–23;257). Zemsky and Massey (2004:9) support the S-curve principle of Rogers, showing the number of technology users that have adopted a technology over time (EDS sa:1). The S-curve is typically followed with e-learning patterns of modernisation, change and adoption (Zemsky & Massey 2009:4) as it is illustrated in diagram 4.1 below and indicates how academics will adapt to e-learning.

Diagram 4.1: S-curve of technology adoption



Source: Zemsky & Massey (2004:9).

According to Chen (2007) the speed of technology adoption (as illustrated in diagram 4.2) is influenced by the infrastructure required for implementation. If no new infrastructure is needed, the technology will be adopted more rapidly and will be presented by a steep S-curve (Rogers 1995:257; Zemsky & Massey 2004:9; EDS sa:1). At most HEIs, including the Potchefstroom campus of the NWU, the infrastructure (bandwidth, hardware and software) is already in place (Le Roux 2009). Other influences such as the needs of the academics, the perceived usefulness of the innovation (e-learning), the risk attached to the adoption, perceived benefits and the extent of behavioural change required, will also impact on the speed of adoption (Chen 2007; EDS sa:1; Zemsky & Massey 2004:9). It is exactly factors such as the perceived usefulness (discussed in chapter 2) and the behavioural change required that is of importance to this study. Both these human factors influence the pace and style of technology adoption of academics. Perceptions regarding the perceived usefulness of e-learning and the behavioural change required to adapt to the change, need to be dealt with through training and development as it can be argued that these perceptions, if negative, contribute to resistance to e-learning.

A central component to a person's attitude towards a new technology is in its diffusion (Rogers 1995:6). Roger's Innovation decision process theory suggests that the innovation adoption process takes place over time through five phases: Knowledge, Persuasion, Decision, Implementation and Confirmation (Rogers 1995:20–22;35). The innovation-decision process is the process through which an individual or other decision-making component goes:

- from initial *information* of an innovation
- to form an *outlook* toward the innovation
- to a *decision to adopt* or reject
- to *implementation* of the new idea
- to *confirmation* of this decision (Rogers 2003:20–22,35).

Academics, confronted with e-learning for the first time, will go through the abovementioned phases of adoption. The first phase, receiving *information* on the innovation, is a potential online learning facilitator's first contact with e-learning. Thus, a rationale is provided for e-learning. It could be argued that the more detailed and timely information is provided by institutional management and line managers, the more likely it is that the employee will adapt to or feel comfortable with the change in job demand, especially when the purpose and advantages of the new job demand is clearly communicated (as alluded to in chapters 2 and 3). Based on the information and decide to adapt to or reject the innovation. It can further be argued that if the phase of providing information is neglected, academics will not understand the use and purpose of e-learning in its totality and may therefore resist it. This in return, will delay the whole process of technology adoption.

Employees have different rates of technology adoption; therefore Rogers (1995:262) indicates various adopter categories. The diffusion model of Rogers (1995:262) proposes five adopter categories, namely *innovators*, *early adopters*, *early majority*, *late majority* and *laggards*, illustrated in diagram 4.2 below.



Diagram 4.2: Adopter categories defined by Rogers

Source: Adapted from: Rogers (1995:262).

A chasm or time gap develops between the *early adopters* and the *early majority* because of their different expectations (Moore 2002:5;12–13; Schang 2007:iv; Iskold 2007). Failure to identify this gap, which is wedged between pioneering and mainstream employees, or to take up the challenge of closing the gap has resulted in the failure of many potentially successful technology products (Oliver 2001:6; Carr 2007:3). Zemsky and Massey (2004:9–10) use these categories to illustrate e-learning's pattern of innovation and change. The categories can be summarised as follows:

- **Innovators** (2%): These are learners⁸ who enjoy exploring new ideas and are driven by intrinsic motivators.
- **Early adopters** (13%): These are learners who adopt once the concept has been proven. They are viewed as opinion leaders and decision-makers who have the vision to adapt an emerging technology to an opportunity, and they are driven by extrinsic motivators. They have the foresight to match an emerging technology to a strategic opportunity (Oliver 2001:6).
- **Early majority** (35%): These are the eventual users of technology who do not like to take the risks of pioneering, but see the advantages of tested

⁸ Learners in this context refer to academics as learners.

technologies driven by usability and success of the technology; they are the beginning of the mass market.

- Late majority (35%): These are learners who adopt when half of the population has already done so. They are followers who dislike the disruptions of new technologies and are more conservative.
- Diehards (laggards) (15%): These are learners who resist adopting innovations and perform the valuable service of regularly pointing out the discrepancies between the day-to-day reality of the product and the claims made for it (Beshears sa).

Carr (2007:9) explains that for *innovators* and *early adopters*, the existence of a technology in itself will be reason enough for them to pursue it, since they have a natural willingness and interest to explore. One can argue that *perceived usefulness* and *perceived enjoyment* (as discussed in section 2.3.4) will be high for these groups. Their *perceived ease of use* may also be high since these are the people that are willing to face challenges and overcome obstacles easier than people in the other categories. It can further be argued that the following career anchors (discussed in chapter 3), will appeal to these employees: pure challenge (innovation, assignments, problem-solving); autonomy and independence (freedom in decision-making and carrying out responsibilities); lifestyle incorporation (work-life balance); technical/functional competence (proper person-job fit); managerial competence (development towards more responsibility); and entrepreneurship (ownership and responsibility for one's work).

However, the *early majority* (and the other adopter categories to follow them) have a tendency to find use in something usually only when it relates to problems in their own disciplines (Zemsky & Massey 2004:9–10). Only if it can be confirmed that the new technology is effective, efficient and effortlessly applied, to their focused needs, they will be likely to adopt it (Zemsky & Massey 2004:10). It could be argued that employees falling in the *early majority* and *late majority* categories of the technology adoption cycle may need to be encouraged to use technology in teaching and learning by the use of motivators such as incentives or rewards, as there is

resistance to some extent, and they need a certain degree of convincing or motivation.

However, Rogers (1995:221) indicates that employees who receive incentives for adoption to e-learning possibly will change the patterns of adoption. The use of incentives will probably lead to faster adoption by employees who would have otherwise adopted slowly or not at all, and can negatively affect sustainability of adoption (Rogers 1995:221). Incentives can thus enhance the rate of adoption, but it is possible that it may cause a reduction in quality (Elgort 2005:183). It is therefore necessary that line managers do not haphazardly use incentives and rewards to motivate academics towards increased use of e-learning, but that it is well-planned, thought through and connected to an employee's e-profile. Important in this respect, is that training and development should be focused broader than the obtainment of a technical skill, but should also include the philosophy of e-learning pertaining to its use and purpose (as indicated in chapter 3). The benefits and use of e-learning as learning tool should rather be the motivating factor.

It can further be argued that *perceived enjoyment* will be low for these categories as they are slower to adapt and will probably only start enjoying the new technology, once they obtained the skill and experience the benefits thereof. Therefore, their *perceived usefulness* may be moderate to high, as they may be able to understand the benefits of the new technology, but their *perceived ease of use* will be low until the technology has been mastered. It can further be argued that career anchors such as technical/functional competence (proper person-job fit); security and stability (sustaining motivators such as income, benefits, and recognition); and service/dedication (making a difference) will appeal to these employees.

Employees of the *innovator* category will typically be intrinsically motivated and eager to take on challenges. An *innovator* therefore, will most likely embrace the change and job demand and not portray a lack of e-readiness. The *early adopter* category will also accept the new challenge relatively easily, as they are visionary thinkers, opinion leaders and change agents. It is important that leaders in HEIs focus on how to cross the gap between these two groups (*innovators* and *early adopters*) and the *early majority* (mass market), since the *early majority* will adopt

slower and portray resistance to some extent to the new technology. This is usually the biggest component of employees in the organisation and it is therefore imperative that managers should consider how to motivate these employees to firstly adapt to the new technology and secondly to determine training and development interventions to enhance performance.

It can be argued that line managers are responsible for determining the stage/s of technology adoption in which academics are through e-readiness assessment during the performance appraisal process. The assessment process is usually followed with a development plan and subsequent relevant development interventions and employees should be involved in the goal-setting process. Section 2.3.2 in chapter 2 recorded that there is a connection between self-set goals, self-efficacy and enhanced performance. Motivating employees and providing a rationale as to the relevance of e-learning within the framework of organisational objectives, is necessary for the eventual performance of an academic as an online learning facilitator.

According to Carr (2007:5) there are two primary approaches to the use of technology: a determinist (developer-based) focus and an instrumentalist (adopterbased) are different (Yates 2001; Yates 2004:3). For determinists (developers) technology is the primary reason of social change; determinists view the process as innovative progress that is thought to be out of direct human control; and determinists focus on an innovation's technical features and perceive successful adoption/diffusion as the result of an innovation's technological supremacy and regard the innovation's developer as the primary change agent (Carr 2007:5; Yates 2001; Yates 2004:3; Couros 2003:10-11). On the other hand instrumentalists (adopter-based) view the process as evolutionary; they perceive reasons for change to lie in social conditions and in human goals for change and development; and their focus is therefore on the user (adopter) of a technology and the user's value as an instrument to result in preferred change (Carr 2007:5; Yates 2001; Yates 2004:3; Couros 2003:10-11). Central to the instrumentalist's view is human control over the innovation and it is furthermore considered vital to appreciate the social context in which it will be used and the function that it will serve (Carr 2007:5).

109

For the purpose of this study the *instrumentalist (adopter-based)* approach is relevant, where the focus is on the academic (person) using the technology. This emphasises the notion that the consideration of key human factors in the innovation process can be helpful to determine a person's pace and style of technology adoption. Since there is currently no e-readiness assessment instrument available which assesses the e-readiness of the person using the technology, this study emphasises the need for including the e-readiness assessment of academics in performance appraisals. This should be followed with training and development interventions to assist in improving employees' e-readiness.

In addition to the level of technology adoption, each employee has a different style of approaching learning and adapting to a new concept or job demand (Dewhurst & FitzPatrick 2007:14). A number of these learning styles will be outlined in the following section and it will be determined how these learning styles impact on an academic's e-readiness.

4.2.2 Learning styles

Winn (1997) and Jonassen et al (1997:28) in Rautenbach (2007:11) are of the opinion that human beings are unpredictable and have different needs. This premise supports the necessity to identify as many as possible elements that can influence employees' learning needs (Rautenbach 2007:11). In this study the learning needs of academics are explored by means of amongst others preferred learning styles. The inspiration of individualised learning styles and the domain of cognitive learning styles has been emphasised for over 40 years (Cassidy 2004;419; Duff 2003:30). The term *learning style* has been used interchangeably with terms such as *thinking styles, learning modalities* and *multiple intelligences* (CIPD 2008:1). "Learning is a cognitive process of taking in information, processing, organizing, and storing it in rich connected schema in ways that knowledge formed is easily recalled when required" (Ormrod 1999 in Akdemir & Koszalka 2008:1452). Learning styles refer to various approaches or techniques of learning and involve educating methods, which seemingly allow learners to learn effectively (Guild 2001; Park & Lee 2001:656; Felder & Henrique 1995:21; Azhar 2008:5). The way in which learners prefer a

particular way to interact with stimuli and information are termed their learning style (Palade, Howlett & Jain 2003:1230; Felder & Henrique 1995:21).

Research done by Swinton (2006:1) reveals that everybody has a preferred learning style; employees will learn more successfully if they can use their preferred learning style (Swinton 2006:1; cf Robbins 2009:459). On the other hand, employees is likely to be discouraged if the only learning opportunities available to them do not allow them to use their preferred learning style and it will therefore affects their ability to learn (Swinton 2006:1; Robbins 2009:459). Learning styles influence the manner in which individuals attach their own meaning to the subject matter or skill being taught (Roy 2006:22). Online teaching and learning must therefore be structured in such a manner that it makes provision for all learning styles (Salmon 2003:110).

Academics' learning style preferences play a significant role in their adoption to the use of technology in teaching and learning and therefore, their e-readiness. For instance, right brain dominated people may prefer a holistic and visual approach, while left brain dominated people may prefer a systematic, step-by-step approach (Vermeulen 2005:8). "Dominance by the left cortex means that an individual is rational, logical, analytical, fact-based, and quantitative. Dominance by the left limbic means that someone is organised: sequential, planned, and detailed. Dominance by the right limbic makes someone interpersonal: feeling-based, kinaesthetic, and emotional. Finally, dominance by the right cortex means that someone is imaginative: holistic, intuitive, integrating, and synthesising" (Simons 2008:424). It is evident that employees have different traits, such as skills, aptitudes and preferences for processing information and constructing knowledge from information, and therefore provision should be made that they can learn in the manner with which they are most comfortable (Magoulas & Chen 2006:327).

Further, for the purpose of enhancing an employee's e-readiness, it is important to take into consideration that the ideal is to use a combination of active, practical, theoretical and reflective learning activities (Honey & Mumford 1982:25–29; Dewhurst & FitzPatrick 2007:14; Arp, Woodard & Mestre 2006:29; Boyatzis & Kolb 1995:3). There is a universal acceptance amongst researchers in the field of learning styles that a learner's learning approach influences his/her performance and

attainment of outcomes (Cassidy 2004:420). Learning styles are therefore important in the education environment, because it assists educators in understanding how learners learn (Swinton 2006:1). Assessing the learning styles of academics can therefore assist in understanding how employees learn and which approach should be applied to maximise an employees' motivation towards e-learning and their eventual performance. Organisations can use questionnaires to measure an employee's preferred style of learning (Jackson & Lawty-Jones 1996:293; cf Morrison 2007). Learning style questionnaires of Honey and Mumford are progressively more used as an autonomous method to encourage learning and with the aim of increasing the ability of people to learn from their jobs and training courses (Morrison 2007; Jackson & Lawty-Jones 1996:293).

Jackson, Furnham, Forde and Cotter (2000:223) point out that a connection exists between learning styles and occupation personality types (Bakx, Van der Sanden & Vermetten 2002:1229; Jackson & Lawty-Jones 1996:293). Although this study will not focus on individual personality styles, this premise supports the notion that an individual's learning style is to some extent, connected to a person's personality. It also emphasises the usefulness of including the assessment of an academic's learning styles (which will partly indicate the personality type of the employee) in the performance appraisal with a view to draft a training strategy, consistent with the employee's learning style preference.

Various learning styles and models are indicated and researched by psychologists, but for the purpose of this study the learning styles of Honey and Mumford (1982) were chosen. The following learning style categories are identified by Honey and Mumford (1982:25–29):

- **Activists** individuals who prefer to deal with new challenges and experiences and should receive a range of activities to keep them interested.
- **Pragmatists** individuals who require a link between the training and the endresult required of them. They will determine the practical value and use of what they are being taught.

112

- Theorists individuals who require good structure and sufficient time to explore the relevance between ideas and scenarios. They are analytical and detail-conscious and need to think things through in a logical step-by-step manner.
- **Reflectors** individuals who spend a significant amount of time to think intensively about the activities and concepts provided to them online. *Reflectors* fall into the category that probably benefits the most from web-based teaching and learning (Liu 2007:41). Downing and Chim (2004) in Liu (2007:41) find that individuals who are *reflectors* tend to be extroverted in the online environment whereas they may be introverts in the traditional classroom setting.

It can be argued that academics that are *activists* can be compared to the *innovator* and *early adopter* of the technology adoption cycle, who is an employee with a vision of the benefits of technology in teaching and learning. The *activist* will, however, prefer to be the leader and rather give direction (cf Honey & Mumford 1982:56). It can be argued that both the *perceived enjoyment* and *perceived ease of use* of employees belonging to these categories will be high. *Activators* thrive on new challenges and if an activating learning style is combined with an *innovator* category of technology adoption, it can be assumed that the employee will embrace the new job demand and will enjoy trying out online tools and possibilities. Employees, portraying the *activist* learning style will also perceive the technology to be *easy to use* as they do not fear new, unfamiliar challenges and are likely to "play around" with the new technology and *enjoy* the experience while they are mastering it.

The *pragmatist* may be compared to people falling under the *early majority* category of the technology adoption cycle. This group will make the paradigm shift from traditional classroom teaching and learning to e-learning once the concept has been proven. It can therefore be argued that the *perceived ease of use* will be very important to an employee portraying the pragmatist learning style. As long as they are not convinced of the use and purpose of the technology they will not adopt to using it. The same can be said regarding their *perceived enjoyment*, when they realise the possibilities of the use and purpose of e-learning and the rationale behind

it, they will start using it and in the process start enjoying it as they have realised its value in learning.

The *theorist* may be compared to people in the *early majority* or *late majority* category, needing more time to become familiar with the use of technology in teaching and learning. They will analyse and review information before they adopt the use of technology in this way and will eventually make the paradigm shift. Once they adopt the use of technology for e-learning, they tend to flourish in the online environment owing to their systematic and analytical approach (Arp et al 2006:30). The *reflector*, similar to the *theorist*, may be compared to people in the *early majority* and *late majority* category. Prior to adopting the use of technology, they will consider it intensively and listen to others' views, considering them before taking action or incorporating their own views. With both the *theorist* and *reflector* learning styles perceived ease of use will take time as employees belonging to these learning style categories take time thinking concepts over, especially if combined with the early majority and late majority categories, which also take time to determine the use of elearning. Employees belonging to the *late majority* category may remain negative pertaining to *perceived ease of use* and it is possible that they will only eventually adapt to e-learning as it is expected of them by managers and/or learners, as indicated in the previous section. As these employees' ease of use will take long, they will also not *perceive* it to be *enjoyable* as they do not have the skill and do not know the purpose of e-learning.

The above discussion shows that academics who need to take on the role of online learning facilitators naturally tend towards a particular learning style. This learning style relates and indicates an employee's preferred way of interacting with and adapting to new information and demands. These learning styles need to be assessed during a performance appraisal as part of an academic's e-readiness assessment and should be considered during goal-setting and drafting of a personal development plan. By considering these learning styles during the developmental part of the performance appraisal process, training and development interventions can be structured in a manner to suit this learning style. This approach is likely to increase the employee's motivation and will make the technology adoption process

114

easier. The abovementioned learning styles will therefore be included in the framework in which employees can be plotted during performance assessment.

In addition to employees' pace and style of technology adoption and their preferred learning style, employees also have different work profile patterns. The personal work profile patterns as third key human factor in this study is discussed in the next section.

4.2.3 Personal work profile patterns

As indicated in section 4.2.2, employees portray different qualities, skills, abilities and prefer various methods to deal with information, to gather knowledge from information, and to apply their knowledge to real-life problem solving (Magoulas & Chen 2006:327). Thus employees have different approaches and preferences to work. Xu and Tuttle (2004:22) explain that interpersonal aspects are more important to an employee's success in the workplace than technical skill. An interpersonal aspect that is likely to differ between employees is work style; for instance, some employees approach problem solving in a cautious, systematic manner, whereas others favour innovative solutions (Xu & Tuttle 2004:22). Employees' work style therefore influences their behaviour in the workplace. Pearsall (2001:157) describes behaviour as "the way in which one acts or conducts oneself, especially towards others" and style as "a manner of doing something" or "in a manner characteristic of" something (Pearsall 2001:1847). Thus a work behaviour style refers to the characteristic manner in which an employee acts and conducts, especially towards other employees, including managers, in the workplace. In this study the Dominance Influence Steadiness Compliance (DISC) factors of Thomas International will be used to describe personal work profile patterns of academics.

The DISC profiling instrument is a useful and well-known assessment instrument to determine the attributes that will contribute to an employee's personal work profile pattern (Thomas International 2005). The DISC profiling instrument describes human behavioural pattern styles in four dimensions. A DISC profile reports a style or characteristic of behaviour in a work situation. Four dimensions or "typical patterns of interaction" of a person in the working environment are important (Thomas

International sa). All people have all four behavioural preferences but to various extents. The relationship of the four preferences to each other construct a profile pattern which gives information about a person's probable behavioural responses (Mills 2002). Thomas International (sa) identifies the following four dimensions:

- Dominance: This category considers the manner in which problems are addressed. Individuals of this category are concerned with results. They are typically competitive, with high performance standards, and focused on achieving goals, solving problems, and accepting challenges.
- **Influence**: This category considers the manner in which people are dealt with. Individuals of this category like people and want to be liked in return. They are typically charming, optimistic, and outgoing, and focused on networking, conversation, and working with others.
- **Steadiness:** This category considers the manner in which an individual paces him- or herself. Individuals of this category are concerned about relations. They are typically sympathetic, friendly, good listeners, "finisher completers", and team players, who work hard and create a stable environment.
- **Compliance**: This category considers the manner in which rules and procedures are followed. Individuals of this category are concerned with accuracy and research every aspect of a situation, considering each possibility before making a decision (Witt sa). They typically have high standards, particularly for themselves; can be perfectionists; and prefer systems, processes, procedures, as well as predictable and consistent outcomes.

It is evident that an academic displaying a high *dominance* factor profile is likely to be motivated and inspired by a challenging and dynamic environment and enjoys experimenting with new technologies at a fast pace. Further, such an employee is unlikely to be motivated by incentives and rewards but will most probably be intrinsically motivated and have an inclination to set challenging goals (Thomas International sa). It can be argued that employees with a high *dominance* factor will typically be *innovators* on the technology adoption cycle, with the *activist* learning style. When an academic portrays a high *dominance* factor and low *steadiness* and *compliance* factors, it can be assumed that the profile of the online learning facilitator

116

tends to favour the achievement of results irrespective of unfavourable circumstances (Johannes 2007:256). These employees will experience high levels of *perceived ease of use* and *perceived enjoyment* due to the challenging and innovative nature of their profiles.

An academic with a high *dominance* factor will prefer an unstructured environment, which allows for frameworks and directions to guide people on how to act, tolerate innovative thought, creative problem-solving and independence to act, not prescribing strict rules and procedures (Thomas International sa). It can be argued that the development interventions for an online learning facilitator portraying a high *dominance* factor should be innovative, exciting and creative with challenging tasks. It is evident that academics belonging to this category will adapt to teaching and learning with technology easily and will not portray a lack of e-readiness. It can furthermore be argued that academics with a high dominance factor will connect relatively easy with young learners from Generation Y due to the same need for innovation and creativity (Thomas International sa; Naidoo 2005).

Academics' portraying high *steadiness* and *compliance* factors and a low *dominance* factor, will tend to favour standard operating procedures, a traditional approach and maintaining the status quo (Johannes 2007:256-257). Factors such as attention to detail and ensuring quality and standards are important (Johannes 2007:257). It can further be assumed that structure and security within a clearly defined learning environment will appeal to these employees. Academics belonging to these categories can be compared to the *early majority* and *late majority* categories of the technology adoption cycle, as well as the *theorist* and *pragmatist* learning styles. Their *perceived ease of use* of e-learning will initially be negative as it will put their status quo and comfort zones at risk. Likewise, they will not initially *perceive* e-learning to be *enjoyable* and will only change their perceptions over a long period of time.

Shelton, McKenna and Darling (2002:372) state that, not only do employees have different work profile patterns, they regularly also have diverse values and interests. Further employees are inclined to have a preference for various types of job tasks and work cultures (Shelton et al 2002:372; cf Liu 2010a; Switzer 2010). Liu (2010a)

asserts that managers that want to obtain objectives and achieve performance from their subordinates need to understand that employees have different work profile patterns and will therefore be differently motivated. An employee's personal work profile pattern thus plays a significant role in the manner in which he or she will adapt to technology and should therefore be included in an employee's e-profile.

4.3 CONCLUSION

The chapter highlighted the significance of the human factor in e-learning and its impact on academics' e-readiness. Employees portray different qualities, skills, abilities, preferences and behaviours and learn differently. These differences contribute to an employee's e-profile. The chapter advocates for particular key human factors to be considered during an employee's e-readiness assessment with a performance appraisal: pace and style of technology adoption, preferred learning style and personal work profile patterns.

It became evident that the traits portrayed by a person belonging to a certain technology adoption category show similarities to traits portrayed by particular learning styles and personal work profile patterns. Certain profiles could be identified, based on trends and similarities pertaining to interpersonal traits. It for example became evident that an employee who is an *innovator* on the technology adoption cycle will most likely prefer and *activist* learning style and will display a strong *dominance* factor on the DISC profile. These are high achievers who embrace challenges and new opportunities.

Another likely connection that could be identified was that an *early adopter* on the technology cycle may also portray an *activist* learning style and a *dominance* factor on the DISC profile, as these employees also adapt to technology relatively fast. The *early majority* and *late majority* technology adoption categories will most likely be related to the *theorist* or reflector style and will correlate with the *steadiness* and *compliance* factors of the DISC profile. These categories need more time to think about a changed job demand and will only adapt when most others have already done so or when it is expected of them. It is thus evident that these combinations imply various levels of e-readiness. In chapter 6, where results obtained from the

empirical research are discussed, it will be determine whether these trends could also have been identified in practice.

The following chapter outlines the role and purpose of e-learning in the 21st century, necessitating the e-readiness of academics. Attention is therefore also given to the resistance to e-learning and how it relates to key human factors of academics. As the academic plays a pivotal role in e-learning it is also discussed what e-readiness entails and what can be regarded as an adequate level of e-readiness.

CHAPTER 5

E-READINESS: CRITICAL REQUIREMENT FOR EFFECTIVE E-LEARNING IN THE HIGHER EDUCATION ENVIRONMENT

5.1 INTRODUCTION

The previous chapter emphasised the importance of key human factors in e-learning and their impact on an academic's e-readiness. The thesis therefore advocates for the incorporation of an assessment of key human factors in the e-readiness assessment of academics during performance appraisals. The previous chapters also recorded that the performance appraisal should be followed with the drafting of a personal development plan that makes provision for uniquely structures training and development interventions to enhance the employee's e-readiness.

The assessment of academics' e-readiness presupposes pre-determined goals, standards and competency requirements. In this chapter the e-readiness criteria against which academics should be assessed are determined with a view to establish an e-readiness construct. As the perceived lack of e-readiness amongst academics hamper the successful use of e-learning as educational tool, reasons for e-learning resistance are also explored. However, attention is firstly given to the higher education environment of the 21st century, necessitating the use of technology in teaching and learning, and in return, requiring that academics are e-ready.

5.2 THE HIGHER EDUCATION ENVIRONMENT OF THE 21ST CENTURY

The education environment, in particular the 21st century, presents universities with various challenges. Over the centuries universities have portrayed the ability to reinvent themselves when intellectual, political, or technological change take place. Likewise universities need to adjust to the pace of technological change in teaching and learning before they expose themselves to the risks of losing their standing and competitive advantage (Amirault & Visser 2009:62–63.)

Since the early 1980s HEIs have been criticised for their competence and accountability (Meek & Davies 2009:42). During the period mentioned above HEIs had the luxury of examining, debating, and arguing the advantages and disadvantages of advanced technology in teaching and learning. However, in the 21st century, the inevitable progress of technology into a meaningful, ever-present and sustained reality (OECD 2004; Bennet 2002:2; Albright & Nworie 2008:15) has largely cancelled out such debate and arguments in support of and not in favour of the notion (Amirault & Visser 2009:66). The ongoing success of universities as HEIs depends on their ability to adapt to technological change and to integrate such technology into teaching and learning (Amirault & Visser 2009:66; Albright & Nworie 2008:15; cf Hung 2006 5). A number of reasons that makes it necessary for HEIs to adapt to the changing pace of technology, in particular e-learning, are identified and discussed below.

5.2.1 The types of learners and their expectations

Traditional learners usually are well-developed in linguistic and logical-mathematical intelligences and are academically prepared for higher education (Bluestein 2001). Traditional learners portray personality traits such as concrete and logical thinking, rationality, promptness, they are well-organised, function well in a structured environment, and are able to follow rules and procedures (Bluestein 2001). Learners in the 21st century, however, differ from traditional learners (Magolda & Terenzini 2010; cf EIU 2008:5) for a number of reasons of which the following are discussed in this chapter:

- the unique characteristics portrayed by their generation, namely, Generation Y (Bennet 2002:6–7)
- a significant amount of current day learners are underprepared for tertiary learning (Reid 2007; Gabriel & Flake 2008: xii;1)

These 21st century types of learners are discussed below.

5.2.1.1 Generation Y learners

The first type of learners is those referred to as Generation Y learners. Undergraduate university learners in the 21st century are living in a world of technology (Kezi 2009:1; Bennet 2002:6–7). To expect of these learners to yield the realities of their world by providing them with traditional classroom teaching and learning, is to disregard the global role and place of technology (Amirault & Visser 2009:66). Generation Y refers to people born between 1980 and 2009 (Naidoo 2005). This year range refers mostly to Generation Y in the United States of America; in the South African context these learners are those born from 1990 to 2011 (Steyn, Badenhorst & Kamper 2010:177; 185; cf Msimang 2008). According to Song et al (2004:59) Generation Y portray particular characteristics which determine their interaction with others and their environment, how they connect and learn with each other, as well as the assortment of technologies they use to do so (Gonsalves 2006; Rockler-Gladen 2006).

In terms of the *The Generation Y and the workplace annual report* (2010) and EIU (2008:5) generation Y is characterised by a high level of technical literacy (Gen Y Report 2010:7; 17; EIU 2008:5) and some of its commonly used technologies include live virtual classrooms, podcasts, blogs, social networks and collaborative editing (Halse & Mallinson 2008:1; cf Song et al 2004:59; cf EIU 2008:5). Having the right technological platform is pivotal to learners of this generation (Gen Y Report 2010:8). This generation considers the internet indispensable to life, learning, work, and leisure time (Moore, Moore & Fowler 2011). Their behaviours, approaches, skills and abilities differ from those of other generations as a result of their exposure to technology and in many instances the teaching and learning perspective of Generation Y learners differs considerably from that of university management and academics (Moore et al 2011).

This generation, according to Naidoo (2005), prefers work to be fun and relaxed and a traditional approach to working does not appeal to them (cf Hansen 2010; cf

⁹ This estimation is approximate, as members of Generation Y in South Africa is generally considered to be born between 1990 and 2000, but in the United States, this generation is believed to be born between 1980 and 2000 (Naidoo 2005 Steyn et al 2010:177,185).

O'Neill 2010:3). Learners of this generation need to be constantly stimulated to prevent them from getting bored (Rockler-Gladen 2006). They are skilful at multitasking, think fast and are passionate and broad-minded (Naidoo 2005; cf O'Neill 2010:8). They are also an innovative generation and seek reinforcement and constant feedback on a regular basis (Naidoo 2005). Learners in this generation are networked, collaborative, and social (Gen Y Report 2010:24). The manner in which they use these digital technologies and the way in which they prefer to work with technology is often regarded as challenging for those who do not belong to this generation (cf Gen Y Report 2010:24).

These characteristics and technologies preferred and used by Generation Y, can shape the way in which academics teach. According to Ramsden (2003:xii) understanding learners' experiences of learning is and ingredient to effective teaching. Milliken and Barnes (2002:225) indicate that this implies that teaching and learning strategies may have to be adapted to focus on the enhancement of learners' learning. The application of new technology can be brought into play to improve both the teaching and learning experience (Milliken & Barnes 2002:226). It is therefore clear that making provision for Generation Y's learning needs will require academics to be flexible in terms of teaching and learning approaches.

Statistics from the Higher Education Management and Information System (HEMIS 2010) indicate that a total of 371 381 (46,3%) out of a total of 799 490 learner enrolments for 2008 at public HEIs, belong to Generation Y (HEMIS 2010). For 2009 a total of 387 550 (46,6%) out of a total of 837 779 learner enrolments at public HEIs, belong to Generation Y (HEMIS 2010). These statistics are indicated in table 5.1 below and include learner enrolments of ages 18 to 22 during 2008 and 2009.

Age	2008 Enrolment		2009 Enrolment	
	Total	%	Total	%
18	82 742	10,3	93 614	11,2
19	76 665	9,6	80 935	9,7
20	78 780	9,8	80 424	9,6
21	72 855	9,1	72 480	8,6
22	60 339	7,5	60 097	7,5
23	371 381	46,3	387 550	46,6
Total of registrations	799 490		837 779	

Table 5.1: Learner enrolments at public HEIs, by age

Source: HEMIS (2010).

When learner enrolments for learners between 18 to 21 years are considered, (including the vast majority of undergraduate learners), a total of 38,8% of learners in 2008 were between the ages of 18 and 21 and a total of 39,1% of learners were between these ages in 2009. These statistics indicate that almost 40% of the total learner enrolments at public universities in the South African context could be categorised as Generation Y during 2008 and 2009. If the learners of 22 years of age are also considered in the equation, this percentage moves up to 46,3% and 46,6% in 2008 and 2009 respectively, bringing it to a total of almost 50% of the total learner enrolments (including all postgraduate learners) for these years¹⁰.

The HEMIS statistics reveal that the vast majority of undergraduate students at public universities in the 21st century are Generation Y learners. Statistics on the presence of Generation Y learners at the NWU specifically are provided in chapter 6.

5.2.1.2 Underprepared learners

The second type of learners of the 21st century is the so-called underprepared learners. Addressing the needs of underprepared learners is an increasingly grave concern at HEIs throughout South Africa (Parliamentary Monitoring Group 2010;

¹⁰ In most instances the ages of 18 to 24 are considered for undergraduate (Worthen & Jefferson 2009:1, but as the South African Generation Y was born mostly from 1990, ages between 18 to 22 years were considered.

Taljard 2010; Gabriel & Flake 2008:6; Brüssow 2007:126). Globally HEIs face these challenges (NCES 2003:1; Ramsden 2008:3; Ramsden 2003:4).

Based on research conducted by Dzubak (2005:1) underprepared learners are learners whose academic skills are lower than what is regarded as higher education readiness skills, especially in terms of reading, writing, and mathematics. The underprepared learner in the higher education environment has substantial academic and societal problems and complexity that is related to academic underperformance (TSTC 2004:10 in Brüssow 2007:133). Further, underprepared learners portray weak academic skills (Anderson 2004:3 in Brüssow 2007:133). In research conducted for her PhD study by Brüssow (2007:134), she found that the underpreparedness of learners in the higher education environment may also be attributed to a challenging educational history, characterised by learning environments that did not sufficiently prepare students for higher education learning (Brüssow 2007:134).

Academic preparedness presupposes academic proficiency that includes abilities such as reading, writing, note-taking, examination writing (Brüssow 2007:133–134), as well as the ability to effectively study, solve problems, and think critically and analytically to make adequate progress through higher education academic programme work (Dzubak 2005:2). According to Nel (2010) learners should ideally be able to read an amount of 400 words per minute with an 80% correctness level, but 350 words per minute and a 70% correctness level is regarded an acceptable level to pass an academic programme.

HEIs thus face unique challenges to identify initiatives to support learners' underprepared for higher education (Brüssow 2007:127). However, although learners may be underprepared for the first semester of their first year of study, they have the ability to enhance their academic literacy skill once they have mastered the required prerequisites (Dzubak 2005:3). Chapter 6 elaborates on measures and tools being put in place at the NWU to enhance the academic underpreparedness of learners.

125

From the above discussions on the 21st century types of learners, it can be deduced that with Generation Y learners, as a highly technological generation (see section 5.2.1.1), technology should be used maximally to assist these learners in preparing their academic literacy and/or reading. Furthermore, it is advisable to use creative methods such as simulations, games, exercises that are fun, etcetera, as these learners belong to a highly innovative generation and get bored easily (Rockler-Gladen 2006).

Amirault and Visser (2009:75) are of the opinion that Bloom's Taxonomy can be very useful in making decisions regarding instructional technologies and strategies (cf Kanuka 2006:4). In Bloom's Taxonomy (Bloom 1956:201-207) he provided a classification of learning objectives within education, including lower order and higher order skills. The attainment of facts, knowledge, and information are usually best realised and mastered through repetition strategies, which are the powerful collection of an entire class of software based technologies, such as electronic flash cards, memory games, and electronic practice guizzes and tests (Amirault & Visser 2009:75; Adams 2007:72). The lower order skills represent the attainment of facts, knowledge, and information, whereas the higher order skills (analysis; synthesis; evaluation) cannot be achieved without the attainment of the lower order skills (Bloom 1956:20–21;207). When technology is effectively applied and integrated into teaching and learning, through meaningful facilitation and interaction of learners with information, it can assist in developing higher order thinking skills such as comprehension, reasoning, problem-solving and creative thinking (SA 2004b:14). It is thus evident that the preparation phase/s and bridging courses for learners can easily be facilitated and/or enhanced through the use of e-learning.

In the same manner that technology can be applied to improve the academic literacy and/or reading pace and correctness of learners, it can also be aligned with learning outcomes (cf Hoffman 2010:4). Although these repetition and exercise technologies are valuable in providing training opportunities, it is not the best suited method for the development of higher order skills such as analysis, synthesis, and evaluation. For learning outcomes of this nature technologies should be intended to accomplish research, collaboration, information gathering and comprehension, and content construction. Whilst behavioural approaches that are characterised by computer based drill and practice routines are highly effective in achieving Bloom's lower order skills, constructivist approaches that are characterised by computer based research, collaboration, and content construction tools, are exceedingly effective in achieving Bloom's higher order skills. (Amirault & Visser 2009:76.)

It is evident that the 21st century learner's profile requires academics to adjust teaching and learning strategies to those that appeal to learners. The above discussion also implies that academics be skilled in using e-applications to enable them to be effective in the teaching and learning environment.

5.2.2 Changing demographics of learners

Traditionally HEIs targeted learners between 18 and 24 years of age (Worthen & Jefferson 2009:1; Borden 2004:12; cf Chen, Gonyea & Kuh 2008:1). The National Center for Education Statistics (NCES) of the United States of America (USA), however, indicates that this demographic is changing at a fast pace in the educational environment of the 21st century. It is projected that learner enrolments of HEIs learners between the ages of 25 and 34 will far outpace learner enrolments of 18 to 24 years olds over the decade from 2010–2020, due to changing employee and employment requirements (NCES 2008:9). In the USA it is also expected that enrolments among learners of 35 years or older will increase (NCES 2008:9) and this trend may also become a reality in South Africa. These learners already face challenges of studying and working full-time simultaneously and therefore, accommodating working adults requires that innovative instructional technologies are effectively brought into play to make university studies possible to this group of learners (Amirault & Visser 2009:70). A great number of HEIs have already made provision for these working adults as they became aware of the increasing demand of this group of learners and have thus gradually improved their distance learning programmes and e-learning programmes (Amirault & Visser 2009:70).

It is necessary to keep in mind, however, that merely a change in modality, such as offering programmes via e-learning, is not enough to meet the learning needs of these learners. Simply creating the path to the classroom (in this case, the virtual classroom), is not sufficient; a classroom environment that is attractive to learners to choose time and time again is equally important (Amirault & Visser 2009:70). This is where the need for an e-learning skill of academics comes into play again. Not to simply provide an online course through making use of the basic applications (making study material available, post announcements, notes or power point presentations), but to know how to actively and effectively facilitate a course online, constructing knowledge in an interesting an appealing manner to keep the learning experience of the learner valuable and engaging.

5.2.3 Lifelong learning

The traditional way of learners completing higher education, the attainment of a first degree preceding entering the work environment, is no longer sufficient for a knowledge workforce (Amirault & Visser 2009:71; Ramsden 2008:8; cf Markkula 2006:15). The *World Bank report on lifelong learning* in the global knowledge economy¹¹ (2003:16) explains that organisations can no longer rely exclusively on new graduates as the main source of new skills and knowledge in labour market; they will increasingly need workers who are eager and able to continuously update their skills throughout their lifetimes, by way of lifelong learning.

A knowledge economy cannot be possible without the existence of advanced technological infrastructure (Auer & Cazes 2000:379; Amirault & Visser 2009:66; cf Brinkley 2010; cf Markkula 2006:16). This implies a workforce with technological know-how. The application of new technologies has significant implications for learning and the application of knowledge. Lifelong learning becomes an inevitability of which its framework includes learning throughout the lifecycle. Lifelong learning becomes progressively more crucial for countries to be competitive in the global knowledge economy and can be called the education for the knowledge economy. Within this lifelong learning framework, the most important aspect is learning and meeting learners' needs. (World Bank 2003:xiii.)

It is evident that the use and application of technology play a significant role in preparing learners for the knowledge workforce which compels the adoption of

¹¹ The knowledge economy includes bringing together powerful computers and well-educated people in order to meet the growing demand for knowledge-based goods and services (Brinkley 2006:3).

lifelong learning. Through the internet learners have fast access to resources that were not traditionally available with classroom teaching as only means of teaching and learning (cf Herselman & Britton 2002:270–273).

5.2.4 Increased competition among HEIs

It is necessary to take cognisance of the augmented competition from other educational institutions such as private sector training institutions, corporate universities, and content brokers (World Bank 2003:16;18–19). The number of corporate universities has grown by approximately 500% in the period of 1988 to 2002 (World Bank 2003:19). Therefore, if HEIs do not make provision in this regard, they will lose their competitive edge.

The Organisation for Economic Co-operation and Development (OECD) has already, since the 1990s, indicated that education will be pivotal to the knowledge-based economy, and learning will be the instrument of both individual and organisational development (OECD 1996:14; cf Markkula 2006:4). The demand for higher education is progressively increasing as employees seek a competitive advantage in the knowledge economy (Pusser 2002:105; cf ESRC 2008:2;4). It can be argued that the increased need for higher education among working individuals/adults and the changing demographics of learners resulted in a higher demand for higher education of non-traditional learners. It can further be argued that public HEIs, such as universities, must ensure they provide in the needs of the adult learner to prevent them from loosing learners to private sector HEIs.

From the above discussion on the types of learners and unique challenges faced by HEIs in the 21st century it can be deduced that HEIs are facing problems different from what they have traditionally experienced. It is further also apparent that HEIs need to embrace the use of technology in teaching and learning, considering not only the technological advancement, but also the changing educational environment, the rise of the knowledge economy, the emphasis on and need for lifelong learning and the changed profile of the 21st century learner. From the abovementioned discussion it is clear that technology has a significant impact on higher education, not only to enhance the teaching and learning experience for learners that prefer this

way of learning (Generation Y), but also to prepare learners for the technologically advanced knowledge workforce.

It becomes more and more apparent that academics at HEIs no more have a choice in using technology in teaching and learning. It is also evident that merely providing learners with the opportunity to use the e-learning platform is not sufficient (eg by posting study material or power point presentations on the e-learning platform); elearning should be used as pedagogical tool (through the use and adaptation of various interactive teaching and learning strategies), facilitating learning, collaboration and knowledge construction and providing opportunity for optimum interactivity and engagement. It is thus evident that academics that lack the ereadiness to take up this challenge must undergo training and development to become competent for the new job demand. It is the responsibility of the HEI as employer to ensure that employees receive the necessary training and development opportunities to become e-ready.

Throughout the discussion it remains evident that new, innovative teaching and learning approaches are required, including the use of technology in teaching and learning, that necessitates the e-readiness of academics. The next section subsequently focuses on the e-readiness of academics.

5.3 E-READINESS FOR ONLINE LEARNING FACILITATORS

With a view to emphasise the importance of e-readiness of academics in HEIs, the following paragraphs focus on e-readiness indicators of academics to determine the acceptable level of e-readiness for employees to act as online learning facilitators. As the perceived lack of e-readiness is usually the common contributor to resistance to e-learning, the section also finds it appropriate to reflect on resistance to e-learning as a critical requirement for effective teaching and learning by HEIs.

In order to determine e-readiness indicators for an academic's online learning duties, it is necessary to determine the role and responsibilities of an online learning facilitator as these will give an indication of the required e-readiness indicators.

List of research project topics and materials

5.3.1 Role and responsibilities of an online learning facilitator

The role and responsibilities of an online learning facilitator have been outlined in chapter 1 (section 1.8). Online learning facilitation skills include:

- engaging the learner in the learning process, particularly at the beginning
- appropriate questioning, listening and feedback skills
- the ability to provide direction and support to learners
- skills in managing online discussion
- ability to build online teams
- a capacity for relationship building
- motivational skills (ANTA 2003:3; AFLF 2003:3).

Skills required of an online learning facilitator such as being a good listener, providing feedback, communication and encouraging learners and building relationships, mentioned above, all relate to the human nature of e-learning. Online learning facilitators should possess the ability to let learners feel comfortable about participating in online discussions and sharing information (UKNCSL 2006; Sutton 2004).

It is therefore evident that an online learning facilitator should be focused on not only the technical aspects of online learning, but also on the learning experience of the learner and social nature of learning, thus the human side of learning. According to Salmon (2003:4) the primary role of an online learning facilitator is to promote human interaction and communication through knowledge construction and modelling, as well as transferring of skills and knowledge. Salmon (2003:4) further explains that the task of acting as an online learning facilitator requires a combination of fresh insights and technical skill. More so, it calls for understanding and appreciating the management of online learning and group working, thus focusing on the human aspects and variables of learning (Salmon 2003:4).

From the various traits and attributes indicated in chapter 1 and this chapter, an online learning facilitator can be defined for the purpose of this study as follows:

An online learning facilitator is a person who guides the learning process, facilitates knowledge construction, encourages learners to participate, promotes interaction and communication between facilitator and learner, as well as between learner and learner, offers technical assistance, enhances and promotes collaborative learning, manages online meetings, designs and develops online courses and creates a friendly and comfortable environment for learners.

It is apparent that the job of an online learning facilitator is comprehensive and requires not only a technical skill, but also interpersonal skills. Being an online learning facilitator also requires having the ability to adapt teaching and learning strategies to suit the needs and preferences of the learner and the medium of learning (e-learning), as emphasised in chapter 3 and this chapter. These skills, competencies and abilities, required of an online learning facilitator, demand e-readiness of academics. The following paragraphs therefore discusses the indicators of e-readiness, namely what are the requirements for an academic to be regarded as e-ready to take up the role of online learning facilitator. These indicators will become the assessment criteria against which an academic's e-readiness will be assessed during a performance appraisal. It is thus necessary to create an *e-readiness construct* for this purpose.

There is not a one-size-fits-all profile for an online learning facilitator. However, Llorens, Salanova and Grau (2002:208) are of the opinion that particular characteristics have to be present to realise the successful facilitation of e-learning. Primary characteristics and learning skills for e-learning are adaptability to change, creativity and innovation, empathy, identification of opportunities in the given context, effective management of time and aims, potential for teamwork, high self-esteem, and mental flexibility (Llorens et al 2002:208). Subsequently, the e-readiness indicators for an academic's online learning duties can now be determined.

5.3.2 E-readiness indicators

As the thesis argues for the incorporation of the assessment of the readiness of academics to use electronic tools in delivering learning, it becomes appropriate to

determine the indicators of e-readiness, namely which factors indicate academics' level of e-readiness. These discussions intend to inform the development of an e-readiness construct.

• Technical experience and computer literacy

According to Guglielmino and Guglielmino (2003) in Moolman (2007:65), e-readiness of individuals can be assessed by evaluating amongst others an individual's technical experience and competency with computers (cf Schreurs et al 2008:267; cf Lee-Post 2009:66). In addition to these competencies, an individual should have the capability to direct his/her own learning, by means of relevant knowledge, attitudes, skills and habits (Guglielmino & Guglielmino 2003 in Moolman 2007:65). An individual's e-readiness levels may or may not be supported by the readiness of the organisation, therefore managers in organisations need to guide and support e-learning processes for it to be considered a supporting factor to the e-readiness of an individual (Moolman, 2007:65). It is also necessary to emphasise the importance of learning programme (Swinton 2006:1; Roy 2006:22), as discussed in chapter 4. Aptitude, prior knowledge and motivation will also play a role in an employee's e-readiness (Guglielmino & Guglielmino 2003 in Moolman 2007:65; cf Reeves 1997).

• Charisma and communication skills

Miller (2003:1) mentions the need for charisma as trait of an online learning facilitator. She describes charisma as a "magnetic charm or appeal" to attract learners. A charismatic online learning facilitator will have the ability to communicate an in-depth understanding of the subject matter as well as a wide spectrum of life experiences and will, through this behaviour, bridge the gap created by distance and time. A charismatic online learning facilitator therefore portrays personality traits that will enhance the online learning experience of the learner. Outstanding written communication skills is an asset for online learning facilitation, but communicating with charisma will include a personal touch to compensate for what is lacking visually and verbally. Furthermore, online messages should ideally not be written with firm

devotion to grammatical guidelines, but rather be informal and with more emotion. An online learning facilitator can in this regard also make use of emoticons. A charismatic online learning facilitator may be able to help build the bond between teacher and learner due to his/her particular personality traits. (Miller 2005:2–8.)

• Self-awareness

Salmon (2003:56) identifies self-awareness, consideration, and the ability to influence as important characteristics of an online learning facilitator.

In addition to the abovementioned e-readiness indicators, Moolman (2007:79) identifies the following indicators that will give an indication whether a person is e-ready:

- access and connectivity to an IT infrastructure
- experience of technology
- utilisation of the platform provided by the organisation
- motivation to become a self-driven learner
- exercising of learning styles and attitudes to take responsibility of training
- experiencing a culture of learning and support.

The first three indicators mentioned by Moolman (2007:79) refer to the physical readiness for e-learning: bandwidth, hardware, software and the technical skill to teach online, assess assignments online, compile e-tests and/or quizzes online, upload material to an e-learning platform, facilitate discussions on the forum, post messages and announcements on the e-learning platform and compile and produce podcasts.

The last three indicators of Moolman (2007:79) focus on the intrinsic human factors such as motivation, learning style, attitude and a culture of support (which creates a sense of belonging). These indicators support the significance of human factors in the e-readiness of academics and should be considered during e-readiness assessment and the subsequent training and development interventions.

In addition to the abovementioned indicators of e-readiness, indicated by various authors on e-learning, the following additional indicators can be identified for the purpose of this thesis:

- Comfortableness with teaching online: E-readiness implies that a person should be comfortable with the online environment (Information Technologies Group 2000). Usually the better the technical skill of an online learning facilitator, the more comfortable he/she is with teaching online (Le Roux 2009; Focus Group 2006). One can argue that a high level of comfortableness with teaching online will most likely result in a motivated and enthusiastic online learning facilitator, which in turn will motivate and encourage learners to be enthusiastic about the online course and will enhance their performance.
- Willingness to teach online: One can assume that an academic should be willing to teach online to be regarded as e-ready (Mitra 2010; Wolf 2006:49; Kosak, Manning, Dobson, Rogerson, Cotnam, Colaric & McFadden 2004; Palloff & Pratt 2000:4; Information Technologies Group 2000). A lack of willingness will portray a lack of e-readiness. The willingness refers to psychological aspects and an attitude towards e-learning (Juutinen & Saariluoma 2009; Clarke & Mayer 2007:29; Information Technologies Group 2000). Human factors such as the learning style, pace and style of technology adoption and the personal work profile pattern of an employee will influence the employee's willingness to teach online, as discussed in the previous chapter. This study argues that, although it is difficult to measure an inherent quality or preference such as a learning style or personal profile pattern, these human factors will give an indication of the most preferred employee development approach that should be followed.
- Use and purpose of e-learning: As e-learning was initially used mostly as distance learning or remote learning tool (Fuller 2009), the perception of this being the only manner in which to use e-learning may still be prevalent amongst academics. This perceived ignorance can cause resistance to e-learning and therefore calls for the awareness of the philosophy of e-learning. Academics should undergo e-learning training, broader that learning a technical
skill, but should also be schooled in the use and purpose of e-learning. Previous chapters recorded that e-learning has become included into a broader debate on the enhancement of learning by more effective and comprehensive use of digital technologies (JISC 2009:8). The focus of e-learning is therefore not to replace the teacher, but to use technology to increase the focus on pedagogic skills (JISC 2009:5). It is also important that academics are aware of the pedagogical use of e-learning and do not merely view it as technology. The "e" in e-learning, referring to "electronic", adds a technological edge to the "learning", but the focus remains on *learning* as the critical element and e-learning can therefore be seen as enhanced learning (JISC 2009:8). The focus of *e-learning* is thus not on technology, but on *learning*, supported by technology (Armstrong 2006b:583).

Knowledge and adaptability of teaching and learning strategies: As indicated in section 3.2.3, people shape their personal theories regarding teaching and learning early on in their lives and carry it over from generation to generation (Elgort 2005:184). A person usually teaches in the manner in which a person is taught and that teachers' principles, approaches, attitudes, and practices is likely to be connected to prior experiences (Thomas & Pederson 2003:319). Academics also have a tendency to use ICT tools only if these tools are aligned with their own philosophy of teaching and learning (Robertson 2004 in Elgort 2005:184). A cademics are usually subject specialists and not educational specialists. In many instances academics are not sure what is expected of them in terms of pedagogical expectations. Thus, it is advisable that e-learning training goes hand in hand with training pertaining to the use and adaptability of teaching and learning strategies. In this regard the constructivist approach, as outlined in section 3.2.3 can be considered.

Online learning facilitators at HEIs should meet a minimum level of e-readiness in order to fulfil their online duties satisfactory. This minimum level of e-readiness refers to extrinsic factors such as internet accessibility, adequate bandwidth, organisational and technical support and the necessary hardware and software. These factors need to be provided by the organisation and are in place to an acceptable level at the NWU, as allowed for within the parameters of bandwidth, accessibility and

connectivity in the South African environment (Le Roux 2009). In addition to these external factors, a technical skill and particular human factors are also necessary.

Also required for an acceptable level of e-readiness is to have the required technical skill to teach online: be able to access the e-learning platform, post messages, announcements and documentation and communicate asynchronously with learners. Although this can be regarded as an acceptable level of e-readiness, to excel at teaching online and to make the learning experience valuable and effective, interactive learning is necessary, which calls for the use of forums, blogs, podcasts, wikis, etcetera and to design on an e-learning platform (Le Roux 2009).

However, for this study, most important are the human factors of e-readiness: emotional, essential, personal issues, which are intrinsic and related to a person's beliefs and attitudes (Bozarth 2006:2). Aspects such as concept of work role, beliefs about practice, beliefs about quality of e-learning, personality factors, and the perception of technology as a supporting device in learning rather than an enabler and enhancer of learning as indicated in the previous section. It is also apparent that Bozarth's second order factors (soft skills) can be related to Honey and Mumford's learning styles, Thomas International's DISC personal profile patterns and Rogers' technology adoption cycle, outlined in chapter 4, section 4.2. The specifics of these indicators will be indicated in the e-readiness construct.

Durrheim and Painter (2009:142) describe a *construct* as "attributes that have been conceptualised and defined in language, and which have been theoretically elaborated in terms of how they are related to other constructs". For example, the attributes of people that a researcher wishes to assess is seen as a methodically arranged set of ideas, that is as constructs (Durrheim & Painter 2009:142). Cronbach and Meehl (1955:3) support these notions as they define a construct as an assumed attribute of people that is believed to be revealed in assessment performance. Babbie (2005:124) explains that constructs are not real, but they are valuable and helpful to the researcher to systematise, and comprehend things that are real and to communicate about these things. Constructs therefore assist in making predictions regarding real things (Babbie 2005:125), as they stand in an explicit relationship to things that are real and observable (Johannes 2007:12).

An **e-readiness construct** therefore consists of the attributes/indicators that will provide an indication of an employee's level of e-readiness, systematically arranged as a set of standards regarding e-readiness. This arranged set of ideas can be compared to other constructs, in this case key human factors of academics. The e-readiness construct assisted the researcher in understanding the contributing factors to the e-readiness of academics. Thus it could be determined what the solutions are to a lack of e-readiness. As the e-readiness construct as people attribute reveals itself in performance assessment, the study argues for the inclusion of e-readiness assessment in performance appraisals of academics. The e-readiness construct is indicated in table 5.2 below:

Table 5.2: The e-readiness construct

Indicator	Specifications			
Technical skill	Basic computer skills			
	Skill to use the e-learning platform tools			
	Basic tools – announcements, messages, schedule, resources,			
	assignments, gradebook			
	 Interactive/advanced – forum, e-testing, podcasts, wikis, bloggers, 			
	simulations			
	Technical support to learners			
Subject competency	Subject expert – knowledge			
	Curriculum development			
	Development of course material			
	Facilitate knowledge construction			
	Well-structured assessments			
	Guidance and assistance in terms of subject related problems			
	Providing feedback			
Comfortableness	Ease of use			
	Lack of fear/resistance			
	Enjoyment of online activities			
Willingness	Positive attitude			
	Adaptability and flexibility to adjust to change			
	Focusing on outcome of learning			
	Considering learner profile			
Understand use and purpose of e-	Philosophy of e-learning – understand e-learning as pedagogical tool			
learning	Use technology to achieve learning			
Knowledge and use of teaching and	Knowledge and skill of pedagogical strategies an approaches			
learning strategies	Flexibility with teaching and learning strategies			
	Educational "savvy"			
	Considering learner profile			
Interpersonal skills	Communication – sharing information; provide direction & support; online			
	charisma			
	Motivation – encourage learners			
	Providing feedback			
	Accommodating			
	Counselling skills			
	Good rapport with learners			
	Ability to bridge the gap – building relationships			
	Mental flexibility			
Demonstrativiliantes	Compassion/empathy			
Personal attributes	Diligence			
	Perseverance			
	• Commitment			
	Openness Openticity and inclusion of the section			
	Creativity and innovation			
	Time management			

For the effective pedagogical use of technology in teaching and learning, an academic's e-readiness should comply with these requirements. It can be argued that initially basic technical skills will suffice like basic computers skills and the use of basic e-learning platform tools (announcements, messages, schedule, resources, assignments, gradebook). Advanced use of e-learning tools can be included in the employee's development programme, once the employee is comfortable with the basic tools and had sufficient time to practice it. In this way, the advanced development course will be a natural progression for the employee.

Subject competency should be regarded as a fundamental, irrespective of the learning modality. A lack of comfortableness and willingness can be regarded as the indicators that usually are some of the biggest contributors to e-learning resistance. Therefore, these factors should receive attention. Resistance to e-learning is usually experienced before the skill is obtained and after training to master the skill, most academics are more comfortable to teach online (Le Roux 2009), unless an online learning facilitator falls in the category of *late majority* or *laggards* according to the technology adoption cycle as discussed in section 3.2. The *late majority* category is likely to resist e-learning as long as possible and will not necessarily be comfortable with it when they eventually are trained and using technology (Zemsky & Massey 2004:10). The *laggards* will never adapt to technology (Zemsky & Massey 2004:10).

Two other categories vitally important are the use and purpose of e-learning and the use and knowledge of teaching and learning strategies. These two components should be included in development programmes from the start. Addressing the use and purpose of e-learning will immediately create awareness of what is expected and how technology should be viewed in education, addressing the resistance and thus, the lack of e-readiness. Knowledge about the use of teaching and learning strategies is also imperative. It can be argued that the knowledge of these strategies will enhance an employee's e-readiness as they will have a clearer idea of how to approach online teaching and learning. It can be argued that it should be made compulsory as it is the only way to guarantee that employees have the knowledge and skill. It should, however, be done with a great amount of care and motivation from line managers. As indicated in chapter 2, sections 2.2.1, 2.3.2 and 2.3.3.1, the

provision of a rationale enhance an employee's sense of autonomy, which positively impact on motivation and ultimately task performance.

As for interpersonal skills and personal attributes, there is not a one-size-fits-all. Chapter 4 and this chapter emphasised the differences of employees in terms of human factors, which in return impacts on their behaviours, attitudes and approaches in the workplace. Therefore, based on how employees rated with regard to technology adoption, learning style and personal work profile pattern, it will give an indication as to the presence of these indicators.

Based on the discussion above and the e-readiness construct, it can be argued that the minimum required level of e-readiness of an academic can be regarded as the inclusion of the following indicators: technical skill in the form of basic computer skills and basic e-learning platform skills; subject competency; willingness; comfortableness; an understanding of the use and purpose of e-learning; and knowledge and skill to use various teaching and learning strategies.

It is evident that all the indicators, indicated in table 5.2, will be influenced by the key human factors identified in this study: pace and style of technology adoption, preferred learning style and personal work profile pattern. For example, an employee with a high influence DISC factor is likely to do well with the human side of elearning, for example communication, motivation, compassion and counselling skills, as a high *influence* DISC factor is concerned about the manner in which people are dealt with and are typically charming, optimistic, and outgoing, and focused on networking, conversation, and working with others (Thomas International sa). An employee with a high *dominance* DISC factor or an *innovator* technology adoption style is likely to do well with the technical skills, subject competency, comfortableness and willingness, as a high dominance factor and innovators are competitive, with high performance standards, and focused on achieving goals, solving problems, enjoy exploring new ideas and accept challenges (Thomas International sa; Zemsky & Massey 2004:9). The same can be said about an employee with an *activist* learning style that also prefers to deal with challenges (Honey & Mumford 1982). However an employee who is an *innovator*, prefer an

141

activist learning style or portray a high *dominance* factor, does not necessarily have good interpersonal skills.

These indicators are linked to an employees' e-profile, that will be determined during e-readiness assessment (by assessing the employee's key human factors), and will provide insight into which attributes will need attention during training and development interventions. As it will indicate areas for development and should thus be included in a development plan.

It is apparent that a number of factors can contribute to an academic's e-readiness or lack thereof, which may cause resistance to the changed job demand (cf Robbins 2003:559–560). The next section will outline some of these barriers to e-learning, relevant to this study.

5.3.3 Resistance to e-learning

It can be argued that employees resist e-learning because of a lack of e-readiness. As mentioned in previous chapters, employees usually resist change in the workplace due to various reasons such as amongst others fear of the unknown, habits (comfort zones), inadequate information and communication, threats to status, fear of failure and lack of perceived benefits (Proctor & Doukakis 2003:268; Robbins 2003:559–560). It is therefore necessary to explore the barriers to e-learning with the aim of addressing it and to determine whether it will simultaneously provide insight into the e-readiness of academics.

Bozarth (2006:2) indicates that barriers to change (such as employees' resistance to the inclusion of technology in teaching and learning) can be classified into first order or second order barriers. First order barriers refer to extrinsic factors such as organisational support or access to equipment and second order barriers refer to more emotional, essential, personal issues, which are intrinsic and related to a person's beliefs and attitudes (Bozarth 2006:2). Important to this study is the underlying, second order barriers of resistance to e-learning amongst academics. As mentioned in the previous section, factors that cause first order barriers are in place at the NWU, Potchefstroom campus: access to equipment, the internet, hardware

and software, development support and technical support (Le Roux 2009). This study therefore does not focus on first order barriers, but places the focus on second order barriers that relate to the human factors in technology.

Second order barriers relate to aspects such as concept of work role, beliefs about practice, beliefs about quality of e-learning, personality factors, and the perception of technology as a supporting device in learning rather than an enabler and enhancer of learning (Bozarth 2006:2). These human factors (second order factors) is addressed in chapter 4 of this study by explaining an online learning facilitator's learning style, personal profile pattern and pace of adoption to technology.

When factors that cause resistance to e-learning are identified, one option of minimising their impact is to increase participation in employee development (Barak & Cleveland-Innes 2006:118). The study supported this view in chapter 3, indicating that academics should take responsibility for their own development and jointly set goals with their managers. Olgren (2000:7) in Barak and Cleveland-Innes (2006:118) argue that all employees are unique and multifaceted and therefore go into a learning experience with different backgrounds, attitudes, skills, and motivations. Burge (2000:90) in Barak Cleveland-Innes (2006:118) proposes that managers focus on finding the blocking factors of employees' resistance to e-learning and remove them. It is important to consider not only the motivating factors that encourage employees to embrace e-learning, but also the factors that cause resistance to e-learning (Barak & Cleveland-Innes 2006:118).

Bozarth (2006:3–8) indicates a number of reasons why online learning facilitators are not e-ready, namely work roles, practice and beliefs about teaching, beliefs about quality of e-learning, personality factors, and vision. The following barriers to ereadiness can be identified (Bozarth 2006:3–8):

• Work roles

E-learning brought a shift from the traditional hierarchy between teacher and learner from the traditional view of the teacher as authority and expert, to a role of the teacher as guide and the learner as explorer. Many facilitators fear that their specialised status and position as "expert" will be downgraded to the roles

of production worker, simply typing out content to be put online, and customer service representative that should be available 24/7 to assist learners with technical problems. Also creating resistance are a belief of loss of routine and the "old way" of doing things, losing control and a fear of being replaced by technology. (Bozarth 2006:3–4; cf Billings sa.)

Academics, viewing e-learning from this perspective, will most likely adopt very slowly to the use of technology in teaching and learning and will in all likelihood resort under the *late majority* category on the technology adoption cycle, since they dislike the disruptions of new technologies, have conservative thinking and therefore are likely to resist a change in work demand or job requirements (Zemsky & Massey 2004:10). It is clear that academics, resisting e-learning for this reason are not well-informed about the role and purpose of e-learning. As alluded to in various sections of this study, the focus of e-learning is not to replace the teacher, but to use technology to increase the focus on pedagogic skills and learning (JISC 2009:5;8). Academics should be made aware of the pedagogical use of e-learning and should not merely view it as technology or a threat to their work roles.

Practice and beliefs about teaching

Online learning facilitators that are regarded as technologically advanced, judged by the extent and manner of their use of technology and online learning approaches, tend to use constructivist strategies such as inquiry learning and collaborative learning. These online learning facilitators plan lectures to meet the needs of individual learners and wish to instil in learners a sense of curiosity and desire to learn. For them, the goal is not to provide big chunks of information, but to equip learners to find answers themselves. These online learning facilitators are also likely to adjust their teaching practice in answer to learner needs. They furthermore view teaching and learning as an everchanging activity that develops over time, and are excited by trying innovative methods of teaching and learning. They are eager to participate in professional development opportunities, even without encouragement or incentive. (Bozarth 2006:4–5.)

It can be argued that this type of academic will be categorised as an *innovator* on the technology adoption cycle since they enjoy exploring new ideas and are driven by intrinsic motivators (Zemsky & Massey 2004:9), as outlined in section 4.2.1 of chapter 4. One can also assume that such an employee will portray the *activist* learning style of Honey and Mumford (1982) since they prefer to deal with new challenges and experiences, is action oriented, and are good adaptors, as outlined in section 4.2.2 of chapter 4. This type of academic will most likely be categorised with a high *dominance* factor on Thomas International's (sa) DISC personal profile patterns, since they are competitive, have high performance standards, are focused on achieving goals and solving problems, and easily accept challenges as described in section 4.2.3 of chapter 4.

On the other hand, academics that resist new approaches and technologies are more often than not operating from a teacher-centred perspective, viewing themselves as the one and only source of knowledge (Bozarth 2006:5; Hanley 1994:3). These employees are typically fearful that technology may interfere with their association of control and authority (Hanley 1994:3). Whereas learner-centred instructors make use of freely-structured lecture plans and innovative learning techniques, instructors who use a more teacher-centred approach follow the textbook and lesson plan strictly (Jonassen 1991:28). These online learning facilitators therefore follow a traditional, behaviourist approach (Bozarth 2006:5; Jonassen 1991:28).

It can be argued that academics that follow the teacher-centred approach will be more likely to portray a lack of e-readiness than those who follow a learnercentred approach. This type of employee portrays distinct similarities with the *late majority* category on the technology adoption cycle since they dislike the disruptions of new technologies and are conservative in their thinking (Zemsky & Massey 2004:10), as outlined in section 4.2.1 of chapter 4. One can furthermore assume that such an employee will portray the *theorist* learning style of Honey and Mumford (1982), since they prefer and need sufficient time to explore the relevance between ideas and scenarios and need to review information, analyse, and form abstract concepts and generalisations before acting, as outlined in section 4.2.2 of chapter 4. This type of employee will most likely be categorised with a high *compliance* factor on Thomas International's DISC personal profile patterns, since they consider the manner in which rules and procedures are followed, are concerned with accuracy, consider each possibility before making a decision, and prefer systems, processes, procedures, as well as predictable and consistent outcomes (Thomas International, sa), as described in section 4.4.3 of chapter 4.

Beliefs about quality of e-learning

For many academics, the start of e-learning brought many challenges and many ideas and philosophies about adult learning and what constitutes up-tostandard acceptable teaching (Robertson 2008:824). Implementation of elearning also often requires that academics should challenge their belief systems and ideas regarding what comprise teaching and learning (Sharpe, Benfield & Francis 2006:135; Robertson 2008:824). It is advisable that educators engage "in dialogue about personal practical theories of learning and teaching" (Errington 2004:43) in order to review and revise their theories and practices to support sustainable changes to teaching practice (Robertson 2008:824). It is again emphasised that the willingness to use e-learning usually goes hand-in-hand with the willingness to adapt teaching and learning strategies.

Personality factors

Distinct patterns between personality traits and approaches to work relative to the use of technology in teaching and learning can be seen (Bozarth 2006:6). Academics who successfully integrate technology with teaching and learning are often described as pioneers, explorers and risk-takers (Zemsky & Massey 2004:9) and also display a broad-mindedness for uncertainty and willingness to make mistakes (Bozarth 2006:6–7). Early adopters to technology can further be described as finding excitement in innovative ways of working and perceiving use of a fresh approach as an intellectual challenge (Zemsky & Massey 2004:9).

It is evident that human factors such as personality play a role to the use of technology and thus e-readiness. Section 4.2.2 recorded a relationship between personality and learning styles. In this regard an employee's position on the technology adoption cycle and his/her personal work profile pattern will also play a role as they are human factors, relating to personality (Thomas International sa).

Vision for education and training

Online learning facilitators who successfully integrate technology in teaching and learning have a common belief that technology provides a valuable tool to realise their visions of teaching and learning. These facilitators have strong visions of classroom technology use and therefore do not appear to be easily disturbed by typical implementation obstacles. On the contrary, many of these online learning facilitators achieve high levels of use despite the lack of equipment, training, or time. Their refusal to give up in the face of difficulty allows them to overcome obstacles that usually keep others from proceeding. On the other hand, online learning facilitators who view technology as a presentation tool, or "add-on", rather than the means of enhancing practice, are far more likely to report barriers. (Bozarth 2006:7–8.)

Most of the academics at the NWU, Potchefstroom campus, use the e-learning platform as an "add-on" to classroom teaching, where information (announcements, study notes, power point presentations and other documents) are distributed and not as an interactive learning tool (Le Roux 2009). As it is used merely as a distribution mechanism, e-learning is not applied to its fullest extent as a learning tool and will therefore not produce enhanced learning results.

The abovementioned barriers to e-readiness supports various notions/conclusions made in the previous chapter with regard to personal profile patterns, learning styles and pace and adoption to technology:

147

- Employees' position on the technology adoption cycle will influence their perceptions of teaching and learning online. Employees who adapt easily to the use of technology in teaching and learning will be unlikely to portray a lack of ereadiness and will in most probability use the e-learning platform to enhance the teaching and learning experience of learners. On the other hand, employees who adapt slowly or resist the use of technology in teaching and learning will be more likely to portray a lack of e-readiness and will in most probability need more motivation and development opportunities to use the elearning platform maximally.
- Employees' personal work style preferences will influence their adaptability to the use of technology in teaching and learning and therefore, their level of resistance to e-learning. Some employees (eg innovators) will eagerly embrace new challenges and changes in the workplace, whereas others (eg late majority) will slowly adapt to changes in the workplace.
- Employees' learning styles will influence their preferred manner in which employee development should take place. Learning styles also indicate how easily or slowly an employee will adapt to new job demands.

It is thus evident that addressing barriers to e-readiness should be approached differently with each online learning facilitator, based on his/her personal profile pattern, learning style and pace and style of technology adoption. It is furthermore evident that an online learning facilitator's practices and beliefs pertaining to teaching and learning in general, will influence his/her perception on the use of e-learning. It may be necessary to first address an online learning facilitator's teaching and learning theories/beliefs before the technology is introduced, as argued in the previous section.

5.4 CONCLUSION

This chapter highlighted the 21st century higher education environment and focused on various challenges posed to HEIs, regarding the changing demographics of students, a shift from traditional learning towards lifelong learning and the increased competition among HEIs. Given the 21st century higher education context, the chapter also focused on two types of learners and their expectations: Generation Y and underprepared learners.

Generation Y learners have teaching and learning needs and preferences, different from traditional learners, that should be catered for by HEIs if they do not want to lose their competitive edge. One of the most significant preferences of Generation Y learners, identified in this chapter, is their preference for the use of technology. It is thus apparent that HEIs need to embrace the use of technology in teaching and learning, considering not only the technological advancement, but also the changing educational environment, the rise of the knowledge economy, the emphasis on and need for lifelong learning and the changed profile of the 21st century learner.

Furthermore, in many instances, learners entering the higher education environment are underprepared for higher education. Thus, it is required of HEIs to find ways in which to improve the academic literacy of learners to enhance their learning and enable them to live up to the demands expected of them in their academic programmes. Academic underpreparedness especially relates to inadequate reading and writing skills. It can therefore be argued that HEIs should make provision for academic literacy programmes, enhancing these abilities of learners.

It became evident that the increased need for higher education among working individuals/adults and the changing demographics of learners resulted in a higher demand for higher education of non-traditional learners. It can further be argued that public HEIs, such as universities, must ensure they provide in the needs of the adult learner to prevent them from loosing learners to private sector HEIs.

Enabling factors to e-readiness were outlined and discussed and the following indicators of the e-readiness of academics have been established and included in an e-readiness construct:

- technical experience and computer literacy
- charisma and communication skills
- self-awareness

- access and connectivity to an IT infrastructure
- utilisation of the platform provided by the organisation
- motivation to become a self-driven learner
- exercising of learning styles and attitudes to take responsibility of training
- experiencing a culture of learning and support
- comfortableness with teaching online
- willingness to teach online
- use and purpose of e-learning
- knowledge and adaptability of teaching and learning strategies

Since a lack of e-readiness usually causes resistance to e-learning, various barriers to e-learning have been discussed. The underlying causes for e-learning resistance could be linked to a lack of e-readiness and emphasised the need for adequate and comprehensive e-learning training and development.

An empirical study has been conducted to verify the theoretical framework and assumptions made in this chapter and the preceding chapters. The next chapter discusses the results of the empirical study.



CHAPTER 6

PROFILING THE NORTH-WEST UNIVERSITY

6.1 INTRODUCTION

In preceding chapters a review of scholarly literature has been conducted on the various aspects related to key human factors in the e-readiness of academics. This chapter applies the theory derived from the literature review to the human resource practices at the NWU, Potchefstroom campus. Consequently, this chapter provides a profile of the current human resource performance appraisal processes and practices at the NWU. Further the NWU practices pertaining to employee development and related human resource practices such as career management, career development and talent management are also reviewed. The profiling of the performance management and employee development practices are being done with a view to determine whether provision is made for the assessment of the e-readiness of academics during their performance appraisals, as well as the subsequent employee and development interventions to follow these performance appraisals.

The chapter also profiles the NWU learner in terms of e-learning preference and to determine whether the average undergraduate learner can be categorised as a Generation Y learner.

6.2 HUMAN RESOURCE PERFORMANCE APPRAISAL AT THE NWU

As explained in chapter 2, the process of human resource performance appraisal takes place within the human resource performance management process of an organisation. At the NWU the performance appraisals of academics take place within the University's *Performance management policy*, which is aligned with the NWU *Institutional plan, 2010–2012.* One of the goals set in the *Institutional plan* is to optimise the integrated performance management system with suitable development opportunities, recognition programmes and incentive programmes for employees (2009b:9).

The NWU has approved its *Performance management policy* on 6 December 2005 with the following two primary objectives (NWU 2005a:2):

- to provide an integrated framework for the enhancement of performance and outputs of the University and all its sections
- to provide a structured process and framework for the identification of performance problems and developmental needs for employees

The *Performance management policy* of the NWU thus echoes the principle of identifying and providing in developmental needs of employees as part of the performance management process, as identified in literature and explained in chapter 2. This principle is also indicated as one of the guiding principles in the NWU *Performance management policy*: "A performance management system is primarily aimed at development, and should therefore be distinguished from other systems such as remuneration and promotion" (NWU 2005a:2) and the *Institutional plan, 2010–2012*, of the NWU: "Optimize the integrated performance management system, with appropriate development opportunities, recognition programmes and incentive programmes for staff at identified levels" (NWU 2009b:7). It is thus evident that the NWU management acknowledges the need for employee development. The policy (NWU 2005b:2) also makes provision for and describes the need for personal development plans for academics and indicates that the performance appraisals of academics is based on a performance agreement (Annexure G) and a personal development plan (Annexure H) (NWU 2005a:2).

Another guiding principle of the *Performance management policy* relevant to this study is: "Performance management should encourage creativity and innovation within an ordered system with a clear and agreed understanding between managers and staff on all levels" (NWU 2005a:2). To "encourage creativity and innovation" in the academic environment is a vital principle for the successful implementation of elearning. It is thus evident that the use of creative and innovative teaching methods, including the use of e-learning, is supported by the NWU. This principle is further supported by the University's *Teaching and learning framework* (NWU 2009a:16). The personal development plan makes provision for training and development in the

category "Teaching and learning" (see Annexure H). E-learning training and development can therefore be included in this section of an academics' personal development plan.

The *Performance management policy* (NWU 2005a:7) also makes provision for a self-evaluation form (Annexure I) to be completed by academics. This forms the basis of a development oriented assessment discussion with the relevant line manager. However, provision for the inclusion of technology in teaching and learning as such is not made on the self-evaluation form, but provision is made for the "Effective use of appropriate teaching media" (see Annexure I). Academics thus can assess themselves pertaining to the use of technology in teaching and learning. However, if an academic lacks e-readiness, this category will merely be used for the use of other appropriate teaching media. In this respect the line manager plays a significant role. To ensure that teaching and learning is optimised through e-learning and to comply with the University's mission, the line manager must ensure that academics are well-skilled and trained to provide quality teaching and learning.

From the *Performance management policy* and the *Teaching and learning framework* of the NWU, it is evident that e-readiness assessment is not included in the performance appraisals of academics at the NWU. The Director: Human Resource Management (HRM) at the NWU confirmed this and explained that the lack of e-readiness assessment of academics can be attributed to the fact that the use of technology in teaching and learning is not compulsory at this stage (De Wit 2010). He is, however, of opinion that this has to change due to the changing needs of university students and the technological era which necessitates that a mind shift needs to take place amongst academics in order to accomplish this (De Wit 2010).

The inclusion of the e-readiness assessment of academics in performance appraisals calls for new and adjusted policies. The current *Performance management policy* of the NWU will have to be adjusted to make provision for the inclusion of e-readiness. The introduction of e-readiness assessment in the performance appraisals of academics will also impact on other human resource policies:

153

- Recruitment if e-learning becomes a compulsory job requirement, it must be included in the Recruitment policy and indicated in job advertisements. Academics would have to be assessed on their e-readiness before they are appointed or they should be appointed on a probationary period and their permanent appointment should be subject to an acceptable level of ereadiness. In the latter case an academic can undergo a development programme, once he or she is appointed.
- Service contracts of academics If compulsory, e-learning should be included in the service contracts of academics and be described in *Human resource management policy* of the NWU.
- Teaching and learning framework The current Teaching and learning framework should be adjusted to make provision for the e-learning training and development of academics, broader than providing merely basic technical skills. Academics should also be trained how to incorporate and adjust new teaching and learning strategies and what the most appropriate ways of the use of e-learning will be to enhance teaching and learning training should be incorporated of a broader teaching.

It can be argued that, in addition to the above mentioned adjustments to policies, a separate policy for e-learning will be advisable and also be of value as it will outline the use and purpose of e-learning; the University's position on e-learning; as well as guidelines and directives of e-learning as teaching and learning tool. Prior to the adjustment or introduction of the abovementioned policies, it is necessary for Institutional Management of the NWU to consult all stakeholders, including the Workplace Forums and labour unions (SA 1995:63–64). It is thus evident that the inclusion of e-readiness assessment in performance appraisals will have a broader impact on the human resource management of the academics of the NWU than merely adjusting the human resource performance appraisal process and the *Performance management policy* of the NWU.

In chapters 2 and 3 of this study it has been alluded to that performance appraisal consists of not only an assessment component, but also a developmental

154

component. The next section subsequently reviews the employee development practices at the NWU.

6.3 EMPLOYEE DEVELOPMENT AT THE NWU

Chapter 3 pointed out that within the broader human resource management system, the process of employee development logically flows from an employee's performance appraisal from which an employee's development needs are identified (Rademan & De Vos 2001:54; cf Grobler et al 2002:260;266). It has also been mentioned that employee development is aimed at assuring that all employees possess the competence required to optimally perform their duties (Jacobs & Washington 2003:344). Academics who teach online, but are not trained as online learning facilitators, experience difficulty to meet the demands of the e-learning environment (Johannes 2007:6). E-learning training and development is therefore necessary.

In the interview with the Director: HRM at the NWU, Potchefstroom campus, he indicated that, given the fact that the world is using technology more and more and students are expecting e-learning to be included in their academic programmes, the NWU has to ensure that its academics are skilled to use e-learning. Therefore, academics need to be trained and developed to obtain this skill. The Director: HRM further indicated that, in his experience, according to their research/experience, academics, especially older employees, do not like the disruptions of new technology. He referred to the fact that current university students belong to a different generation than most of the academics and is therefore of the opinion that academics need to be made aware of the different generations and their needs. Alternatively the NWU stand the risk of losing its students if it does not adapt to and include e-learning in academic programmes. This necessitates the need for the assessment of e-readiness to ensure that academics are equipped to fulfil their online teaching duties. (De Wit 2010.)

The Director: HRM further confirmed that e-readiness assessment and the subsequent e-learning training have to be coupled with a personal development plan for each academic (De Wit 2010). He is of the opinion that it is the university's

responsibility to its staff to develop and empower them, even if they may not be employed by the NWU for the rest of their careers (De Wit 2010). This broad minded and visionary remark from the Director: HRM indicates that he comprehends the importance of training and development as motivational tool, leading to better performance. It is also refreshing that he is willing to make this investment in employees, irrespective whether they will be retained. The Director: HRM clearly grasps the global tendency to move away from being employed to being employable. Literature (Judhi, Pa'Wan, Othman & Moksin 2010:1; Vermeulen 2010:12; UNECE 2010:13; Sieber 2008:2) reveals that although employees mostly no longer can rely on organisations for long-term employment, they are increasingly expecting employability. This is likely to have a positive influence on employees' motivation and consequently enhance performance.

At the NWU, employee development, in terms of teaching and learning, is integrated in the *Institutional plan, 2010–2012*, of the NWU, calling for the continuous improvement of the quality of teaching and learning (NWU 2009b:14). It is therefore compels the NWU to have well-skilled and competent academics to provide such quality teaching. It can be argued that line managers, as being responsible for academics' performance appraisals, have to support academic staff in skills development, including an acceptable level of e-readiness.

6.3.1 Employee training and development programmes at the NWU

Employee development at the NWU is guided within the parameters of the *Teaching* and learning framework (2009), the *Teaching and learning policy* (2007) and the *Staff development policy* (2005) of the university. The *Teaching and learning* framework of the NWU outlines the broad approach of the University towards teaching and learning (NWU 2009a:i) and the alignment of teaching and learning goals, objectives, principles and practices to the goals and objectives of the *Institutional plan, 2010–2012* of the university (NWU 2009b:2–4). The *Teaching and* learning and learning distribution of the university is practiced within the parameters of national legislation and adheres to the principles for effective outcomes-based education in the Higher Education Sector (NWU

2007a:1). Secondly, the *Teaching and learning policy* aims to implement the mission statement of the NWU, to: "Develop, educate and empower through quality teaching and learning, well-rounded graduates, able to think laterally and critically and to serve the country and its people" (NWU 2007a:1). The objectives of the *Staff development policy* are to make the availability of competent, efficient and effective human resources for the NWU possible; and to provide a structured framework for the development and learning of employees (NWU 2005b:1).

The guiding principles stated in the *Staff development policy* (NWU 2005b:1–2) have already been identified as core aspects relating to employee development and performance management in literature and outlined in preceding sections of this study:

- "Opportunities for learning and development will be accessible to all staff members, provided that it is aligned with the needs and goals of the University and within budgetary constraints" (NWU 2005b:1). See sections 2.2.1 and 2.2.2 of this study.
- "The allocation of resources for staff development will be fair, as determined by the strategic aims and objectives of the University" (NWU 2005b:1). See section 2.2.1 of this study.
- "Staff development programs will be competency based, i.e. focusing on the development of knowledge, skills and attitudes/values as appropriate for the staff member in a specific position" (NWU 2005b:1). See sections 2.2.1, 2.2.2 and 3.2.1 of this study.
- "Staff development interventions must be planned and executed in a structured way. The line manager in participation with the staff member is responsible for the identification of staff development needs, in co-ordination with and based on assistance by the Human Resources department. Training may be provided in-house or sub contracted as needed" (NWU 2005b:1). See sections 2.2.1, 2.2.2, 3.2.1 and 3.2.2 of this study.
- "Staff development should follow from the annual formulation of a personal development plan for every staff member in a systematic and integrated manner" (NWU 2005b:1). See sections 2.2.1, 2.2.2, 3.2.2 and of this study.

- "This personal development plan results as the output from the performance management system" (NWU 2005b:1). See sections 2.2.1, 2.2.2 and 3.2.2 of this study.
- "The outcome of staff development interventions must add value to the attainment of the goals and objectives of the University" (NWU 2005b:2). See sections 2.2.1, 2.2.2, 3.2.2 and 3.2.4 of this study.

It is thus apparent that the NWU is committed to the enhancement of employee performance through structured and well-planned employee development opportunities. From the perspective of this study however, specifically exploring the e-readiness of academics, it is necessary to make adjustments or additions to the current policies and programmes to enhance the e-readiness of academics. It will include the consideration of key human factors such as an employee's personal work profile pattern, preferred learning style and pace and style of technology adoption, which are discussed in chapter 4. Further, provision for development of only the technical skill is currently included in e-learning training and should be expanded to include the use of teaching and learning strategies to optimally use e-learning as learning tool.

The *Teaching and learning framework* (2009a:17–18) outlines two development programmes currently offered by the Potchefstroom campus of the NWU:

• The Institutional course for new lecturers (ICNL)

The Institutional course for new lecturers (ICNL) is a compulsory programme for newly-appointed academics (NWU 2009a:17). The ICNL provides basic knowledge and skills pertaining to outcomes-based teaching principles and practices, and includes brief, concise training in research (NWU 2009a:17). The ICNL runs in two phases (NWU 2009a:17):

 Phase I is offered by the Institutional academic and development and support (ADS) office, consists of information training sessions and workshops, focusing on various aspects of both research and of teaching and learning, including mini-contact sessions presented by the new lecturers themselves.

- Phase II, a mentoring programme, is arranged by the ADS office of a particular campus and requires the presentation of a contact session by a new academic, under the guidance and leadership of a mentor from the particular campus' ADS office and a subject-specialist from the particular school in which the new academic is employed (see Annexure J) for the evaluation form of the contact session. Phase II also includes the attendance of three compulsory workshops (NWU 2009a:17):
 - the writing and use of a study guide
 - group-work as teaching tool
 - e-learning (basic training on the e-learning platform)

It is apparent that the ICNL is addressing sound teaching and learning principles and practices. However, the basic training of the use of the e-learning platform's tools can be identified as a shortcoming. Academics, as subject experts and not educational experts, should be schooled in the use of teaching and learning strategies and how to apply these with the use of technology in teaching and learning. Although teaching and learning strategies are included on a foundational level, it is not specifically targeted for the use of technology in teaching and learning and is mostly focused on classroom teaching and learning.

• The Institutional teaching excellence award (ITEA)

Another development programme offered by the NWU is the Institutional teaching excellence award (ITEA). The aim of is to encourage academics to enhance their teaching skills and to give evidence-driven classes (see Annexure K) for contact session requirements and Annexure L for the assessment scorecard for contact sessions) and a portfolio with study material, assessments, the use of teaching and learning strategies, the use of media and innovation, peer assessment, student evaluation and the academic's teaching philosophy (see Annexure M) for portfolio requirements and Annexure N for the assessment scorecard of the portfolio) to prove

that their teaching practices are outstanding reflections of outcomes-based teaching principles that should result in effective learning (NWU 2009a:17).

The NWU *Teaching and learning framework* (2009a:17) makes, in addition to the ITEA as award for excellent teaching practice, also provision for a developmental opportunity to young and inexperienced academics to optimally improve their teaching practices by means of structured and planned workshops and under the guidance of an skilled academic advisor and an academic peer from the same academic discipline. For this developmental leg of ITEA the Development ITEA is awarded for noteworthy improvement of teaching capability (NWU 2009a:17). Academics that are eligible for participation in the ITEA programme must be nominated by their dean/school director (NWU 2009a:18). The ITEA process includes (NWU 2009a:18):

- observation and evaluation of three contact sessions (one scheduled and two unscheduled) by the evaluation panel
- student feedback on the lecturer's teaching
- the evaluation of a teaching portfolio compiled by the academic

A financial incentive is allocated to the ITEA as follows (NWU 2009a:18):

- The prestigious ITEA amounts to R30,000 for academics who achieve 80% and above.
- Academics who achieve between 75% and 79% in the programme receive an amount of R25,000.
- Academics who complete the Development ITEA programme successfully with an average mark between 70% and 74% are awarded an amount of R20,000.

The focus of the ITEA programme is precisely on the teaching aspects that are argued for in this study: applying innovative teaching strategies, including technology, facilitation rather than lecturing, interactivity and high class participation, teaching on the relevant National Qualifications Framework (NQF) level and applying Bloom's taxonomy applicably to these levels. Also a positive aspect to the ITEA

programme is the financial incentive attached to it, serving as motivation. The shortcomings that can be identified, however, is that the use of teaching and learning strategies to enhance the use of e-learning as learning tool is not specifically dealt with in this programme and it is assumed that the academic already is competent in this respect, unless the employee entered the Development ITEA. However, as with the ICNL, when an employee entered the Development ITEA, the development is not specifically targeted on the adjustment of teaching and learning strategies for the use of technology in teaching and learning and is mostly focused on classroom teaching and learning.

In addition to the ICNL and the ITEA, the University also offers the following employee development programme:

• A management skills development programme (MSDP)

The management skills development programme (MSDP) was introduced in 2008 as an exclusive programme for top management and since 2009 it was also offered to deans of faculties and school directors (De Wit 2010). This programme focuses on management skills and not academic skills (De Wit 2010). The following topics are included in the programme content (NWU 2010a):

- Managing both functional and dysfunctional conflict for optimal outcomes.
- Establishing a value-driven work environment.
- Establishing the empowerment and competencies for effective planning and successful execution.
- Learning to negotiate for win-win outcomes; also enhancing relationships.
- Developing an ability to know when and what to change, and the competence to complete the change process.
- The practical application of values.
- The building up of customer relations.
- The valuing of diversity.
- The importance of competitive intelligence.
- The valuing of teamwork.

- The building of trustworthiness.
- The value of performance orientation.
- The importance of the 360° managing spectrum.

In consideration of the purpose of this study it may be sensible to include change management for line managers, guiding them as to how to lead academics to the acceptance of the changed job requirement of including technology in teaching and learning. It would therefore also be helpful if line managers were trained in motivational skills to better motivate employees. Line managers should also be made aware of their roles and responsibilities pertaining to the identification of training and development needs of academics, as well as given the opportunity to attend such training and development sessions and providing the necessary support to academics.

According to Erasmus et al (2005:297) career development and career management should be part of the development planning process. Therefore, the next section reviews the role and place of career management at the NWU.

6.3.2 Career management programmes at the NWU

The NWU do not have a career management policy as such, but does to some extent make provision for it through the *Institutional plan, 2010–2012*, and the *Staff development policy*, which are discussed below.

• The Institutional plan, 2010–2012

In terms of career management, the *Institutional plan, 2010–2012* indicates that the NWU intends to:

 continue with employee development as currently is the case (see section 6.3.1), and will manage individual personal development plans and career plans (NWU 2009b:28)

- define mechanisms for employee retention, including mentoring and coaching, to ensure qualification for career development and promotion (NWU 2009b:28)
- improve the existing talent management strategy to integrate practices such as capacity building, succession planning, mentoring and coaching (NWU 2009b:28)
- keep on implementing the talent management programme, which is connected to development opportunities and rewards, in order to retain high performers (NWU 2009b:28)

Although personal development plans are in practice to some extent at the university, the use of career plans is not commonplace (De Wit 2010). Further, mentoring and coaching, as with career management, is currently not widely practiced at the NWU (De Wit 2010). The university is currently in the process of drafting a talent management strategy that will be applied for academics (De Wit 2010). The strategy is, however, in its infant phase and not being implemented yet as it still has to be approved at the university's Senate. It can however, be argued that a talent management strategy will positively enhance career management practices such as mentoring and coaching, succession planning and employee development at large, especially as an integrated approach will be followed.

The literature review in chapter 3 revealed that mentoring and coaching can be integrated into a career management plan to develop academics' levels of e-readiness and their skill to effectively make use of e-learning. In this case the coach will not necessarily be the line manager, but a person that is skilled in the technical ability of e-learning. The coach can help the employee to develop online skills and to master the tools of the e-learning platform. In other words, the coach will contribute to the "e" in e-learning. The mentor can be the line manager, a senior academic that is experienced in educational strategies, or a teaching-learning advisor from the ASS Unit. The mentor's role will be to develop the academic's educational knowledge and school the employee in the philosophy and use of e-learning. The mentor should also provide guidance regarding the use of various teaching and learning strategies, most relevant to the learners' needs and preferences with the ultimate goal of optimising learning. The mentor will thus contribute to the "learning" in e-learning.

It can be argued that an integrated approach (where human resource practices such as mentoring and coaching, career development, succession planning and talent management are all aligned to career management), is most likely to strengthen the effectiveness of a career management programme and will prevent managers from following a silo approach.

• Staff development policy

The *Staff development policy* indicates that:

"Career planning and staff development is primarily the responsibility of the individual staff member, and supported by management" (NWU 2005b:1). This principle also appears in the *Academic staff promotion policy*, stating that "the organisation should create the environment and framework within which individual staff members can optimise their potential" (NWU 2005a:2). Line managers are responsible for adhering to the *Staff development policy*, including the process and procedures thereof, whereas the Human Resources Department is responsible for th*e administration process*, and the provision of management information (NWU 2005b:2).

The principle of career planning being the responsibility of the employee, within a conducive and supportive environment provided by the organisation, is supported by the recent literature in the field of career management (see sections 3.3.1 and 3.3.2). It will be necessary, however, to clearly communicate this policy to employees and provide an opportunity for open discussion and questions to clarify what is expected of the employee and how the university will assist in terms of career management. Also the principle pertaining to responsibilities allocated to the line manager and the Human Resources Department respectively corresponds with literature (see sections 3.2.3 and 3.2.4).

It can be argued that the implementation of the planned talent management programme will make a significant difference in terms of career planning and development of academics. The integrated approach will ensure that monitoring and review of the talent management programme are done against the Human Resource Plan and the *Institutional plan* (DPSA 2008:7). Further, the holistic integrated approach of talent management, career management, succession, planning and mentoring and coaching will ensure that all areas of the academic's capabilities and development areas are addressed and that it is aligned to organisational goals (cf DPSA 2008:9). Important though, is that the talent management programme must be correctly implemented and monitored. Correct implementation entails buy-in and commitment from senior managers who need to pilot the programme (Vermeulen 2008:413).

Best practices for talent management reveal that it should be a strategic approach, driven by senior managers (McCauley & Wakefield 2006:5). Best practice research (Albertsson 2003:3; McCauley & Wakefield 2006:5) further reveals that line managers must be aware of their role and responsibilities and will in all likelihood play a significant role in terms of mentoring and motivating staff as they are responsible for getting the work done as well as for developing the employees they manage and are therefore essential to making talent management succeed. It can be argued that with regard to the inclusion of development for e-readiness of academics in their development plans, line managers will have to be particularly attentive to motivating employees as the new job demand of e-learning may not necessarily be congruent with the employee's personal career goals.

In addition to senior managers and line managers, the Human Resource Management Department also has a significant role to play. The Human Resource Department will be responsible for the administration, policy-making and support to managers. Best practice research revealed that organisations that excel at talent management connect the critical owners of talent (senior managers, the Human Resource Management Department and line management) together to drive the organisation to increased performance (Mucha 2004:100).

It can be argued that the talent management programme will also contribute to employee performance. Surveys held with employees, employed by institutions that excel at talent management and retention indicate that leaders in their institutions go the extra mile to develop subordinates and grow leaders to follow in their footsteps

165

(Corporate Leadership Council 2003:3). Thus, not only the employee, but also the NWU at large will benefit from talent management programmes.

As indicated in this section a number of employee development programmes are already in place at the NWU, but in none of these the enhancement of the ereadiness of academics as part of a broader teaching and learning approach is dealt with. Therefore, attention must be given to the drafting of a personal development plan that makes provision for uniquely structured training and development interventions for each employee. It is advisable that these training and development interventions not only be focused on the technical skill of e-learning, but follow a broader teaching and learning approach, including the philosophy of e-learning and the use and adaptation of teaching and learning strategies to obtain optimal student learning. These training and development opportunities can be incorporated in a career management programme or a talent management programme.

6.4 THE NWU LEARNER

Chapter 5 described the 21st century learner as mostly belonging to the so-called Generation Y. Chapter 5 also indicated that the 21st century higher education environment is characterised by underprepared learners. This section profiles the NWU learners in terms of their generation and preparedness for higher education.

6.4.1 Generation Y learners

It has been determined in chapter 5 that Generation Y learners have teaching and learning needs and preferences that differ from traditional learners. One of the most significant preferences of Generation Y learners, identified in this chapter, is their preference for the use of technology

The vast majority of undergraduate learners at the NWU are Generation Y learners. Table 6.1 below indicates the statistics for undergraduate learners at the Potchefstroom campus of the NWU, as well as the total amount of undergraduate students across all three campuses of the NWU: The Potchefstroom campus, the Mafeking campus and the Vaal Triangle campus:

Year of birth	2010 Undergraduate contact learners		2010 Undergraduate learners	
	(Potchefstroom Campus)		(All campuses)	
	Total	%	Total	%
1992	89	1,2	555	4,4
1991	2 774	37,8	4 033	32,2
1990	876	11,9	1 668	13,3
1989	298	4,1	730	5,8
1988	605	8,2	919	7,3
1987	383	5,2	623	5
Total	5 025	68,4	8 528	68
Total of undergraduate	7 342		12 537	
registrations				

Table 6.1: Undergraduate learner enrolments at the NWU

Source: NWU (2010b).

The information in table 6.1 reveals that at least a total of 68,4% of undergraduate learners at the NWU, Potchefstroom campus can be categorised as Generation Y and 68% of undergraduate learners at the NWU as a whole (including all three campuses) can be considered as belonging to Generation Y. As mentioned, this generation's behaviours, approaches and preferences differ from those of other generations as a result of their exposure to technology, and in many instances the outlook of Generation Y learners also differs considerably from that of university management and academics (Westerman 2007). This reality implies that teaching and learning strategies may have to be adapted to focus on the enhancement of Generation Y learners' learning as indicated in the discussion above. Academics have to come to the realisation that traditional classroom teaching and learning strategies will not appeal to these learners and will not obtain optimal learning results. It is also necessary for senior managers at HEIs to take cognisance of the facts and put structures and policies in pace to formalise the use of technology in teaching and learning.

The Academic support services unit of the Potchefstroom campus of the NWU took a step in this direction by conducting a survey pertaining to e-learning during 2010. The aim was to amongst others determine learners' preferences and perceptions

pertaining to the use of technology in teaching and learning¹². A total of 1 284 undergraduate learners participated in the survey of which 88,45% were under the age of 25 years. Both male (46,11%) and female (52,63%) learners participated in the survey and mother tongue languages were represented as follows: 78,95% of learners were Afrikaans, 7,23% Setswana, 4,01% English and 3,69% Sesotho. Therefore, learners who participated in the survey represented the demography of the NWU, Potchefstroom campus to a great extent. (NWU 2010e:1.)

Some of the results obtained in the survey, relevant to this study are:

- On a question whether learners would prefer to receive all study material electronically if money was not an issue (adding to printing costs), 50,7% of learners indicated that they would prefer this manner of distribution. The other 49,3% indicated that they would still refer to receive study material in hard copy (NWU 2010e:3).
- When asked to which extent learners want e-learning incorporated in their courses, 9,2% indicated they prefer modules only using technology, 25,8% prefer modules that mostly use technology, 50,6% prefer modules that have a relative average use of technology, 10,8% prefer modules with a limited number of technology and 3,6% prefer modules where technology is not at all incorporated (NWU 2010e:6).

The abovementioned results prove the majority of learners (96%) prefer the inclusion of e-learning in their courses to various extents, thus supporting the discussion in chapter 5 pertaining to Generation Y's preferences.

It is evidently clear from the abovementioned statistics that the vast majority of learners at the NWU are Generation Y learners. These learners prefer new, innovative methods of learning and the traditional approach of teaching and learning does not appeal to them. These learner preferences, together with global technological advancement and the shift towards e-learning (OECD 2004; Bennet 2002:2; Albright & Nworie 2008:15), compel academics to adapt their teaching and

¹² More aspects of e-learning were also included in the survey, but are not discussed as it does not form part of the scope of the study.

learning practices to better address their learners' needs. In this regard, e-learning training that includes the use of appropriate teaching and learning strategies for the use of technology in teaching and learning is important. The learner profile calls for well-skilled online learning facilitators who are willing to adapt their teaching and learning approaches to accommodate 21st century learners.

As a step in moving towards making provision for the needs of generation Y learners, the NWU piloted an e-learning project at the beginning of 2011. The next section elaborates on this pilot project.

6.4.2 Task group and pilot project: technology in teaching and learning

The Faculty of Theology on the Potchefstroom campus of the NWU commenced with a teaching and learning with technology project at the beginning of 2011. The aim of the pilot project is to determine whether the initiative taken at the Faculty of Theology can be implemented in the rest of the campus, based on the successes and failures experienced in this project (NWU 2010c:3). The project is undertaken by the Teaching and learning with technology task group of the Potchefstroom campus which was established during 2010 (NWU 2010c:1). The task group functions under the leadership of the Vice Rector: Teaching and Learning of the Potchefstroom campus and is chaired by the Dean of Theology (NWU 2010c:1). The mandate of the task group is to:

- establish a broad decision-making criteria framework pertaining to the use and integration of technology in teaching and learning at the Potchefstroom campus of the NWU, in consideration with national initiatives in this regard
- determine current teaching and learning policy and processes and revise it within the framework of strategic and tactical considerations
- determine whether teaching and learning with technology initiatives were taken on the other campuses of the NWU to benchmark and learn from such initiatives with the aim of optimal synergy (without reducing the rate of development at the Potchefstroom campus)
- plan and implement specific projects on the short and long term

At the beginning of the 2011 academic year each first year Theology students were issued with a laptop. The laptops have WiFi¹³ connectivity. Before classes commenced students received training in the use of the laptop, the software uploaded to it and the use of the e-learning platform (NWU 2010c:3;5). First year Theology academics also received e-learning training (NWU 2011:2). Progress reports will be drafted throughout the course of the project to determine challenges and find solutions. The lessons learnt from the challenges experienced during the pilot project can be used to improve the implementation of the project at other faculties at the NWU. The establishment of the task group and the subsequent pilot project in the Faculty of Theology are thus positive steps towards the enhancement of learning through the use of technology. The increased use of technology in teaching and learning is likely to appeal to Generation Y learners.

The NWU's commitment to enhance the learning of the second type of 21st century learner, the underprepared learner, is discussed in the next section.

6.4.3 Underprepared learners

In chapter 5, section 5.2.1.2, it has been determined that academic preparedness for higher education presupposes learners' academic proficiency that includes abilities such as reading, writing, note taking, exam writing (Brüssow 2007:133–134). Academic preparedness also includes the ability to effectively study, solve problems, and think critically and analytically to make adequate progress through a higher educational academic programme (Dzubak 2005:2).

Learners on the first year level at higher education should ideally be able to read an amount of 400 words per minute with an 80% correctness level, but 350 words per minute and a 70% correctness level is regarded an acceptable level to pass an academic programme (Nel 2010; Beukes 2010). However, a significant number of learners entering their first year of study at the NWU are not able to read at these

List of research project topics and materials

¹³ WiFi is "used to certify the interoperability of wireless computer networking devices" (Webster 2011).

levels (Nel 2010; Taljard 2010). Therefore, the NWU, as other HEIs, face challenges to identify initiatives to support students' underprepared for higher education (Brüssow 2007:127). The NWU, Potchefstroom Campus, has established the Centre for Academic and Professional Language Practice (hereafter referred to as the Centre). The Centre provides support programmes such as: learner academic literacy, a reading laboratory and a course in computer and technology skills. (Taljard 2010.)

According to Taljard (2010) the academic literacy support programme at the NWU commences with a test (Test for Academic Literacy Levels – TALL) that all first time first year learners at the university should write. The TALL is classified as a medium stakes test, meaning that it determines academic programmes to which learners will be able to learn most suitably and not to merely disqualify learners. If learners fail the TALL they are obliged to register for two academic literacy courses in their first year of study. However, learners that do pass the TALL need to register for one prescribed academic literacy course which is compulsory to all first year learners. The content of these academic literacy courses focus on amongst others, empowering learners to function in the academic writing environment, academic vocabulary, nominalising (using active and passive voice), and written argumentation for the writing of assignments. (Taljard 2010.)

According to Beukes (2010), the reading laboratory as the second component of the support programme, assists learners with reading speed and reading correctness. All reading and tests are conducted and assessed on computers. The reading course is, however, due to a lack of adequate staff capacity, at this stage voluntary (Beukes 2010). As mentioned above, learners should be able to read an amount of 400 words per minute with an 80% correctness level, but 350 words per minute and a 70% correctness level is regarded an acceptable level to pass the course (Nel 2010). Nel (2010) further explains that in previous years, excluding the past two years, the majority of learners needed approximately 10 sessions of 45 minutes to reach this level of reading. During the past two years, however, the majority of learners needed an average of 15 sessions to master these levels (Nel 2010). This decline in performance supports the notion that HEIs increasingly deal with underprepared learners.

171
The third component of the three support programmes, the computer literacy and technology skills component, is compulsory to all first year learners (Taljard 2010). This component focus on the following: the use of computer programmes such as Microsoft Office, the management of electronic files, basic computer applications for the writing of assignments, the drafting of power point presentations, using the internet to search for information for assignments with sophisticated search techniques, and the sending of e-mails (Van Aswegen 2010).

In addition to the three support programmes, a writing laboratory is also available for both learners and staff, providing support in academic writing. The writing laboratory makes use of 14 writing consultants, which are mostly NWU postgraduate learners. These writing consultants receive training from the Centre for Academic and Professional Language Practice in the writing of academic text and academic text editing. (Taljard 2010.)

As indicated in chapter 5, section 5.2.1.2, technology can play a significant role in the process of preparing underprepared learners for the academic challenges of tertiary education, but is, however, not yet employed to its fullest extent at all universities (cf Amirualt & Visser 2009:72).

At the NWU, Potchefstroom Campus, first year learners are introduced to the university's e-learning platform, eFundi, during the two weeks orientation programme, prior to the commencement of the academic year (Van Aswegen 2010). eFundi is currently used to some extent in the abovementioned programme of the Centre for Academic and Professional Language Practice: announcements are made on eFundi and test results, general course information and text examples for assignments are posted on the e-platform (Taljard 2010). Future planning in this regard includes online exercises for learners to practice the areas in which they are not competent yet, as well as previous exam papers and memorandums of the programme (Taljard 2010).

It is thus evident that the NWU has put definite mechanisms in place to support learners in improving their academic literacy. Although learners need a longer period of time to reach an acceptable level of reading speed and correctness than was the case a few years ago, pointing to underpreparedness of learners, the decline in performance is likely also be worsened by a shortage of staff. It will benefit the NWU to strengthen their staff base in the academic literacy support programmes and make all programmes, including the reading programme compulsory to learners who are not on the adequate academic preparedness level for higher education.

6.5 CONCLUSION

The chapter reviewed the current performance management and employee development programmes and policies at the NWU. It became evident that although the NWU have policies and practices in place for the performance appraisals of academics, including personal development plans and self-assessment, no provision is made for the assessment of the e-readiness of academics as e-learning is currently not a compulsory job requirement. Once it becomes an integral and compulsory part of an academic's job, it should be included in performance appraisals and be reflected in all related policies such as the *Performance management policy*, the *Recruitment policy*, the *Human resource management policy*, the *Teaching and learning framework* and the *service agreements* of academics. A separate e-learning policy should also be drafted.

It is further evident that academics at HEIs who are expected to teach online will benefit from training and development. Through employee development academics will be provided with a skill that will enable them to teach online and to be able to use the tools of the e-learning platform effectively. As an academic will be appraised for this skill against job requirement standards, it is necessary for the line manager to be involved in the development process by taking responsibility therefore and ensuring that the employee has the opportunity to attend the training. The line manager should also provide an employee with the opportunity to implement the new skill in the workplace.

It was further determined that career management programmes and talent management programmes can play a positive role in employee performance. The integrated approach of a talent management programme, as currently planned by the NWU, including human resource management development practices such as career management, succession planning and mentoring and coaching, is perceived to have a positive effect on developing the e-readiness of academics. The holistic approach of a talent management programme will ensure that all relevant aspects of an academic's job requirements are addressed.

The chapter also profiled the NWU learner, determining that the majority of undergraduate learners at the NWU can be categorised as Generation Y learners and are underprepared for higher education. As a high preference for the use of technology is one of the outstanding characteristics of Generation Y, academics are compelled to include the use of technology in teaching and learning. The use of technology in teaching and learning in return requires academics to be well-skilled in this regard, calling for employee development pertaining to e-learning.

The underpreparedness of learners compels the NWU to make provision for enhancing the academic literacy of their learners. In this regard the NWU has established a number of programmes/mechanisms, offered by the Centre for Academic and Professional Language Practice: a learner academic literacy programme, a reading laboratory and a course in computer and technology skills.

As the profiling of performance management and employee development practices and policies at the NWU revealed shortcomings, empirical research has been conducted to determine how these shortcomings could be improved to contribute to the e-readiness of academics. The empirical research discussed in the next chapter thus determines how provision can made for the assessment of the e-readiness of academics during their performance appraisals and for the subsequent training and development interventions. The next chapter also determines how the learning needs of the 21st century learner can be provided for through the use of technology in teaching and learning. The empirical research thus determines how the scholarly literature review on the various aspects related to key human factors in the ereadiness of academics can be applied in the higher education environment, specifically at the Potchefstroom campus of the NWU.

CHAPTER 7: EMPIRICAL RESEARCH: METHODOLOGY AND FINDINGS

7.1 INTRODUCTION

Whereas the preceding chapters have outlined the theoretical framework for the study, this chapter focuses on the results obtained from the empirical research. With the aim of achieving the research objectives and answering the research problem outlined in chapter 1, sections 1.3 and 1.4, an in-depth literature study and an empirical study was conducted. This chapter discusses the findings of the empirical research and subsequently interpret these findings with the aim to make recommendations on the inclusion of e-readiness assessment of academics in performance appraisals. Recommendations will also be made on the development plan and consequent e-learning training and development interventions. For this purpose a framework, considering an employee's key human factors, namely, personal work profile pattern, preferred learning style and technology adoption pace and style, is developed. Further, the application and implementation of the framework towards attaining optimal results is discussed.

The research methodology followed with the research is outlined and justified in terms of the purpose, paradigm, techniques and context of the study. Further, particular challenges were experienced during the data collection of the study, which in itself led to certain observations and deductions and is therefore also outlined in this chapter, following in the next section.

7.2 CHALLENGES EXPERIENCED DURING DATA COLLECTION

A number of obstacles were experienced during the data collection for the empirical component of the research. Data collection for the empirical research was done through a questionnaire, a focus group discussion and personal interviews¹⁴. The researcher's initial aim with regard to data collection through the questionnaire was to distribute the questionnaire to between 500 and 600 permanent academics at the

¹⁴ These data collection methods are discussed in detail in section 7.3 as part of the research methodology.

Potchefstroom campus of the NWU. Due to certain challenges experienced (as discussed later in this section) it was ultimately distributed to 300 academics. Participation was voluntary and was stated clearly in the cover letter, accompanying the questionnaire (see Annexure C). In addition to the cover letter which explained the purpose of the study and calling upon employees' support in this regard, and irrespective of School Directors' urging and encouraging their staff to complete the questionnaire, only 85 academics completed and returned the questionnaire. Not even an extension of the deadline made a significant difference in the increase of responses. A number of reasons may have caused this low response rate:

- The questionnaire was distributed during October 2009, two weeks before the final annual academic exams. The rationale was that academics would mostly have concluded semester classes and would be scaling down on academic activities for the semester. At this time they would also not yet be busy with marking exam scripts, therefore this was considered to be a suitable time to complete a questionnaire without interfering too much with work commitments, as would be the case during the midst of the semester. This was also the time that the researcher has reached the point in the research to be ready to distribute the questionnaire. However, judging by the feedback response, this turned out to not be the ideal time. It appeared that many academics were still busy with last minute academic activities and calculation of semester marks.
- Before the questionnaire was distributed amongst academics of a particular School, the School Directors were contacted to obtain permission for the distribution of the questionnaire in their schools (see Annexure A). Three School Directors did not grant permission for the questionnaire to be distributed amongst their staff, due to exam preparations and the finalisation of academic matters for the semester. In some instances School Directors never responded to the request, however, this was the exception rather than the rule. On the other hand, a number of School Directors, who indicated that they were not comfortable with e-learning, completed the questionnaire in a spirit of academic and collegial courtesy and agreed that it could be distributed amongst their staff.

- It became evident that the chosen method of distribution of the questionnaire (electronically) caused a lower feedback return than expected. The researcher chose to distribute the questionnaire via electronic mail for three reasons: Firstly, the delivery of the questionnaire to academics was immediate; secondly, the customary slow and poor response to mail-delivered questionnaires could be thus avoided; and thirdly, it was applicable to distribute a questionnaire pertaining to e-learning electronically. In retrospect, however, it appears that in all probability employees lacking e-readiness to teach online did not complete the questionnaire. It was received in a format that they are not most comfortable with (electronically). Although the cover letter (see Annexure B) clearly indicated that hard copies of the questionnaire would be provided if preferred, only five employees made use of this option.
- It can be argued that the low response rate can also be attributed to a tendency of apathy amongst employees, if a matter does not affect them directly, or if it is perceived to not affect them directly. Currently the use of technology in teaching and learning is voluntary and academics who do not make use of elearning supposedly do not see the need to get involved in a study in this regard and maintain an apathetic stance towards it. This lack of response in itself provides food for thought. The notion of apathy is supported in literature. A research essay on the My best essays website explains that apathy is the indifference of an individual or society to the activities and events of the world around them (Anon 2009:1; cf McNulty 2009; cf DeFiore & DeFiore 2005). Apathy is a lack of interest, enthusiasm or concern (Pearsall 2001:75). Apathy is also described as indifference (Webster 2011) to anything that does not directly affect a person or his/her environment and therefore people usually are apathetic towards issues that they feel do not concern them (Anon 2009:1). "In fact, this is one of the major causes of apathy, along with a lack of knowledge of the subject matter, a perceived superfluity and complexity of information and a simple lack of interest" (Anon 2009:1). Helen Keller (sa) once said: "Science may have found a cure for most evils: but it has found no remedy for the worst

of them all, the apathy of human beings". Apathy evidently played a role in the feedback to the questionnaire.

- With the literature on the low response to surveys (Sivo, Saunders, Chang & Jiang 2006; Cummings, Savitz & Konrad 2001:1348–1349; cf Sax, Gilmartin & Bryant 2003:409–410) in mind it can be argued that a lack of e-readiness, and perhaps even a fear of, or resistance to e-learning, may have been responsible for the low response rate of the questionnaire:
 - The title of the e-mail, as well as the covering letter indicated the title and nature of the study. Concepts such as *e-learning* and *e-readiness* itself could be a push factor for employees lacking e-readiness.
 - It is likely that an employee, already not in favour of using technology in teaching and learning did not want to complete the questionnaire since they have perceived it as supporting the cause of e-learning.
 - They may have been under the impression that the questions are of a technical nature.
 - They are merely not interested in the topic.

It is interesting to note that the apathetic stance of academics imply that they are under the impression that e-learning does not affect them directly. Academics may hold this believe for the reason that the use of technology in teaching and learning is still optional at this stage. However, what this impression is also indicative of is a lack of understanding for learners' needs and preferences, as discussed in chapter 5. Judged by the apathy of a significant amount of academics (as reflected in the poor feedback response of the questionnaire), it appears that academics either are not aware of how favourable learners perceive the use of e-learning as part of their academic courses, or they choose to ignore this fact due to their own biases and preferences.

If the phases of technology adoption as recorded in chapter 4 are considered in this regard, it can be argued that the considerable reaction of apathy points to the possibility that a significant amount of academics may belong to the *early*

majority category of the technology adoption cycle (the eventual users of technology who do not like to take the risks of pioneering, but see advantages of tested technologies are driven by usability and success of the technology [Zemsky & Massey 2004:9]; or they may belong to the *late majority* category (those who adopt when half of the population has already done so; they are followers who dislike the disruptions of new technologies and are more conservative [Zemsky & Massey 2004:9]). Some may even belong to the *diehards* category (those who resist adopting innovations [Zemsky & Massey 2004:9]).

Although the response rate was not what the researcher has hoped for, valuable data of significance for obtaining the study's research objectives was still obtained through the questionnaire and focus group discussion. These results are discussed in section 7.4.

7.3 RESEARCH DESIGN

To enable the researcher to address the research objectives outlined in chapter 1, data was gathered through a literature study, followed by an empirical study. The study followed both a quantitative and qualitative approach with the empirical research. A research design, most applicable to the purpose, paradigm, techniques and context of the study was developed.

A research design is described as a planned framework for action that serves as a connection between research questions and the execution or implementation of the research (Durrheim 2009:34; cf Babbie & Mouton 2004:72). The research design thus directs the research activities to ensure that valid conclusions are attained (Durrheim 2009:36).

The research design was firstly developed in consideration of the **purpose of the study**, which is to determine the role of key human factors in the e-readiness assessment of academics and how their e-readiness can be assessed. The study follows an **explanatory** approach. Explanatory research aims to provide explanations of phenomena (Durrheim 2009:44). During this study the researcher

explains the role of particular key human factors on the e-readiness of academics. The researcher further indicates and explains the interrelatedness of these human factors and how these factors contribute to an employee's e-profile. It is further argued that the e-profile of an academic should be used to draft a development plan for each employee. Section 7.6 provides and explains the framework, drafted for this purpose.

Secondly, the research design was developed in consideration of the interpretive paradigm. The interpretive paradigm sustains the belief that the reality which is studied consists of people's subjective experiences of the external world (Terre Blanche & Durrheim 2009:7). In this study it is argued that the e-readiness of academics, consist of their subjective perceptions of the use of technology in teaching and learning. This notion is supported by the results obtained from the questionnaire and the focus group and is discussed in section 7.4.

Thirdly, to suit the interpretive paradigm, a *qualitative* method, was chosen for data collection in order to explain the subjective reasons and meanings that lie behind employees' e-learning perceptions. For this purpose the technique of a focus group discussion was used. In section 7.3.5.2.1 a detailed account is given on the procedure followed with the focus group discussion. A quantitative method was however also chosen for data collection. For this purpose the survey technique by means of a questionnaire was used to collect data for determining academics' subjective perceptions towards e-learning, which impacts on their e-readiness. These subjective perceptions are rooted in key human factors such as personal profile patterns, preferred learning style, and technology adoption style and pace. Particular questions posed in the questionnaire, determined these key human factors by means of respondents' choices. Responses further indicated the interrelatedness of these key human factors, leading to certain trends and patterns that could be identified and related to employees' level of e-readiness.

The fourth consideration in the development of the research design was the **context** in which the study took place. Research always takes place in a specific context and it is the researcher's prerogative to decide to which extent the context of the study will be manipulated and considered (Durrheim 2009:53). Thus, it is necessary to VE List of research project topics and materials

consider the context of the study. Extraneous, or nuisance **variables**, that can manipulate a participant's performance and thus influence the outcome of the study, can be controlled (Durrheim 2009:53; Van der Riet & Durrheim 2009:91). The researcher, however, chose not to control extraneous variables to avoid misleading, misinterpretation and unrepresentative results. Instead, the researcher followed a naturalistic approach, considering these nuisance variables as an essential part of the real-world setting that can impact on the outcomes of the study (Van der Riet & Durrheim 2009:91). Variables connected to the context of this study are: the e-readiness of academics as the dependent variable; the key human factors of academics as the independent variable; e-learning training and development as the mediating variables; and the challenges and changes in the higher education environment as the extraneous variable.

7.3.1 Sampling

Sampling, the selection of research participants, is influenced by the unit of observation and the unit of analysis (cf Durrheim 2009:49). In this study the unit of analysis is the e-readiness construct and the unit of observation the permanent academics at the Potchefstroom campus of the NWU. These staff members cannot be regarded as representative of all HEI academic staff, but serves as a case study.

The first consideration with sampling is representativeness (Durrheim 2009:49; Babbie 2008:210; Babbie & Mouton 2004:170; May 2006:93). The researcher had to ensure a sample that will represent the academics of the Potchefstroom campus of the NWU, is chosen. A second consideration with sampling is the size of the sample (Durrheim 2009:49). The sample size must be large enough to make deductions and assumptions about the population (Durrheim 2009:39). In some instances, as was the case with this study, the sample size is determined by practical considerations (Durrheim 2009:49). These practical considerations were discussed in section 7.2.

Whereas random probability sampling from the identified unit of observation has been used for the questionnaire, purposeful non-probability sampling, convenience sampling and maximum variation sampling were used for the focus group. For the interviews with specialists, informants were selected. Durrheim and Painter (2009:134–135) explain that random probability sampling is done when each constituent (academic) in the sampling parameter has an equal chance of being selected for the sample. The questionnaire was distributed amongst 300 academics and was completed on a basis of availability and willingness to participate.

With the focus group discussion the researcher purposefully selected 20 academics to participate, of which 10 was able to participate. Kelly (2009:288) indicates that convenience sampling refers to sample selection, based on the availability of participants. The availability of academics played a role in their selection for the focus group discussion. Durrheim and Painter (2009:139) explain that with purposeful non-probability sampling, sampling does not only depend on the availability and willingness of participants to participate, but that the sample should comprise of cases (academics) that are typical of the population. For this purpose participants were purposefully selected to include employees at various levels of ereadiness. The focus group thus included employees who are relatively comfortable with the use of technology in teaching and learning and portray a certain level of ereadiness, as well as employees who are either not using the e-platform, or are using it, but simply because it is expected of them, and in general portray a lack of ereadiness and a lack of the understanding of the optimal use of technology in teaching and learning to enhance the learning experience. The same reason has led to the use of maximum variation sampling for the selection of focus group participants. Kelly (2009:290) explains that maximum variation sampling refers to sampling when the researcher aims to acquire a wide range of information and perceptions on the topic, which result in the selection of participants who have different experiences, skills and viewpoints about the topic. This was accomplished through the selection of academics on various levels of e-readiness, and with various perceptions and experiences in this regard.

For the interviews informants were selected. Babbie (2008:206) describes an informant as someone that is knowledgeable in the phenomenon a researcher is studying and who is prepared to share his or her knowledge pertaining to the particular phenomenon with the researcher.

7.3.2 Validity

In order to comply with sound research principles and being able to draw certain deductions and conclusions of the findings, the researcher had to ensure the validity and reliability of the results. According to Van der Riet and Durrheim (2009:90) "validity refers to the degree to which the research conclusions are sound". When research produces credible results that can be used to make certain generalisations, it can be regarded as valid (Van der Riet & Durrheim 2009:90). Babbie and Mouton (2004:122) further indicate that results are valid when it provides the real and precise meaning of the concept it is intended to measure (cf Babbie 2008:160).

Results obtained from the questionnaire can be regarded as valid as it produced the occurrence of particular trends. Particular patterns and interrelatedness between academics' personal profile patterns, preferred learning style, and technology adoption pace and style could be identified. Results obtained from the focus group discussion can also be regarded as valid as the same trends and patterns, identified through the questionnaire, were confirmed through the focus group discussion. Further, the lack of e-readiness and a lack of a clear understanding of the role and purpose of e-learning could be identified and related to the same factors in both the questionnaire and the focus group discussion. Certain deductions and generalisations could therefore be made from these results.

7.3.3 Reliability

Reliability of research is obtained when a specific technique, applied repetitively to the same object, would produce the same result repeatedly (Babbie & Mouton 2004:119; May 2006:92). However, Van der Riet and Durrheim (2009:93) explain that in interpretive research (such as this study) it is acknowledged that the researcher is not investigating a stable and static reality (as in the case with positivists which believe they study a stable and static reality), and are therefore not anticipating the same results repeatedly. Interpretive researchers rather anticipate that the actions and views of individuals, groups and organisations will alter and vary in changing contexts (Van der Riet & Durrheim 2009:93). It is proposed by Van der Riet and Durrheim (2009:93) that dependability can be used as criterion in the place

of reliability. Dependability will be achieved when the reader can be convinced that the findings do in fact arise in the manner the researcher indicates and explains it (Van der Riet & Durrheim 2009:93). Rich and thorough descriptions that reveal how certain behaviours, beliefs and attitudes are rooted in, and developed out of, contextual interaction, will support and attain dependability of research (Van der Riet & Durrheim 2009:93–94).

In this study it became evident that certain trends and patterns between the key human factors of an academic could be identified and could be related to an employee's level of e-readiness. One could argue that once the level of e-readiness is dealt with through training and development interventions, the same results would not be obtained with the same academics as their level of e-readiness would have increased. Therefore the same results will not be attained repeatedly. One can further argue that even without a training and development intervention, the results will vary in some instances. When a new job demand is introduced in the work environment, employees portraying the *dominance* DISC factor, with an *activist* learning style and an *innovator* style of technology adoption, would most likely adapt to the new job demand without a training and development intervention. The results will therefore depend on the changing context.

Therefore, a research plan, entrenched in the problem statement of the study, was drafted to optimise the attainment of valid data from observations. The research design ensured that the research objectives, as set out in chapter 1, section 1.4, would be obtained and the research problem would be addressed in order to make certain deductions and recommendations.

7.3.4 Review of scholarly literature

An in-depth literature study enabled data collection of the most significant variables of the study: public human resource performance appraisal and the consequent employee development process; career management and career development; the key human factors included in this study, namely personal profile patterns, preferred learning style and technology adoption pace and style; e-learning and e-readiness.

The literature study also assisted in data collection pertaining to the most likely motivators of academics and their preferences of and reactions to goal-setting.

Kaniki (2009:19) points out that a literature review should put one's research study into context by demonstrating how it fits into a particular field. According to May (2006:28), facts do not exist separately from the basis through which they are interpreted. A literature study allows for the ability to provide an explanation and identify with the findings of research within a particular conceptual framework that clarifies the data (May 2006:29). The literature study was thus done with the aim to outline the theoretical frameworks of the most important variables in the study and to enable the researcher to view the empirical results in the context of these frameworks.

A further aim of the literature study was to determine shortcomings in current literature. Snieder and Larner (2009:133) support this notion and indicate that it is essential to learn what research has already been done in the chosen field of research in order to carry out original research. In order to contribute to the scholarly knowledge in a field, it is necessary to be updated on the current state of knowledge and its limitations (Snieder & Larner 2009:133). The literature research enabled the researcher to identify a gap in the literature of both human resource performance appraisal and e-learning with regard to e-readiness of academics at HEIs. Current e-readiness assessment instruments make provision for the e-readiness assessment of countries, governments, businesses, and to a limited extent HEIs (EIU 2007:1–3; Bridges.org 2005:1–10; Mutula 2006; Machado 2007:73–74; McConnell International 2001:1–23; Maugis et al 2005:313–342; Choucri et al 2003; Ifinedo 2005; SchoolNet Africa 2003; Bridges.org 2005:1–4; 2002; Brendan 2006:276). However, no instrument for the e-readiness assessment of employees in HEIs exists, which caused the researcher to embark on this study.

7.3.5 Empirical study

As already referred to, quantitative as well as qualitative methods were used for data collection.

7.3.5.1 Quantitative data collection

Quantitative research strives towards two standards: generalisable findings and objective data (Durrheim & Painter 2009:132). Generalisability refers to the degree to which the data of a research study can be generalised to broader populations and situations (Van der Riet & Durrheim 2009:91). Quantitative methods start with a string of prearranged categories that are typically embodied in standardised quantitative measures (Durrheim 2009:49). This data is used to make broad and generalisable comparisons (Durrheim 2009:47).

Babbie (2008:25) is of the opinion that quantification of data often provides better clarity on observations. Phenomena are thus measured by assigning numbers to the qualities of things (Babbie & Mouton 2004:49; cf Durrheim 2009:47; Blaikie 2004:21). It also simplifies data collection and data review (Babbie 2008:25). With quantitative research the emphasis is placed on the quantification of constructs (Babbie & Mouton 2004:49).

A *construct* is defined as "attributes that have been conceptualised and defined in language, and which have been theoretically elaborated in terms of how they are related to other constructs" (Durrheim & Painter 2009:142). For example, the attributes of people that a researcher wishes to assess is seen as a methodically arranged set of ideas, in other words as constructs (Durrheim & Painter 2009:142). Babbie (2008:135) describes constructs as "theoretical creations that are based on observations but that cannot be observed directly or indirectly". Constructs are not real, but they are valuable and helpful to the researcher to systematise and comprehend things that are real and to communicate about these things (Babbie 2005:124). Constructs assist in making predictions regarding real things (Babbie 2005:125), as they stand in an explicit relationship to things that are real and observable (Johannes 2007:12). Cronbach and Meehl (1955:3) define a construct as an assumed attribute of people that is believed to be revealed in assessment performance. They further explain that a construct is defined by a network of relations or systems in which it arises (Cronbach & Meehl 1955:3).

An e-readiness construct therefore consists of the systematically arranged set of ideas regarding e-readiness. This arranged set of ideas can be compared to other constructs, in this case key human factors of academics. The e-readiness construct assisted the researcher in understanding the contributing factors to the e-readiness of academics. Thus it could be determined what the solutions are to a lack of e-readiness. As the e-readiness construct as a people attribute reveals itself in performance assessment, the study argues for the inclusion of e-readiness assessment in performance appraisals of academics.

For this study, a self-administered questionnaire was used as quantitative data collection technique in order to determine the variables in the e-readiness of academics.

7.3.5.1.1 Quantitative data collection by means of a questionnaire

A self-administered questionnaire (see Annexure C), was designed and distributed to the primary unit of observation (academics at the Potchefstroom campus of the NWU) via electronic mail. As indicated in section 7.2, the questionnaire was distributed via electronic mail to 300 permanent academics from a population of 771 academics at the Potchefstroom campus of the NWU. Eighty five (85) completed questionnaires were returned.

The online encyclopedia (sa) defines a questionnaire as a collection of written questions used to gather information from respondents. A self-administered questionnaire can be described as a questionnaire that respondents complete themselves (Babbie 2008:286). A self-administered questionnaire can only be used if the population is sufficiently literate (Babbie & Mouton 2004:258).

As indicated in chapter 1, section 1.6.5.3, the questionnaire does not include a professional psychological analysis of personal profile patterns, preferred learning style and pace and style of technology adoption of academics. Particular questions have, however, been asked to get an indication of an employee's personal profile pattern, learning style preference and pace and style of technology adoption.

The questionnaire used in this study comprises of closed and open-ended questions, known as a semi-structured questionnaire (Westburn 2002). Open-ended questions provide for respondents to communicate their experiences and/or beliefs about a specific matter in their own words, without constraint (Kanjee 2009:486; May 2006:102). The researcher interprets the implication and significance of responses, which has the possibility to lead to misinterpretation and bias (Babbie & Mouton 2004:233). In order to limit bias and misinterpretation, the researcher refrained from interpreting responses that were not clearly representative of certain preferences. Furthermore, to validate responses to open-ended questions, these responses were compared to a respondent's work style preference, preferred learning style and technology adoption pace and style, as gathered from the closed questions. Open-ended questions were used to probe respondents on their perceptions pertaining to the use of e-learning and the manner in which they are currently using e-learning.

Open-ended questions were also directed about the e-learning training and development of academics, as well as their perceptions of the inclusion of e-readiness assessment in performance appraisals. Responses to the open-ended questions provided insight into respondents' perceptions on e-learning, e-readiness assessment and e-learning training and development, and could be related to respondents' e-learning profiles. Respondents' e-learning profiles are subsequently connected to training and development interventions on a matrix.

Kanjee (2009:487) explains that closed questions do not provide respondents with the opportunity to answer in their own words, but oblige them to choose one or more choices form a set list of answers. The advantage of closed questions is that it brings forth a standardised set of responses from all the respondents, creating patterns, and thus allow for comparative data analysis (Kanjee 2009: 487; cf Babbie & Mouton 2004:233; cf May 2006:103). Closed questions were used to determine respondents' work style patterns, preferred learning styles, style and pace of technology adoption, motivating factors, goal-setting preferences, preferences pertaining to the use of e-learning and opinions on e-learning training and development interventions. The feedback provided for comparative data analysis on these aspects and certain conclusions and trends could be identified from the quantitative data.

7.3.5.2 Qualitative data collection

Qualitative data is collected by means of written or spoken language, or by way of observations that are documented in language, and are analysed by recognising and classifying categories or trends (Durrheim 2009:47; cf May 2006:193). The aim of qualitative research is to study a particular phenomenon in-depth, which enables the researcher to recognise and identify the categories of information that present itself from the data (Durrheim 2009:47). Qualitative research is naturalistic, holistic and inductive (Durrheim 2009:47). It is *naturalistic* in the sense that it does not manipulate information or control the surroundings (cf Babbie 2008:321), but is open to whatever emerges; it is *holistic* as it focuses on multifaceted interdependencies and the phenomenon as a whole is viewed as a complex system that is more than its various components (Durrheim 2009:47); and it is *inductive* as central "categories, dimensions and interrelationships" are revealed, which originated with the exploration of open questions (Durrheim 2009:47-48; cf Terre Blanche, Kelly & Durrheim 2009:272–273; cf Babbie 2008:24), and often is "intuitive reasoning, driven by observations" (Snieder & Larner 2009:18).

In this study, qualitative research was conducted through the use of a focus group of academics and interviews with employees in key specialist positions at the NWU, Potchefstroom campus.

7.3.5.2.1 Focus group discussion

A focus group is a group of individuals who have a similar type of experience in common, but is usually not necessarily part of the same social group (Kelly 2009:304). Therefore, with a focus group the researcher gets access to intersubjective experience, namely. a group of people's mutual experience (Kelly 2009:304). A focus group discussion often reveals facets of a topic that would not have been likely to have surfaced in individual interviews (Babbie 2008: 339). A focus group is not merely a method to get a number of people together to talk about a certain topic, it has a particular *purpose, size, composition* and *procedures* (Krueger & Casey 2009:2), accordingly discussed below.

• Purpose

The purpose of conducting a focus group discussion is to promote self-disclosure, to listen to participants and collect information and a range of opinions pertaining to a particular topic/theme, with the aim to understand people's perceptions, feelings and thoughts regarding a particular matter (Krueger & Casey 2009:2;4;7;17–20). The purpose of the focus group is not to reach consensus, make recommendations or decide between alternatives, but to have a focused discussion, which is solicited through open-ended questions and a natural environment where variables are not controlled and manipulated (Krueger & Casey 2009:7). In this study the purpose of the focus group discussion was to determine academics' perceptions about elearning and e-readiness in the higher education environment, as well as their perceptions on learners' expectations in this regard and the role that academics should play in terms of e-learning. This was done through posing open-ended questions to participants, followed by their discussions.

• Size

The ideal size for a focus group is between five and 12 people (Krueger & Casey 2009:6). The researcher invited 20 academics to attend the focus group discussion, of which 12 confirmed attendance and 10 attended and participated in the focus group discussion.

Composition

Participant selection of a focus group is based on specific common characteristics, relating to the topic of the focus group (Krueger & Casey 2009:2). The nature of these similar characteristics is determined by the purpose of the study (Krueger & Casey 2009:7). For the purpose of this study the focus group participants were selected from permanent academics at the Potchefstroom campus of the NWU (the unit of analysis). Participants represented lecturers, senior lecturers, associate professors and professors from two faculties. All participants are involved in tertiary education of undergraduate and/or postgraduate learners.



Procedures

The focus group discussion was held on 10 December 2010, from 10:30 to 12:30 at the Potchefstroom campus of the NWU. The following procedure was followed with the focus group discussion:

- Participants were informed that participation is voluntary and that they may quit the discussion at any time. They were further informed that results obtained from the focus group discussion will be presented in a collective manner and no individual will be identified or implicated in the study. All participants signed a consent form (see Annexure D).
- Participants were informed of the procedure and method that would be followed during the focus group discussion.
- The background and context of the study was provided by means of a brief power point presentation (see Annexure E), focusing on the key variables of the study and providing the background to the discussion to follow.
- A video clip¹⁵, pertaining to Generation Y (which constitute the majority of the learner population at the NWU, as indicated in section 5.2.1.1), was played to contextualise the learner population in terms of e-learning.
- Questions (see Annexure F) were posed to participants and they were requested to commence with discussions.

The researcher did not take part in discussions to refrain from manipulating of the situation and from steering the discussion in a particular direction, following in other words the naturalistic approach. This approach allowed for honest discussion and freedom of participants to disclose their true feelings, experiences and prejudices pertaining to e-learning. The discussions and interpretations emerging from the focus group discussion are outlined in section 7.4.2.

7.3.5.2.2 Individual interviews

The individual interview as qualitative research method is a suitable method for the interpretive approach to research (Kelly 2009:297). Interpretive research view

¹⁵ The video clip is available at: <u>http://www.youtube.com/watch?v=SEZM6nUhKW8</u>.

interviews as a means to determine people's true feelings and perceptions about particular experiences and/or matters (Kelly 2009:297). Semi-structured interviews with open-ended questions were conducted with various experts to obtain specialist information on particular variables of the study. The following specialists were interviewed for data collection:

- The Director: Human Resource Management, Institutional Office, NWU.
- The Section Head: Information Technology Support, Academic Support, NWU, Potchefstroom Campus.
- The Head: Writing Laboratory, Centre for Academic and Professional Language Practice, NWU, Potchefstroom Campus.
- The Head: Learner Learning and Reading Development, Centre for Academic and Professional Language Practice, NWU, Potchefstroom Campus.
- The Manager: Reading Laboratory, Centre for Academic and Professional Language Practice, NWU, Potchefstroom Campus.
- A subject specialist: Computer Science and Information Systems, NWU, Potchefstroom Campus.

Results obtained through these interviews are discussed in chapters 3 and 5.

7.4 EMPIRICAL RESEARCH FINDINGS

As explained in section 7.3, the data collection for the empirical research component of this study was done through a self-administered questionnaire and a focus group. Both were conducted with permanent academics at the Potchefstroom campus of the NWU. The terms *respondent* and *participant* will interchangeably be used to refer to academics who participated in the study by means of completing the questionnaire or participating in the focus group discussion. In addition to the questionnaire and focus group discussion with academics, specific specialists have been interviewed with regard to their particular expert knowledge and experience pertaining to certain aspects dealt with in the study. The sections to follow outline the results obtained from the questionnaire and the focus group.

7.4.1 Results obtained through the self-administered questionnaire

As indicated in section 7.3.5.1.1, the questionnaire consists of both open-ended and closed questions. The closed question component of the questionnaire consists of five sections: biographical information, work style preferences of academics, academics' motivation in the workplace, the use of and perceptions of an e-learning platform, and training and development for e-readiness. Two of the sections of the questionnaire also contain open-ended questions, namely section D: The use of and perceptions of an e-learning platform, and section E: Training and development for e-readiness. Respondents had to answer these questions, indicating their personal opinions, beliefs and perceptions regarding a particular matter, and were requested to motivate their answer.

The following sections give the results obtained through the closed questions of sections A–E of the questionnaire. Thereafter the discussions and interpretations of the results on the closed questions follow. Lastly, the discussions and interpretations on the results of the open-ended questions of the questionnaire (sections D and E of the questionnaire) follow. The statistics given at all questions are based on valid percentages¹⁶. The frequencies¹⁷ are indicated at each question. The results obtained through the questionnaire are followed by the results obtained from the focus group discussion, as well as the analysis of the focus group discussion results.

7.4.1.1 Section A: biographical information

Respondents were requested to indicate their designation, faculty and/or school where they are employed; years of teaching experience; and their highest qualification obtained and their age. The purpose of including the biographical information of respondents is to possibly explain certain trends in their preferences or behaviour, for example, to determine whether older academics portray a higher lack of e-readiness, or whether employees in a particular academic field portray a greater tendency to make use of e-learning. Also, a particular academic field may

¹⁶ Valid percentages "express the number responding as a percent of those who responded" (SFU sa:2).

¹⁷ The frequencies refer to the number of respondents who answered each question (SFU sa:2).

arguably not lend it itself ideally to the use of e-learning. The following paragraphs discuss the results obtained pertaining to biographical information.

7.4.1.1.1 Designation

Participants were requested to indicate their designation, for example professor, associate professor, senior lecturer, junior lecturer. Diagram 7.1 below illustrates the results obtained in this regard.





More than a third (35%) of the respondents are senior academics, namely professors (16%) and associate professors (19%). Eleven percent (11%) of respondents did not indicate their designation, but did, however, indicate that they have obtained doctoral degrees. Although they did not indicate their designation, one can deduce they must be senior lecturers, associate professors or professors, due to the university's qualification requirements for these levels (with the exception of a few, eg the Faculties of Law and Nursing which can promote an academic to associate professor without the obtainment of a doctoral degree as prerequisite) (NWU 2005c). It is thus evident that **a significant number of the respondents are senior academics**. If the 21% of senior lecturers (who also mostly possess doctoral degrees due to the university's qualification requirements for this level, again with the exceptions of the Faculties of Law and Nursing) (NWU 2005c), are added to the number it brings the total of respondents on senior lecturer level and higher to 67%. The high response rate (67%) of respondents that occupy senior positions can be interpreted as an understanding of the importance of data collection for research amongst these

academics, as opposed to a lower understanding of this nature amongst junior academics.

7.4.1.1.2 School/faculty

Participants were requested to indicate the faculty and/or school in which they are employed. Diagram 7.2 below illustrates that the vast majority of respondents are academics from the **Faculty of Arts (40%)** and secondly, from the Faculty of Economic and Management Sciences (30%).

Diagram 7.2: Faculties of academics



The high number of respondents from the Faculty of Arts may be attributed to the fact that the researcher is employed in this faculty and therefore received support from colleagues. As for the Faculty of Economic and Management Sciences, the second best response rate, it can be argued that it may be due to the fact that most of the School Directors in this Faculty agreed to the questionnaire being distributed amongst their staff. A further reason may be the fact that the researcher often collaborate with colleagues from the School of Human Resource Sciences with regard to one of the programmes jointly run by them and the School for Social and Government Studies. Almost 50% of respondents from the Faculty of Economic and Management Sciences are from the School of Human Resource Sciences.

7.4.1.1.3 Years of teaching experience

The purpose of this section was to determine how many years of teaching experience respondents have. Academics' years of teaching experience are also

brought into relation of their age in section 7.4.1.6.2. Diagram 7.3 below indicates respondents' years of teaching experience.



Diagram 7.3: Years of teaching experience

Results obtained from the questionnaire revealed that academics with less than five years of teaching experience comprised 27% of the total number of respondents. Employees with teaching experience of five to 10 years accounted for 24% of the total number of respondents, thus indicating that a total of 53% of respondents have less than 10 years of teaching experience. Academics with 20 to 30 years of teaching experience accounted for 15% of the respondents and 17% of the respondents had teaching experience of more than 30 years.

7.4.1.1.4 Age

Diagram 7.4 below illustrates the respondents' age distribution.



Diagram 7.4: Age of academics

Respondents mostly represented the age groups between 30 to 39 years of age (27%) and 50 to 59 years of age (27%). These two age groups account for more

than half (54%) of the total number of respondents. It is interesting to note that the age group between these two groups, 40 to 49 years of age, had a significantly lower response rate (16%). Academics' ages are also brought into relation to their preference pertaining to e-readiness assessment in section 7.4.1.6.2.

7.4.1.1.5 Highest qualification

The vast majority of respondents, 63%, are in possession of a doctoral degree, 27% are in possession of a master's degree, and 10% are in possession of honours degrees or equivalent degrees (see diagram 7.5 below).





These results correspond with the designations of academics discussed in section 7.4.1.1.1, where it was also indicated that the majority of respondents are in possession of doctoral degrees.

The biographical information provided an indication of the personal profiles of respondents. Further, a correlation between age and e-readiness could be found: the older academics are, the more likely they are to have a low level of e-readiness. The next section of the questionnaire, discussed in the following section, will provide an indication of the professional profile of respondents.

7.4.1.2 Section B: work style preferences of academics

Questions asked in this section probed respondents to indicate their work style preferences pertaining to their adoption to new technologies/innovations and their preferred attitude and approach towards learning a new skill. The rationale was to form an impression of an academic's e-profile by asking questions pertaining to particular key human factors, like DISC factors, as outlined by Thomas International, preferred learning style and pace and style of technology adoption, as outlined and discussed in chapter 4.

Although the answers to these questions cannot be regarded as conclusive of an employee's personal work profile preference, learning style or pace and style of technology adoption, it gives an indication of which of these styles and/or preferences an employee is most likely to embrace in the workplace. It also gives an indication of the typical e-profiles eminent at the NWU, Potchefstroom campus. Further, the results will show whether certain trends and similarities exist between the various key human factors. These styles and preferences can in all likelihood also give an indication of the manner in which an employee will be motivated in the workplace.

7.4.1.2.1 Style and pace of technology adoption

In the first question in this section participants were requested to choose their most likely feeling/reaction when confronted with a new technology/innovation in the work environment. The aim of this question was to determine their style and pace of technology adoption. Results obtained showed that 8,5% of respondents immediately embrace and welcome a new technology when confronted with it; 49% adapt to the new technology/innovation relatively fast; 34% immediately see the risks involved, but will eventually see the advantages of the technology innovation over a period of time when the usability and success of the technology/innovation is proven; 8,5% dislike the disruptions of new technology/innovations and will be very slow to adapt; and none of the respondents chose the option that they will never adapt to a new technology.

In section 4.2.1 of this study the various categories of technology adoption style and pace, as indicated by Zemsky and Massey (2004:9) are outlined: *innovators, early adopters, early majority, late majority and the diehards*. Zemsky and Massey (2004:9) further allocate certain universal percentages of each category. In diagram 7.6 below a comparison is drawn against the universal percentages indicated by

Zemsky and Massey (2004:9) and the results obtained at the NWU (Potchefstroom campus).





As illustrated in diagram 7.6, it is evident that there is a difference in results obtained from respondents of the questionnaire (academics at the NWU, Potchefstroom campus), and Zemsky and Massey's general category weights. The *innovators* (those who enjoy exploring new ideas and are driven by intrinsic motivators [Zemsky & Massey 2004:9]) are significantly more amongst respondents to the questionnaire (8,5%) than the 2% indicated by Zemsky and Massey. This may be due to the fact that the tertiary environment is a high performance environment, leading to the high percentage of 8,5% of *innovators*.

Further, according to Zemsky and Massey (2004:9), the vast majority of individuals usually lie within the *early majority* and *late majority* categories (35% each). However, 49% of the respondents to the questionnaire lie in the *early adopter* category, 14% more than Zemsky and Massey's 35%. This category refers to those individuals who adopt once the concept has been proven (Zemsky & Massey 2004:9). According to Oliver (2001:6) *early adopters* are opinion leaders and decision-makers who have the vision to adapt to an emerging technology to an opportunity, and they are driven by extrinsic motivators. They have the insight to match an emerging technology to a strategic opportunity (Oliver 2001:6).

Results obtained from the questionnaire indicated that 34% of respondents belong to the *early majority* category (the eventual users of technology who do not like to take the risks of pioneering, but see advantages of tested technologies, are driven by usability and success of the technology; they are the beginning of the mass market [Zemsky & Massey 2004:9]). This result of 34% is close to Zemsky and Massey's 35% for this category. The late majority category is represented by only 8,5% of respondents, compared to Zemsky and Massey's 35% for this category. They describe individuals belonging to this category as people who adopt when half of the population has already done so; they are followers who dislike the disruptions of new technologies and are more conservative (Zemsky & Massey 2004:9). None of the respondents represented the *diehard* category (those who resist adopting to innovations and regularly pointing out the discrepancies between the day-to-day reality of the product and the claims made for it) (Beshears sa).

The vast majority of respondents, 83% (of which 49% are in the early adopter category and 34% are in the *early majority* category), will according to theory adopt to the use of technology in time, once the concept is proven and the technology showed usability and success (Zemsky & Massey 2004:9). Usually a high rate of respondents in the *early adopter* category can be attributed to a high *dominance* factor (of Thomas International's DISC profiles) amongst the group, as discussed in section 4.2.3 of this study. The *early majority* category can usually be attributed to a high compliance factor of an employee. The compatibility and validity of these notions will be determined, once the results to the question pertaining to the DISC profile of employees are also outlined and can thus be compared to the results obtained in this question. Results pertaining to the DISC profile will be discussed in section 7.4.1.2.3.

7.4.1.2.2 Preferred learning style

In the second question of this category of the questionnaire participants were requested to indicate their preferred approach and attitude towards learning a new skill such as teaching and learning online. The purpose of this question was to determine their preferred learning style. The results obtained give an indication of the respondents' learning styles, as outlined in chapter 4, according to Honey and List of research project topics and materials

Mumford's (1982) learning style categories. Fourteen percent (14%) of respondents indicated that they adapt easily and prefer to deal with new challenges and experiences (representing the *activist* learning style); 51% indicated that they want to understand the practical value and use of what being taught first (representing the *pragmatist* learning style); 33% indicated that they want to see good structure and a logical step-by-step approach (representing the *theorist* learning style); and only 2% of respondents indicated that they want to have ample time to think intensively about the activities and concepts provided to them (representing the *reflector* learning style). Diagram 7.7 below illustrates the learning styles of respondents.





The diagram illustrates that the vast majority of respondents (almost 51%) prefer the *pragmatist* learning style. Individuals who prefer this learning style will require a link between e-learning training and the end-result required of them (Honey & Mumford 1982:28). They will determine the practical value and use of what they are being taught (Honey & Mumford 1982:28). The result of 51% further shows a relationship to the results of style and pace of technology adoption (in the previous section), where 49% of respondents can be placed in the *early adopter* category. The *activist* learning style can be compared to the *innovator* category of the technology adoption cycle, to which 8,5% of respondents belong. Employees who prefer this learning style adapt easily to new technologies/innovation and prefer to deal with new challenges and experiences (Honey & Mumford 1982:25).

The 33% of respondents, preferring the *theorist* learning style, corresponds with the 34% of respondents that can be categorised in the *early majority* category of the

technology adoption cycle. Employees who embrace the *theorist* learning style require good structure and sufficient time to explore the relevance between ideas and scenarios; they are analytical, detail-conscious and need to think things through in a logical step-by-step manner (Honey & Mumford, 1982:27). This sentiment is echoed in the technology adoption styles of the *early majority* category that are described as the eventual users of technology (Zemsky & Massey, 2004:9). Employees in the *early majority* category do not like to take the risks of pioneering innovations and technology, but see the advantages of tested technologies; they are driven by the usability and success of the technology and are the beginning of the mass market. The relation between these learning styles and technology adoption styles is illustrated in diagram 7.8 below.



Diagram 7.8: Relation between learning styles and technology adoption

It is evident that the vast majority of academics are *pragmatists* (51%) and second to that, *theorists* (33%). Employees, preferring either one of these learning styles, will not adapt immediately and without reservations to new technology/innovations in the workplace.

7.4.1.2.3 Personal work profile patterns

In the third question of this section, respondents were requested to choose the statement that provides the most accurate description of them in the workplace. The purpose of this question was to determine employees' personal work profile patterns.

Four statements were provided to respondents, of which one should have been chosen. Each one of the four statements represents one of the DISC profiles as discussed in chapter 3. The results obtained were as follows and are also indicated in diagram 7.9:

- Dominance I am concerned with results, competitive, focused on achieving goals, like to solve problems and love challenges – 18,4%.
- Influence I like people and want to be liked in return, am optimistic and outgoing – 11,1%.
- Steadiness I am sympathetic, friendly, a good listener, a great team player, work hard and create a stable environment – 31,6%.
- Compliance I have high standards, am a perfectionist, and prefer systems, processes, procedures and consistency – 39%.



Diagram 7.9: Personal work profile patterns

From the above results it is evident that the majority of respondents (39%) are portraying the *compliance factor* of the DISC profile, followed by the *steadiness* factor (31,6%). When a comparison is made between the relation between an employee's pace and style to technology adoption, preferred learning style, and personal work profile patterns, certain comparisons and deductions can be made (see diagram 7.10 below).

Diagram 7.10: Comparison between pace and style of technology adoption, preferred learning style, and personal work profile patterns



From the comparisons drawn between pace and style of technology adoption, preferred learning style, and personal work profile patterns (DISC factor) in diagram 7.10 above, it is apparent that certain parallels appear. The majority of respondents portray an *early adopter-pragmatist-compliance* profile. This implies that the majority of respondents will adopt to a new technology/innovation in the workplace, once the concept is proven (*early adopter*); they want to see the practical value of e-learning and would like to see a link between what they are taught and the end-result (*pragmatist*); and they have high standards, particularly for themselves, can be perfectionists, and prefer systems, processes, procedures, as well as predictable and consistent outcomes (*compliance*).

Interesting deductions can be made from the abovementioned results. The *pragmatist* learning style match with the *compliance* DISC factor proves to be logical, as both these profiles are cautious about (although not necessarily in opposition to) changes in the workplace and like to see stability and practicability before they will support a changed job demand (Honey & Mumford 1982:28; Thomas International sa). Secondly, the high rating of the *early adopter* category of the technology adoption cycle would be a more logical match with the *dominance* factor of the DISC profiles, as *early adopters* adapt relatively fast and are innovative (Zemsky & Massey 2004:9–10). In this case, however, the *early adopter* category is coupled with the

compliance factor of the DISC profile. Individuals who portray a high *compliance* factor, usually tends to favour standard operating procedures, support a traditional approach and prefer to maintain the status quo (Thomas International sa). The researcher is of opinion that these are not qualities usually associated with the *early adopter* profile, but would be a more likely match with the *early majority* profile. It can be argued that, although a significant number of respondents portray the *compliance* DISC factor (supporting a traditional approach), the high performance environment of academia led to the earlier acceptance of technology. Thus, it led to a higher precedence of the *early adopter* style of technology adoption than the *early majority* style of technology adoption.

The second highest profile that can be identified amongst respondents is the *early majority-theorist-steadiness* profile. This implies that employees belonging to this profile are the eventual users of technology who do not like to take the risks of pioneering, but see advantages of tested technologies, and are driven by usability and success of the technology (*early majority*); they require good structure and sufficient time to explore the relevance between ideas and scenarios, are analytical and detail-conscious and need to think things through in a logical step-by-step manner (*theorist*); and they are sympathetic, friendly, good listeners, "finisher completers", and team players, who work hard and create a stable environment (*steadiness*).

The second profile identified amongst respondents appears to be more logical than the first. Individuals portraying a high *steadiness* factor on the DISC profile, usually are, as with the *compliance* factor, in favour of standard operating procedures, support a traditional approach and prefer to maintain the status quo, but are also to a great extent concerned about relations and want to create a stable environment (Thomas International sa). These qualities are well-aligned to the *theorist* learning style and the *early majority* category, as both the *theorist* and the *early majority* usually portray a sense of preferring a stable environment without too many disruptions and are in favour of a well-defined step-by-step process. However, it would have also made sense if the *theorist* learning style and the *steadiness* DISC factor was coupled with the *late majority* category of the technology adoption cycle. For an academic displaying the *early majority-theorist-steadiness* profile, factors such as attention to detail and ensuring quality and standards are important (Thomas International sa). Thus, structure and security should be provided by a clearly defined learning environment and the development intervention should be predictable and stable.

It is thus evident that e-learning training and development interventions should mostly be focused on two types of academics: firstly those who portray an *early adopter-pragmatist-compliance* profile, and secondly those who portray an *early majority-theorist-steadiness* profile as almost 80% of employees portray one of these two profiles. This does not mean that the other 20% should be neglected. They should also receive the necessary training and motivation.

7.4.1.3 Section C: academics' motivation in the workplace

Questions in this section of the questionnaire relates to an academic's personal preferences and perceptions pertaining to goal-setting, preferences towards receiving goals, approaches pertaining to new work challenges and their most likely motivators in the workplace. These questions were aimed at determining whether the manner in which an employee is motivated can be connected to the employee's personal work style preference, preferred learning style and/or pace and style of technology adoption.

7.4.1.3.1 New challenging goals

In the first question of this section respondents were requested to indicate which will be their most likely reaction when new challenging goals are assigned to them from the choices provided to them. The following choices were provided and results obtained are illustrated in diagram 7.11:

- I resist it, because I already carry a heavy workload.
- I resist it because it is given in brief, abrupt instructions.
- I resist it because a rationale is not provided.
- I do it because it is expected of me.

• I embrace the opportunity as it motivates me to develop my skills and demonstrate competence.



Diagram 7.11: Most likely reaction when new challenging goals are assigned

It is evident that almost half of the respondents (49,4%) embrace new challenging goals. This outcome corresponds with the outcome of 49% of respondents that indicated that they adapt to new technology/innovations fast (*early adopters*). Given these results, at least half of academics should be e-ready. In section 7.4.1.6.2 this result is compared to the percentage of respondents who are willing to undergo e-readiness assessment as part of their performance appraisals in order to support the use of e-learning.

7.4.1.3.2 Goal preference

The second question in the section on motivation probed respondents to indicate the type of goal they prefer to receive. A significant number of 57,7% of respondents indicated that they prefer to receive difficult and specific goals; 13,2% indicated that they prefer difficult and general goals; 25,3% prefer easy and specific goals and 3,6% prefer easy and general goals. It can be argued that the high number of respondents in favour of difficult and specific goals (almost 62%), are willing to take on new challenges, provided that the goals and expected outcomes are clarified. Diagram 7.12 illustrates these results.
Diagram 7.12: Goal preference



The results obtained is supported in literature that indicate that *difficult* and *specific* goals result in higher performance as opposed to easy and general goals or no goals (Latham & Locke 2006:232; Li & Butler 2004:37; Callaghan et al 2003:2517; Perry et al 2006:509; Locke & Latham 1990; Wegge & Haslam 2005:400).

Chapter 2, section 2.3.2, further recorded that research on goal commitment has found that the same level of commitment and performance are reached when goals are assigned to employees by managers, as when employees take part in the setting of their goals or set their own goals (Latham 2007:112; Curtis 1994:41; cf Elston & Ginis 2004:500; Feltz et al 2008:99–100; cf Lycette & Herniman 2008:27). The only exception where assigned goals do not contribute to better performance, but actually lead to poorer performance, is when the assigned goals are given with brief, abrupt instructions and with the absence of a rationale (Locke & Latham 2002:708; Locke & Latham 1990:241). Providing employees with a rationale for a challenging goal can assist in increasing goal commitment (Li & Butler 2004:38; Locke & Latham 1990:241). The importance of providing a rationale, as outlined in chapter 2, sections 2.2.2. 2.3.2 and 2.3.4, is thus emphasised as it provides an employee with a sense of autonomy which in return enhances performance (Guay et al 2006:238; Markland et al 2005:815–816).

Further, assigned goals strongly influence personal goals of employees (Locke & Latham 2002:709; Fiske, Gilbert & Lindzey 2009:271). The fact that new goals are now assigned to academics in terms of e-learning should according to literature not be a restricting factor to motivation. However, in practice, resistance to some extent is usually prevalent when changed job demands are introduced (Robbins 2003:559–

560). The various personal profile patterns, preferred learning styles and differences in pace and style of technology adoption also influence the adaptation and acceptance of e-learning goals. It is thus necessary that managers approach goalsetting for a job demand such as e-learning that may be experienced as challenging, in the most appropriate manner for each employee.

The fact that almost 60% (57,7%) of respondents indicated a preference to difficult and specific goals can therefore be seen as positive in terms of task performance and self-efficacy. Lock and Latham (2002:707) explain that self-efficacy improves goal commitment and that it is more likely that employees with high self-efficacy will develop effective task strategies than employees with low self-efficacy. Managers can increase the self-efficacy of their employees by inter alia seeing to it that they receive adequate training to increase mastery that results in achievement (Locke & Latham 2002:707). They can further increase self-efficacy through effective communication that creates a sense of confidence with the employee that the goal can be achieved (Locke & Latham 2002:707). Almost 60% of respondents have expressed a preference to difficult and specific goals, which, combined with high self-efficacy may ultimately in the long term positively impact on the acceptance of elearning as core function of academics. It is clear, however, that line managers have a responsibility in this regard to provide guidance and to ensure that goals are aligned to organisational goals.

Research further proposes that the associations between goal-setting and performance processes are moderated by many contextual aspects, such as task arrangement and an employee's profile (Perry et al 2006:509). Literature is thus supporting the arguments proposed in this study: key human factors such as personal profile patterns, preferred learning style and pace and style of technology adoption informs an academic's e-profile; and when the e-profile (which reflects the level of e-readiness) is determined, it should guide the e-learning training and development intervention as well as suitable motivational techniques to be used for a particular academic towards the enhancement of e-readiness.

7.4.1.3.3 Motivators in the workplace

In the last question on motivation respondents were asked to indicate the factor by which they would be motivated most in the workplace. Respondents could have indicated more than one motivator. The following results were obtained:

- A financial incentive 49,4%
- Promotion 46%
- Recognition from my School Director/Dean 43,5%
- Support with resources (eg provision of a laptop, cell phone, etc) 40%
- Flexitime or the opportunity to work from home 37,6%



Diagram 7.13: Motivators (more than one chosen)

The results indicated in diagram 7.13 that a financial incentive took top priority, although it is not considerably higher. Literature supports the importance of financial incentives to some extent. A good salary assists in attracting quality employees and retaining them (Mason & Watts 2009:1; Lee 2007:1–2). Marshall and Harrison (2005:4) confirm this notion and indicate that financial incentives lead to better work performance.

It is, however, surprising since the majority of respondents (67%) are senior academics and one would have expected that other motivators would be more significant if compared to Maslow's hierarchy of needs. Maslow's higher order esteem needs refer to the desire to stability, having a high valuation of oneself, self-

respect, self-esteem, based on true capability, accomplishment and respect from others (Maslow 1946:33). Maslow (1946:33) places esteem needs in two categories: firstly, the need for *strength, success, competence, confidence, autonomy* and *freedom*; and secondly, the *desire for status, prestige, recognition, attention, importance* and *appreciation*. Fulfilment of self-esteem results in *self-confidence, value, strength, ability,* and *competence for being constructive* (Maslow 1946:34). The other higher order need on Maslow's hierarchy, self-actualisation, refers to the *desire for self-fulfilment, the tendency to become actualised in what one latently is and potentially can be,* and *to grow to be all one is able of becoming* (Maslow 1946:34). In this framework one would expect motivators such as recognition, promotion and flexitime to have rated higher than financial incentives.

When asked to indicate the single most important motivator of the various motivators indicated by respondents, the following results were obtained:

- Promotion 23,6%
- A financial incentive –19,4%
- Flexitime or the opportunity to work from home 19,4%
- Recognition from my School Director/Dean 16,6%
- Support with resources (eg provision of a laptop, cell phone, etc) 16,6%
- Other 4,2%



Diagram 7.14: Motivators: single most important motivator

From diagram 7.14 it is evident that when respondents were forced to choose only one motivator, promotion took precedence over financial incentives. Financial incentives, however, are still rating the second highest with flexitime. A small number of respondents (4,2%) indicated that none of the above motivates them. They are motivated by aspects such as an inner calling, spiritual beliefs and a challenge to make a difference in a subject field. This stance is supported by McDonald et al (2007:334) when they indicate that professional performance is usually intrinsically motivated as professionals perform because they find an activity inherently enjoyable and rewarding; professional performance is usually not due to an external reward. Further, research found that "extremely imposed incentives" can weaken internal motivation (McDonald et al 2007:334). This is of significance as the conventional view of intrinsic motivation is that it is regarded as a significant attribute of excellence in professional practice (McDonald et al 2007:334).

In chapter 2, section 2.3.2, it was also indicated that there is a relationship between incentives and higher self-efficacy (people's perception of how well they can perform a particular task), as well as higher self-set goals, which is related to intrinsic motivation (Callaghan et al 2003:2518–2519). It was further explained in section 2.3.2 that higher assigned goals lead to higher self-set goals and higher self-efficacy (Locke & Latham 1990:241; Callaghan et al 2003:2515–2517). There is a direct connection between self-efficacy and task performance – individuals with high self-efficacy beliefs perform better, irrespective of the self-set goal level. Self-set goals are also linked directly to task performance (Callaghan et al 2003:2515–2517; cf Madden 1997:412).

In various focus groups (2008a, 2008b, 2008c, 2008d, 2008e, 2009a, 2009b, 2010a, 2010b, 2010c) with senior managers in the Public Service it was indicated that a financial incentive usually is not the most significant motivating factor, in particular amongst senior employees in an organisation. In general senior employees place a higher value on non-financial benefits such as support with resources, flexitime, self-development, acknowledgement, promotion and status (Focus groups 2008a, 2008b, 2008c, 2008d, 2008e, 2009a, 2009b, 2010a, 2010b, 2010c). Although these senior managers and the academics' work environments, job descriptions and challenges differ, one can, to some extent, make this comparison as it is senior professional

employees with substantial work experience in their field, as is the case with the vast majority of the respondents (Focus groups 2008a, 2008b, 2008c, 2008d, 2008e, 2009a, 2009b, 2010a, 2010b, 2010c). It therefore makes sense that promotion took precedence over financial incentives when respondents had to choose only one motivator.

Motivation remains a challenging task and line managers should take cognisance of their responsibility in this regard. Lincecum (2000:1) reminds managers that it is essential to realise that all human action is determined by a blend of cognitive and motivational factors, including with the use of technology. These factors may be intrinsic and hard to discern (Lincecum 2000:1). It is thus important that managers realise they cannot rely only on their employees' intellectual capacity, but should make an effort in motivating staff and providing incentives.

7.4.1.4 Section D: the use of and perceptions of an e-learning platform

Respondents were requested to indicate the extent of their use of the various tools provided on eFundi, whether they use it for undergraduate and/or postgraduate learners, which tools of the e-platform they use, their level of comfortableness with using the e-platform, their initial reasons for not using the e-platform, their willingness to include cutting edge technology in teaching and learning, and their use of social networks, such as *Facebook*, *Twitter*, etcetera. The aim of these questions was to determine academics' general perceptions on the use of and purpose of e-learning as part of tertiary teaching and learning, as well as their level of comfortableness with technology in teaching and learning.

7.4.1.4.1 The use of eFundi

In the first question of this section respondents were requested to indicate for which type of modules (eg undergraduate modules) they use the e-learning platform. A number of respondents, 42,3%, indicated that they use the e-platform for all of their modules; 20% indicated that they use it for all undergraduate learners; 1,2% indicated that they use it for all postgraduate learners; none indicated that they use it for all distance learning learners; 9,3% indicated that they use it in some of their

modules; and 27% indicated that they use it in none of their modules (see diagram 7.15 below).



Diagram 7.15: The use of eFundi for modules

Respondents that indicated they do not use the e-platform at all (27%) provided reasons such as:

- They never had the opportunity to use eFundi.
- They did not have the time to go for training.
- Academics teaching practical music indicated that they cannot use it in their modules.

The fact that eFundi is not used optimally, as revealed in the abovementioned statistics, makes it evident that training is necessary. Further, also the fact that a lack of training was indicated as one of the reasons for not using eFundi, emphasises the need for training. It is, however, also necessary that academics realise the importance of e-learning in order to apply better time management to attend training sessions. The role of line managers is also stressed by these responses as it is their responsibility to identify training needs and to provide employees with the opportunity to attend training (Reid, Barrington & Brown 2004:118). Considering respondents indicated that they never had the opportunity to make use of eFundi, there is reason for concern. It is not clear why these employees have been denied the opportunity to make use of eFundi. However, the researcher is of opinion that line managers must support employees in this regard. The ultimate responsibility for the performance of a

particular school or section lies with the line manager and therefore also the responsibility for employees' learning and training (Reid et al 2004:118).

7.4.1.4.2 Use of e-learning platform tools

The next question in this section pertained to the use of the various tools of the elearning platform. Diagram 7.16 illustrates the results obtained.



Diagram 7.16: The use of eFundi tools

It is evident that the *announcement* and *resources* tools are by far the most used by respondents with 65,8% and 62,3% who respectively indicated the use of these tools. The next most commonly used tools are the *assignment* tool (49,8%), the *schedule* (34,2%), and the *message* tool (30,8%). Tools such as *podcasting* and *bloggers* are not used at all, and *wikis* are used by only 1,5% of the respondents.

From the abovementioned results it is evident that academics still mostly use elearning tools that are not interactive. The only interactive tool used by respondents is the *forum*, with only 10,5% of respondents that indicated the use of this tool. If the use of e-learning is viewed from an educational perspective, as discussed in chapters 3 and 5, teaching strategies need to be adapted to suit learner needs. It appears that a significant number of academics predominantly still use the traditional approach to teaching and learning. This approach, as indicated in chapter 3, called the objectivist approach, hold the idea that educators serve as channels through which learners' thoughts and meanings are transferred to the passive learner (Hanley 1994: 3). It is advisable that academics lean more towards the constructivist approach. Whereas the objectivist approach gives emphasis to observable, external behaviours and therefore, steers clear of reference to meaning, representation and thought, constructivism follows a more cognitive approach (Gergen sa). In the constructivism approach educators are coordinators, facilitators, resource advisors, tutors or coaches (Gergen sa). Tu et al (2008:1143) are of the opinion that users with constructivist-oriented epistemological beliefs are inclined to have greater preferences to engage in meta-cognitive thinking in online environments, as opposed to those who do not have constructivist-oriented epistemological beliefs. It is thus necessary for academics to adjust their teaching and learning strategies to accommodate their learners' needs and preferences. This includes the use of technology in teaching and learning that has been identified as significant to Generation Y, which constitute the vast majority of current learners at the NWU.

7.4.1.4.3 Level of comfortableness with e-learning

The following question of this section probed respondents to choose the statement, most applicable to their level of comfortableness with teaching online. The following results were obtained with a frequency of 97%:

- I am not comfortable with teaching online and therefore do not use the elearning platform – 12,4%
- I am not comfortable teaching online, but use the e-learning platform because it is expected of me by my School Director/Dean – 8,6%
- I was initially not comfortable teaching online, but became more comfortable in time – 11,1%
- I am comfortable with teaching online as long as I can stick to the basics (providing information and resources; assessing assignments; posting details and due dates on the schedule) – 42%
- I am comfortable with teaching online and am using more and more of the online tools – 21%

 I am completely comfortable with teaching online (and use tools such as podcasting and bloggers) – 4,9%

Only 25,9% (21% are comfortable and 4,9% are completely comfortable) of respondents currently portray a comfortableness level that will allow them to use advanced e-learning tools, to use e-learning creatively and to use it interactively. An alarming 42% of respondents indicated that they prefer to make use of the basic tools only. Thus, it is to a great extent used as a distribution mechanism for notes, study material, power point presentations (with the *resources* tool), and one-way communication tools such as the *announcement* tool and the *schedule*. Although the "basic tools" per se is not the issue, it is questionable whether these tools, mostly distributing information and not engaging learners in the learning experience and also not supporting the acquirement of a particular skill, is enhancing learning. E-learning tools should not be used for the sake of including technology in teaching and learning, but to use it as a tool to enhance learning pertaining to a particular skill or concept.

The fact that 4,9% of respondents indicated that they are entirely comfortable with teaching online and are using tools such as podcasting and bloggers, is contradicting to the results obtained in a prior question. At the question where respondents were requested to indicate of which tools on the e-platform they make use, none indicated that they make use of *podcasts* and *bloggers*. One can assume that these respondents currently feel completely comfortable with teaching online and although they are not using tools such as *podcasts* and *bloggers* yet, will make use of it in future. It is also possible that they make use of other cutting edge technology, not mentioned in the questionnaire. (Also see diagram 7.18, indicating respondents' attitude towards and willingness to include cutting edge technology in teaching and learning.)

7.4.1.4.4 Reasons for not using e-learning

The next question probed respondents who do not make use of the e-platform, or who initially did not use it, to provide reasons. Results obtained are indicated in diagram 7.17 below:



Diagram 7.17: Reasons for not using e-learning

Over 27,8% of respondents indicated they initially resisted (or still are resisting) elearning because they perceive it to be an additional workload. Although this may be true in some instances, when an online learning facilitator is well-skilled on the eplatform it will not necessarily contribute to an additional workload. The secret lies in using it skilfully to contribute to both teaching and learning. Also interesting in this regard is that 34,2% of respondents indicated that they did not initially use (or are still not using) the e-platform due to a lack of training, making it the biggest reason for respondents not to have used e-learning. It can be argued that the provision of training may also lower the perception in of an additional workload. It is thus clear that the reasons provided by respondents for not using e-learning can be addressed through training. Such training, however should make provision for not only the technical skill, but should also provide schooling in the philosophy and use of elearning to eliminate misconceptions pertaining to the use of e-learning.

7.4.1.4.5 The use of cutting edge technology

On the question whether respondents are willing to use cutting edge technology, such as podcasting and online games in their courses, 26% indicated that they would use it on a limited basis and none of the respondents indicated that they are already using cutting edge technology in their courses. This again supports the notion that academics are not aware of the possibilities and advantages of technology and teaching and learning, and a lack of skill is probably preventing them

from applying these technologies in their courses, thus emphasising the importance of training and development. Diagram 7.18 illustrates respondents' attitude towards and willingness to include cutting edge technology in teaching and learning.



Diagram 7.18: The use of cutting edge technology

It seems that the majority of respondents are willing to include cutting edge technology in teaching (on a limited basis, 26%; definitely, 16%, maybe, 25,9%). It brings the total of possible users of cutting edge technology amongst the respondents to 67,9%.

The above results are indicative of the various styles and pace of technology adoption. It can be argued that those who have indicated that they will definitely use it (although not using it yet), 16% can be categorised under the group of early adopters that will agree to a new technology, once the concept is proven. As indicated in an earlier section, 49% of respondents fall in this category. Part of this 16% of respondents may also belong to the *Innovators* and merely did not have the opportunity to use it yet. However, one can also argue that an *innovator* will ensure that an opportunity exists to experiment with cutting edge technology as they are risk takers and thrive on new challenges. Respondents who have indicated that they will use it on a limited basis (26%) can in all likelihood be categorised as the early majority, the eventual users of technology who do not like to take the risks of pioneering, but see advantages of tested technologies are driven by usability and success of the technology. They are the beginning of the mass market. The group of respondents that have indicated that they may perhaps use it in future (25,9%) can be categorised as the *late majority*, those who adopt when half of the population has already done so. They are followers who dislike the disruptions of new technologies

219

and are more conservative. From the above results and interpretations thereof, it can be deduced that an academics' style and pace of technology adoption influence their willingness to use cutting edge technology in teaching and learning.

7.4.1.4.6 The use of social networks

The last question of this section requested respondents to indicate which social networks they are using. More than one option could be chosen. Diagram 7.19 indicates the results obtained:



Diagram 7.19: The use of social networks

It is interesting to note that 48,2% of respondents do not make use of any social media. On the other hand 44,7% does make use of *Facebook*. This is almost a 50/50 division. Some of the other social networks are used to a small extent.

One of the ways to increase interactivity in teaching and learning is to make use of the social media such as *Facebook* and *Twitter*. Social media can be well used as communication tool in teaching and learning (Liu 2010b:101–103; Bassendowski 2011:1–2). The use of social media in teaching and learning can be limited to learners enrolled for a specific subject, or learners of various subjects can have access to for example a particular *Facebook* page, depending on the need and applicability. Learners can share their experiences, ask questions to each other and the facilitator, brainstorm and provide information they have found. All on an open

forum, available to all people allowed on the specific *Facebook* page. Through for example *Facebook* and *Twitter* various links to websites with valuable information to the subject can be provided. Both facilitators and learners reading relevant information to the subject on for example a news network or newspaper website, the option to share the information with the *Facebook/Twitter* group can be chosen which will immediately provide the relevant information to all learners on the particular *Facebook/Twitter* page. However, if respondents are not using a social network themselves, it is unlikely that they will use it to enhance their learners' learning. Only one respondent indicated that the use of social media can play a positive role in teaching and learning. Although the use of social media may not be relevant to all subject fields, certain fields will definitely be able to benefit from this type of collaborative learning.

From feedback obtained in this section it became evident that a significant number of academics are using the e-learning platform as a distribution mechanism and not as a learning tool. The e-platform is used interactively by a very small number of employees and the preferred e-tools (announcements, resources) do not make provision for learner participation. A significant number of respondents indicated that they are comfortable with e-learning as long as they can stick to the basics, thus not using interactive tools or cutting edge technology. Further, the fact that almost 50% of respondents do not make use of a social network makes it unlikely that this medium will be applied for learning of learners. From these results it is evident that the majority of respondents do not realise the possibilities and purpose of e-learning, emphasising the need for training.

7.4.1.5 Section E: training and development for e-readiness

Questions in this section probed respondents' views on the necessity of receiving elearning training prior to using an e-learning platform, the type/s of instructions, aids, tools, and applications that should be included in e-learning training, and their initial feelings on the matter of online teaching and learning when they were first taught how to use the e-platform. These questions were asked with the purpose of determining respondents' views and preferences pertaining to the purpose and/or need of e-learning training and development.

7.4.1.5.1 The need for training

Questions in this section probed respondents on their perceptions on the need for elearning training. The first question in this section requested respondents to indicate the importance of receiving training before facilitating online by means of the option they choose. A total of 93% of respondents answered the question. An overwhelming majority of 71% of respondents indicated that training is imperative. It is evident that the vast majority of respondents recognise the importance of being taught how to use e-learning optimally, thus indirectly acknowledging that there is more to it than the manner in which they are currently using it. Further, in section 7.4.1.4.4 34,2% of respondents indicated that a lack of e-learning training was the reason that prevented them from using e-learning.

The second question in this section requested respondents to indicate what should be included in e-learning in terms of the method of e-learning. More than one option could be chosen. Diagram 7.20 below indicates the results.



Diagram 7.20: Included in e-learning training

The high rating of visual support – 70,6% of respondents in favour of an instructor that demonstrates and 56,5% of respondents in favour of the use of visual aids such as videos – is noteworthy. It is evident that respondents prefer to see something to support their understanding thereof. It is also interesting to note that 69,2% of respondents indicated that they would also like to receive paper-based notes or a user manual. One can assume this is because it is something known to them that they are comfortable with and to which they can continuously refer back for assistance.

It can thus be deduced that the vast majority of respondents regard training as imperative to the use of technology in teaching and learning. It is also evident that their preference in terms of the manner in which training should be conducted is through visual support, based on their choice of an instructor that demonstrates and the use of visual aids.

7.4.1.5.2 Perceptions when first taught

In the second question in this section respondents were requested to indicate their perceptions/feelings when being taught to teach online for the first time. Eighty five percent (85%) of respondents answered this question. The results again point to the need for training and development, as 42,3% of respondents indicated that they looked forward to start teaching online, once they have mastered the new skill. Another 32,4% indicated that they also felt more comfortable with teaching online once they have obtained the skill, although e-learning would never be their first choice. It brings it to a total of 74,7% that indicated the need for obtaining the skill through training. The 12,6% of respondents who indicated that they could not wait to begin teaching online will in all likelihood portray a high *dominance* factor on the DISC profile, an *activist* learning style and will fall in the *innovator* category of the technology adoption cycle. Only 12,7% of respondents indicated that they still did not want to teach online once they have been taught how. This group of respondents can be considered as the *late majority* on the technology adoption cycle. In diagram 7.21 below a comparison is drawn between the DISC profiles, the learning styles, the pace and style of technology adoption, and these results.

Diagram 7.21: Comparison between the DISC profiles, learning styles, pace and style of technology adoption, and initial feelings/perceptions after elearning training



It is evident that the majority of respondents portray an *early adopter-pragmatistcompliance* e-profile, as indicated in a previous section. It can now also be deduced that employees who fall in this category will in all likelihood look forward to teaching online, once they have mastered the new skill. Again the need for training is emphasised as this group of respondents first obtained the skill and then became enthusiastic about e-learning. Once this group of employees, who are not opposed to e-learning and are willing to learn, are trained to use the e-learning platform optimally and couple it to sound educational strategies, it can make a significant positive contribution to learners' experience of the use of technology in teaching and learning and specifically to learners' learning experiences itself. The second highest profile amongst respondents, also already identified is *the early majority-theorist-steadiness* e-profile. It can now also be deduced that, although elearning will never be this group's first choice, they will be comfortable with teaching online, once they have mastered the skill. Motivating employees belonging to this group effectively is of particular importance, as they do not naturally lean towards elearning and make use of it only when half of the population has already done so. It can be argued the only reason they are using technology in teaching and learning is because it is expected of them. Extra time and effort should be taken during the training of employees who fall in this category. It is imperative to provide a rationale (see section 2.4.3) and to constantly encourage these employees towards the role that e-learning plays as educational advantage. In this regard the line manager will have to play a significant role.

7.4.1.5.3 Skills included in e-learning training

The last question in this section requested respondents to indicate the skills they wish to be included in e-learning training. More than one option could be chosen. The following results were obtained:





It is evident that the majority (63,5%) of respondents prefer to know only how to use the basic tools of the e-platform. In section 7.4.1.4.3 it has already been indicated that only using these tools will not necessarily enhance learning and that it merely supports the e-platform to be used as a distribution mechanism. It is, however, interesting to note that 57,6% of respondents want to be trained to customise an online site. This high percentage in all likelihood reflects one of the causes for the poor and inefficient use of e-learning, a lack of training. It can be argued that once academics are trained to make effective use of e-learning, the use of eFundi and other e-learning mechanisms will increase.

The high percentage of respondents who indicated that they want training on how to facilitate online discussions to ensure optimal participation (52,9%), also signifies that there is a lack of know-how to use the tools and therefore stresses the need for training in this respect. As an important human resource management function, training should be done by the section responsible for academic training and development, Academic Support Services. The line manager, however, also has a role to play in this regard. The responsibility to ensure that employees are trained and developed to ultimately function productively and effectively in the work place lies with the line manager. The line manager can also play a significant role in terms of motivation.

It is disappointing that only 25,8% of respondents indicated that they want pointers on how to motivate learners that resist e-learning. This result calls for concern as it reveals that academics are of the opinion that their online duties do not include motivation. One would hope that all academics would be concerned about getting all learners on board to participate and benefit maximally from any learning opportunity. This result is indicative of academic's ignorance pertaining to e-learning. In chapter 1 the significant role of the online learning facilitator has been emphasised and in chapters 4 and 5 again. Skills required of an online learning facilitator such as being a good listener, providing feedback, communication and encouraging learners and building relationships, mentioned above, all relate to the human nature of e-learning (cf Hootstein 2002). Online learning facilitators should also possess the ability to let learners feel comfortable about participating in online discussions and sharing information (National College for School Leadership 2006; Sutton 2004). It is thus evident that academics are not aware of the human nature of e-learning and that they in all probability still view it as an add-on to traditional classroom teaching and learning. This result highlights the need for e-learning training.

In addition to the abovementioned closed questions discussed, the questionnaire also contains open-ended questions. The open-ended questions are subsequently discussed in the following section.

7.4.1.6 Open-ended questions

The open-ended questions of the questionnaire pertain to the use of and perceptions of academics on an e-learning platform, as well as perceptions on training and development for e-readiness.

7.4.1.6.1 The use of and perceptions pertaining to an e-learning platform

In these questions respondents were asked to indicate whether they use the elearning platform interactively; they were also requested to indicate in which manner they are using it interactively. Respondents were further requested to indicate whether they are planning on using the e-learning platform interactively, if they do not do so already, and to indicate the manner in which they will do so. Respondents were also requested to indicate their general perception of e-learning.

The rationale behind these questions was to determine whether academics are using the e-platform as an interactive teaching and learning tool, or merely as a distribution mechanism. Answers to these questions will signify employees' perceptions on how e-learning is believed to be used, and what the general personal views pertaining to e-learning are. The answers to these questions can in all probability be linked to an employee's e-readiness and the need for training. As human resources are the university's most important asset, both the unit of Academic Support Services, responsible for academic development, and line managers should take cognisance of the need for training as important human resource development function. The first question in this section posed to respondents is:

• Do you currently use your module/s interactively on eFundi? How?

A total number of 78 respondents (92%) answered the question. The following results were obtained from respondents who use the e-learning platform:

- No 64,5% of respondents indicated that they do not use eFundi interactively.
- Yes 36,5% of respondents indicated that they use eFundi interactively. Of these respondents:
 - 9,2% use if for *assignments* and/or *messages*
 - 10,5% use the *forum* for discussions
 - 2,6% use it for *quizzes* and *tests*
 - 1,2% use it for *wikis*
 - 13% indicated that they use eFundi interactively, however, the tools of the e-platform used by them, as indicated in another question, do not correspond with the interactive use of the e-platform

The following deductions can be made from the abovementioned results:

- The actual number of respondents who use eFundi interactively is 23,5%, as 13% who indicated that they use it interactively, does in actual fact not use interactive tools on the e-platform. Thus the number of respondents who do not use eFundi interactively is an alarming 77,5%. This confirms the notion that the e-learning platform is merely used as a distribution mechanism and not as a tool to enhance learning. It is evident that academics do not realise the value of interactivity and collaborative learning as methods to enhance learning. It once again confirms the need for e-learning training of academics, with the focus on using technology to enhance teaching and learning. In previous sections it has already been stated that it is necessary for academics to adjust their teaching and learning strategies to accommodate the "new" type of learner, Generation Y.
- The number of respondents who indicated that they use eFundi interactively with *messages, assignments* and *wikis* correspond with the percentages obtained in a previous question pertaining to the use of tools on the e-platform. However, 2,6% indicated that they use eFundi for *quizzes and tests*, whereas 5,9% of respondents indicated that they use *quizzes and tests* in the question pertaining to the tools of the e-platform used. This can be interpreted that 3,3% of the respondents who make use of the *quizzes and tests* tool of eFundi do not realise that it is an interactive tool.

Another aspect of concern is some of the reasons provided by respondents for not using the e-learning platform. Some of the reasons included:

"I am lecturing on postgraduate level."

"Ek het geen voorgraadse vakke nie daarom gebruik ek nie die stelsel nie." Freely translated as: "I do not lecture any undergraduate subjects, therefore I do not make use of the system."

These two comments indicate that a number of academics are under the perception that e-learning cannot be used for postgraduate learners. This again focus the attention to the fact that in many instances academics still prefer to make use of traditional teaching and learning methods only. It also confirms the need for training and awareness of what the possibilities with e-learning are. In particular with postgraduate learners, e-learning can be applied to enhance research skills, provided that the academic is skilled to assist the learner in this regard.

Another comment, illustrating a misconception is:

"Yes, especially for feedback after test(s), discussing general mistakes. Learners often reply to this."

This comment was made by one of the respondents who indicated that they use eFundi interactively, but judging by the tools of the e-platform used, in actual fact does not use it interactively. One can assume that test feedback is provided via an announcement (which is the only tool used by this respondent, as indicated in a previous question). If this is the case, it is not an interactive tool. It is also not clear in which manner learners reply to the feedback. One can only assume that this must be orally, in a consultation or in class, as no provision for learner replies is made on the e-platform.

229

The following response emphasises the fact that a number of academics are not aware what is regarded as interactive use of the e-platform. Again this comment is highlighting the need for training and e-learning awareness.

> "No I only use it for announcements, schedules and resources. I lecture Financial Accounting to 3rd year and Honours Chartered Accountancy learners, and these modules do not naturally lend themselves to using more "creative" tools like podcasting and bloggers. It is a very technical subject and application of some of the more creative eFundi tools would be very difficult, if not impossible."

The fact that academics are not always aware of the possibilities with the use of technology, emphasise the need for training. One can understand that the use of eFundi will not be viable with a practical subject such as piano teaching, but surely it can be used for a subject such as Accountancy. It is evident that the lecturer is not aware of links that can be made to the websites of professional associations, scientific journals, and the possibility of subject related discussions on the forum, to mention a few. Surely at honours level a learner should be able to do research and the *forum* tool can be valuable for in-depth discussions of complex issues.

In addition to the direct quotes above, a few more interesting aspects came forth during the responses to this question:

- A number of respondents indicated that they use the gradebook, but do not make use of either quizzes and tests or assignments, and will therefore not be able to use the gradebook (the gradebook only reflects online assessments). This signifies ignorance of the use of the *gradebook* tool.
- One respondent indicated that the learners enrolled for his/her particular subject are part time learners and are not computer literate. He/she could therefore not continue to use eFundi, but intends to make computer literacy and/or internet access a prerequisite as he/she is op opinion that it is a valuable tool, especially for part time learners. This seems to be a sensible decision, List of research project topics and materials

given the fact that the lecturer has limited contact time with part time learners and that especially the *forum* tool can be effectively used in this regard for continuous discussion, communication and direction to learners.

One respondent indicated that he/she provided the opportunity to learners to add content and discuss topics on the forum, but they did not sufficiently make use thereof. This lack of participation can be due to a number of reasons: These particular learners are not e-ready; they are not sure what is expected of them; they do not find the tools provided valuable; what is discussed on the forum is not linked to learning outcomes and exam preparation; or they are not sufficiently guided and encouraged by the lecturer to participate. If it is one of the last two reasons, this supports the need for training. It thus may also refer to the teaching and learning strategy that needs to be adjusted.

In brief, the feedback obtained from this question confirmed that the majority of academics use the e-platform as a distribution mechanism and not as an interactive learning tool. It further showed that in many instances academics are not aware of what is regarded as interactive e-learning. The feedback emphasised the need for elearning training for employee development towards optimal performance. In order to optimally use e-learning for an enhanced learning experience, training should firstly focus on creating awareness of the philosophy of e-learning and the possibilities with the use of e-learning, and secondly to provide academics with the skill to do so. Included in the training should be new approaches to teaching and learning. To ensure learners' needs are provided for and that their learning is enhanced through e-learning, it is necessary to create an awareness of the type of learners and their needs and that academics receive training on the best suitable methods to service these learners. Considering the fact that the vast majority of academics are not qualified and trained educationalists but subject experts, it is essential that training is focused on teaching strategies. In this regard the constructivist approach, which is learner centred, can be useful. It remains the responsibility of line managers to ensure that their subordinates are well-trained and capable to provide quality teaching and learning to learners.

The second question in this section posed to respondents is:

• Do you plan to use your module/s on eFundi interactively? How?

Results obtained from this question are indicated in diagram 7.23 below:





Fourty percent (40%) of respondents indicated that they do not plan on using eFundi interactively in future. This high percentage signifies a disturbing level of unwillingness to adapt to learners' needs and new teaching and learning strategies. The 11% of respondents who indicated that they may perhaps use eFundi interactively probably are not sure exactly what it entails and therefore are not sure whether they should indicate that they will use it interactively.

Six percent (6%) of respondents indicated they are not sure about the meaning of "interactively", which supports the deduction made from feedback from the previous questions that all academics are not aware of the possibilities of the use of interactive tools. Of the 49% of respondents who indicated that they will use eFundi interactively, 17% also portrayed ignorance pertaining to the concept of interactivity. Although they have indicated that they will use it *interactively*, judging by their comments on *how* they will do it, it is clear that they will use the basic tools such as *announcements* and *resources*, which are not interactive tools. They have therefore merely indicated that they will use eFundi to some extent and it brings the total of respondents that are willing to use eFundi interactively to 32%.

Another interesting aspect that came forth was that 23% of those who have indicated that they will use eFundi interactively further indicated that they need training and some that they are not sure how to use it interactively. It is thus confirmed that in many instances academics do not use e-learning tools due to a lack of training and ignorance regarding the use and possibilities. It once again emphasises the need for

the Academic Support Services unit, responsible for human resource development of academics, to provide more information and targeted training.

Respondents who indicated that they will use eFundi interactively indicated that it will be used for *wikis, bloggers, podcasts, the forum, assignments, quizzes and tests,* setting up open learning platforms, videos, video conferencing, and *chatrooms*. What is interesting to note is that most respondents indicated the type of e-tool they wish to use, but did not indicate for which reason or in which way it will enhance learning in their specific subject field, with the exception of one respondent. This may be a matter of academics being willing to incorporate interactive technology in their teaching strategies as they realise it is important and expected by their learners in this time and age. However, they will most likely not use it in a focused effective manner without the necessary training of how to use it for enhanced learning, and not merely for the sake of using technology.

Comments from respondents that indicated they will use eFundi interactively portrayed further ignorance. Some indicated they would use it interactively in the same manner as they currently are, which is, judging by the tools they use, not interactive e-learning. Another comment referred to technical problems with eFundi. As the researcher uses eFundi on a daily basis, she is convinced that technical problems cannot be used as an excuse for not using the e-platform as these problems are minimal and the exception. This comment further signifies the need for training.

Another comment supporting the need for better information and awareness of the possibilities of e-learning was made by a respondent who indicated that he/she was not aware that podcasting and bogging is possible through the e-platform. Also a comment referring to computer labs for e-tests not being available underlines the ignorance in this regard as 24 computer labs of various sizes are available on campus (Pienaar 2011). A comment indicating that as soon as part time learners have access to eFundi the respondent will make use of it, further indicates a lack of awareness as all part time learners, as registered learners do have access to eFundi (Le Roux 2009). It is thus evident that judging the feedback of this question from a

233

human resource management perspective, as well as from a line management perspective, it is imperative to provide e-leaning training to academics.

The third question in this section posed to respondents is:

What is your general perception of e-learning?

A total number of 77 respondents (91%) answered this question. Feedback on this question indicated that 66,2% of respondents perceive it as positive with various degrees of embracing it. Some are extremely positive, others view it as a valuable additional source and some see it as an important communication tool. A number of respondents, 18,2%, portray a fully negative attitude towards e-learning and 15,6% of respondents indicated a view that is neither positive nor negative.

The fact that 19,5% of respondents indicated that e-learning cannot replace a more traditional learning environment or contact classes, and should only be used supplementary to traditional contact or where learners are not available for contact learning, indicates one of the biggest misconceptions about e-learning. These comments signify a fear that e-learning may replace the lecturer's role in teaching and learning and is probably one of the biggest reasons why e-learning is not used or underutilised as teaching and learning tool. A number of respondents also indicated that e-learning is ideal to use for distance learning, which also illustrates the ignorance of the use of e-learning for full time learners as learning tool. As recorded in chapter 1, section 1.1, e-learning is no longer merely associated with distance learning, but is consciously chosen and used to provide the best and most suitable ways to support and encourage effective learning (JISC 2009:9). The "e" in e-learning attaches technology to learning, but the focus remains on leaning as the key element (cf JISC 2009:8–9). These misconceptions emphasise the need for not only obtaining the skill, but also creating an awareness of the manner in which elearning can be integrated with contact learning.

Respondents who indicated a high level of positivity towards the use of e-learning made inter alia, the following comments:

234

"... more and more important in 21st century."

"... integral part of teaching today ..."

"The future."

"It saves time."

"...necessity to engage with new type of learner."

"Necessary for new learning environment."

"Important for era in which we live."

"Important for netgeneration."

"Should expand. All lecturers should receive training."

"Great opportunity if used efficiently."

"Use of social media"

"I can see how it is a tool that has many advantages and that the new student¹⁸ will prefer this method to more traditional methods. Therefore I am prepared to eventually make more use of it. There are for example master classes in piano conducted via the internet and students can receive valuable input from teachers elsewhere."

"Platform for instant information, and downloading of resources for students."

¹⁸ Although the concept *learner* is preferred and used in the study, as opposed to *student* (see chapter 1), the concept *student* has been retained where it is given in a direct quote from a respondent.

"Valuable"; "useful".

The abovementioned comments indicate that a number of respondents are aware of the need to adjust teaching and learning strategies to suit the non-traditional learner from Generation Y. General comments in favour of e-learning included:

"One should not follow a "one-size-fits-all" approach."

"The application possibilities vary between various subjects."

"It has value but should not be overstated."

"Additional resource and communication method."

"Useful if all parts available are employed."

Some comments made in favour of e-learning, but indicating the ignorance with regard to the possibilities of e-learning, or the solutions to the perceived problems are:

"If you have many students registered for your module or if your module is presented part time."

"It does help as students do not attend the formal classes."

"Students perceive e-learning as replacement for attending class, and expect all material to be posted online."

"Positive but not for postgraduate students."

"Services a certain type of client; short courses easier to administer."

"Students may find it difficult to get access to computer resources or understand and subsequently use it."

The majority of the abovementioned comments support the need for focused training and e-learning awareness as part of employee development to deal with misconceptions. It is evident that a number of employees are still under the impression that e-learning is a medium to be used for part time learners and cannot be used for postgraduate learners. The lack of understanding of the type of learners and their needs is once again also emphasised by the abovementioned comments.

Some of the comments made by respondents who perceive e-learning as negative or who stood neutral towards it are:

"Not positive because it is unknown to me."

"It is very new and I am not very technologically literate."

"Takes time to set up eFundi."

"Just another tool to not be overused, overestimate or used for entertainment."

"Not interested in it although I guess it may be useful."

"My experience of e-learning is too scanty to make a balanced judgement."

"It feels to me like the UNISA modules we have to present. The learners are faceless and to me the personal contact is very important."

"It will only be effective if all learners have regular access to the internet, campuses will update software, it is not realistic in SA at the moment as SA has one of the lowest computer literacy and internet usage in the world."

Again the abovementioned comments signify ignorance and need for targeted elearning training to enhance e-readiness and the use of technology in teaching and learning. Many of the reservations that respondents hold towards e-learning are not based on factual information but perceptions, ignorance and sometimes fear. Fear is a common reaction to change in the workplace. Due to employees' different eprofiles they will not adapt to change in the workplace in the same manner and at the same pace (cf Robbins 2003:559–560; Hunsaker 2001:380). From a human resource management view it is, however, important to ensure that change is managed and that it is ensured that all staff are skilled and prepared to deal with the changes in the workplace, as organisations achieve objectives through its human resources (Ivancevich 2004:4). In this regard motivation from line managers will also play a significant role. Knowing an academic's e-profile can assist managers with regard to employee motivation.

The last comment above, referring to South Africa's computer literacy and Internet access, holds truth to it as indicated by the digital economy rankings of the Economist Intelligence Unit (EIU). The EIU does an annual benchmarking study of countries' digital development and assesses countries' ability to use ICT to develop its economy and to provide in its welfare (Quantec 2010). In the annual digital rankings of the EIU of 2010 South Africa rated 40th in the world in terms of digital economy rankings (EIU 2010:4). In 2009 South Africa rated 41st and have therefore moved up only one place in a year's time (EIU 2009:5). South Africa scored a rating of 5.61 out of 10, compared to 8.49 of Sweden, the highest rating country, and is the highest rating African country on the list (EIU 2010:4). Although South Africa's internet usage and connectivity cannot be compared to those of developed countries, a new Norwegian mobile software company, indicated that webpage-views in Africa's top ten mobile internet markets (which is led by South Africa), increased approximately four times in 2009, and that unique users and the data they used almost doubled (EIU 2010:9).

The Internet World Stats (2009) indicates that 10,8% of the population in South Africa used the internet in 2009. Furthermore, in the year 2009 the internet user base of South Africa grew at a double-digit rate (Research & Markets 2009). Data reveals that the internet user base grew by 15% in 2009, from 4,6-million to 5,3-million (Research & Markets 2009). Thus, although the usage of internet is growing in South Africa, it is still not at an acceptable rate. What is important though is that and that 24 computer laboratories are available, free of charge, for learners' use and that the e-learning platform can be accessed and used free of charge (Pienaar 2011). Further, internet access on campus costs only 20 cents per megabyte between 6:00 am to 5:00 pm, 10 cents per megabyte from 5:00 pm to 12:00 pm, and is free of charge between 12:00 pm and 06:00 am (Pienaar 2011). It is, however, important that learners who are not e-ready be trained. As indicated in chapter 4, training to use the e-learning platform is part of first year learners' orientation in the beginning of their first year.

From the feedback obtained in this section it is evident that the majority of academics does not use the e-platform to optimally enhance learning and is in many instances merely used as a distribution mechanism. This underutilisation or misuse of the e-platform is caused by the high level of ignorance with regard to the possibilities and utilisation of e-learning. The feedback further emphasised the need for e-learning for employee development towards optimal performance. The training should be focused on providing academics with the skill to teach online and optimally use e-learning. Included in the training should be new approaches to teaching and learning.

It is also necessary that an awareness and sensitivity of the type of learners and their needs is created and that academics receive training on the best suitable methods to service these learners. Considering the fact that the vast majority of academics are not qualified and trained educationalists but subject experts, it is essential that training is focused on teaching strategies. In this regard the constructivist approach, which is learner centred, can be useful. Further, line managers must fulfil their responsibility to ensure that their academics are well-

239

trained and capable to provide quality teaching and learning of world class to learners.

7.4.1.6.2 Training and development for e-readiness

In this section respondents were requested to indicate whether they consider training and development to become an online learning facilitator to be included in their personal career development plans. Respondents were also requested to indicate the type of assistance they need from their School Directors and/or Deans with regard to teaching online. Respondents were further probed to indicate whether they are of the opinion that the e-readiness of academics should be assessed during a performance appraisal and subsequently be included in the development of an academic that is not yet e-ready. They were requested to motivate their answer. The aim of these questions was to determine academics' commitment to, and understanding of the importance of e-learning in the 21st century, the need for their e-readiness, and their willingness to be assessed on it in the same manner in which they are assessed for their (offline) teaching and research competencies and output. The first question posed in this section is:

• Do you think training on becoming an online learning facilitator should be part of your personal career development plan?

The results obtained for this question are indicated in diagram 7.24:



Diagram 7.24: E-learning training as part of personal career development plan

The fact that 23% of respondents do not want e-learning included in an academic's personal career development plan is indicative that they want e-learning to not

List of research project topics and materials

become a compulsory job requirement but to remain optional. They will in all likelihood also prefer to stick to traditional methods of teaching and learning. This group of respondents will most likely mostly belong to the *late majority*, with a few perhaps to the *early majority*. Zemsky and Massey (2004:9) describe members of the *late majority* as those who adopt when half of the population has already done so. They are followers who dislike the disruptions of new technologies and are more conservative (Zemsky & Massey 2004:10).

As indicated in diagram 7.24, the high number of respondents that indicated that elearning should be part of an academic's personal career development plan, which is almost two thirds (66%), points towards a realisation amongst academics of the necessity to be skilled in e-learning. It is also evident that they realise the need to be trained in order to optimally use technology in teaching and learning. The majority of reasons provided by this group of respondents related to the need to keep up with technological advancements, the use of technology is not voluntary any more, and there is a need to keep abreast with new and innovative methods of training.

One of the respondents indicated that it is to the university's advantage when academics embrace technology and that in the long term it has the potential to add great value to the academic environment. The respondent further indicated that

> "... we also need to prepare our students for the business environment, where the use of technology forms an important part of doing business."

This is a very sensible response as the respondent is able to envisage the role of elearning in the greater whole and the benefits that can be reaped from it. It is also necessary that the NWU position itself in the competitive tertiary environment. As indicated in chapter 5, traditional universities have to adapt to technological changes if they want to survive against other HEIs.

Another significant response was:

241

"In some respects I feel we are already far behind some of our colleagues at other, comparable teaching institutions. Also, I am aware of different faculties and schools on our own campus that apparently utilise very innovative tools and techniques in conjunction with eFundi. This makes me wonder if there is any coherent vision on campus regarding e-learning for the NWU and how they intend to ensure evenhanded and proper implementation of such policy. In a word, I am confused in this respect and somewhat jealous of some of my colleagues across campus."

It is not clear why this respondent, who is clearly e-ready and realising the value of e-learning, is not able to use technology. The respondent is able to use eFundi as the e-platform is available to all academics. However, it appears that he/she is unable to use additional e-learning tools in combination with eFundi. It can be assumed that the necessary support in terms of resources is not available in this respondent's School/Faculty. This comment stresses the need for line managers to firstly realise the importance of e-learning and secondly support their subordinates to use it effectively in the learning environment. It also points to the need of having a consistent e-learning vision, supported by policy. This is exactly what this study is arguing for, that, in the light of technological development and the 21st century learner, e-learning should be a compulsory part of an academic's duties. It is though necessary to first determine an employee's e-readiness in order to determine the most applicable route to follow in terms of training and developing an employee to become e-ready.

As indicated in chapter 5, section 5.3.2, and earlier in this chapter, in many instances the biggest obstacle to e-readiness is not necessarily a lack of skill, but a lack of willingness, a lack of knowledge (ignorance) and an unwillingness to adapt teaching and learning strategies to suit the new type of learner. It is evident that in many instances the paradigm shift from traditional teaching methods to the incorporation of technology in teaching and learning has not been made. Further, the unfounded fear of being replaced by technology that is prevalent due to ignorance regarding e-learning, causes e-learning to not be optimally used to enhance learning. If the NWU

wants to be a world class institution, providing high quality learners, it has to adapt to the global technological advancement.

The second question posed in this section is:

• What type of assistance do you expect from your School Director/Dean to support you for online teaching and learning?

Various types of expectations were indicated by respondents. However, diagram 7.25 indicates those who came forth the most in respondents' replies.

Diagram 7.25: Assistance expected from School Directors/Deans



With almost half of the respondents (48%) indicating the need for opportunity to undergo training, it is evident that there is a definite need for training and that a considerable number of respondents has come to this realisation. Of those that indicated they need the opportunity to attend training, 5,3% specifically made reference to professional training. This is interesting as it implies that they do not perceive the current training provided by Academic Support Services as professional or they are simply not aware of the content of this training and assume that it is not professional. It is also interesting to note that in addition to the opportunity to attend training, respondents also raised the issue of awareness of courses, refresher courses and new updates/possibilities of e-learning. These comments support the notion that more effort should be put in place to create awareness of the possibilities of e-learning, not only by Academic Support Services, but also by line managers. Training and development should be a continuous human resource development function and it is necessary that not only employees realise this, but also line managers and Academic Support Services.
Time, for training and/or to explore and use e-learning or to develop online material was indicated by 25% of respondents, which indicates that a significant number of academics feel they are pressured for time. It is evident that employees are of the opinion that implementing e-learning will be time consuming and add to an already heavy workload. Johannes (2007:71) is of the opinion that the workload demands of teaching online are considerably more than those in traditional teaching practice. Although mind-sets towards e-learning (varying between technophobia and technophilia) will constantly influence the e-learning approach, participation and interest of the academic, it is widely recognised that institutions need policies and benchmarking procedures to provide for changing job demands (Johannes 2007:71). If policies are not in place, making provision for standardisation, the e-readiness of employees, or lack thereof, will continuously determine the use of e-learning.

Once employees have obtained the skill and have a broader understanding and awareness of e-learning and its possibilities, they will have a better understanding that, if well-planned, e-learning does not necessarily create an extra workload. As mentioned earlier, it is also necessary that e-learning training is coupled with teaching and learning strategies. This will assist employees in applying e-learning as a learning tool and not merely a technological tool. This should be regarded as an essential element of employee development.

Fourteen percent (14%) of respondents indicated they don't want any support from their School Director/Dean. It is not clear from their replies whether they are satisfied with current support provided or whether they merely are not interested in any support because they are not interested in e-learning.

Thirteen percent (13%) of respondents indicated other aspects needed with regard to assistance such as inter alia, the provision of resources, hardware and software, soft skills required of managers, motivation and an e-learning policy were also mentioned. Whereas one respondent indicated he/she wants the "option to employ e-learning, or not", another one indicated that it should be made compulsory to all lecturers. The fact that one respondent requested an "assistant that can put my lectures on eFundi" again reflects a lack of e-readiness and an unwillingness to adapt to new teaching and learning strategies. The following comments indicate the need for the soft skills expected from line managers:

"Motivation and information."

"Empathy, patience and recognition in the long run ..."

"Encouragement to lecturers to make use of e-learning platforms."

"I expect of my School Director/Dean to assist me in overcoming any obstacles I might experience in any way possible."

These soft skills are typically the factors that will motivate some employees. The extent to which these should be applied will depend on an employees' position on the technology adoption cycle, personal profile patterns and learning style preferences.

Another comment that reflects the need for motivation from and soft skills required of line management is:

"... the necessity to undergo such training should not be enforced on individuals, but rather be presented as an opportunity to improve the capabilities of a lecturer (within their own time schedules and based on their own developmental needs within their careers)."

In chapter 2, section 2.3.2, the importance of providing a rationale and not forcing employees to do a certain task has been emphasised. When employees are coerced into doing something, they usually become less attracted to the task and will perform it only as long as there is some form of supervision (Gagné et al 2000:1843). This notion is supported by various authors (Roca & Gagné 2008:1588; Pugno

2008:1332; Vansteenkiste et al 2007:252–253). Cognitive evaluation theory, a subtheory of self-determination theory, has revealed that an external control such as coercion will most likely reduce an employee's intrinsic motivation (Roca & Gagné 2008:1588). In contrast to coercion, when a rationale is provided for doing a task, people usually become more attracted to doing the task and are more likely to keep on engaging in it (Moller et al 2006:105–106;110; Deci & Ryan 2000:227).

The following comments signify that the use of technology in teaching and learning should become a broader management issue and that policy and uniform policy implementation should be in place:

"A technology-friendly culture – support paper-free initiatives, move from meetings to online conferencing, embrace social media for student- and staff support – in essence, making all staff confident about the positive impact of technology even if it means to take staff out of their comfort zone."

"It would be mostly financial: buying technology that would improve e-teaching/make it easier to make use of online teaching; paying assistants to help with the input of questions on the database on eFundi."

"Policy and innovative thought regarding the adoption, availability and use of tools and techniques of e-learning."

The abovementioned comments also suggest that in cases where academics are eready they do not necessarily receive the required support in terms of funds, resources and motivation from line management. If e-learning is included as compulsory job requirement, as part of an academic's service level contract, ereadiness should be assessed, a personal development plan should be drafted, and consequently training should follow. It is, however, important that line managers buy into this process as they will have to steer it, together with Academic Support Services. A lack of understanding, commitment or motivation from line managers will lead to failure and negativity. In this regard the role of line managers pertaining to employee motivation cannot be emphasised enough. As some academics are already not inclined to the use of e-learning – typically those who fall in the *early majority* and *late majority* categories, it is advisable that this process is handled with empathy, consideration and applicable motivation.

It can be deduced that academics expect their School Directors/Deans to ensure they receive training, as well as time to attend training. Other factors indicated as required by academics from their School Directors/Deans were the provision of resources, hardware and software, policies for e-learning, soft skills of managers, in particular motivation and providing a rationale for e-learning.

The third question posed in this section is:

 Do you think e-readiness should be assessed during a performance appraisal and subsequently be included in the development plan of an academic that is not e-ready?

The results obtained from this question are indicated in diagram 7.26.

Diagram 7.26: E-readiness assessment preference



The number of respondents indicating that they are willing to undergo e-readiness assessment as part of performance appraisal (44%), is very close to the number indicating that is not in favour of e-readiness assessment as part of performance appraisal (39%).

In the following paragraphs respondents' e-readiness assessment preferences are linked to their teaching experience and age with the purpose to determine whether certain trends could be identified.

• Teaching experience and e-readiness assessment preference

Of the academics with less than five years teaching experience, almost 70% indicated that they believe the assessment of e-readiness should be part of the performance appraisal of an academic. One respondent indicated that e-readiness assessment should only be included in a performance appraisal if it is clearly indicated as part of an employee's duties in a job description, and another respondent indicated that it should only be included in a performance appraisal if it is connected to an employee's personal development plan. Thirty percent (30%) of respondents indicated that e-readiness assessment should not be part of the performance assessment of an academic. Respondents that indicated a condition for e-readiness assessment or who opposed e-readiness assessment, were mostly between the age of 36 to 40 and one is almost 50 years of age. One can presume that these three employees, irrespective of their age, having less than five years of teaching experience, were probably employed in their respective industries in practice. In their professional practices they were probably not confronted with elearning and are therefore relatively new to the phenomena. It is also evident that age play a role in the willingness to be assessed for e-readiness as most of the respondents in this group are in their twenties and almost 70% are willing to undergo e-readiness assessment as part of their performance appraisals.

With employees that have teaching experience between five and 10 years, 47% indicated that the assessment of e-readiness should be part of an employee's performance assessment; 26,5% indicated that they are not sure or have reservations of some kind; and another 26,5% indicated that they do not think that the assessment of e-readiness should be part of a performance appraisal. Therefore, in this group (of five to 10 years teaching experience), 53% of respondents were against the assessment of e-readiness or stated conditions to it. The respondents in this group are mostly in their thirties, thus supporting the notion that age plays a role in the willingness to be assessed on e-readiness.

Forty two percent (42%) of those respondents with 11 to 20 years of teaching experience indicated that they believe that e-readiness should be assessed during a

performance appraisal of an academic, 50% indicated that it should not be part of a performance appraisal and 8% indicated that it should only be part of a performance appraisal if the development of e-readiness is part of the personal development plan of an academic.

Of those respondents who have between 21 to 29 years teaching experience, 42,5% of respondents were in favour of an e-readiness assessment, 47,5% were against it and 10% indicated that they do not support such an assessment, but will also not object to it if management decides to follow this route. With the group with teaching experience of 30 years and more 31,5% indicated that they believe e-readiness of academics should be assessed during a performance appraisal and 68,5% indicated that they do not support the idea of e-readiness assessment during a performance appraisal. The notion of younger colleagues, more in favour of e-readiness assessment as opposed to older colleagues, is again confirmed by these results, as employees with more than 30 years teaching experience are mostly in their fifties or sixties. Diagram 7.27 illustrates the relation between teaching experience and willingness to undergo e-readiness assessment.



Diagram 7.27: Teaching experience and e-readiness assessment preference

A total number of 46,6% of respondents, across all years of teaching experience, indicated that they believe e-readiness assessment to be part of a performance appraisal of an academic, whereas over 44,5% of respondents therefore do not

believe that this should be the case. The number of respondents in favour of ereadiness assessment and those against it are therefore divided almost 50/50. If this is representative of the broad spectrum of academics at the NWU, Potchefstroom campus, it can be argued that almost 50% of academics will resist the introduction of e-readiness assessment, if made compulsory. This anticipated resistance implicates that both line managers and institutional managers will be obliged to apply management of change during the initial phases of including e-readiness assessment in performance appraisals. In this respect motivating employees will be crucial. In chapter 2 the importance of keeping employees motivated during changing work circumstances was outlined, in particular providing a rationale for the changed job demand.

It is further evident that with the increase in teaching experience a decline in favour of e-readiness assessment as part of a performance appraisal of an academic has occurred. It can be argued that the longer an employee is teaching, the more they want to stick to what is known and the less they want to be confronted with new challenges, new technologies and new teaching methods. These results, to a great extent, reflect a lack of willingness to be assessed on the use of technology in teaching and learning amongst a significant number of experienced academics. In chapter 4 it has been indicated that a willingness to teach online portrays a significant aspect of e-readiness. One can therefore deduce that a willingness to be assessed on e-skills can be regarded as one of the core components of being eready. When the willingness is lacking, it is highly likely that the e-readiness is also lacking and that it will be a challenging task to convince these employees to make use of e-learning, let alone use it interactively.

Age and e-readiness assessment preference

Seventy three percent (73%) of respondents in the age group of 20 to 29 years are in support of e-readiness assessment as part of a performance appraisal, therefore the vast majority of this age group. In the age group of 30 to 39 years, 33% of respondents are in favour of e-readiness assessment, 43% are not in favour of it and another 27% are not sure or will support such an assessment, provided that it is part of an employee's service level agreement and personal development plan. Fifty List of research project topics and materials

percent (50%) of respondents in the age group of 40 to 49 years support e-readiness assessment, whereas the other 50% is opposed to it. In the age group of 50 to 59 years, only 38% of the respondents responded positive to e-readiness assessment and 62% negative. In the age group of 60 years and older 66% are in favour of e-readiness assessment as part of a performance appraisal and 33% are against it.



Diagram 7.28: Age and e-readiness assessment preference

Diagram 7.28 above illustrates that there is a general decrease in support of ereadiness assessment as part of a performance appraisal of academics with the age increase of academics. These results correspond to a great extent with the results pertaining to the years of teaching experience of an academic, indicated in the previous section. It is thus evident that the younger the academics, the more willing they are to be assessed on their e-readiness and the older they become, the less willing they are to be assessed on their e-readiness. It can be argued that the willingness to be assessed on e-readiness is due to the fact that most of the younger employees are already e-ready and are comfortable with using technology in teaching and learning. On the other hand, the older employees are, the less they want to be confronted with a new challenge. This trend is evident in diagram 7.28 above with all the age groups, with the exception of those of 60 years and older. Surprisingly, 66% of the respondents in the age group of 60 years and older responded positively towards e-readiness assessment. This positive attitude amongst this group of respondents can in all likelihood be attributed to the fact that these respondents are all School Directors, and view e-readiness from a strategic point of view, and therefore has an understanding of the need for and benefits of eready academics.

In addition to the deductions that could be made between the influence of respondents' years of teaching experience and age on their e-readiness assessment preference, it is noteworthy that 49% of respondents has been categorised as *early adopters* on the technology adoption cycle in section 7.4.1.2.1. In section 7.4.1.3.1, 49,4% of respondents indicated that they usually embrace the opportunity if new challenging goals are assigned to them as it motivates them to develop their skills and demonstrate competence. However, when confronted with a specific challenging goal in this question (e-learning), the same result is not obtained. Five percent (5%) less of respondents who earlier indicated they will embrace new challenges and new technology relatively fast, do not support the assessment of e-readiness, which will lead to training, preparing them to achieve this goal. Several reasons can be responsible for the 5% of respondents who are willing to get involved in e-learning, but are not in favour of being assessed on their e-readiness. Some of these possible reasons are discussed in the following sections. An aspect that could also have caused the 5% difference is that all respondents did not answer all questions.

Reasons provided by respondents in favour of e-readiness assessment as part of a performance appraisal portrayed that, although indicating in favour of it, some still have reservations about it. A number of respondents indicated that e-readiness assessment should not be compulsory and that academics should have the freedom of choice to make use of e-learning or not. The following comment illustrates this notion:

"Yes, on a voluntary basis. If not voluntary, people will be negative towards e-learning."

If e-readiness assessment is done voluntary, the status quo will remain and no progress will be made in terms of e-learning. It is, however, of the utmost importance that e-readiness assessment is introduced with care, consideration, a personal development plan and a personalised approach of motivation.

252

The fear of change again came forth in a number of responses, indicating that ereadiness can be assessed but that e-learning should not replace classroom teaching. As long as academics are not properly trained and informed about the use of e-learning, the perception of it being a distance learning medium or a medium that will replace the lecturer as person, will remain. This in itself causes the underuse of e-learning which could have been effectively applied to enhance learning to the modern learner who thrives on technology.

Some of the respondents indicated that the need for e-readiness assessment depends on the nature of a subject. This can be taken into consideration as it is unlikely that a practical music subject such as violin lessons will find use of e-learning to the extent that it will make a difference in the learning of a learner.

Reasons provided by respondents not in favour of e-readiness again highlighted the resistance to change and the misconceptions regarding the use of e-learning. Aspects such as e-learning not being suitable for postgraduate learners and the importance of personal contact classes have been mentioned. A number of respondents also indicated that they are not interested in e-learning and do not want to use it. However, reasons provided are based on misconceptions such as:

"No, we are not a distance education university. Class attendance is still the most important part of a lecturer's task."

Unless it is the aim of an academic to reduce contact time with learners, class attendance is probably poor due to the fact that the e-platform is used as a distribution mechanism where all information is placed, making class attendance unnecessary to a learner. Another reason for poor class attendance may be the approach followed in class. If a lecturer is not using a teaching and learning strategy that is most applicable to the type of learner in his/her class, and also when a valuable proposition is lacking in class, which make it necessary for a learner to attend class in order to fully comprehend the syllabus and answer test and exam questions, it may lead to poor class attendance.

253

"Students are being spoon-fed too much."

This comment illustrates the ignorance about how e-learning should be used. This misconception probably originates from the manner in which e-learning is used by most academics, as distribution mechanism. In this sense it can be seen as spoon feeding as notes material and power point presentations are posted on eFundi without any responsibility placed on the learner and no tasks requested from them. By using e-learning correctly, the learner will have to take responsibility for learning as well and a skilled online learning facilitator will be able to give online tasks that are challenging and relevant. An example that can be mentioned is: to place a video clip, relevant to a particular subject matter on eFundi, request learners to watch it and answer certain questions afterwards as preparation for class; or learners can be referred to news websites or subject related websites, researching for specific aspects, based on the assignment and provide feedback by means of an essay, assignment, report, debate, presentation, to name a few.

As indicated in chapter 5, section 5.2.1.2, Bloom's lower order thinking skills can be developed by repetition and drilling, which can be done through online quizzes, tests, simulations and games. On the other hand, higher order skills can be developed through online assignments and forum discussions. The key to the level of the skill to be obtained still lies within the manner in which the task is articulated (through the action words used) and what is expected of the learner (eg synthesise, evaluate, discuss, compile, draft, analyse). Misconceptions like the one mentioned above, arise due to a lack of e-readiness, a lack of comprehending the role of e-learning and a lack of knowledge regarding different teaching and learning strategies. All these aspects can be addressed through employee development.

Other comments indicating the lack of knowledge of the use of e-learning are:

"I am not sure it will contribute towards anything."

"Although I realise that we have to keep up with the times, we should be careful that the e-learning approach does not erode the character of a good teacher." Clearly, the last comment indicates ignorance regarding the manner in which elearning can enhance the teaching skills of a good teacher. One can argue that the ability of a good teacher does not lie within being a good teacher in class, but to ask the question, "Did learning take place?" and being able to answer yes without a doubt, irrespective of which teaching and learning medium was used. These comments further highlight the fear of the unknown. Again these misconceptions can be addressed through training and development. It is evident, however, that an employee's e-profile has to predict the approach followed. The same approach cannot be followed with an employee categorised in the *late majority* category as one in the *early adopter* category as more motivation and more intense training and development will be necessary with the employee in the *late majority* category. In addition to the style and pace of technology adoption, the employee's personal profile pattern and preferred learning style also needs to be taken into consideration.

Another comment, illustrating ignorance is:

"No. It is not part of core activities."

It is clear that this respondent does not realise that e-learning is in actual effect teaching and learning, merely with a different medium. If teaching and learning are included in the core business of a university, surely e-learning will also be regarded as core business. Learners, as the core business of HEIs are expecting to learn. If e-learning are withheld from a learner that thrives on using technology, as is the case with Generation Y, it is the responsibility of academics to ensure that they provide a quality service to this type of learners in the manner which will make most sense to them and that is likely to contribute most to learning. As indicated in chapter 4, to expect of these learners, the so-called generation Y, to yield the realities of their world when they enter the classroom is to disregard the global role and place of technology (Amirault & Visser 2009:66). Yet another comment signifying not only the role of e-learning, but also the needs of the learners is:

"No, no it is just there to help lecturers."

From the majority of comments made by respondents not in favour of e-readiness assessment, it became evident that these comments are founded in fear, ignorance or misconceptions. These comments revealed the following beliefs of respondents: that e-learning does not contribute towards learning; that it may erode the character of a good teacher; that it is meant to be used for distance learning; that learners are spoon-fed through e-learning; and that e-learning will cause a decline in class attendance of learners. A number of comments also illustrated the belief that e-learning is not part of an academic's core activities and should remain optional.

A number of respondents in favour of e-readiness assessment indicated that as they are already spending work time on incorporating e-learning in their subjects, and the fact that it took them time and effort to become e-ready, they are in favour of ereadiness assessment as it will be included in their performance agreement and will affect their performance rating positively.

A number of respondents also indicated the matter of worldwide technological development and trends, the new type of learner, Generation Y, and social media. Some also indicated that e-learning is part of the duties of the modern lecturer and that e-readiness assessment and the consequent training will assist academics in increasing their effectiveness and efficiency with regard to teaching. Some of the comments of respondents in favour of e-readiness were:

"The importance of being able to use web-technology is evident in most disciplines that our students will move into after their studies. Academics should thus also ensure that they have mastered that skill, both to prepare the students for their future careers and also to ensure that the academics themselves stay up to date with the changing times."

"An institution's strength lies in the skills and knowledge of its academics. If any institution wishes to have a competitive edge it, and therefore its employees, need to be on par with international standards and methods of teaching for quality education."

"... our students expect us to be e-ready."

These comments reveal that a number of academics are able to view the role and place of e-learning in the broader environment and understand the need for including it in teaching and learning. These respondents are in all probability e-ready to a great extent and will in all likelihood be ready for advanced a-learning training. However, a significant number of academics appear to not be e-ready, mostly because of misconceptions and ignorance and in some instances a lack of skill.

It is apparent that the Academic Support Services unit has a significant role to play with regard to e-readiness training and development. Not only should this training be focused on obtaining an e-learning skill, but it should also be aimed at providing information on the philosophy and possibilities of the use of e-learning. It is further of the utmost importance that academics are trained and developed to understand the role and place of e-learning as part of a broader teaching and learning strategy. It is necessary to school academics in pedagogy as most academics are subject experts and not educationalists.

It is further evident that the majority of academics are not aware of the various teaching and learning strategies and do not realise the need to adjust a teaching and learning strategy to provide for learners' needs and preferences. Training should furthermore significantly place the focus on optimal learner learning as end-result of any teaching and learning strategy, irrespective of the medium used (classroom, e-learning, mobile learning). As indicated in chapter 4, e-learning is no longer only linked to distance or remote learning, and is also no longer regarded as a specialist area, but has become an element in a conscious decision to use the best and most suitable ways to enhance effective learning through the use of suitable technology (JISC 2009:6;9). The "e" in e-learning, referring to "electronic", adds a technological edge to the "learning", but the focus remains on leaning as the critical element (JISC 2004:8).

The Joint Information Systems Committee (JISC) of the Higher Education Funding Council for England explains that although learners' are familiar and comfortable with technology, they lack the "critical and evaluative skills" required to construe information from online resources. This highlights the need to support learners in developing digital literacy skills. Thus the skill of the online learning facilitator is vital to the effectiveness of learning. Online learning facilitators need to be aware of the various tools and media that are increasingly available and how to select and use various tools and media to best serve the learning purpose. The blended approach to teaching and learning, with the skilful and suitable incorporation of technology with face-to-face classroom teaching, is a significant element of 21st century teaching and learning practice. When planning and designing learning, online learning facilitators must ensure that they explicitly make use of the most appropriate learning strategies and technology that will have the biggest impact on the way learners learn. (JISC 2009:7.)

It can be assumed that once academics are schooled in teaching and learning strategies, they will be better able to understand and appreciate the value and use of e-learning and the focus will naturally shift to the enhancement of learning and away from the fear of or resistance to technology. The notion of e-learning being a distance learning medium will also be addressed with comprehensive training.

As this venture is broader than merely e-learning training, and will also focus on the complete development of an academic to be a modern, well-skilled, qualified and successful lecturer, providing world class education to learners, a personal development should be drafted for each academic. Line managers should also realise the significant role they have to play in this regard. As line managers are jointly involved with employees in drafting an employee's performance agreement, they have the responsibility to do the same with regard to the personal development plan.

As indicated earlier in this chapter, the majority of respondents portray a pragmatic learning style, combined with a compliance DISC factor and an early adopter technology adoption style, which indicates they want to understand the practical value and use of what are being taught and therefore line managers should ensure that this is clear and understood. Line managers further have the very important role of motivating staff, not an easy task as employees, based on their various learning style preferences, personal profile patterns and pace and style of technology adoption, are motivated by different factors. For introducing something as ereadiness assessment to performance appraisal for the first time, motivation, communication and rationalising are crucial as many academics that are not e-ready will portray a negative attitude and resist the new job demand.

Academic Support Services currently have an academic staff development programme called the Institutional Teaching Excellence Award (ITEA), which is aimed at developing academics to become excellent teachers, as discussed in chapter 3. The focus of the programme is precisely on the teaching aspects that are argued for in this study: applying innovative teaching strategies, including technology, facilitation rather than lecturing, interactivity and high class participation, teaching on the relevant National Qualifications Framework (NQF) level and applying Bloom's taxonomy applicably to these levels. This programme is however, to a great extent, experienced as negative amongst a significant number of academics and strong resistance to participate in it is experienced. It can be argued that academics strongly resist participating in ITEA as it forces them out of their comfort zones, poses new challenges, expects of them to apply new teaching and learning strategies and provide constructive criticism of how to improve teaching and learning strategies. It will typically be employees portraying a high *dominance* factor on the DISC profile, with an activist learning style and an innovator technology adoption style and pace who will volunteer to participate in ITEA. Participation in ITEA is at this stage by nomination from the School Directors and the Dean (NWU 2009a:18).

The shortcomings that can be identified, however, is that the use of teaching and learning strategies to enhance the use of e-learning as learning tool is not specifically dealt with in this programme and it is assumed that the academic already is competent in this respect, unless the employee entered the Development ITEA. In the latter case, however, as with the ICNL the development is not specifically targeted on the adjustment of teaching and learning strategies for the use of technology in teaching and learning and is mostly focused on classroom teaching and learning.

Although ITEA highly encourage the use of technology and innovative teaching, it does not make provision for e-learning training. Another section of Academic

Support Services, Information Technology Support, is responsible for e-learning training. This training is, however, only focused on training for the use of the elearning platform, eFundi (NWU 2009a:11). It does not provide information on the philosophy of e-learning and its role and place in the broader teaching and learning environment. A further shortcoming is that the use of teaching and learning strategies to enhance the use of e-learning as learning tool is not specifically addressed in this programme and it is assumed that the academic already is competent in this respect. From the feedback obtained from the questionnaire, it is evident that more than obtaining a technical skill is necessary. The literature research also confirmed that e-learning is no longer regarded as a distance learning or remote learning tool, or a specialist area, but that it has become part of the broader aim of enhancing teaching and learning through the use of suitable technology (cf Birkbeck 2005:1). The focus of e-learning is therefore not to replace the teacher, but to use technology to increase the focus on pedagogic skills (JISC 2009:5). To a great extent a paradigm shift still needs to be made amongst academics from traditional class room teaching and learning to including more innovative teaching and learning methods with the 21st century learner.

The literature study further indicated that the skills required of an online learning facilitator such as being a good listener, providing feedback, communication and encouraging learners and building relationships, mentioned above, all relate to the human nature of e-learning (cf Hootstein 2002). Online learning facilitators should also possess the ability to let learners feel comfortable about participating in online discussions and sharing information (National College for School Leadership 2006; Sutton 2004). It is thus evident that academics are not aware of the human nature of e-learning and that this awareness should be created during training and development.

The feedback further emphasised the need for e-learning training for employee development towards optimally use e-learning for an enhanced learning experience. Training should firstly focus on creating awareness of the philosophy of e-learning and the possibilities with the use of e-learning, and secondly to provide academics with the skill to do so. Included in the training should be new approaches to teaching and learning. To ensure learners' needs are provided for and that their learning is VI List of research project topics and materials

enhanced through e-learning, it is necessary to create an awareness of the type of learners and their needs and that academics receive training on the best suitable methods to service these learners. Considering the fact that the vast majority of academics are not qualified and trained educationalists but subject experts, it is essential that training is focused on teaching strategies.

The feedback obtained further revealed that the majority of academics use the eplatform as a distribution mechanism and not as an interactive learning tool. It showed that in many instances academics are not aware of what is regarded as interactive e-learning and should thus be included in the e-learning training and development.

It also became eminent that an academics' age and years of teaching experience influence their preference to e-readiness assessment. Younger employees with less teaching experience are in general more inclined to e-readiness assessment as opposed to older employees with more teaching experience. E-learning training and development programmes for older employees can thus be customised to perhaps follow a slower pace and to focus more attention on the use, philosophy and advantages of e-learning as it can be argued that the resistance to e-readiness assessment may be born out of fear for the new job requirement.

7.4.2 Focus group

As indicated in section 7.3.5.2.1 a focus group discussion was held with academics to determine their perceptions on e-learning, learners' expectations, and the role of the academic in e-learning. The researcher follows an approach of providing the feedback obtained from participants on each question, where after the researcher's interpretation on the feedback follows. The following discussions emerged from the questions posed to participants:

7.4.2.1 Question 1: How do you perceive the current role of e-learning in the higher education environment?

The purpose of the question was to determine focus group participants' views on the role of e-learning in higher education. Their views would give an indication whether they have an accurate understanding of the role and purpose of e-learning.

7.4.2.1.1 Summary of feedback on question 1

The discussion was started by a participant, depicting a strong resistance to elearning and a significant lack of e-readiness due to technological illiteracy. The participant indicated that he only started using power point presentations in lectures recently and is far from comfortable to engage in e-learning. Throughout the discussion he retained a stance of resistance to e-learning and a preference to a traditional approach to teaching and learning, namely "talk-and-chalk". Another participant indicated that e-learning is intimidating when you are older. This participant left academia for a few years and when he returned was frightened by the idea of mastering e-learning skills.

One participant is of the opinion that e-learning can make the life of both learners and academics easier due to the depth of its resources, but the drawback is the affordability and accessibility of internet connection in South Africa. Other participants also indicate that some of their learners do not have access to the internet and therefore not to the e-learning platform. It is mentioned that two economies or two poles are evident with the learner population: those who portray true Generation Y characteristics (embracing technology) and those who do not have access to various electronic means. An example is given of a learner that searched for something in class quickly on his *Blackberry* smart phone and gave the answer within seconds. However, all learners are not equipped with smart phones and in some instances older learners on postgraduate level, especially those who are employed in the local sphere of government, do not always have access to email, let alone the internet. Further, the context of a computer environment was not known to all learners from a young age, especially those who are from previously disadvantaged groups. One participant drew the attention to the fact that the higher education environment has changed. Academics learned to become facilitators to adjust to changing teaching and learning expectations. Learners do not receive lectures as before but should take responsibility for their own studies as well. They need to search for information between the richness of sources on the internet and come prepared to class. This participant foresees that the use of e-learning will drastically increase in the next few years and that it will eventually replace contact classes. He received significant resistance from a number of other participants in the group regarding the notion that all classes will become online.

It is also mentioned that e-learning should be regarded as an enabler. It is merely another modus of delivery in addition to classes and can never replace the class situation. It should only carry a weight of approximately 25% to the total teaching and learning at tertiary level. This is supported by another participant, indicating that indepth learning cannot be obtained though e-learning. The participant gives an example of an interactive approach that was followed with a second year group of learners, making provision for using the study guide as a road map, referring learners to the e-platform and from there to certain activities to be completed. He experienced that it was not successful. He further experienced that learners do not have the ability to think analytically and to refer them to an online source to gather information and to complete online activities would merely confuse them. He stated that lecturers should rather use scheduled class periods to improve learners' skills and higher order thinking. There is a general consensus that learners are perceived to be on a low academic level and the concern is raised that e-learning will not be able to assist in this respect.

During the discussion reference was made to the psyche of Generation Y, indicating that they are not interested in personal relationships and that they will send a text message regarding an important happening in their lives rather than giving the news in person. The concern that social skills will be lost is raised. It is also mentioned that learners are merely interested in passing a subject/course. They are not interested in the depth of a subject, the underlying philosophies and where different aspects really fit into the broader context. It is also mentioned that a sense of curiosity is not

created on school level. They simply want to know what to study to pass the exam. It is further mentioned that learners do not participate in class, which creates a oneway communication. One participant, previously employed as an academic/researcher at a university in the United States of America (USA), stated that one of the problems pertaining to the inefficient use of e-learning is that outcomes-based education (OBE) is not fully functioning in South Africa and that neither learners nor lecturers know how to practice OBE. A number of other participants made comments opposing OBE.

The fact that academics are not trained and qualified educators, but are rather subject specialists, is mentioned as another reason for the insufficient use of e-learning. The participant who raised this issue is, however, of the opinion that the subject content suffers when academics try to improve themselves as educators. It is also mentioned that all academics do not grow and become better with years of teaching experience and that a significant number of academics oppose e-learning.

Another participant mentions the role of the programme ITEA (discussed in section 6.3.1). He is of the opinion that e-learning definitely plays a role in this regard. The participant further indicates that he has completed the ITEA programme but does not really want to use all the tools and methods that are expected of an academic when taking part in ITEA.

The difference between subject content was also mentioned as a cause for not using e-learning. Some academics are of the opinion that e-learning cannot be used for postgraduate learners. Others are of the opinion that due to the nature of the subject it is more important to discuss matters with learners in class. However, one of the participants referred to a particular subject where animation videos were used to illustrate particular aspects, with great success.

Participants are further of the opinion that the type of learner, the level of control an academic has over them (residential vs distance learning), the NQF level and the demography will determine the use of e-learning.

7.4.2.1.2 Researcher's interpretation on question 1

From the abovementioned discussions, emerging from question 1, it is evident that a significant number of misconceptions pertaining to e-learning still persistently stick with academics. The following deductions could be made from the discussion:

- Age plays a role in e-readiness: the participant who openly and honestly resist e-learning in its totality, and the participant who indicated that he finds e-learning intimidating, are both approximately 60 years old. This confirms the trend of the higher the age, the less interest in e-learning, as identified through the questionnaire results.
- The two types of learners, those who perfectly fit the Generation Y profile, and those who are either not e-ready or do not have access to e-learning, is a reality experienced in the higher education environment. However, the participants' discussion pertaining to learners who do not have access or who are not e-ready referred to approximately 20 master's learners in the School of Social and Government Studies. These 20 learners are hardly representative of the majority of the learner population and are mostly mature individuals who are employed and study part time. The statistics given in chapter 6, section 6.4.1 proves the majority of learners (96%) prefer the inclusion of e-learning in their courses to various extents. It is a fact that certain subjects such as practical music, for example piano lessons do not lend itself to e-learning in a manner that will enhance learning in the particular subject field.
- The comment pertaining to the affordability and accessibility of internet connection in South Africa holds truth. The income inequality affects affordability and access to various social resources including ICT services (Oyedemi 2009:153). However, as indicated in section 7.4.1.6.1, all residential learners have access to the e-learning platform on campus (Pienaar 2011).
- Academics do not realise the role that e-learning can play in the higher education environment. Indicating that the university will be a complete virtual campus in a few years is not a realistic expectation. Firstly, the NWU is predominantly a residential university and not a distance learning university. As indicated in previous sections, the aim of e-learning should not be to do away

with contact sessions, rather to enhance the total learning experience and development of particular skills. As indicated in various chapters, e-learning is no longer merely linked to distance or remote learning, but has become an element in a conscious decision to use the best and most suitable ways to enhance effective learning (JISC 2009:9).

- When academics prefer to reduce lectures through the use of e-learning it should be the lecturer's prerogative, provided that the learning outcomes are obtained and that the amount of notional hours, as expected by the South African Quality Authority (SAQA), is still adhered to. As with the results obtained from the questionnaire, feedback from the focus group illustrate that widespread ignorance regarding the role and place of e-learning is evident amongst academics. Very few academics are aware of the possibilities of e-learning and its role and purpose in the broader teaching and learning environment. It is further apparent that focus group participants did not realise that the crux of the matter is to adjust teaching and learning strategies to adapt to modern learners' needs.
- The mentioning of e-learning that should not replace the class situation, which was also evident from the questionnaire feedback, can be interpreted as fear and ignorance regarding the role of e-learning. As recorded in chapters 1, 4 and 5 (sections 1.1, 4.2 and 5.3), it is known that one of the reasons employees resist change is due to fear (Robbins 2003:559-560). This may include fear of job loss, fear of changing expectations, fear of the unknown, fear of technology, and fear of giving up a comfort zone (Proctor & Doukakis 2003:268; Robbins 2003:559-560). In chapter 5 the connection between resistance to change and a lack of e-readiness was made. Bozarth (2006:2) indicates that barriers to change (such as employees' resistance to the inclusion of technology in teaching and learning) can be classified into first order or second order barriers. First order barriers refer to extrinsic factors such as organisational support or access to equipment and second order barriers refer to more emotional, essential, personal issues, which are intrinsic and related to a person's beliefs and attitudes (Bozarth 2006:2). Important to this study is the underlying, second order barriers of resistance to e-learning amongst online learning facilitators. At the NWU (Potchefstroom campus), factors that could cause the first order

barriers are in place: access to equipment, the internet, hardware and software, development support and technical support (Le Roux 2009). The barriers to elearning are therefore caused by the second order barriers, which refer to the intrinsic personal preference, beliefs and attitudes. Hence, the notion that the key human factors as discussed in chapter 4 have an influence on employees' e-readiness.

- The researcher is of the opinion that putting a percentage limit of 25% to the use of e-learning is defying the purpose. The focus should rather be on how it should be used, for which purposes (which learning outcomes should be achieved), and whether it is the best teaching method to obtain these.
- The notion that in-depth learning cannot be obtained though e-learning, reflects ignorance to what e-learning is what how it can be used. In chapters 3 and 5 it has been indicated that e-learning can be effectively applied to make decisions regarding the use of Bloom's Taxonomy. For the development of higher order skills such as analytical thinking, technologies should be intended to accomplish research, collaboration, information gathering and comprehension, and content construction (Amirault & Visser 2009:76). Constructivist approaches that are characterised by computer based research, collaboration, and content construction tools, are exceedingly effective in achieving Bloom's higher order skills (Amirault & Visser 2009:76).
- The general consensus amongst participants that learners are perceived to be on a low academic level is supported by various authors and other stakeholders (Parliamentary Monitoring Group 2010; Taljard 2010; Gabriel & Flake 2008:1). Globally universities face this concern (NCES 2003:1; Ramsden 2008:3; Ramsden 2003:4). However, the concern that e-learning will not be able to assist in this respect is ungrounded. Chapter 5, section 5.2.1.2, has referred to the fact that technology can play a significant role in the process of preparing underprepared learners for the academic challenges of tertiary education, but is, however, not yet employed to its fullest extent at all universities (Amirualt & Visser 2009:72). It is evident that academics are not aware of the ways in which technology (eg practice exercises, quizzes and simulations on the e-platform) can be applied to assist with enhancing particular skills of underprepared

learners. The type of skill to be improved will determine the type of technology to be used and/or the manner in which it should be applied.

- The comment pertaining to the psyche of Generation Y, indicating that they are not interested in personal relationships is another misconception amongst participants. One of the key characteristics of Generation Y is that they are collaborative, extremely social, and desire to be continually connected to their social networks (Gen Y Report 2010:24). This misconception raises concern whether academics are aware of their learners' needs and preferences. It certainly explains why e-learning is not used to a greater extent.
- The notion that learners are merely interested in passing a subject/course and are not interested in the underlying philosophies and broader context can be related to the underpreparedness of learners. As one participant has mentioned it can indeed be due to a lack of curiosity created during their school education. On the other hand it may also be that learners do not find a particular subject interesting or, it may be that the lecturer's teaching strategy does not appeal to learners. As alluded to in a previous section, poor class participation can in all likelihood be attributed to the academic's teaching style and strategy. It has been recorded in chapters 3 and 5 (sections 3.2.3 and 5.2.1.1) that traditional teaching and learning methods are unlikely to appeal to the 21st century learner or enhance learning in the best possible way. In chapter 3 (section 3.2.3) it has been indicated that academics usually continue to teach in the same manner they always have. If an academic therefore believes in the approach of conveying information, he/she will use e-learning to facilitate this manner of learning, and any tools on the e-learning platform that are not aligned with this approach will be either ignored or misrepresented (Elgort 2005:184).
- The perception of OBE not being fully functioning in South Africa and that neither learners nor lecturers know how to practice OBE, may signify a truth. As indicated in chapter 3 (section 3.2.4), the majority of academics appear to still be following the behavioural, objectivist approach to teaching and learning, driven by "talk-and-chalk" (Hanley 1994:3). Objectivists hold the idea that there is a fixed world of knowledge that the learner must come to know and educators serve as channels through which their thoughts and meanings are

transferred to the passive learner (Hanley 1994:3). Whereas the objectivist approach gives emphasis to observable, external behaviours and therefore, steers clear of reference to meaning, representation and thought, constructivism follows a more cognitive approach (Gergen sa). In the constructivism approach educators are coordinators, facilitators, resource advisors, tutors or coaches. (Gergen sa). Literature indicates that people's epistemological beliefs mirror their perceptions about the nature of knowledge and knowing (Tu et al 2008:1143). These perceptions are found to be related to their common learning habits, or their approaches to processing learning tasks (Tu et al 2008:1143). Users' epistemological beliefs direct their cognitive as well as meta-cognitive activities in web environments. Users with constructivistoriented epistemological beliefs are inclined to have greater preferences to engage in meta-cognitive thinking in online environments, as opposed to those who do not have constructivist-oriented epistemological beliefs (Tu et al 2008:1143).

- A correct observation made by participants is that academics are not trained and qualified educators, but rather subject specialists. However to assume that the subject content suffers when academics try to improve themselves as educators, does not carry weight. If academics do not make themselves available to improve their teaching skills, it may become cause to another comment made by participants: "All academics do not grow and become better with years of teaching experience". It can also be argued that the subject content will not be applied to its fullest extent if approached with a teaching strategy that does not appeal to learners. Ramsden (2003:xii) supports this notion by telling academics that the first step to become a good teacher is to understand learners' experiences of learning. Milliken and Barnes (2002:225) indicate that this implies that teaching and learning strategies may have to be adapted to focus on the enhancement of learners' learning. The application of new technology can be brought into play to improve both the teaching and learning experience (Milliken & Barnes 2002:226).
- The fact that one of the participants refers to the credibility of the ITEA programme, but at the same time indicates an unwillingness to comply with all

269

its requirements, points towards unwillingness to adapt a teaching and learning strategy when being taken out of a comfort zone.

- Comments referring to the difference between subjects and subject content not always necessarily ideal for the use of e-learning reflect a reality. It can be argued that in a practical music subject (eg practical piano; practical flute; etc), it does not make sense to use e-learning. With the theoretical music subjects, however, one may be able to use e-learning. In some instances academics use subject content as an excuse for not using e-learning. This can be due to the fact that they do not want to use e-learning or do not know how to use it.
- The fact that participants are of the opinion that the NQF level will determine the use of e-learning, further indicates ignorance with regard to e-learning. As indicated earlier in this section, academics are sometimes under the impression that e-learning cannot be used for postgraduate learners. This proves the necessity of e-learning training to create awareness and skill for the use of elearning for learners at all levels.

The feedback obtained from this question illustrated that participants do not have a clear understanding of the role and purpose of e-learning. It further indicated that a number of misconceptions, rooted in ignorance, are still eminent amongst academics. It is necessary to address these misconceptions by creating awareness of the role of e-learning and its possibilities, coupled with the best suitable teaching and learning strategies for learners.

7.4.2.2 Question 2: How do you perceive the future role of e-learning in the higher education environment?

Question 2 was posed to participants in the focus group discussion to determine their perceptions on the role that e-learning will play in future in the higher education environment. The purpose of the questions was to determine whether their expectations were realistic and also to give an indication of how they perceive the role of e-learning to evolve.



7.4.2.2.1 Summary of feedback on question 2

Participants indicated that e-learning will be more used and more integrated in coursework in future. One of the participants indicated that the section that deals with master's and doctoral learners, are planning on performing all administrative work, including applications for ethical approval, electronically. As indicated in the previous question's feedback, one of the participants foresees a complete virtual classroom. The same participant focused the other participants' attention on the fact that academia are moving away from flipcharts and black boards in class to learners bringing their own laptops and being able to actively engage online during classes.

It is also mentioned by one participant that it is necessary to convince all academics to make use of e-learning. The participant who mentioned this point was of the opinion that a great amount of apathy exits amongst academics.

Participants further indicated that they would like to see the free use of internet for all learners.

7.4.2.2.2 Researcher's interpretation on question 2

The researcher agrees that e-learning will play an increasingly integrated role in future teaching and learning. It is however, of utmost importance that awareness of the use of e-learning is created by academics and that it is followed with thorough training. The researcher is in complete agreement with the statement that academics are apathetic towards matters that they perceive do not influence them, as pointed out earlier in this chapter. An eminent level of unwillingness to change work methods and move from comfort zones is observable.

It is interesting to note that the essence of the feedback on this question was merely focused on the expansion of e-learning to be used to a greater extent in future, with the focus of the discussion on technological features (iPads; smart phones, internet, etc). No mention has been made of the adaptation of teaching and learning strategies or the profile of the future learner. It is evident that academics are not informed about the possibilities of e-learning, and are particularly ignorant as to the

adaptation their teaching and learning strategies to accommodate the 21st century learner.

In chapter 5, section 5.2.2, it was indicated that projected learner enrolments of university learners between the ages of 25 and 34 will far outpace learner enrolments of 18 to 24 year olds over the next decade, due to changing employee and job requirements (NCES 2008:9). It is also expected that enrolments among learners of 35 years or older will increase (NCES 2008:9). Accommodating working adults requires that innovative instructional technologies are effectively brought into play to make university studies possible for this group of learners. It is necessary to keep in mind, however, that merely a change in modality is not enough to meet the learning needs of these learners (Amirault & Visser 2009:70). Merely creating the path to the classroom (in this case, the virtual classroom), is not sufficient, a classroom environment that is attractive to learners to choose time and time again is equally important (Amirault & Visser 2009:70). This is where the need for an elearning skill of online learning facilitators comes into play. Not to simply provide an online course through making use of the basic applications (making study material available, post announcements, notes or power point presentations), but to know how to actively and effectively facilitate a course online in an interesting an appealing manner to keep the learning experience of the learner valuable and engaging.

7.4.2.3 Question 3: What should be done to ensure that e-learning is able to fulfil this role?

The purpose of this question was to determine participants' views on what should be done for e-learning is to fulfil its role.

7.4.2.3.1 Summary of feedback on question 3

The following were mentioned by participants as essential for e-learning to be successfully implemented:

• A notebook/laptop should be issued to each lecturer.

- The electronic infrastructure in offices should be upgraded; some academics have to use outdated technology.
- The electronic infrastructure in lecture halls should be upgraded; a significant number of lecture halls do not have a data projector installed and none of the lecture halls provides network connectivity.
- Institutional management should move away from the surplus approach and appropriate funds for the upgrade of infrastructure and the provision of free internet access to all learners.
- Management should give the opportunity to experiment with technology in a pilot study (eg if the lecturer and learners both are issued with iPads, it can be explored how to apply this to their learning).
- Learners should indicate how they prefer to use e-learning; this suggestion was opposed by another colleague, indicating that lecturers should not be prescribed by learners on the teaching methods to be used and that the lecturer, as subject expert, should remain the authority in the class.
- Buy-in from institutional management into the uses of e-learning; a number of participants are of the opinion that institutional management is not supporting elearning as yet and do not make provision for the most recent updated hardware and software (one of the participants disagrees with this opinion).
- There is a need for e-learning training, but also for information sessions on the possibilities of e-learning.

7.4.2.3.2 Researcher's interpretation on question 3

The researcher is in complete agreement with the need for upgraded electronic equipment in offices, as well as data projectors and internet connectivity in all lecture halls. The pilot study also seems like a good idea in principle, if used as best practice example to those academics who lack e-readiness and should still be convinced of its use. Financial viability will also be a concern. However, if the NWU is in support of progressive teaching and learning (as is clear from the ITEA programme), and wants to be a world leader and in the forefront with innovation (as indicated in the university's vision). It becomes imperative to supply its staff with the means to accomplish these goals.

The fact that the suggestion of involving learners to indicate their needs and preferences pertaining to e-learning was opposed emphasises unwillingness to compromise and to adjust teaching strategies. To a great extent academics still cling to the behavioural, objectivist approach (teacher-centred teaching and learning) and judging from the feedback obtained from the questionnaire and focus group, a vast number of academics are not willing (or perhaps ready) to embrace the constructivist approach (learner-centred teaching and learning).

A further matter of concern is that participants are of the opinion that institutional management has not yet bought into the use of e-learning and is supporting old software programmes. In an interview with the HR director of the NWU it became evident that the use of technology in teaching and learning (and for administrative matters) is undoubtedly supported by institutional management (De Wit 2010). Management's support is also demonstrated through their support and promotion of the e-platform's training sessions and the introduction of the ITEA programme. It is however, perceived by participants that management is not committed to e-learning, which is something that management may consider dealing with in order to create a positive spin-off. This possible misconception stresses the importance of better information and communication regarding the use and role of e-learning.

Lastly, participants' expression of the need for training (these comments were in particular from the older participants) is encouraging. A willingness to undergo elearning training is definitely a positive aspect that should be supported. Participants have however also expressed the need for more information and awareness of elearning, indicating that Academic Support Services will need to increase and improve their marketing campaigns.

7.4.2.4 Question 4: What are your perceptions pertaining to the expectations of learners in terms of e-learning?

The question was posed to determine participants' perceptions on the expectations of learners in terms of e-learning. The purpose of the question was further to determine whether participants were aware of their learners' expectations.

7.4.2.4.1 Summary of feedback on question 4

The underpreparedness (low academic quality), earlier mentioned in the discussion, was emphasised again. Participants are of the opinion that learners are not interested in class participation and want to be spoon-fed. Learners want to be entertained and want all of their senses stimulated for this purpose. It appears that learners do not enjoy the classes and simply comply with requirements as it is expected of them and because they want to pass a subject/course. In the discussion of this question it was evident that participants was unsure about how to respond to the question and the discussion did not flow as with the other questions. Participants eventually confessed that they are actually not sure about what their learners' expectations were.

7.4.2.4.2 Researcher's interpretation on question 4

The underpreparedness of learners has already been dealt with in section 7.4.2.1.2 and is accepted as a current reality. The belief that learners are not interested in class participation and want to be spoon-fed may be true to some extent. However, one has to keep in mind that teaching style and strategy may also play a role in learners' indifference, as indicated in section 6.4.1.6.2. It may be that learners are merely bored and are therefore not enjoying classes. The drawback is that in such cases learners probably also do not listen and/or remember what has been said or done in class. Thus, no learning has taken place. Therefore the belief that they want to be entertained and want all of their senses stimulated for this purpose may be true. As already identified in chapter 5, section 5.2.1.1, and earlier in this chapter, learners belonging to Generation Y learn differently from other generations. They want learning to be fun and relaxed and the traditional approach does not appeal to them (cf Naidoo 2005). Learners belonging to Generation Y are inquisitive, skilful at multi-tasking, think at a high speed and are fervently broad-minded in terms of diversity (Naidoo 2005; cf O'Neill 2010: 8). They are also an astonishingly innovative generation and will seek reinforcement and constant feedback on a regular basis (Naidoo 2005).

What was most evident during this discussion was that academics were not aware of their learners' characteristics and preferences. A logical deduction to be made is that it is impossible to cater for learners' needs if one is not aware of these needs. According to the researcher this, to a great extent, explains learners' indifference and apathy in class. If a stimulation of senses is what they need to be interested in their subjects and classes, it is necessary to trigger their interest and keep their attention. Once an interest and understanding of a subject has been developed the focus can shift to teaching higher order skills. However, as long as academics are persistently lecturing in a manner that is most preferred to them (usually the traditional objectivist teaching and learning approach), and not using strategies that capture learners' attention, learners will remain indifferent. The constructivist approach, mentioned in section 7.4.2.1.2, remains one of the key factors to addressing and motivating the 21st century learner.

7.4.2.5 Question 5: How should the role of academics change to accommodate learners' expectations?

The purpose of this question was to determine participants' perceptions on how the role of academics should change to accommodate learners' expectations. Feedback to the questions would also reveal whether participants know learners' expectations and what is expected of them in this regard.

7.4.2.5.1 Summary of feedback on question 5

A number of the participants indicated that it is difficult to answer the question as they are not aware of learners' expectations. It is further mentioned that Academic Support Services should provide e-learning training to academics, including a refresher course each year. Lecturers need continuous support from Academic Support Services in terms of e-learning. They state that there is no exposure to elearning wider than eFundi. One of the participants indicated that he was unaware of the fact that e-learning is broader than eFundi.

A number of the participants are of the opinion that eFundi is not user friendly. One participant indicates that eFundi should be an add-on, that it takes too much time

and creates an additional workload. Other participants supported it and added that they have full schedules with lecturing duties, research pressure and administration, and therefore do not get time to explore e-learning as well. It is also indicated that the use of e-learning should not be duplication, but it should play an additional role.

One of the participants mentioned that today's academics, belonging to the older generations, were raised with the perception that children should be seen and not heard. However, in this time and age they are confronted with a new generation of children that question things and they (as academics) do not necessarily have the answers.

The final word came from the participant who to the greatest extent resists elearning, indicating that he pays someone to post messages and power point presentations on eFundi.

7.4.2.5.2 Researcher's interpretation on question 5

The need for e-learning training is once again confirmed, emphasising the importance thereof. Also considering the fact that one of the participants was not aware of the fact that e-learning is broader than eFundi, and that others realised that they do not know much about e-learning other than eFundi, it is necessary that training, as well as awareness and wider exposure to e-learning is provided.

The fact that participants indicated that they need support from Academic Support Services in terms of e-learning, indicates that they are not aware of the support this department is already providing. Academic Support Services is advertising weekly training sessions on a daily basis on the NWU's intranet. They also can be contacted at a helpdesk throughout the day and an email address is available for after hours enquiries, although e-mails will only be responded to the next working day. What is evident though is that participants are not aware of these services and support available to them.

Comments regarding the lack of ease of use of eFundi demonstrate the extent of the lack of skill. Compared to other LMSs the researcher found eFundi to be significantly

simpler and more user friendly to use than LMSs such as Blackboard and Moodle for example. The researcher is of opinion that the ease of use of eFundi is so good and the navigation system so clear and simple that it can be used without any training.

Comments pertaining to additional workload again indicated an unwillingness to make use of technology in teaching and learning. It also indicates ignorance as the use of e-learning does not necessarily has to mean additional workload. On the other hand, line managers may also benefit by it to take cognisance of employees' comments pertaining to workload and research. The more emphasis is placed on are overloaded with administrative research and the more employees responsibilities, the less time and will is left for the use of e-learning, which is in fact directly related to a university's core business - education. Knowles, Holton III and Swanson (2005:10) define education as an action taken on and/or initiated to result in changes in the knowledge, skills and attitudes of individuals, groups or communities. One can assume that for these changes to be accomplished, learning should take place. E-learning is one of the methods the educator, as change agent (Knowles et al 2005:10), can apply to accomplish these changes. This does not mean that research is not significantly important, but its promotion should not harm teaching and learning practices, including the use of e-learning.

It was also indicated that the use of e-learning should not be duplication, but it should play an additional role. The comment that e-learning should play an additional role was made before. However, it was now indicated that e-learning should not be used as duplication. This is ironic since academics themselves use the e-platform to duplicate. As indicated in chapter 3 and earlier in this chapter, the e-platform is mostly used for the distribution of notes and power point presentations and only 10,5% of academics who completed the questionnaire make use of the forum, which is an interactive tool. No other interactive tools are used and none of the respondents indicated any manner in which they use the e-platform to develop higher order skills. Again this can be related to a lack of skill and ignorance about the possibilities of e-learning.

One of the participants mentioned that today's academics, belonging to the older generations (Generation X and the Baby Boomers), were raised with the perception

that children should be seen and not heard. However, in this time and age they are confronted with a new generation of children that question things and they (as academics) do not necessarily have answers. This is again ironic, as in a previous question participants indicated that this generation of learners are not interested in participating in class and merely want to know what to study to pass. In section 7.4.2.4.2 the importance of knowing the learners' needs and preferences and adjusting teaching strategies accordingly to maximise learning, has already been emphasised.

The remark from the participant indicating that he pays someone to post messages and power point presentations on eFundi is priceless. This remark in itself best demonstrates the lack of e-readiness. The same perceptions and misconceptions held by respondents of the questionnaire surfaced during the focus group discussion. In addition to a lack of technical skill, a resistance to e-learning could also be observed with an age increase. Furthermore, misconceptions and ignorance pertaining to the use and purpose of e-learning is widespread. It is apparent that a fear that technology will take over and to a great extent replace the lecturer is still withholding a significant number of academics from exploring the possibilities of elearning. This misconception also illustrates that employees are not aware of the human role in e-learning, the requirement of an online learning facilitator to promote communication and motivation online. Thus it illustrates the need for comprehensive training, including awareness of the use, purpose and philosophy of e-learning. Furthermore, respondents who revealed e-readiness in terms of willingness to incorporate technology in teaching and learning dismally failed to use technology as an interactive learning tool. The manner in which the e-platform is currently used by the vast majority of respondents, as a distribution mechanism, does not promote effective learning.

It is further evident that a significant number of academics stubbornly want to stick to traditionally known methods. The resistance to threatening of a comfort zone is supported by literature and needs to be addressed by institutional managers and line managers alike. Institutional managers are responsible to draft an institutional policy to ensure standardisation and implementation. Line managers have a significant role to play in performance appraisals, training identification, providing training and
development opportunities and supporting employees through effective motivation. Overall line managers should take more responsibility for the learning and development of their subordinates.

An interesting paradox can be observed. A 49% rate of *early adopters*, compared to a 39% of respondents who resist e-readiness assessment during performance appraisals, with a further 17% of respondents indecisive or sceptical, which brings the total number of respondents, not unconditionally in favour of e-readiness assessment to 56%. The 44% of respondents that responded positively to the inclusion of e-readiness assessment in the performance appraisals of academics did not necessarily do so without reservation. In many instances it was indicated that such an assessment should not be compulsory. Therefore, although 49% of respondents indicated aspects relating to the early adopter category of the technology adoption cycle, it appears that this selection was to a great extent driven by a performance orientation and not necessarily by an eagerness to engage with technology in teaching and e-learning. This notion is supported by Dweck (1990) and Dweck and Leggett (1988), guoted in Whitmore (2000:8) by explaining the unspoken expectations that people hold about self-evaluation when asked to complete difficult tasks. Two focal points are eminent: a focus that is positive, described as the "mastery-orientation"; and a focus that can often be self-defeating, described as the "performance-orientation". Employees who embrace the mastery orientation enthusiastically look for challenges, as the complex challenges are perceived by them as the best opportunities for learning (as with the profile of an early adopter and *innovator* on the technology adoption cycle). On the other hand, employees who embrace the performance orientation portray a preference to problems that support their opinion of themselves as primarily intelligent. When performance takes precedence these employees tend to implement strategies that limit the possibility of being unsuccessful, even if that destines them to fairly monotonous and routine pursuits (Dweck 1990; Dweck & Leggett 1988 quoted in Whitmore 2000:8). Thus the high performance culture of academics may be the cause of preferring the old trusted ways of traditional teaching and learning to new innovative methods of teaching and learning.



7.5 SUMMARY AND DEDUCTIONS FROM EMPIRICAL RESULTS

The empirical results obtained through the feedback of the questionnaire and the focus group discussion shed valuable light on the e-readiness of academics. It became evident that a lack of e-readiness is to a great extent evident amongst participants in the study. It is further apparent that a number of misconceptions pertaining to e-learning, its use and purpose remain, in many instances leading to resistance to the use of e-learning. Typical misconceptions include the notion that e-learning is to be used for part time learners and distance learning learners and cannot be used for postgraduate learners or small groups of learners.

Fear of the changed job demand is evident, especially amongst, but not exclusive to, older academics. The fear of change also became eminent in the constant reference to contact education to not being replaced by e-learning, during the questionnaire results and focus group discussion. This feedback supports the notion that academics do not realise the role, purpose and importance of e-learning in residential education. Thus, a fear based on misconceptions withholds academics from exploring the richness and pedagogical value of e-learning.

Feedback and results furthermore reveal that academics are not informed about the needs and preferences of their learners. The majority of learners at the NWU belong to Generation Y, known for their high interest in and preference to the use of technology. This generation thinks differently, act differently, and grew up in different circumstances (the technological age) from previous generations and therefore the traditional classroom approach to teaching and learning does not appeal to them. Educational experts such as Eble (1988:9), Ramsden (2003:xii), and Milliken and Barnes (2002:225–226) underline the importance of teachers adapting to learners' needs and preferences. This adaptation includes the adjusting of teaching and learning strategies. It became evident that a significant number of academics still embrace the objectivist approach to teaching and learning, which is driven by "talk-and-chalk" and strongly depends on textbooks for the structure of the course (Hanley 1994:3; cf Kinchin 2004:302). According to objectivists there is a fixed world of knowledge that the learner must come to know and educators serve as channels through which their thoughts and meanings are transferred to the passive learner

(Hanley 1994: 3). Whereas the objectivist approach gives emphasis to observable, external behaviours and therefore, steers clear of reference to meaning, representation and thought, constructivism follows a more cognitive approach (Gergen sa). In the constructivism approach educators are coordinators, facilitators, resource advisors, tutors or coaches (Gergen sa).

Within the constructivist approach the lecturer will introduce new ideas or cultural tools where necessary and provide the support and guidance for learners to make sense of these for themselves (Murphy sa). The lecturer will also identify the ways in which the instructional activities are being interpreted to inform further action (Murphy sa). Results obtained from the questionnaire and focus group discussion (discussed in section 7.4), indicate that to a great extent academics are struggling to, or resisting to, follow a constructivist approach to teaching. It appears that a significant number of academics are unwilling to change their teaching and learning strategies to suit their learners' needs.

It became clear that e-learning should become a broader management issue. Institutional management should take responsibility for formal e-learning policy. In the absence of policy, e-learning will remain optional and learners would thus be withheld of optimal learning possibilities and experiences. The university may also lose its competitive edge, as well as its learners to other HEIs.

Further, line managers should realise and embrace their responsibility towards employee development. Not only should training needs be identified, but time and opportunity for training should also be granted. One of the most significant roles of line managers, motivation, should take precedence in this endeavour of a changed job demand and in particular if e-readiness assessment is introduced to the performance appraisals of academics. It is imperative that line managers are able to act as skilled change agents and motivate staff. It may be necessary to provide particular training to line managers in this regard. Best practice pertaining to talent management, career management and employee retention showed that incentives such as flexitime, recognition, support with resources, training and mentoring and coaching support not only the retention of talent and their development, but also their performance and sense of belonging (Corporate Leadership Council 2003:3). The use and importance of both monetary and non-monetary incentives should also not be underestimated and the value of goal-setting should be understood.

It is clear that certain key human factors play a role in an academic's e-readiness. Key human factors considered for this study are the personal profile patterns of employees, namely employees' DISC factors, employees' preferred learning styles, and employees' position/category on the technology adoption cycle, indicating employees' pace and style of technology adoption. Certain trends and similarities could be observed between these human factors. For example, a particular learning style revealed by an employee, in most instances typically portrayed a typical DISC factor and a specific category of technology adoption pace and style. The following particular matches became evident, indicated in table 7.1 below:

Table 7.1: Matches between personal profile patterns, learning styles andpace and style of technology adoption

Personal profile pattern (DISC factor)	Learning style	Technology adoption (pace and style)
Dominance	Activist	Innovator
Influence	Reflector	Late majority
Steadiness	Theorist	Early majority
Compliance	Pragmatist	Early adopter

The last two matches indicated in table 7.1, the *early majority-steadiness-theorist* and the *early adopter-pragmatist-compliance* matches represent the e-profiles of the majority of respondents, 79% (also see diagram 7.10). These e-profiles can thus be related to employees' level of comfortableness with the use of e-learning, as indicated in table 7.2 below (also see diagram 7.21).

Table 7.2: Matches between personal profile patterns, learning styles, pace and style of technology adoption, and e-learning perceptions when trained and first being used

Personal profile	Learning	Technology	Perceptions towards e-learning
pattern	style	adoption	after trained and used for the first
(DISC factor)		(pace and style)	time
Dominance	Activist	Innovator	Very eager to teach online
Influence	Reflector	Late majority	Still did not want to teach online
Steadiness	Theorist	Early majority	Comfortable, but will never be first choice
Compliance	Pragmatist	Early adopter	Looked forward, once mastered the skill

Table 7.3 below indicates these profiles, placed in order of the perceived highest level of e-readiness to the lowest level of e-readiness.

Table	7.3:	E-read	liness	levels

E-readiness	Personal	Learning	Technology	Perceptions towards e-	Percentage
levels	profile	style	adoption	learning after trained and	of
(highest to	pattern		(pace and	used for the first time	respondents
lowest)	(DISC factor)		style)		%
Very high	Dominance	Activist	Innovator	Very eager to teach online	13
Relatively high	Compliance	Pragmatist	Early adopter	Looked forward, once mastered	46
				the skill	
Moderate	Steadiness	Theorist	Early majority	Comfortable, but will never be	33
				first choice	
Low	Influence	Reflector	Late majority	Still did not want to teach online	8

Respondents to the questionnaire portrayed a relatively high (46%) or moderate (33%) level of e-readiness. If these results are considered to be generally representative of the academics of the NWU, Potchefstroom campus, it is clear that successfully training and developing employees portraying the *early adopter-compliance-pragmatist* and the *early majority-steadiness-theorist* pertaining to the use and purpose e-learning, the bulk of the academic workforce at the NWU will be well-skilled and able to implement e-learning optimally to support world class learner learning. If the 13% of respondents who portray a dominance-activist-innovator e-profile is considered to the bulk of academics portraying the abovementioned two e-

profiles (79%), it brings the total of e-ready academics to 92%. Undoubtedly this will make a difference in the effectiveness of use of e-learning.

Table 7.4 indicates the matches between personal profile patterns, learning styles, pace and style of technology adoption, e-learning perceptions when trained and first being used and type of motivation.

Table 7.4: Matches between personal profile patterns, learning styles, pace and style of technology adoption, e-learning perceptions when trained and first being used and type of motivation

E-	Personal	Learning	Technology	Perceptions towards e-	Motivation	Percentage
readiness	profile	style	adoption	learning after trained		of
levels	pattern		(pace and	and used for the first		respondents
(highest to	(DISC		style)	time		%
lowest)	factor)					
Very high	Dominance	Activist	Innovator	Very eager to teach online	Intrinsic	13
Relatively	Compliance	Pragmatist	Early adopter	Looked forward, once	Mostly	46
high				mastered the skill	extrinsic, but	
					also intrinsic	
Moderate	Steadiness	Theorist	Early majority	Comfortable, but will never	Extrinsic	33
				be first choice		
Low	Influence	Reflector	Late majority	Still did not want to teach	Extrinsic	8
				online		

Employees belonging to the *innovator-activist-dominance* profile will be intrinsically motivated. This category of employees will be intrinsically motivated as they perform an activity for its own sake and enjoy performing it (Gagné & Deci 2005:331; Vansteenkiste et al 2007:253). Intrinsic motivation refers to a condition in which an individual feels interest, pleasure and enthusiasm by taking on task-related activities (Gagné & Deci 2005:331; Vansteenkiste et al 2007:253). Kyan & Deci 2000:56). Employees belonging to this profile will not portray a lack of e-readiness and will embrace the challenge of a new job demand.

Another e-profile category that is also intrinsically motivated (to some extent), but can also rely on extrinsic motivation, is the *early adopter-pragmatist-compliance* category, representing 46% of the respondents of the questionnaire. Literature has

identified *early adopters* to be extrinsically motivated, but since they are also regarded as strategic thinkers (Zemsky & Massey 2004:9–10), some of them may be intrinsically motivated. It can be argued that *pragmatists* will be extrinsically motivated as the technology adoption will not be a natural process of grabbing opportunities to them as with the *innovators* or *activists*, as they will first determine the practical value and use of what they are being taught and will require a link between the training and the end-result required of them (Honey & Mumford 1982:25). Further, the *compliance* DISC factor indicates that, an employee portraying this e-profile prefers predictable and consistent outcomes (Thomas International sa) and may therefore need encouragement in the form of incentives or rewards to adapt to the new job demand.

The other e-profile, represented by a significant number of respondents (33%), the *early majority-theorist-steadiness* profile, is mostly extrinsically motivated, implying that external rewards and incentives may be necessary to encourage these employees towards goal accomplishment pertaining to the new job demand. This is due to the fact that the *late majority* only adopt when half of the population has already done so and they dislike the disruptions of new technologies (Zemsky & Massey 2004:10). Therefore, they will need external motivation to adapt to the changed job demand.

If these profiles are considered to be generally representative of the academics of the NWU, Potchefstroom campus, it implies that the categories representing the majority of academics will mostly need extrinsic motivation. If the university lacks the necessary funds to provide financial incentives, they will have to think creatively of other incentives to motivate employees towards goal accomplishment. This also places a significant responsibility on the line manager or support in this respect, as employees will need support and encouragement throughout the learning curve.

It is important to keep in mind that other variations of e-profiles can emerge with ereadiness assessment and that these profiles are used as examples to draft a strategy, as they surfaced as most eminent during the empirical research.

286

Before the training and development interventions can take place it is necessary to draft a personal development plan for each academic, based on the employee's e-readiness assessment during a performance appraisal. The next section will discuss the implementation of the e-readiness assessment and the subsequent training and development by means of a framework, developed for this purpose.

7.6 FRAMEWORK FOR THE E-READINESS ASSESSMENT OF ACADEMICS

The study argues that due to the nature of technological development, the increasing use of technology in teaching and learning and the profile of the 21st century learner (Generation Y), academics must understand and embrace the importance and role of e-learning and no longer have the option to resist e-learning. For e-learning to become an integral part of an academic's job requirements, it necessitates their e-readiness. Particularly also in the light of the significant role that the online learning facilitator play in e-learning, not only with regard to facilitation, subject content and technical assistance, but also with regard to learner motivation, communication and encouragement.

The study further argues that to determine an academic's e-readiness, it should be assessed during a performance appraisal. As the performance appraisal process consists of not only an evaluative component, but also a development component, the e-readiness assessment should be followed with a personal development plan, indicating uniquely structured training and development opportunities. The study indicated that key human factors (pace and style of technology adoption, preferred learning style and personal work profile patterns) play a role in the e-readiness assessment and therefore argues that these human factors should be included in the e-readiness assessment of academics. An academic's key human factors will be indicative of his/her e-readiness indicators (as explained in chapter 5) and the training and development approach to be followed. An e-readiness construct was constructed to determine the minimum acceptable level of e-readiness for an academic (by means of particular indicators) to effectively be able to use e-learning as pedagogical tool.

Once the key human factors are assessed, and an e-profile is drafted, the employees' e-readiness indicators will be apparent and the training and development interventions can be structured.

Tables 7.5, 7.6 and 7.7 indicate the e-readiness framework, based on the outcome of an employee's e-readiness assessment. Potential matches that may emerge after an academic's key human factors (technology adoption, preferred learning style and personal profile pattern) have been assessed are indicated, as well as the unique training and development approach that should be structured for an employee. The table cells that are not filled are highly unlikely matches:

- Innovator/Influence the *innovator* is only a logical match with the *dominance* factor due to the shared characteristics of exploring new ideas, being driven by internal motivation and accepting challenges (Zemsky & Massey 2004:9; Thomas International sa).
- Innovator/Steadiness the *innovator* thrives on challenges and exploring new ideas, whereas the *steadiness* factor portrays a preference for a stable structured environment (Thomas International sa).
- Innovator/Compliance the *innovator* prefers challenges and new experiences, as opposed to the *compliance* factor that portrays a preference for rules, procedures, structure, predictability and consistency (Zemsky & Massey 2004:9; Thomas International sa).
- Early adopter/Steadiness an *early adopter* is an opinion leader and decisionmaker, matching opportunities with strategic goals, whereas the *steadiness* factor portrays a preference for a stable structured environment and the status quo (Zemsky & Massey 2004:9; Thomas International sa).
- Early majority/Dominance the *early majority* belongs to the mass market and will be relatively slow to adapt to technological change, whereas the *dominance* factor are part of a minority of competitive, visionary leaders (Zemsky & Massey 2004:10; Thomas International sa).
- Late majority/Dominance the *late majority* is slow to adapt to technological change, whereas the *dominance* factor are part of a minority of competitive,

visionary leaders who will adapt immediately to technology (Zemsky & Massey 2004:9; Thomas International sa).

- Diehard/Dominance these two categories are exact opposites, whereas the *diehard* will never adapt to technology and the *dominance* factor will adapt immediately to technology (Zemsky & Massey 2004:10; Thomas International sa).
- Diehard/Influence the *diehard* will never adapt to technology, but as the *influence* factor portrays an optimistic and outgoing nature, focused on networking and working with others it is likely that the *influence* factor will at some point adapt to technology (Zemsky & Massey 2004:10; Thomas International sa).
- Innovator/Pragmatist an *innovator* takes challenges, whereas a *pragmatist* wants to see the practical value before engaging into a learning activity (Zemsky & Massey 2004:9; Honey & Mumford 1982:28).
- Innovator/Theorist an *innovator* takes challenges and new opportunities, whereas a *theorist* carefully thinks things through before engaging in a learning activity (Zemsky & Massey 2004:9; Honey & Mumford 1982:27).
- Innovator/Reflector An *innovator* takes challenges and new opportunities, whereas a *reflector* spends a significant amount of time to think intensively about the activities and concepts provided to them online (Zemsky & Massey 2004:9; Honey & Mumford 1982:26).
- Early adopter/Theorist an *early adopter* is an opinion leader and decisionmaker, matching opportunities with strategic goals, whereas the *theorist* will first think things through before engaging in an activity (Zemsky & Massey 2004:9; Honey & Mumford 1982:27).
- Early adopter/Reflector an *early adopter* acts as a decision-maker, matching opportunities with strategic goals, whereas the *reflector* will take a significant amount of time to think intensively about the activities and concepts (Zemsky & Massey 2004:9; Honey & Mumford 1982:26).
- Early majority/activist whereas the *early majority* adapts to technology relatively slowly, the *activist* will adapt to technology immediately (Zemsky & Massey 2004:10; Honey & Mumford 1982:25).

- Late majority/activist whereas the *late majority* adapts to technology slowly and only when half the population has already done so, the *activist* will adapt to technology immediately (Zemsky & Massey 2004:10; Honey & Mumford 1982:25).
- Diehard/Activist these two categories are exact opposites, whereas the *diehard* will never adapt to technology and the *dominance* factor will adapt immediately to technology (Zemsky & Massey 2004:10; Honey & Mumford 1982:25).
- Diehard/ragmatist whereas the *diehard* will never adapt to technology, the *pragmatist* will adapt to technology as soon as its practical value can be determined (Zemsky & Massey 2004:10; Honey & Mumford 1982:28).
- Dominance/ragmatist the *dominance* factor is a competitive, visionary leader, taking on challenges without considering the risks involved, whereas the *pragmatist* wants to see the practical value of what being taught before engaging in the learning activity (Thomas International sa; Honey & Mumford 1982:28).
- Dominance/theorist the *dominance* factor thrives on challenges and will immediately engage in a new opportunity, as opposed to the *theorist* that will think it through before engaging in the activity (Thomas International sa; Honey & Mumford 1982:27).
- Dominance/reflector the *dominance* factor thrives on challenges and will immediately engage in a new opportunity, wheras the *reflector* will take a significant amount of time to think about the learning activity before engaging in it (Thomas International sa; Honey & Mumford 1982:26).
- Influence/activist the *influence* factor is people oriented, whereas the *activist* is goal-oriented (Thomas International sa; Honey & Mumford 1982:25).
- Influence/pragmatist the *influence* factor is concerned with people, as opposed to the pragmatist that is practical and task-focused (Thomas International sa; Honey & Mumford 1982:28).
- Steadiness/activist the steadiness actor prefers a stable environment, whereas the activist functions best in an unstructured environment where being confronted with new challenges (Thomas International sa; Honey & Mumford 1982:25).

—vtd List of research project topics and materials 290 Compliance/activist – although both these two categories are task-focused, the *compliance* factor is comfortable with rules, procedures, structure, predictability and consistency, as opposed to the *activist* that functions optimally without the constraints of rules and procedures and do not want predictability (Thomas International sa; Honey & Mumford 1982:25).

The rest of the key human factor matches, indicated in the framework (tables 7.5, 7.6 and 7.7) below, are likely matches that may emerge from an e-readiness assessment of academics.

Personal work profile pattern		Dominance	Influence	Steadiness	Compliance
(DISC profile)					
	Innovator	Unstructured; freedom to explore; internal motivation; visual aids; fast pace			
E C H N	Early Adopter	Structured/unstructured; freedom to explore; internal motivation; visual aids; instructor that demonstrates; relatively fast pace	Structured; external motivation; visual aids; instructor that demonstrates; instructor that gives verbal instructions; paper-based notes; opportunity for socialisation; relatively fast pace		Structured, step-by-step approach; external motivation; visual aids; instructor that demonstrates; instructor that gives verbal instructions; relatively fast pace
O L O G	Early majority		Structured; external motivation; visual aids; instructor that demonstrates; instructor that gives verbal instructions; paper-based notes; opportunity for socialisation	Structured, step-by-step approach; external motivation; visual aids; instructor that demonstrates; instructor that gives verbal instructions; paper-based notes	Structured, step-by-step approach; external motivation; visual aids; instructor that demonstrates; instructor that gives verbal instructions; paper-based notes
Y A D O P	Late majority		Structured; external motivation; visual aids; instructor that demonstrates; instructor that gives verbal instructions; paper-based notes; opportunity for socialisation; enough opportunity to reflect and ask questions	Structured, step-by-step approach; external motivation; visual aids; instructor that demonstrates; instructor that gives verbal instructions; enough opportunity to reflect and ask questions	Structured, step-by-step approach; external motivation; visual aids; instructor that demonstrates; instructor that gives verbal instructions; enough opportunity to reflect and ask questions
T I O N	Diehard/Laggard			Structured; external motivation; visual aids; instructor that demonstrates; instructor that gives verbal instructions; paper-based notes; enough opportunity to reflect and ask questions	Structured; external motivation; visual aids; instructor that demonstrates; instructor that gives verbal instructions; paper-based notes; enough opportunity to reflect and ask questions

Table 7.5: The technology adoption – personal work profile pattern match

Learning style		Activist	Pragmatist	Theorist	Reflector
т	Innovator	Unstructured; trial and error; freedom to explore; internal motivation; visual aids; fast pace			
E C H	Early Adopter	Structured/unstructured; freedom to explore; internal motivation; visual aids; instructor that demonstrates; match strategically; relatively fast pace	Structured; match strategically; holistic picture; future benefits; step-by-step approach; external motivation; instructor that demonstrates; visual aids; relatively fast pace		
N O L O G	Early majority		Structured; match strategically; picture; future benefits; step-by-step approach; external motivation; instructor that demonstrates; visual aids ; paper-based notes	Structured; external motivation; step-by-step approach; external motivation; instructor that demonstrates; instructor that gives verbal instructions; visual aids; paper-based notes	Structured; unhurried process; ample opportunity to ask questions and review; opportunity to reflect on progress; external motivation; instructor that demonstrates; instructor that gives verbal instructions; visual aids; paper-based notes
G Y A D O P T I O N	Late majority		Structured; holistic picture; step-by-step approach; external motivation; instructor that gives verbal instructions; instructor that demonstrates; visual aids	Structured; step-by-step approach; external motivation; instructor that demonstrates; instructor that gives verbal instructions; visual aids; paper-based notes	Structured; step-by-step approach; ample opportunity to ask questions and review; opportunity to reflect on progress; external motivation; instructor that demonstrates; instructor that gives verbal instructions; visual aids; paper-based notes
	Diehard/laggard			Structured; step-by-step approach; external motivation; external motivation; instructor that demonstrates; instructor that gives verbal instructions; visual aids; paper-based notes	Structured; unhurried process; ample opportunity to ask questions and review; opportunity to reflect on progress; external motivation; instructor that demonstrates; instructor that gives verbal instructions; visual aids; paper-based notes

Table 7.6: The technology adoption – preferred learning style match

Table 7.7: Personal work profile pattern – preferred learning style match

Learning style		Activist	Pragmatist	Theorist	Reflector
	Dominance	Unstructured; trial and error; freedom to explore; internal motivation; visual aids; fast pace			
	Influence			Structured; step-by-step approach; external motivation; visual aids; instructor that demonstrates; instructor that gives verbal instructions; paper-based notes; opportunity for socialisation; enough opportunity to ask	Structured; step-by-step approach; external motivation; visual aids; instructor that demonstrates; instructor that gives verbal instructions; paper-based notes; opportunity for socialisation; enough opportunity to reflect and
DISC	Steadiness		Structured; match strategically; holistic picture; practical step-by- step approach; external motivation; visual aids; instructor that demonstrates; instructor that gives verbal instructions; paper- based notes	questions Structured; step-by-step approach; external motivation; visual aids; instructor that demonstrates; instructor that gives verbal instructions; paper-based notes; enough opportunity to ask questions	ask questions Structured; step-by-step approach; external motivation; visual aids; instructor that demonstrates; instructor that gives verbal instructions; paper-based notes; enough opportunity to reflect and ask questions
	Compliance		Structured; match strategically; holistic picture; future benefit; practical step-by-step approach; external motivation; visual aids; instructor that demonstrates; instructor that gives verbal instructions; paper-based notes	Structured; step-by-step approach; external motivation; visual aids; instructor that demonstrates; instructor that gives verbal instructions; paper-based notes; enough opportunity to ask questions	Structured; step-by-step approach; external motivation; visual aids; instructor that demonstrates; instructor that gives verbal instructions; paper-based notes; enough opportunity to reflect and ask questions

Once an employee has been plotted within the abovementioned framework (provided by the three tables), the employee's e-profile will be known and uniquely structured training and development interventions for each academic can be planned and implemented. The framework outlines to line managers, human resource managers and Academic Support Services, the training and development approach that should be followed with each academic and the preferred nature and structure of the training and development intervention, that is structured or unstructured; step-by-step or freedom; visual aids, paper-based notes, verbal instructions from an instructor or an instructor that demonstrates; internal or external motivation; and opportunity to ask questions and reflect or a relatively fast pace.

In addition to the abovementioned framework, the e-readiness construct, constructed in chapter 5, should also be taken into consideration, indicating all the indicators and specifications for e-readiness. The indicators and specifications indicated in the ereadiness construct pertain to the content of the training and development interventions and will give an indication of how much time should be spend on which indicators, given employees' e-profiles that will portray for which indicators they will have a natural inclination.

The planning of training and development interventions for academics that portray one of the two most prominent e-profiles, as obtained through the questionnaire results, can be approached as follow, based on the e-readiness framework:

• Early adopter-compliance-pragmatist e-profile

An academic with this e-profile would, according to the above framework have been plotted on table 7.5 as an *early adopter-compliance* match, which indicates a structured, step-by-step approach; external motivation; the use of visual aids during training, as well as an instructor that demonstrates and gives verbal instructions; and a relatively fast pace should be followed during training and development.

The employee would have been plotted as an *early adopter-pragmatist* on table 7.6, which indicates a structured step-by-step approach; training should be matched with strategic goals; a holistic picture should be provided; future benefits should be

underlined; external motivation should be provided; training should be given through an instructor that demonstrates, use visual aids and should be at a relatively fast pace.

On table 7.7 the employee would have been plotted as a *compliance-pragmatist*, indicating a structured and practical step-by-step approach; training should be matched with strategic goals; a holistic picture should be provided; the future benefit of the training should be explained; external motivation should be provided; training should be done through the use of visual aids, an instructor that demonstrates and that gives verbal instructions, as well as paper-based notes.

Therefore, the results (*early adopter-compliance; early adopter-pragmatist* and *compliance-pragmatist*) obtained from the three tables in the framework reveal an *early adopter-compliance-pragmatist* e-profile. This profile was portrayed by 46% of questionnaire respondents. An academic portraying this e-profile should therefore receive e-learning training and development in a structured manner with a practical step-by-step approach. A holistic picture of the role and place of e-learning in the university's teaching and learning framework and goals, as well as within the institutional plan and strategic goals and objectives should be provided. Employees should understand how the training will benefit them in future and why it is imperative to their teaching and learning skills. The person responsible for e-learning training should make use of visual tools such as podcasts and simulations, give verbal instructions, demonstrated the tools to employees and provide paper-based notes for referral when practicing. An employee portraying this e-profile will need to be externally motivated, therefore the line manager and the human resource manager should decide on the method of motivation that will be followed.

An employee that belongs to the *early adopter* category of technology adoption, as well as the *pragmatist* learning style preference of an employee portraying this profile, it can be assumed that an employee with this profile will adapt to e-learning relatively fast, once the skill is mastered. Therefore once the technical skill is obtained, training should focus on e-readiness indicators such as communication, motivation, compassion and counselling (the social nature of e-learning), as these indicators may need to be developed. A focused should also be placed on personal

attribute indicators such as innovation and creativity, commitment and time management. Due to the *compliance* DISC factor of this profile (favouring standard operating procedures and the status quo), attention should be given to the practical use, purpose and philosophy of e-learning.

• Early majority-steadiness-theorist e-profile

An academic with this e-profile would, according to the above framework have been plotted on table 7.5 as an *early majority-steadiness* match, which indicates a structured, step-by-step approach; external motivation; the use of visual aids during training, as well as an instructor that demonstrates and gives verbal instructions; and a relatively fast pace during training and development.

The employee would have been plotted as an *early majority-theorist* on table 7.6 and a *steadiness-theorist on* table 7.7, which both indicate a structured step-by-step approach; external motivation should be provided; training should be given through an instructor that demonstrates the use of e-learning tools, the use of visual aids and paper-based notes, and ample opportunity should be given to ask questions.

Therefore, the results (*early majority-steadiness; early majority-theorist* and *steadiness-theorist*) obtained from the three tables in the framework reveal an *early majority-steadiness-theorist* e-profile. This profile was portrayed by 33% of questionnaire respondents. An academic with this e-profile should therefore receive e-learning training and development in a structured manner with a practical step-by-step approach. The person responsible for e-learning training should make use of visual tools such as podcasts and simulations, give verbal instructions, demonstrated the tools to employees and provide paper-based notes for referral when practicing. An employee portraying this e-profile will need to be externally motivated, therefore the line manager and the human resource manager should decide on the method of motivation that will be followed.

Training and development for employees that portray the *early majority-steadinesstheorist* e-profile will differ from the *early adopter-compliance-pragmatist* e-profile: Firstly, information on strategic linkages and the practical value of the training will not be such a high priority for employees that portray the *early majority-steadiness-theorist* e-profile as for those portraying the *early adopter-compliance-pragmatist* e-profile. However, this information should ideally be provided to all employees that receive e-learning training, irrespective of their e-profiles. Secondly, employees that portray the *early majority-steadiness-theorist* e-profile will in all likelihood have a more natural ability for the social nature of e-learning and the personal attribute indicators of e-readiness will be more eminent, due to the *steadiness* DISC factor of the profile, that reflects good people skills and good team players (Thomas International sa). Thirdly, training and development for employees that portray the *early majority-steadiness-theorist* e-profile will follow a slower pace than training and development for employees portraying the *early adopter-compliance-pragmatist* e-profile for two reasons:

- An employee that is categorised in the *early majority* category will portray a bigger lack of e-readiness than an *early adopter* and will be less comfortable with change and new technologies, and will therefore need more technical guidance.
- Secondly, *theorists* will tend to ask more questions than *pragmatists* and will need time to think about activities and challenges posed to them during training.

Only 8% of questionnaire respondents portrayed the *late majority-influence-reflector* e-profile. Employees portraying this profile will receive similar training as the *early majority-steadiness-theorist* profile, but even more time must be provided to reflect on activities and challenges and opportunity for socialisation must also be provided. Due to the high *influence* DISC factor employees revealing this profile are likely to do well with the human side of e-learning, as employees with a high *influence* DISC factor are concerned about the manner in which people are dealt with and are typically charming, optimistic, and outgoing, and focused on networking, conversation, and working with others (Thomas International, sa), they will have a natural ability for the following indicators of the e-readiness construct: communication, motivation, compassion and counselling skills. As this category of technology adoption will resist e-learning as long as possible and be hard to

convince of its use, employees in this category should firstly be focused on ereadiness indicators such as the philosophy, use and purpose of e-learning, as well as the needs and preferences of Generation Y learners. A focus on the use and adaptation of teaching and learning strategies is also important. Further, training must be focused on the technical skill due to the *late majority* aspect of this profile, portraying a significant lack of e-readiness. Focus should be placed on training strategies and techniques that will make an employee belonging to this category comfortable, another indicator of the e-readiness construct.

Thirteen percent (13%) of questionnaire respondents portrayed the innovator*dominance-activist* profile. Employees portraying this e-profile will in all likelihood not need to be trained (as they would have explored the e-learning platform on their own, played around and managed to use the tools). Therefore they will in all probability not have a development need or subsequent need for training and development in this regard. However, if these employees did not master all the tools of the e-learning platform on their own, and from a guality control point of view, line managers can consider, together with the employee, to include e-learning training in their personal development plans. An employee with a high *dominance* DISC factor or an *innovator* technology adoption style is likely to do well with the technical skills, subject competency, comfortableness and willingness, as a high *dominance* factor and *innovators* are competitive, with high performance standards, and focused on achieving goals, solving problems, enjoy exploring new ideas and accept challenges (Thomas International sa; Zemsky & Massey 2004:9). The same can be said about an employee with an *activist* learning style that also prefers to deal with challenges (Honey & Mumford 1982). However an employee who is an *innovator*, prefer an activist learning style or portray a high dominance factor, does not necessarily have good interpersonal skills. Therefore, particular focus can be placed on e-readiness indicators such as learner motivation, communication and compassion.

7.7 CONCLUSION

This chapter outlined the research methodology followed in the study and indicated particular challenges experienced during data collection. Data were obtained through quantitative data collection with a questionnaire and through qualitative data collection with a focus group discussion. Interviews were held with particular specialists in fields relating to the study. The empirical results obtained shed valuable light on the e-readiness of academics. It became evident that a lack of ereadiness is to a great extent evident amongst academics. It is further apparent that a number of misconceptions pertaining to e-learning, its use and purpose remain in many instances, leading to resistance to the use of e-learning. Fear of the changed job demand is also evident, especially the fear of contact education being replaced by e-learning. This feedback supports the notion that academics do not realise the role, purpose and importance of e-learning in residential education.

Feedback and results furthermore reveal that academics are not informed about the needs and preferences of their learners. The majority of learners at the NWU belong to Generation Y, known for their high interest in and preference for the use of technology. Educationalist experts underlined the importance of teachers adapting to learners' needs and preferences. This adaptation includes the adjusting of teaching and learning strategies. It became evident that a significant number of academics still embrace the objectivist approach to teaching and learning, whereas the study proposes the use of a constructivist approach to teaching and learning.

It further became apparent that e-learning should become a broader management issue and that institutional management should take responsibility for formal elearning policy. Also, line managers should realise and embrace their responsibility towards employee development. Not only should training needs be identified, but time and opportunity for training should be granted. One of the most significant roles of line managers, motivation, should take precedence in this endeavour of a changed job demand and in particular if e-readiness assessment is introduced to the performance appraisals of academics. It is imperative that line managers are able to act as skilled change agents and motivate staff. It may be necessary to provide particular training to line managers in this regard.

It became clear that certain key human factors (employees' DISC factors, employees' preferred learning styles, and employees' position/category on the technology adoption cycle) play a role in an academic's e-readiness. Certain trends and similarities could be observed between these human factors. The following List of research project topics and materials

particular e-profile categories were revealed through questionnaire results: an *early adopter-compliance-pragmatist* profile, an *early majority-steadiness-theorist* profile, an *innovator-dominance-activist* profile and a *late majority-influence-reflector* profile.

Lastly, an e-readiness framework was drafted in this chapter. After their e-readiness assessment (during which particular key human factors are assessed), academics' results can be plotted on the framework, consisting of three tables. Based on their results, which will indicate a particular e-profile, academics' training and development interventions can be structured to suit their specific profiles.

The next chapter provides a summary of the study and outlines the key findings of the study. Particular recommendations pertaining to the e-readiness of academics, performance appraisals and the use of technology in teaching and e-learning are also made in this chapter.

CHAPTER 8

REFLECTIONS AND RECOMMENDATIONS

8.1 INTRODUCTION

The first five chapters of the study provided a theoretical framework, constructed through an in-depth literature research. The preceding chapter provided results of the empirical research, conducted to verify the theoretical framework and to test the assumptions made in the study. In the previous chapter an e-readiness framework was developed, making provision for determining how academics' e-learning training and development interventions should be structured to suit their particular e-profiles. The aim of the preceding chapters was to achieve the research objectives of the study and to answer the research problem outlined in chapter 1.

In this chapter a summary is provided of the main arguments and topics discussed in each chapter. The most significant conclusions, resulting from the study are indicated and particular recommendations in this regard are made. This chapter also indicates the contribution made by this study, as well as further research that can derive from it.

8.2 SUMMARY OF CHAPTERS

In **chapter 1** the primary objective of the study is indicated: to determine the role of key human factors in the e-readiness of academics, with specific focus on personal work profile patterns, preferred learning style and pace and style of technology adoption. The research problem that the study focused on therefore is: How to determine the role of key human factors in the e-readiness of academics.

Chapter 1 provided an introduction and background to the study. The chapter explained the rationale for the study, indicating why the particular topic for research was chosen by highlighting the importance of e-readiness of academics at higher education institutions (HEIs) and by identifying the shortcomings in the literature of both human resource performance appraisal and e-readiness. The context of

technological development, the increased use of e-learning and the adaptation needs for HEIs were outlined. The NWU's stance on e-learning, as well as its support for e-learning by means of a mission statement and various policies was stated. The use and purpose of e-learning in the 21st century was highlighted, as well as the significant role of the online learning facilitator in this regard. A brief profile of the 21st century learner, known as Generation Y, was provided, emphasising the need for changed teaching and learning approaches to accommodate the needs and preferences of these learners.

It was indicated that the changed job demand (to increasingly incorporate technology in teaching and learning) posed challenges to academics and in some instances resistance to the change is experienced, caused by a lack of e-readiness. It was argued that the e-readiness of academics should be assessed during their performance appraisals and be followed with a personal development plan. Further, it was argued that the e-readiness of academics are influenced by particular key human factors, namely style and pace of technology adoption, preferred learning style and personal work profile patterns, that should be considered during the ereadiness assessment of academics. Once an employee's e-profile has been determined through the assessment of these key human factors, uniquely structured training and development interventions, included in the personal development plan can follow to deal with the perceived lack of e-readiness. It was indicated that a framework should be developed for this purpose (which was done in chapter 6).

The chapter also provided a brief outline of the research methodology followed in the study and indicated ethical aspects considered during the study. Study title concepts and other related concepts were clarified to provide clarity as to what the focus of these concepts are and to highlight the primary areas of discussion of the study. Finally, a chapter outlay was provided, giving an indication of the chapters and respective topics to follow.

In **chapter 2** a literature review of public human resource performance appraisal and its role and place in addressing the e-readiness of academics at HEIs was provided. The chapter explored critical elements of a performance management system and performance appraisals of employees. It was recorded that the performance

management system of an organisation is directly connected to the organisation's vision and objectives and that the performance appraisal process derives from the performance management system of the organisation. It was highlighted that it is therefore also necessary to align individual performance goals and objectives with organisational objectives, and the line manager should provide guidance and assistance in this regard. Emphasis was placed on the performance appraisal process of consisting not only of an assessment component, but also of a developmental component, and that the drafting of a personal development plan should follow a performance appraisal.

Employee performance is directly or indirectly related to motivation, therefore the role and importance of employee motivation was discussed and was brought into relation with the e-readiness of academics. The significance of motivating factors such as incentives, goal-setting, intrinsic and extrinsic motivation and self-determination were outlined. Specific attention was also given to the role of self-determination in the motivation towards technology adoption.

Chapter 3 discussed employee development. As human resource performance appraisal entails a developmental component, attention was given to the role of employee development in increasing the e-readiness of academics. The relevance of competency-based training to enhance the e-readiness of academics was explained, the significance of employee development in increased performance and the role of employee development in the use of technology in teaching and learning, were discussed.

Further the rationale for including e-learning training in employee development, career management programmes and talent management programmes received attention. The role that a talent management programme, integrated with other development practices such as career management, succession planning and mentoring and coaching can play in e-learning training and development of academics was explored. The chapter also reviewed the uniqueness of the academic as an adult learner and highlighted the role of the line manager in employee development.

304

In **chapter 4** the key human factors that impacts on academics' e-readiness, were outlined and explained, namely personal work profile patterns, preferred learning style and pace and style of technology adoption. The various technology adopter categories were recorded – innovators, early adopters, early majority, late majority and laggards – and the difference in the pace and style with which employees adapt to technology was explained. It was stressed that an employee's learning style play a significant role in his/her motivation towards training and development and that employees' learning styles differ. The following learning styles were discussed within the context of employee e-readiness: activist, pragmatist, theorist and reflector. Each learning style indicates unique preferences of an employee in the learning environment which will positively impact on the success of the employee's learning if he/she is allowed to use his/her learning style. Personal work profile patterns were discussed by means of Thomas International's DISC profiles, which are dominance, influence, steadiness and compliance. Each one of the DISC factors indicates an employee's preferred work style, influencing the employee's behaviour, attitudes and approaches in the workplace.

The interrelatedness of these human factors and how they contribute to an employee's e-profile was also determined and explained. It was argued that these key human factors should e included in the e-readiness assessment of academics during their performance appraisals.

Chapter 5 discussed the 21st century higher education environment and the typical 21st century learner. The chapter also discussed the role and purpose of e-learning in this environment and in relation to the 21st century learner. Several significant challenges faced by modern day HEIs were discussed, including the type of learners (Generation Y) and their expectations; underprepared learners, inadequately equipped for tertiary education; and the focus on lifelong learning within the context of the knowledge economy. It became evident that HEIs will increasingly have to make use of technology in teaching and learning and academics need to adapt, prepare and be ready for the 21st century learner, which requires the e-readiness of academics.

Further, it was explained what e-readiness encompasses. Enabling factors to ereadiness were outlined and discussed and indicators of the e-readiness of academics were established and included in an e-readiness construct. Since a lack of e-readiness usually causes resistance to e-learning, various barriers to e-learning were discussed. The underlying causes for e-learning resistance could be linked to a lack of e-readiness and emphasised the need for adequate and comprehensive elearning training and development.

Chapter 6 provided a profile of the NWU. Current human resource performance appraisal practices and policies of the NWU were discussed. It appeared that the NWU has sound performance management policies and practices in place, but its performance management system does not make provision for the inclusion of e-readiness assessment of academics. The inclusion of the e-readiness of academics in their performance appraisals is supported by the Director: HRM of the NWU, given the higher education context with technological advancement and the learner profile, in order for the university to remain competitive.

Attention was also given to current employee development and career management practices at the NWU and it was determined that career management, succession planning and mentoring and coaching are not commonplace at the NWU, although it is supported in policies.

The NWU learners have also been profiled in terms of their e-learning preference. It was found that the majority of learners (96%) prefer the inclusion of e-learning in their courses to various extents and that they belong to Generation Y. In a quest to support the needs and preferences of Generation Y learners the NWU, Potchefstroom campus, has launched a pilot project on teaching and learning with technology in the Faculty of Theology. The aim of the project is to determine whether the initiative taken at the Faculty Theology can be expanded to the rest of the campus, based on the successes and failures experienced in this project.

Discussions also revealed that the 21st century learner is increasingly underprepared. The NWU, Potchefstroom campus, has established the Centre for Academic and Professional Language Practice that provides support programmes such as: learner academic literacy, a reading laboratory and a course in computer and technology skills.

In **chapter 7** research results obtained from the questionnaire and focus group discussion were discussed and interpreted. A framework which can be used to determine an employee's e-profile after academics' e-readiness assessment was developed, making provision for the unique structuring of training and development interventions.

8.3 PRIMARY FINDINGS OF THE STUDY

With the aim of achieving the research objectives and answering the research problem outlined in chapter 1, an in-depth literature study and an empirical study was conducted. The study revealed the following findings in terms of the set research objectives:

Determine the role of human resource performance appraisal in addressing the e-readiness of academics

With the profiling of the NWU in chapter 6 it became evident that the performance management system of the NWU does not make provision for the e-readiness assessment of academics. In chapter 2, section 2.2.1, of the study it was determined that a performance management system is integrated with an organisation's vision and objectives, therefore line managers should ensure that individual performance objectives are aligned to the organisational objectives. Section 2.2.1, further revealed that strategic managers should ensure that policies are in place and complied with to accomplish strategic objectives. The inclusion of the e-readiness assessment of academics in performance appraisals therefore calls for new and adjusted policies.

After profiling the NWU policies and practices pertaining to performance management in chapter 6, section 6.2, it became evident that the current *Performance management policy* of the NWU will have to be adjusted to make provision for the inclusion of e-readiness. E-readiness assessment will also impact

on other human resource policies such as the *Recruitment policy*, the *service contracts of academics* and the *Teaching and learning framework*. In addition to the abovementioned adjustments to policies, a separate policy for the use of technology in teaching and learning should also be drafted. This *E-learning policy* should outline the use and purpose of e-learning, the University's position on e-learning, as well as guidelines and directives of e-learning as teaching and learning tool.

In chapter 7, section 7.4.1.6.2 results from the questionnaire portrayed that 44% of respondents were willing to undergo e-readiness assessment as part of their performance appraisals. This number is very close to the number indicating that they are not in favour of e-readiness assessment as part of a performance appraisal, namely 39%. It further became evident that with an increase in age and teaching experience (which also imply an increase in age), the willingness to be assessed on e-readiness declined. Therefore, thorough consultation and communication will be necessary when e-readiness assessment is introduced to the performance appraisals of academics.

As indicated in various chapters of the study, people tend to resist change for various reasons such as fear of the unknown, habits (comfort zones), inadequate information and communication, threats to status, fear of failure and a lack of perceived benefits. In this case where e-learning is being introduced as new job requirement, a lack of understanding of the use and purpose of e-learning, fear of an increased workload, and a threat to the comfort zone (cf Bozarth 2006:2–4; cf Kottolli 2008:1) may cause resistance to the new job requirement and subsequently to e-readiness assessment. It is thus clear that the process of introducing e-readiness assessment to the performance appraisals of academics should be managed properly and employees should be motivated to obtain optimal results and to enhance positivity towards the endeavour.

Determine the role of motivation in employee performance and its relevance to enhancing the e-readiness of academics

In chapter 2, section 2.3, it was revealed that motivation has a significant impact, positively or negatively, on an employee's performance. Therefore, in the NWU's

endeavour to increase employees' performance it is necessary to understand what specifically motivates employees to optimally perform. Line managers need training for employee motivation and management of change.

In chapter 2, section 2.3.2, it was further determined that goal-setting proved to be one of the best motivational strategies as it leads to enhanced performance, provided that employees remain committed to their goals. Both self-set and assigned goals can lead to increased performance and both are positively connected to selfefficacy, which in return leads to increased performance. Further, difficult and specific goals are more likely to lead to performance than easy goals. Goal-setting should be done in conjunction of line managers, providing clear expectations and a rationale for the changed job and how it fits into the organisational strategy and objectives.

In chapter 7, section 7.4.1.3.3, results obtained from the questionnaire revealed that financial incentives take top priority when respondents could choose more than one motivator, although it was not considerably higher than other motivators. Literature supports the importance of financial incentives to some extent. A good salary assists in attracting quality employees and retaining them and according to research financial incentives lead to better work performance (Shadare & Hammed 2009:7–8; Oyedele 2010; Shadare & Hammed 2009:8; cf Cooke & Meyer 2007:1–2; Bruce & Pepitone 1999:118).

When asked to indicate the single most important motivator of the various motivators indicated by respondents, promotion took top priority with 23,6%. Other motivators indicated by respondents were a financial incentive (19,4%), flexitime or the opportunity to work from home (19,4%), recognition from their School Directors/Deans (16,6%), support with resources, for example provision of a laptop, cell phone, etcetera (16,6%) and other (4,2%).

It was also indicated in chapters 2 and 7 (sections 2.3.2 and 7.4.1.3.2) that there is a relationship between incentives and higher self-efficacy (people's perception of how well they can perform a particular task), as well as higher self-set goals, which is related to intrinsic motivation (Callaghan et al 2003:2518–2519). It was further

explained that higher assigned goals lead to higher self-set goals and higher selfefficacy (Locke & Latham 1990:241; Callaghan et al 2003:2515–2517). There is a direct connection between self-efficacy and task performance – individuals with high self-efficacy beliefs perform better, irrespective of the self-set goal level. Self-set goals are also linked directly to task performance (Callaghan et al 2003:2515–2517; cf Madden 1997:412).

Motivation remains a challenging task and line managers should take cognisance of their responsibility in this regard. In particular as the job requirement of using technology in teaching and learning becomes compulsory, leading to the necessary assessment of academics' e-readiness, it will be necessary to motivate employees in order to successfully implement e-learning as learning tool.

• Determine the role of employee development in increasing the ereadiness of academics

The profile of the NWU in chapter 6, section 6.3.1, revealed that the current employee development programmes at the NWU do not address most of the aspects causing a lack of e-readiness. Employee development programmes should make provision for e-readiness training and development. These training and development interventions should not only focus on the technical skill of e-learning, but follow a broader teaching and learning approach, also including the philosophy of e-learning and the use and adaptation of teaching and learning strategies to obtain optimal learner learning. It is also argued that these training and development opportunities can be incorporated in a career management programme or a talent management programme.

It was further determined in chapter 6 that career management programmes and talent management programmes can play a positive role in employee performance. The integrated approach of a talent management programme, as currently planned by the NWU, including human resource management development practices such as career management, succession planning and mentoring and coaching, is expected to have a positive effect on developing the e-readiness of academics. The holistic

approach of a talent management programme will ensure that all relevant aspects of an academic's job requirements are addressed (see chapter 3).

Results obtained from the questionnaire in chapter 7, section 7.4.1.5.1, indicated that an overwhelming majority of 71% of respondents indicated that e-learning training is imperative. It is evident that the vast majority of respondents recognise the importance of being taught how to use e-learning optimally, thus indirectly acknowledging that there is more than the manner in which they are currently using it. Furthermore, 34,2% of respondents indicated that a lack of e-learning training was the reason that prevented them from using e-learning. These results obtained from the questionnaire, together with results obtained from the focus group discussion in chapter 7, section 7.4.2, as well as the current employee development policies and practices at the NWU, emphasised the importance of training and development.

Determine the role of key human factors in the e-readiness of academics, with specific focus on personal work profile patterns, preferred learning style and style and pace of technology adoption

In chapter 4, section 4.2, it was determined that particular key human factors play a significant role in determining the indicators for e-readiness. These key human factors are an employee's pace and style of technology adoption, preferred learning style and personal work profile pattern (DISC factor). It became evident that the traits portrayed by a person belonging to a certain technology adoption category show similarities to traits portrayed by particular learning styles and personal work profile patterns. Certain profiles could be identified, based on trends and similarities pertaining to interpersonal traits, for example an employee who is an *innovator* on the technology adoption cycle will most likely prefer and *activist* learning style and will display a strong *dominance* factor on the DISC profile. These are high achievers who embrace challenges and new opportunities.

Another likely connection that could be identified in chapter 4 was that an *early adopter* on the technology cycle may also portray an *activist* learning style and a *dominance* factor on the DISC profile, as these employees also adapt to technology relatively fast. The *early majority* and *late majority* technology adoption categories

will most likely be related to the *theorist* or reflector style and will correlate with the *steadiness* and *compliance* factors of the DISC profile. Employees falling in these categories need more time to think about a changed job demand and will only adapt when most others have already done so or when it is expected of them. It is thus evident that these combinations imply various levels of e-readiness.

The following prominent e-profiles could be identified in chapter 7, section 7.4.1.2.3, for respondents of the questionnaire: The majority of respondents portrayed an *early adopter-pragmatist-compliance* e-profile. This implies that the majority of respondents will adopt to a new technology/innovation in the workplace, once the concept is proven (*early adopter*); they want to see the practical value of e-learning and would like to see a link between what they are taught and the end-result (*pragmatist*); and they have high standards, particularly for themselves, can be perfectionists, and prefer systems, processes, procedures, as well as predictable and consistent outcomes (*compliance*).

It can be deduced that employees who fall in this category will in all likelihood look forward to teaching online, once they have mastered the new skill. Again the need for training is emphasised as this group of respondents first obtained the skill and then became enthusiastic about e-learning. Once this group of employees, who is not opposed to e-learning and is willing to learn, is trained to use the e-learning platform optimally and couple it to sound educational strategies, it can make a significant positive contribution to learners' experience of the use of technology in teaching and learning and specifically to learners' learning experiences itself.

The second highest profile amongst respondents identified in chapter 7, section 7.4.1.2.3, was *the early majority-theorist-steadiness* e-profile. This implies that employees belonging to this profile are the eventual users of technology who do not like to take the risks of pioneering, but see advantages of tested technologies, and are driven by usability and success of the technology (*early majority*); they require good structure and sufficient time to explore the relevance between ideas and scenarios, are analytical and detail-conscious and need to think things through in a logical step-by-step manner (*theorist*); and they are sympathetic, friendly, good

312

listeners, "finisher completers", and team players, who work hard and create a stable environment (*steadiness*).

It can be deduced that, although e-learning will never be this group's first choice, they will be comfortable with teaching online, once they have mastered the skill. Motivating employees belonging to this group effectively is of particular importance, as they do not naturally lean towards e-learning and make use of it only when half of the population has already done so. It can be argued the only reason they are using technology in teaching and learning is because it is expected of them. Extra time and effort should be taken during the training of employees who fall in this category. It is necessary to provide a rationale and to constantly encourage these employees towards the role that e-learning play as educational advantage. In this regard the line manager will have to play a significant role.

It became evident that the key human factors highlighted in the study influence the ereadiness of academics. It was further revealed that the traits portrayed by a person belonging to a certain technology adoption category show similarities to traits portrayed by particular learning styles and personal work profile patterns. Thus an elearning profile for each academic can be drafted.

• Determine the reasons for resistance to e-learning and a lack of ereadiness of academics

As indicated above, people tend to resist change. The introduction of technology in teaching and learning is no exception. Chapter 5, section 5.3.3, revealed that e-learning brought a shift from the traditional view of the teacher as authority and expert, to a role of the teacher as guide and the learner as explorer (Bozarth 2006:3–4). Many facilitators fear that their specialised status and position as "expert" will be downgraded to the roles of production worker and customer service representative that should be available 24/7 to assist learners with technical problems (Bozarth 2006:3–4).

Also creating resistance, as determined in section 5.3.3, is a belief of loss of routine and the 'old way' of doing things, losing control and a fear of being replaced by technology (Bozarth 2006:3–4; cf Billings sa). These misconceptions and more in many instances withhold academics from making use of e-learning as teaching and learning tool. This supports the notion, mentioned earlier in this section, that the process should be managed with care and that employees should be kept motivated throughout the process, placing a significant responsibility on line managers.

Section 5.3.3 further revealed that online learning facilitators that are regarded as technologically advanced, judged by the extent and manner of their use of technology and online learning approaches, tend to use constructivist strategies such as inquiry learning and collaborative learning (Bozarth 2006:4–5). It appears that the majority of academics still tend to follow the objectivist (teacher-centred approach) and that many do not use the constructivist (learner-centred approach) or does not have adequate knowledge of the constructivist approach to apply it to their teaching and learning strategies. Further, academics are subject experts, not trained educationalists, and therefore in most instances lack the know-how to apply elearning for pedagogical purposes. A lack of willingness to adapt teaching and learning and learning strategies also appears to be eminent.

As determined in chapter 5, section 5.3.1, for many academics, the start of elearning brought many challenges and many ideas and philosophies about adult learning and what constitute up-to-standard acceptable teaching (Robertson 2008:824). Implementation of e-learning also often requires that academics should challenge their belief systems and ideas regarding what comprise teaching and learning (Sharpe et al 2006:135; Robertson 2008:824). All academics did not adapt to including technology in teaching and learning and in many instances the paradigm shift should still take place. Again, the need for training and development, coupled with employee motivation, should be emphasised.

Chapter 4, section 4.2, also revealed that distinct patterns between personality traits and approaches to work relative to the use of technology in teaching and learning can be seen. Online learning facilitators who successfully integrate technology in teaching and learning have a common belief that technology provided a valuable tool to realise their visions of teaching and learning. These facilitators have strong visions of classroom technology use and therefore do not appear to be easily disturbed by typical implementation obstacles. On the contrary, many of these online learning facilitators achieve high levels of use despite the lack of equipment, training, or time (Bozarth 2006:8). Based on both the literature research and the empirical research of the study, these type of facilitators will in all likelihood belong to the *innovator* category of technology adoption (employees who immediately adapt to technology and change and thrive on challenges and new opportunities), and perhaps to the *early adopter* category of technology adoption (employees who adapt once the concept is proven and the strategic linkages have been made).

On the other hand, online learning facilitators who view technology as a presentation tool, or "add-on", rather than the means of enhancing practice, are far more likely to resist e-learning (Bozarth 2006:8). These will in all likelihood be facilitators who can be categorised as the early majority category of technology adoption (the mass market that will initially resist the introduction of a new technology, but that will eventually adapt to it), and the late majority category of technology adoption (employees that will strongly resist the introduction of a new technology for a long period of time and will adapt to the use of the technology very slowly).

Results obtained from the questionnaire and focus group discussion in chapter 7, sections 7.4.1 and 7.4.2, revealed that in some instances academics place the focus of e-learning on the technology instead of on learning. It became evident that academics are not familiar with the philosophy of e-learning and do not know how to optimally use it as learning tool with residential learners. The perception that e-learning should be used for distance learning learners only remains steadfast in many instances.

Further in chapter 7, sections 7.4.1.4, 7.4.1.5.3, 7.4.1.6 and 7.4.2, results obtained from both the questionnaire and focus group discussion revealed that academics are ignorant about the profile of their learners, namely Generation Y. This generation is known for their high interest in and preference for the use of technology. They think differently, act differently, and grew up in different circumstances (the technological age) from previous generations and therefore the traditional classroom approach to teaching and learning does not appeal to them. Yet, in many instances, provision is not made for their needs and preferences by academics, belonging to different
generations. A significant result, emanating from the focus group discussion was that academics are not necessarily willing to adapt their teaching and learning strategies to provide for the needs and preferences of their learners.

It became evident that the causes for a lack of e-readiness and resistance to the use of technology in teaching and learning can be addressed through employee development. Training and development interventions should, however, not only focus on technical skill, but should also focus on the philosophy of e-learning and the use of technology as learning tool. Training and development should further guide academics in the use and adaptation of teaching and learning strategies.

• Determine the indicators for e-readiness of academics by creating an ereadiness construct

The e-readiness construct, constructed for this study in chapter 5, section 5.3.2, consists of the attributes/indicators that provide an indication of an employee's level of e-readiness, systematically arranged as a set of standards regarding e-readiness. This arranged set of ideas can be compared to other constructs, in this case key human factors of academics. The e-readiness construct assisted the researcher in understanding the contributing factors to the e-readiness of academics. Thus it could be determined what the solutions are to a lack of e-readiness. As the e-readiness construct specifies people attributes pertaining to e-readiness, it will relate to the key human factors discussed in this study. Therefore, when these key human factors are assessed during the performance appraisals of academics, employees' e-profiles can be determined, that will give an indication of the presence or lack of the indicators, showed in the e-readiness construct below.

Table 8.1:	E-readiness	construct

Indicator	Specifications		
Technical skill	Basic computer skills		
	Skill to use the e-learning platform tools		
	 Basic tools – announcements, messages, schedule, resources, 		
	assignments, gradebook		
	 Interactive/advanced – forum, e-testing, podcasts, wikis, bloggers, 		
	simulations		
	Technical support to learners		
Subject competency	Subject expert – knowledge		
	Curriculum development		
	Development of course material		
	Facilitate knowledge construction		
	Well-structured assessments		
	Guidance and assistance in terms of subject related problems		
	Providing feedback		
Comfortableness	Ease of use		
	Lack of fear/resistance		
	Enjoyment of online activities		
Willingness	Positive attitude		
	Adaptability and flexibility to adjust to change		
	Focusing on outcome of learning		
	Considering learner profile		
Understand use and purpose of e-	Philosophy of e-learning – understand e-learning as pedagogical tool		
learning	Use technology to achieve learning		
Knowledge and use of teaching and	Knowledge and skill of pedagogical strategies an approaches		
learning strategies	Flexibility with teaching and learning strategies		
	Educational "savvy"		
	Considering learner profile		
Interpersonal skills	Communication – sharing information; provide direction and support; online		
	charisma		
	Motivation – encourage learners		
	Providing feedback		
	Accommodating		
	Counselling skills		
	Good rapport with learners		
	Ability to bridge the gap – building relationships		
	Mental flexibility		
	Compassion/empathy		
Personal attributes	Diligence		
	Perseverance		
	Commitment		
	Openness		
	Creativity and innovation		
	Time management		

Based on the e-readiness construct, the minimum required level of e-readiness of an academic can be regarded as the inclusion of the following indicators: technical skill in the form of basic computer skills and basic e-learning platform skills; subject competency; willingness; comfortableness; an understanding of the use and purpose of e-learning; and knowledge and skill to use various teaching and learning strategies. These indicators are linked to an employees' e-profile, that will be determined during e-readiness assessment (by assessing the employee's key human factors), and will provide insight into which attributes will need attention during training and development interventions.

• Develop a framework in for e-readiness assessment of academics during performance appraisals to determine their level of e-readiness and subsequent training and development needs that will be specified in a personal development plan

As the study argues for the inclusion of e-readiness assessment of academics during their performance appraisals, the research was focused on how this could be done and which role particular key human factors play in this regard. It became evident that particular key human factors (pace and style of technology adoption, preferred learning style and personal work profile patterns) impacts on an academic's e-readiness. Thus, when these key human factors are assessed, an e-profile can be drafted for each academic. In chapter 7, section 7.6, a framework was developed for this purpose. Once the employees have been assessed, their results can be plotted on the framework (consisting of three matrixes), which provides an employee's e-profile. The e-profile in return provides an indication of the presence or lack of particular e-readiness indicators of the employee, as outlined in the e-readiness construct. Once the employee's e-profile is known, training and development interventions, best suitable for the employee can be planned and implemented. Tables 8.2, 8.3 and 8.4 indicate the e-readiness framework.

Person	al work profile pattern	Dominance	Influence	Steadiness	Compliance
	(DISC profile)				
	Innovator	Unstructured; freedom to explore; internal motivation; visual aids; fast pace			
T E C H N	Early Adopter	Structured/unstructured; freedom to explore; internal motivation; visual aids; instructor that demonstrates; relatively fast pace	Structured; external motivation; visual aids; instructor that demonstrates; instructor that gives verbal instructions; paper-based notes; opportunity for socialisation; relatively fast pace		Structured, step-by-step approach; external motivation; visual aids; instructor that demonstrates; instructor that gives verbal instructions; relatively fast pace
O L O G	Early majority		Structured; external motivation; visual aids; instructor that demonstrates; instructor that gives verbal instructions; paper-based notes; opportunity for socialisation	Structured, step-by-step approach; external motivation; visual aids; instructor that demonstrates; instructor that gives verbal instructions; paper-based notes	Structured, step-by-step approach; external motivation; visual aids; instructor that demonstrates; instructor that gives verbal instructions; paper-based notes
Y A D O P	Late majority		Structured; external motivation; visual aids; instructor that demonstrates; instructor that gives verbal instructions; paper-based notes; opportunity for socialisation; enough opportunity to reflect and ask questions	Structured, step-by-step approach; external motivation; visual aids; instructor that demonstrates; instructor that gives verbal instructions; enough opportunity to reflect and ask questions	Structured, step-by-step approach; external motivation; visual aids; instructor that demonstrates; instructor that gives verbal instructions; enough opportunity to reflect and ask questions
T I O N	Diehard/Laggard			Structured; external motivation; visual aids; instructor that demonstrates; instructor that gives verbal instructions; paper-based notes; enough opportunity to reflect and ask questions	Structured; external motivation; visual aids; instructor that demonstrates; instructor that gives verbal instructions; paper-based notes; enough opportunity to reflect and ask questions

Table 8.2: The technology adoption – personal work profile pattern match

L	earning style	Activist	Pragmatist	Theorist	Reflector
т	Innovator	Unstructured; trial and error; freedom to explore; internal motivation; visual aids; fast pace			
E C H N O L O G Y A D O F T I O N	Early Adopter	Structured/unstructured; freedom to explore; internal motivation; visual aids; instructor that demonstrates; match strategically; relatively fast pace	Structured; match strategically; holistic picture; future benefits; step-by-step approach; external motivation; instructor that demonstrates; visual aids; relatively fast pace		
	Early majority		Structured; match strategically; picture; future benefits; step-by-step approach; external motivation; instructor that demonstrates; visual aids ; paper-based notes	Structured; external motivation; step-by-step approach; external motivation; instructor that demonstrates; instructor that gives verbal instructions; visual aids; paper-based notes	Structured; unhurried process; ample opportunity to ask questions and review; opportunity to reflect on progress; external motivation; instructor that demonstrates; instructor that gives verbal instructions; visual aids; paper-based notes
	Late majority		Structured; holistic picture; step-by-step approach; external motivation; instructor that gives verbal instructions; instructor that demonstrates; visual aids	Structured; step-by-step approach; external motivation; instructor that demonstrates; instructor that gives verbal instructions; visual aids; paper-based notes	Structured; step-by-step approach; ample opportunity to ask questions and review; opportunity to reflect on progress; external motivation; instructor that demonstrates; instructor that gives verbal instructions; visual aids; paper-based notes
	Diehard/laggard			Structured; step-by-step approach; external motivation; external motivation; instructor that demonstrates; instructor that gives verbal instructions; visual aids; paper-based notes	Structured; unhurried process; ample opportunity to ask questions and review; opportunity to reflect on progress; external motivation; instructor that demonstrates; instructor that gives verbal instructions; visual aids; paper-based notes

Table 8.3: The technology adoption – preferred learning style match

Table 8.4: Personal work profile pattern – preferred learning style match

L	earning style	Activist	Pragmatist	Theorist	Reflector
	Dominance	Unstructured; trial and error; freedom to explore; internal motivation; visual aids; fast pace			
	Influence			Structured; step-by-step approach; external motivation; visual aids; instructor that demonstrates; instructor that gives verbal instructions; paper-based notes; opportunity for socialisation; enough opportunity to ask	Structured; step-by-step approach; external motivation; visual aids; instructor that demonstrates; instructor that gives verbal instructions; paper-based notes; opportunity for socialisation; enough opportunity to reflect and
DISC	Steadiness		Structured; match strategically; holistic picture; practical step-by- step approach; external motivation; visual aids; instructor that demonstrates; instructor that gives verbal instructions; paper- based notes	questions Structured; step-by-step approach; external motivation; visual aids; instructor that demonstrates; instructor that gives verbal instructions; paper-based notes; enough opportunity to ask questions	ask questions Structured; step-by-step approach; external motivation; visual aids; instructor that demonstrates; instructor that gives verbal instructions; paper-based notes; enough opportunity to reflect and ask questions
	Compliance		Structured; match strategically; holistic picture; future benefit; practical step-by-step approach; external motivation; visual aids; instructor that demonstrates; instructor that gives verbal instructions; paper-based notes	Structured; step-by-step approach; external motivation; visual aids; instructor that demonstrates; instructor that gives verbal instructions; paper-based notes; enough opportunity to ask questions	Structured; step-by-step approach; external motivation; visual aids; instructor that demonstrates; instructor that gives verbal instructions; paper-based notes; enough opportunity to reflect and ask questions

8.4 CONCLUSION

Given the technological advancement of the 21st century, the needs and preferences of the 21st century learner, the increasing use of technology in teaching and learning and the significant role of the online learning facilitator in e-learning, the e-readiness of academics became crucial to successful teaching and learning in the 21st century and for universities to remain competitive. For academics to effectively use e-learning tools to optimally enhance learning, necessitates their e-readiness.

The study argued for the inclusion of e-readiness assessment during the performance appraisals of academics. During the e-readiness assessment particular key human factors (pace and style of technology adoption, preferred learning style and personal work profile pattern) should be assessed. The results from this assessment will be plotted on the e-readiness framework to determine the employee's e-profile. Once an academic's e-profile has been determined, the most suitable training and development interventions can be planned and included in the e-readiness construct, can to a greater or lesser extent be included in employees' training and development, based on their e-profiles.

From the questionnaire results it was clear that respondents portrayed a relatively high (46%) or moderate (33%) level of e-readiness. If these results are considered to be generally representative of the academics of the NWU, Potchefstroom campus, it is clear that successfully training and developing employees portraying the *early adopter-compliance-pragmatist* and the *early majority-steadiness-theorist* pertaining to the use and purpose e-learning, the bulk of the academic workforce at the NWU (79%) will be well-skilled and able to implement e-learning optimally to support world class learner learning. If the 13% of respondents who portray a *dominance-activist-innovator* e-profile is considered to the bulk of academics portraying the abovementioned two e-profiles (79%), it brings the total of e-ready academics to 92%. Undoubtedly this will make a difference in the effectiveness of the use of e-learning.

8.5 CONTRIBUTION OF THE STUDY

The study addresses the gap in the literature of both human resource performance appraisal and e-readiness by making provision for the e-readiness assessment of academics during performance appraisals, thus contributing to the scholarly knowledge in the field of human resource performance appraisal.

The importance of key human factors, impacting on the e-readiness of academics at HEIs, has been recognised, providing for addressing the resistance to e-learning as the assessment of the key human factors provides an e-profile of the employee. The e-profile will indicate the employee's shortcomings in terms of e-readiness indicators, as outlined in the e-readiness construct, thus revealing where the training and development focus should be placed and what the minimum accepted level of e-readiness is. A framework has been developed for this purpose.

The results obtained from the study can be used during institutional planning of the NWU for e-learning and in the drafting of an e-learning policy. It can also be used for the incorporation of e-readiness assessment during the performance appraisals of academics, as well as for planning and drafting their personal development plans and subsequent training and development interventions.

8.6 **RECOMMENDATIONS**

Given the research problem, to determine the role of particular key human factors in the e-readiness of academics, to establish a framework for the assessment of the ereadiness of academics and to structure unique training and development interventions to enhance their level of e-readiness, and the research objectives, outlined in chapter 1, the following recommendations can be made:

• The assessment of academics' e-readiness during their performance appraisals

Presently the higher education environment is faced with technological advancement, 21st century learners (predominantly belonging to Generation Y),

preferring the use of technology in teaching and learning and not comfortable with the traditional teaching and learning environment, underprepared learners, the growing needs of non-traditional learners, the paradigm shift from "one time" learning to "lifelong learning", and an increasingly competitive higher education environment. These factors necessitated the increasing use of technology in teaching and learning. The online learning facilitator plays a significant role in this regard, thus highlighting the need for the e-readiness of academics. Given these realities, the ereadiness of academics at higher education institutions must be assessed during their performance appraisals.

• The assessment of particular key human factors to determine an academic's e-readiness

A framework should be used to assess the e-readiness of academics. The study proposes that particular key human factors (pace and style of technology adoption, preferred learning style and personal work profile patterns) be assessed to determine and employee's e-profile by plotting the results on a framework (consisting of three matrixes), developed by the researcher for this purpose.

• The drafting of a development plan, including suitable e-readiness training and development interventions for each academic

Once an academic's e-readiness assessment results have been plotted on the framework, training and development interventions, suitable for each employee can be planned and implemented. Line managers should play an important role in identifying training needs, providing opportunities to attend training and development sessions, assist the employee in goal-setting, ensuring that performance goals are aligned to the university's strategic goals and objectives, communicating, supporting and motivating the employee.

• Suitable training and development interventions to enhance the ereadiness of academics

Training and development to enhance the e-readiness of academics should include the philosophy of e-learning. This will provide academics with a better understanding of the use and purpose of e-learning and will create awareness of its benefits. Further, it will eliminate ignorance and resistance to e-learning, which to ultimately enhance the e-readiness of academics which in return will result in the optimal use of e-learning as pedagogical tool, maximising the learning experience for learners and addressing their needs and preferences.

The presence or absence of e-readiness indicators, outlined in the e-readiness construct, which will be revealed by an employee's e-profile, should be included in their training and development. E-readiness training and development should also focus on particular teaching and learning strategies that can be used and how these strategies can be adapted and applied to suit the needs and preferences of learners to optimally enhance learning.

Training and development strategies should be incorporated into broader career management and talent management plans, ensuring that all the areas of an academic's job requirements will be addressed during training and development interventions. By incorporating employee development in a talent management programme, provision can also be made for succession planning which will enhance employee motivation and retention.

Academic Support Services should be responsible for the course development of training and development interventions. They should also take responsibility for the training of academics. Training and development can take place in various phases, from basic training, progressing through the levels to the most advanced phase. Further, teaching and learning advisors at Academic Support Services can act as coaches to academics who are struggling with e-learning.

• The drafting of policies for e-learning and adjustment of performance management policies

Although the NWU has sound performance management policies, none of these policies make provision for the inclusion of e-readiness assessment of academics during their performance appraisals. The inclusion of the e-readiness assessment of academics in performance appraisals therefore calls for new and adjusted policies. The current *Performance management policy* of the NWU will have to be adjusted to make provision for the inclusion of e-readiness. E-readiness assessment will also impact on other human resource policies:

- Recruitment policy If e-learning becomes a critical evaluation area for academics, their job specifications (indicating the competencies to perform their job successfully), need to be adjusted. It will thus implicate that the matter of an e-ready academic should already be addressed in the recruitment and selection phases. Academics will have to be assessed on their e-readiness before they are appointed or they should be appointed on a probationary period and their permanent appointment should be subject to an acceptable level of e-readiness. In the latter case an academic can undergo a development programme, once he or she is appointed. The *Recruitment policy* should thus make provision for e-learning and e-readiness to fulfil the job requirements and should be indicated in job advertisements.
- Service contracts of academics If compulsory, e-learning should be included in the service contracts of academics and be described in *Human resource management policy* of the NWU.
- Teaching and Learning Framework The current Teaching and learning framework should be adjusted to make provision for the e-learning training and development of academics, broader than providing merely basic technical skills. Academics should also be trained how to incorporate and adjust new teaching and learning strategies and what the most appropriate ways of the use of e-learning will be to enhance teaching and learning training should be incorporated of a broader teaching. These aspects should be reflected in the Teaching and learning framework of the NWU.

In addition to the above mentioned adjustments to policies, a separate policy for the use of technology in teaching and learning should also be drafted. This *E-learning policy* should outline the use and purpose of e-learning, the University's position on e-learning, as well as guidelines and directives of e-learning as teaching and learning tool. Prior to the adjustment or introduction of the above mentioned policies, it is necessary for Institutional Management of the NWU to consult all stakeholders, including the workplace forums and labour unions (SA 1995:63,64).

8.7 FURTHER RESEARCH

Further research can include the drafting of an instrument that makes provision for the assessment of all three key human factors discussed in this study (if current instruments for assessing each of these human factors separately will not be used). Further, a study can be done to determining the effectiveness of an e-ready academic as opposed to an academic that portrays a lack of e-readiness on the successful use of e-learning.

BIBLIOGRAPHY

Abraham, S. 2007. Employee retention requires people-centered plan. CementAmericas.Availablehttp://cementamericas.com/mag/cement employee retention requires/.Dateaccessed: 8 September 2010.0

Achimugu, P, Oluwagbemi, O & Oluwaranti, A. 2010. An evaluation of the impact of ICT: diffusion in Nigeria's Higher Educational Institutions. *Journal of Information Technology Impact* 10(1):25–34.

Adam (Sn). 2010. *Employee training and development*. MBA Tutorials, Online MBA Community. Available at: <u>http://www.mba-tutorials.com/human-resource-management/505-employee-training-and-development.html</u>. Date accessed: 7 January 2011.

Adams, NB. 2007. Toward a model for knowledge development in virtual environments: strategies for student ownership. *International Journal of Social Sciences* 2(2):71–77.

Adams, P. 2005. A strategic migration of telephony into an internet protocol world – a South African perspective. MA dissertation. University of Pretoria, Pretoria. Available at: <u>http://upetd.up.ac.za/thesis/available/etd-04252005-</u> 142015/unrestricted/00dissertation.pdf. Date accessed: 6 January 2011.

Adams, DA, Nelson, RR & Todd, PA. 1992. Perceived usefulness, ease of use, and usage of information technology: a replication. *MIS Quarterly* 16(2):227–247.

Adendorff, DE. 2004. An investigation of the roles and competencies of an online facilitator. PhD thesis. University of Pretoria. Available at: http://upetd.up.ac.za/thesis/available/etd-08232004-071632/. Date accessed: 15 February 2007.

Akdemir, O & Koszalka, TA. 2008. Investigating the relationships among instructional strategies and learning styles in online environments. *Computers & Education* 50:1451–1461.

Albertsson, C. 2003. The talent management game, *Strategic HR Review* 2(5):3.

Albright, MJ & Nworie, J. 2008. Rethinking academic technology leadership in an era of change. *Educause Quarterly* 1:15–23.

Alessi, SM & Trollip, SR. 2001. *Multimedia for learning: methods and development*. 3rd edition. Needham Heights, Mass: Allyn & Bacon.

Alexander, S. 2001. E-learning: developments and experiences. *Education and Training* 43 (4/5):240–248.

Alexandrou, M. 2011. *Technology adoption life cycle*. Information Technology Blog. Available at: <u>http://www.mariosalexandrou.com/blog/technology-adoption-life-cycle/</u>. Date accessed: 5 January 2011.

Alvaro, F. 2010. Using ICT and electronic music to reduce school dropout in Europe, *eLearning Papers* 19:1–13, April. Available at: <u>http://www.elearningeuropa.info/files/media/media22477.pdf</u>. Date accessed: 11 September 2010.

Ambler, SW. 2010. Strategies for effective training and education in InformationTechnology(IT).Ambysoft.Availableat:http://www.ambysoft.com/essays/trainingAndEducation.htmlDateaccessed:28October 2010.

Amirault, RJ & Visser, YL. 2009. The university in technological change: a historically grounded perspective. *Computing of Higher Education Journal* 21:62–79.

Amos, S. 2009. Performance appraisal: a means of development or control? Available http://www.satc.org.uk/performance-appraisal-a-means-ofat: development-or-control/. Date accessed: 28 January 2009.

Andrew, MD & Durand, AJ. 2001. Information and communication technology policy: framework for use in the education system. A draft ICT policy for the Ministry of Education, Sports and Youth Affairs of Barbados. A policy framework of the United Public Administration Nations Network (UNPAN). Available at: http://unpan1.un.org/intradoc/groups/public/documents/TASF/UNPAN024791.pdf. Date accessed: 13 March 2011.

Anonymous. 2009. My best essays. Available at: http://mybestessays.com/apathyessay-research-paper-apathyscience-may-have/. Date accessed: 27 November 2010.

Available Anonymous. 2008 Student VS learner. Blog. 1 Dec. at: http://newkidonthehallway.typepad.com/new kid on the hallway/2008/12/student-vlearner.html. Date accessed: 17 September 2009.

Answers Corporation. 2007. Answers.com. Available at: http://www.answers.com. Date accessed: 14 November 2007.

Archibugi, D & Pietrobelli, C. 2003. The globalisation of technology and its implications for developing countries: windows of opportunity or further burden? Technological Forecasting & Social Change 70:861–883.

Armstrong, M. 2006a. Strategic human resource management: a guide to action. Philadelphia: Kogan Page.

Armstrong, M. 2006b. A handbook of human resource management practice. 10th edition. Philadelphia: Kogan Page.

Arp, L, Woodard, BS & Mestre, L. 2006. Accommodating diverse learning styles in an online environment, Reference and User Services Quarterly 46(2):27-32. Hereian List of research project topics and materials

Aswathappa, K. 2005. *Human resource and personnel management*. 4th edition. New Delhi: Tata McGraw-Hill.

Attwell, G. 2004. *E-learning and sustainability*. Report produced for the University of Bremen as a contribution to the Lefo Learning Folders project. Available at: <u>http://www.guidance-</u>

<u>research.org/knownet/writing/papers/sustainabilitypaper/attach/sustainibility4.doc.pdf</u> Date accessed: 12 January 2011.

Auer, P & Cazes, S. 2000. The resilience of the long-term employment relationship: Evidence from the industrialized countries. *International Labour Review* 19(4):379–408.

Australian National Training Authority (ANTA). 2003. *Effective online facilitation: Australian flexible learning quick guide series*. Available at: <u>http://http//:www.flexiblelearning.net.au</u>. Date accessed: 27 November 2009.

Australian Flexible Learning Framework (AFLF). 2003. *Effective online facilitation: Australian flexible learning quick guide series*. Available at: <u>http://pre2005.flexiblelearning.net.au/guides/facilitation.pdf</u>. Date accessed: 6 January 2011.

Azhar, MUSN. 2008. Learning styles and their impact. Paper presented for methods to business writing. Available at: <u>http://www.docstoc.com/docs/11517765/Learning-Style-And-Their-Impacts</u>. Date accessed: 14 January 2011.

Baard, PP, Deci, EL & Ryan, RM. 2004. Intrinsic need satisfaction: a motivational basis of performance and well-being in two work settings. *Journal of Applied Social Psychology* 34:2045–2068.

Babaita, IS. 2010. Productivity as a driving force for investment in training and management development in the banking industry. *European Journal of Social Sciences* 13(2):278–290.

Babbie, E. 2008. *The basics of social research.* 4th edition. Belmont, Calif: Thompson Wadsworth.

Babbie, E. 2005. *The basics of social research.* 3rd edition. Belmont, Calif: Thompson Wadsworth.

Babbie, E & Mouton, J. 2004. *The practice of social research*. Cape Town: Oxford University Press.

Bacal & Associates. 2008. *Is there a difference between training and development?* Performance Management and Appraisal Help Centre. Available at: <u>http://performance-appraisals.org/faq/trainingdevelopment.htm</u>. Date of access: 31 August 2009.

Bailey, TR, Hughes, K & Moore, DT. 2004. *Working knowledge: work-based learning and education reform.* New York: RoutledgeFalmer.

Bainbridge, C. (Sa). Intrinsic motivation. *About.com: a part of the New York Times Company*. Available at: <u>http://giftedkids.about.com/od/glossary/g/intrinsic.htm</u>. Date accessed: 15 October 2007.

Bakx, AW, Van der Sanden, JMM & Vermetten, YJM. 2002. Personality and individual learning theories: a cross sectional study in the context of social-communicative training. *Personality and Individual Differences* 32(7):1229–1245.

Bandyopadhyay, P. 2007. Career anchor based HR practices. *NHRD Journal* 1(4):34-37.

Bartridge, T. 2004. *Competence-based training and development*. AMEinfo.Com: The Ultimate Middle East Business Resource. Available at: <u>http://www.ameinfo.com/43457.html</u>. Date accessed: 11 October 2009.

Bartridge, T. 2004. *Manager's role in a competence-based training and development system*. AMEinfo.Com: The Ultimate Middle East Business Resource. Available at: <u>http://www.ameinfo.com/44529.html</u>. Date accessed: 16 October 2010.

Bassendowski, S. 2011. How are you using social networking tools? *Canadian Journal of Nursing Informatics* 6(1), Technology in Education Column. Available at: http://cjni.net/journal/?p=1199. Date accessed: 1 May 2011.

Bates, T. 2010. *Innovate or die: a message for higher education institutions*. Elearning and distance education blog, 28 June 2010. Available at: <u>http://www.tonybates.ca/2010/06/28/innovate-or-die-a-message-for-higher-</u> <u>education-institutions/</u>. Date accessed: 4 January 2011.

Bennet, RE. 2002. Inexorable and inevitable: the continuing story of technology and assessment. *The Journal of Technology, Learning, and Assessment* 1(1):1–24. Available at: <u>http://escholarship.bc.edu/jtla/vol1/1/</u>. Date accessed: 2 November 2010.

Beshears, FM. (Sa). *The technology adoption life-cycle*. Available at: http://ist-socrates.berkeley.edu/~fmb/articles/lifecycle/index.html. Date accessed: 10 October 2005.

Beukes, M. 2010. Interview. Reading Laboratory Manager: North-West University (NWU), Potchefstroom Campus. Personal Interview, 4 November, Potchefstroom.

Billings, R. (Sa). *Managing change*. Available at: <u>http://www.trainersdirect.com/resources/articles/ChangeManagement/ManagingChangeBillings.htm</u>. Date accessed: 1 December 2009.

Birkbeck University of London. 2005. *An e-learning strategy for Birkbeck, 2005–2008.* As presented to the College Teaching Committee on 20 October 2005 and subsequently approved by ABEXCo on 9 November 2005.

Black, AE & Deci, EL. 2000. The effects of instructors' autonomy support and students' autonomous motivation on learning organic chemistry: a self-determination theory perspective. *Science Education* 84:740–756.

Blaikie, N. 2004. Analyzing quantitative data. London: Sage.

Blake, DA. 2009. What I learned from teaching adult learners online. eLearnMagazine.Availablehttp://www.elearnmag.org/subpage.cfm?section=articles&article=105-1.Dateaccessed: 30 October 2010.Date

Bloom, BS. 1956. *Taxonomy of educational objectives, handbook 1: cognitive domain*. New York: David McKay.

Bluestein, J. 2001. The ideal student. Excerpt from *Creating Emotionally Safe Schools.* Available at: <u>http://www.janebluestein.com/handouts/ideal.html</u>. Date accessed: 13 May 2011.

BNet: The CBS Interactive Business Network. 2010. *Career development*. Available at: <u>http://www.bnet.com/topics/Career+Development</u>. Date accessed: 8 September 2010.

Bonadio, S. 2010. The CEO's guide to top performer retention - ensuring you have the right talent to grow business. *Ezine Articles*. Available at: <u>http://ezinearticles.com/?The-CEOs-Guide-To-Top-Performer-Retention----Ensuring-</u> <u>You-Have-The-Right-Talent-To-Grow-Business&id=4959771</u>. Date accessed: 8 September 2010.

Borden, VMH. 2004. Accommodating student swirl: when traditional students are no longer the tradition. *Change* 36(2):10–17.

Boyatzis, RE. & Kolb, DA. 1995. From learning styles to learning skills: the executive skills profile. *Journal of Managerial Psychology* 10(5):3-17.

Bozarth, J. 2006. *Classroom trainer resistance to e-learning*. Available at: <u>http://www.bozarthzone.com/bozarth_classroom_trainer_resistance_lit_review_2006</u>. .pdf. Date accessed: 23 August 2009.

Bradshaw, LK. 2002. Technology for teaching and learning: strategies for staff development and follow-up support. *Journal of Technology and Teacher Education* 10(2002). Available at: http://www.questia.com/googleScholar.qst;jsessionid=3A74F058348D10730E842B4 http://www.questia.com/googleScholar.qst;jsessionid=3A74F058348D10730E842B4 http://www.questia.com/googleScholar.qst;jsessionid=3A74F058348D10730E842B4 http://www.questia.com/googleScholar.qst;jsessionid=3A74F058348D10730E842B4 http://www.questia.com/googleScholar.qst;jsessionid=3A74F058348D10730E842B4 http://www.questia.com/googleScholar.qst;jsessionid=3A74F058348D10730E842B4

Brendan, L. 2006. Defining the digital divide: the role of e-readiness indicators. *Aslib Proceedings: New Information Perspectives* 58 (4):276–291.

Bridges.org. 2005a. *E-readiness assessment: who is doing what and where*? 28 February. Available at: <u>http://www.bridges.org/publications/40</u>. Date accessed: 17 February 2007.

Bridges.org. 2005b. *E-ready for what? E-readiness in developing countries: current status and prospects toward the millennium development goals* – Executive Summary, 11 May. Available at: <u>http://www.bridges.org/publications/18</u>. Date accessed: 17 February 2007.

Brinkley, I. 2010. *Knowledge economy strategy 2020: The Work Foundation submission to the Comprehensive Spending Review*. The Work Foundation. Available at:

http://www.theworkfoundation.com/research/publications/publicationdetail.aspx?olte mld=263&parentPageID=102&PubType=. Date accessed: 2 November 2010.

Brinkley, I. 2006. *Defining the knowledge economy: knowledge economy programme report.* The Work Foundation. Available at: <u>http://www.theworkfoundation.com/research/publications/publicationdetail.aspx?olte</u> <u>mld=263&parentPageID=102&PubType</u>=. Date accessed: 2 November 2010.

Bruce, A & Pepitone, JS. 1999. *Motivating employees*. Madison, Wis: McGraw-Hill.

Brüssow, S. 2007. A learning facilitation framework to enhance academic skills development among underprepared learners in South African higher education. PhD thesis. University of the Free State, Bloemfontein.

Budhiraja, R & Sachdeva, S. 2010. E-readiness assessment (India). India e-Government,KnowledgeExchange.Availablehttp://www.indiaegov.org/knowledgeexchg/ereadiness_Assessment.pdf.Dateaccessed: 6 January 2011.Date

Butler, M. 2004. Outcomes based/outcomes focused education overview. Article contribution to the King Faisel Specialist Hospital and Research Centre (KFSHRC) in Saudi Arabia by PhD candidate. Available at: http://www.kfshrc.edu.sa/files/Outcomes%20Based%20Education.doc. Date accessed: 2 January 2011.

Callaghan, JS, Brownlee, AL, Brtek, MD & Tosi, HL. 2003. Examining the unique effects of multiple motivational sources on task performance. *Journal of Applied Social Psychology* 33(12):2515–2535.

Cano-Hurtado, Carot-Sierra, Fernandés-Prada & Fargueta. (Sa). *An evaluation model of the teaching activity of academic staff*. Organisation for Economic Cooperation and Development (OECD). Available at: <u>www.oecd.org/dataoecd/4/29/43977296.pdf</u>. Date accessed: 15 September 2010.

Cardno, C. 1995. Diversity, dilemmas and defensiveness: leadership challenges in staff appraisal contexts. *School Organisation* 15(2):117–131.

Cardwell, C & Madigan, D. 2004. Promoting e-literacy: an open, blended approach in Higher Education. *Journal of eLiteracy*, 1:23–30. Available at: <u>http://www.jelit.org/7/01/JeLit Paper 2.pdf</u>. Date accessed: 7 January 2011.

Carnesale, A. 2000. Speech delivered by the Chancellor of the University of California, Los Angeles (UCLA), 31 May 2000, at the Milken Institute Forum.

Available at: <u>http://www.milkeninstitute.org/pdf/carnesaletrans.pdf</u>. Date accessed: 6 January 2011.

Carpenter, M, Bauer, T & Erdogan, B. 2009. *Principles of management*. Available at: <u>http://www.web-books.com/eLibrary/NC/B0/B58/000Title.html</u>. Date accessed: 8 September 2010.

Carr, VH Jr. 2007. *Technology adoption and diffusion*. Available at: <u>http://tlc.nlm.nih.gov/resources/publications/sourcebook/adoptiondiffusion.html</u>. Date accessed: 12 March 2009.

Cassidy, S. 2004. Learning style: an overview of theories, models and measures. *Educational Psychology* 24(4):419–444.

Chen, PD, Gonyea, R & Kuh, G. 2008. *Learning at a distance: engaged or not?* Innovate Online. Available at: <u>http://www.innovateonline.info/index</u>. Date accessed: 16 November 2010.

Chen, JS. 2007. Information technology adoption for service innovation practices and competitive advantage: the case of financial firms. *Information Research* 12(3). Available at: <u>http://informationr.net/ir/12-3/paper314.html</u>. Date accessed: 1 September 2010.

Chirkov, V, Ryan, RM, Kim, Y & Kaplan, U. 2003. Differentiating autonomy from individualism and independence: a self-determination theory perspective on internalization of cultural orientations and well-being. *Journal of Personality and Social Psychology* 84(1):97–110.

Choden, A. 2005. *How to succeed as an online facilitator*. Training by design: Timely Topics. Available at: <u>http://www.trainingbydesign.com/Facilitator.htm</u>. Date accessed: 31 March 2009.

Choucri, N, Maugis, V, Madnick, S, Siegel, M, Gillet, S, O'Donnel, S, Best, M, Zhu, H & Haghseta, F. 2003. Global e-readiness – for what? Research paper, Centre for e-

Business at Massachusetts Institute of Technology, Sloan School of Management. Available

http://http//:ebusiness.mit.edu/research/papers/177_Choucri_GLOBAL_eREADINES S.pdf. Date accessed: 19 August 2006.

Chyung, SY. 2007. Invisible motivation of online adult learners during contract The Journal of Educators Online 4(1):1-22. Available learning. at: http://www.thejeo.com/Volume4Number1/ChyungFinal.pdf. of 1 Date access: September 2010.

CIPD. 2008. *Learning, training and development*. Available at: <u>http://www.cipd.co.uk/subjects/Irnanddev/general/Irngstyles.htm</u>. Date accessed: 11 August 2008.

Clark, C. 2010. Written communications with the Deputy Director-general of the Department of Public Service and Administration (DPSA), 30 July 2010.

Clarke, RC & Mayer, RE. 2007. *E-learning and the science of instruction: proven guidelines for consumers and designers of multimedia learning*. New Jersey: John Wiley and Sons.

Cline, L & Kisamore, JL. 2008. The changing nature of careers and career management: recommended roles and responsibilities for employees and organizations. Paper presented at the Annual Conference of the Midwest Academy of Management at St. Louis, Missouri, 2–4 October.

Coach4Growth. 2007. *Time management: the manager's role in employee development*. Available at: <u>http://www.coach4growth.com/coaching-skills/team-coaching/time-management-employee-development.html</u>. Date accessed: 27 October 2010.

Coach4Growth. 2007. Investing wisely in performance coaching: where to spend your coaching time. Available at: http://www.coach4growth.com/coaching-

skills/team-coaching/time-management-employee-development.html. Date accessed: 27 October 2010.

Coates, DE. 2010. Effective leadership skills – leaders learn best on the job. *Ezine Articles*. Available at: <u>http://ezinearticles.com/?Effective-Leadership-Skills---Leaders-Learn-Best-on-the-Job&id=4920012</u>. Date accessed: 28 October 2010.

Cobb, K. 2010. How to get people motivated. *Ezine Articles*. Available at: <u>http://ezinearticles.com/?How-to-GET-People-Motivated&id=4424753</u>. Date accessed: 12 September 2010.

Comsats, H. 2010. Impact of training and development on employee's performance: *a study of general employees*. Available at: <u>http://www.scribd.com/doc/38516457/IMPACT-OF-TRAINING-AND-</u> <u>DEVELOPMENT-ON-EMPLOYEE%E2%80%99S-PERFORMANCE-A-STUDY-OF-</u> <u>GENERAL-EMPLOYEES</u>. Date accessed: 16 October 2010.

Conceição-Runlee, S & Daley, BJ. 2005. *Constructivist theory to web-based course design*. Ser Professor Universitário. Available at: http://www.serprofessoruniversitario.pro.br/ler.php?modulo=9&texto=481. Date accessed: 6 January 2011.

Connor, ML. 2007. How adults learn. *Ageless Learner*. Available at: <u>http://agelesslearner.com/intros/adultlearning.html</u>. Date accessed: 30 October 2010.

Contento, IR. 2011. *Nutrition education: linking research, theory and practice*. Sudbury, MA: Jones and Bartlett.

Cook, M & Jaggers, T. 2005. Strategies for staff retention. Available at: <u>www.shorecapexchange.org/documents/Strategies for Staff Retention.pdf</u>. Date accessed: 12 May 2007.

Cooke, WN & Meyer, DG. 2007. The effects of workplace climates on employee performance capacities: A knowledge-based perspective. Paper presented at the

Sloan Industry Studies Annual Conference, 25–27 April, Cambridge. Available at: <u>http://web.mit.edu/sis07/www/cooke.pdf</u>. Date accessed: 1 September 2010.

Corporate Leadership Council. 2003. Literature key findings: succession planning strategies. Available at: <u>http://www.corporateleadershipcouncil.com.</u> Date accessed: 25 November 2006.

Couros, A. 2003. *Innovation, change theory and the acceptance of new technologies: a literature review*. Available at: http://www.educationaltechnology.ca/couros/publication_files/unpublishedpapers/change_theory.pdf. Date accessed: 14 January 2011.

Cronbach, LJ & Meehl, PE. 1955. Construct validity in psychological tests. *Psychological Bulletin* 52:281–302.

Cubucku, Z. 2008. Preferences on internet-based learning environments in studentcentred education. *Turkish Online Journal of Distance Education* 9(4):154–174.

Cummings, SM, Savitz, LA & Konrad, TR. 2001. Reported response rates to mailed physician questionnaires. *Health Services Research* 35(6):1347–1355.

Curtis, K. 1994. *From management goal setting to organizational results: transforming strategies into action*. Westport, CT: Quorum Books

Dada, D. 2006. E-readiness for developing countries: moving the focus from the environment to the users. *The Electronic Journal of Information Systems in Developing Countries* 27(6):1–14. Available at: http://www.ejisdc.org/ojs2/index.php/ejisdc/article/viewFile/219/184. Date accessed: 6 January 2011.

Deci, EL, Connell, JP, & Ryan, RM. 1989. Self-determination in a work organization. *Journal of Applied Psychology* 74:580–590.

Deci, EL & Ryan, RM. 2000. The "what" and "why" of goal pursuits: List of research project topics and materials 340 human needs and the self-determination of behavior. *Psychological Inquiry* 11(4):227–268.

DeFiore, S & DeFiore, C. 2005. Pretend interest and apathy: what to do about them. *Ezine Articles*. Available at: <u>http://ezinearticles.com/?Pretend-Interest-and-Apathy...-</u> What-To-Do-About-Them!&id=29570. Date accessed: 28 November 2010.

De Moraes, MB, Melo, FC, Oliviera, EAAQ & Cabral, AS. 2010. Analysis of technological innovation strategy for small and medium companies of the aeronautical sector. *Journal of Aerospace Technology Management* 2(2):225–236.

Department of Public Service and Administration (DPSA). 2008. *Strategic human resource planning: guideline and toolkit*. Available at: http://www.dpsa.gov.za/hrp_documents.asp. Date accessed: 23 March 2010.

Department of Public Service and Administration (DPSA). 2002. *Guidelines on integrated human resource planning in the Public Service*. Available at: http://www.dpsa.gov.za/hrp_documents.asp. Date accessed: 12 July 2007.

Department of Public Service and Administration (DPSA). 2001. *Electronic government, the digital future: a Public Service IT policy framework*. Available at: http://www.dpsa.gov.za/documents/acts®ulations/frameworks/IT.PDF. Date accessed: 20 April 2005.

Dewhurst, S & FitzPatrick, L. 2007. Why learning is a matter of personal preference. *Strategic Communication Management* 11(3):14.

De Wit, C. 2010. Interview. Director: Human Resource Management, North-West University (NWU), Potchefstroom Campus. Personal Interview, 2 December, Potchefstroom.

Dickinger, A, Arami, M & Meyer, D. 2008. The role of perceived enjoyment and social norm in the adoption of technology with network externalities. *European Journal of Information Systems* 17(1):4–11.

Dresser & Associates. 2008. *The influence of employee satisfaction on patient satisfaction and safety: the case for automated talent management in healthcare.* Halogen Software Performance Management White Paper. Available at: <u>http://www.dresserassociates.com/halogen-whitepapers-performance.asp</u>. Date accessed: 7 January 2011.

Duff, A. 2003. The role of cognitive learning styles in accounting education: developing learning competencies. *Journal of Accounting Education* 22(1):29–52.

Durrheim, K. 2009. Research design, in *Research in practice: applied methods for the social sciences,* edited by Terre Blanche, M, Durrheim, K & Painter, D. Cape Town: University of Cape Town Press.

Durrheim, K & Painter, D. 2009. Collecting quantitative data: sampling and measuring, in *Research in practice: applied methods for the social sciences,* edited by Terre Blanche, M, Durrheim, K & Painter, D. Cape Town: University of Cape Town Press.

Dziuban, C, Moskal, P & Hartman, J. 2005, in *Elements of quality online education: engaging communities,* edited by Bourne, J & Moore, JC. Needham, MA: Sloan Center for Online Education.

Dzubak, CM. 2005. What skills and whose standards: why are students underprepared? The Association for the Tutoring Profession. *Synergy*:1–13. Available at: <u>http://www.jsu.edu/depart/edprof/atp/Synergy_1/Syn_1.pdf</u>. Date accessed: 13 May 2011.

Eble, K. 1988. *The craft of teaching*. San Francisco: Jossey-Bass.

Economist Intelligence Unit (EIU). 2010. *Digital economy rankings 2010: beyond ereadiness*. A report from the EIU. Available at: <u>http://graphics.eiu.com/upload/EIU Digital economy rankings 2010 FINAL WEB.p</u> <u>df</u>. Date accessed: 29 December 2010.

Economist Intelligence Unit (EIU). 2009. *E-readiness rankings 2009: the usage imperative*. A report from the EIU. Available at: <u>http://graphics.eiu.com/pdf/E-readiness%20rankings.pdf</u>. Date accessed: 29 December 2010.

Economic Intelligence Unit (EIU). 2008. *The future of higher education: how technology will shape learning*. A report from the Economist Intelligence Unit, sponsored by the New Media Consortium. Available at: <u>www.nmc.org/pdf/Future-of-Higher-Ed-(NMC).pdf-</u>. Date accessed: 2 November 2010.

Economist Intelligence Unit (EIU). 2007. *The 2007 e-readiness rankings: raising the bar*. A White Paper from the EIU. Available at: http://graphics.eiu.com/files/ad_pdfs/2007Ereadiness Ranking_WP.pdf. Date accessed: 23 August 2007.

Economic Social and Research Council (ESRC). 2008. *Education, globalisation and the knowledge economy: a commentary by the teaching and learning research programme*. Available at: <u>http://www.tlrp.org/pub/documents/globalisationcomm.pdf</u>. Date accessed: 7 November 2010.

EDS. (Sa) *Technology adoption*. Available at: http://EDS.com or <u>http://www.stuart.iit.edu/courses/EBUS514/Summer2004/Classfiles/s%20curve%20i</u> <u>n%20technology adoption.pdf</u>. Date accessed: 19 October 2005.

Efaw, J. 2005. No teacher left behind: how to teach with technology. *EDUCAUSE Quarterly* 28(4). Available at: <u>http://www.educause.edu/EDUCAUSE+Quarterly/EDUCAUSEQuarterlyMagazineVol</u> <u>um/NoTeacherLeftBehindHowtoTeachw/157372</u>. Date accessed: 6 January 2011.

E-learning facilitators (ELF). 2006. The *roles of e-learning facilitators: skills, competencies and training*. Available at: <u>http://conseil-recherche-innovation.net/wp-content/download/ELF/ELF%20-%20facilitators%20roles%20report%20-</u>%20Greta%20du%20Velay.pdf. Date accessed: 6 January 2011.

Elgort, I. 2005. E-learning adoption: bridging the chasm. Paper presented at the Ascilite Conference in Brisbane, Australia, 4–7 December. Available at: <u>http://ascilite.org.au/conferences/brisbane05/blogs/proceedings/20_Elgort.pdf</u>. Date accessed: 27 October 2010.

Elston, T & Ginis, KAM. 2004. The effects of self-set versus assigned goals on exercisers' self-efficacy for an unfamiliar task. *Journal of Sport and Exercise Psychology* 26(3):500–505.

Encarta. 2009. *World English online dictionary*. Available at: <u>http://encarta.msn.com/dictionary_561547817/piece_rate.html</u>. Date accessed: 8 August 2009.

Erasmus, B, Swanepoel, B, Schenk, H, Van der Westhuizen, EJ, & Wessels, JS. 2005. *South African human resource management for the public sector*. Cape Town: Juta.

Errington, E. 2004. The impact of teacher beliefs on flexible learning innovation: some practices and possibilities for academic developers. *Innovations in education and teaching international* 41(1):39–47.

Eshet-Alkalai, Y. 2004. Digital Literacy: a conceptual framework for survival skills in the digital era. *Journal of Educational Multimedia and Hypermedia* 13. Available at: <u>http://www.questia.com/googleScholar.qst?docId=5006612796</u>. Date accessed: 6 January 2011.

European Commission (EC). 2005. E-learning: designing tomorrow's education. Report on the consultation workshop: The "e" for our Universities – Virtual Campus: Organisational Changes and Economic Models. Available at: http://ec.europa.eu/education/archive/elearning/doc/workshops/elearning%20content /workshop_report_en.pdf. Date accessed: 27 October 2010.

Felder, RM. & Henriques, ER. 1995. Learning and teaching styles in foreign and second language education. *Foreign Language Annals* 28(1):21-31.

Feltz, DL, Short, SE & Sullivan, PJ. 2008. *Self-efficacy in sport: research and strategies for working with athletes, teams and coaches*. United States of America: Human Kinetics.

Fester, EL. 2006. Outcomes-based education and training in South African highereducationinstitutions.Availableat:http://www.che.ac.za/documents/d000173/submissions/24_Fester_12-May06.pdf.Date accessed: 31 October 2010.

Fiske, ST, Gilbert, DT & Lindzey, G. 2009. *Handbook of social psychology*. New Jersey: Johan Wiley & Sons.

Floyd, WW. 2009. Ten lessons about being a learner-centered teacher. *The Alban Institute*. Available at: <u>http://www.alban.org/conversation.aspx?id=8112</u>. Date accessed: 17 September.

Focus group. 2010a. Senior managers: South African Social Security Agency (SASSA). Held on 8 April, Emperor's Palace, Johannesburg.

Focus group. 2010b. Senior managers: Department of Agriculture. Held on 16 September, University of the Free State, Bloemfontein.

Focus group. 2010c. Senior managers: Statistics South Africa (Stats SA). Held on 7 October, Burgers Park Hotel, Pretoria.

Focus group. 2010d. Academic employees: North-West University (NWU), Potchefstroom Campus. Held on 10 December, North-West University, Potchefstroom Campus.

Focus group. 2009a. Senior managers: North-West Province Premier's Office. Held on 16 July, The Willows, Potchefstroom.

Focus Group. 2009b. Senior managers: North-West Province Premier's Office. Held on 27 October, Elgro Hotel, Potchefstroom.

Focus group. 2008a. Senior managers: National Intelligence Agency (NIA). Held on 13 March. Silver Lakes, Pretoria.

Focus group. 2008b. Senior managers: The South African Secret Service (SASS). Held on 27 March, Silver Lakes, Pretoria.

Focus group. 2008c. Senior managers: Western Cape Province Premier's Office. Held on 12 June, Kromme Rhee, Stellenbosch.

Focus group. 2008d. Senior managers: Various national and provincial government departments. Held on 24 July, Roodevallei, Pretoria.

Focus group. 2008e. Senior managers: Kungwini District Municipality. Held on 18 September, Roodevallei, Pretoria.

Focus group. 2006. Academic employees. Tshwane University of Technology (TUT). Held on 23 November. Pretoria.

Fogarty, RJ & Pete, BM. 2004. *The adult learner: some things we know*. Thousand Oaks, Calif: Corwin Press.

Fryer, L. 2010. *How to motivate employees in a recession*. eHow. Available at: <u>http://www.ehow.com/how 5063962 motivate-employees-recession.html</u>. Date accessed: 12 January 2011.

Fuller, C. 2009. *E-learning: New age distance learning*. Article for Creative Commons Licence. Available at: <u>http://www.articlesbase.com/online-education-</u>

<u>articles/elearning-new-age-distance-learning-988113.html</u>. Date accessed: 16 January 2011.

Gabriel, KF. & Flake, S. 2008. *Teaching unprepared students: strategies for promoting success and retention in higher education*. Sterling, Virginia: Stylus Publishing.

Gagné, M & Deci, EL. 2005. Self-determination theory and work motivation. *Journal of Organizational Behavior* 26:331–362.

Gagné, M, Koestner, R & Zuckerman, M. 2000. Facilitating acceptance of organizational change: the importance of self-determination. *Journal of Applied Social Psychology* 30(9):1843–1852.

Gallagher, SP. 2003. Competency-based performance improvement model. *Leadership Coach*. Available at: <u>http://www.leadershipcoach.us/competency.html</u>. Date accessed: 16 October 2010.

Garavalia, LS & Gredler, ME. 2002. An exploratory study of academic goal setting, achievement calibration and self-regulated learning. *Journal of Instructional Psychology* 29(4):221–230.

Garavan, TN & McGuire, D. 2001. Competencies and workplace learning: some reflections on the rhetoric and the reality. *Journal of Workplace Learning* 13(4):144–163.

Generation Y and the Workplace Annual Report. 2010. Issued by Johanson Controls: Global Workplace Innovation. Available at: <u>http://www.johnsoncontrols.com/publish/us/en/search.html</u>. Date accessed: 23 November 2010.

Geoghegan, WH. 1994. Whatever happened to instructional technology? Paper presented at the 22nd annual conference of the International Business Schools Computing Association, Baltimore, Maryland, 17–20 July.

Gergen, KJ. (Sa) Constructivist Epistemology. Available at: <u>http://www.ucs.mun.ca/~emurphy/stemnet/cle2.html</u>. Date accessed: 7 December 2010.

Gibb, S. 2003. Line manager involvement in learning and development: small beer or big deal? *Employee Relations* 25(3):281–293.

Gilbert, C, De Winne, S & Sels, L. 2010. *The influence of line managers and HR department on employees' affective commitment*. Open access publications from Katholiekhe Universiteit, Leuven. Available at: https://lirias.kuleuven.be/bitstream/123456789/271894/1/MO_1004.pdf. Date accessed: 7 April 2011.

Gilley, J, Eggland, S, & Gilley, A. 2002. *Principles of human resource development*. New York: Basic Books.

Gitman, LJ & McDaniel, C. 2008. *The future of business: the essentials*. 4th edition. Mason, OH: South-Westen Cengage Learning.

Gleeson, PB. 2003. Managing and motivating the generations: implications for the student and the employee. Paper presented at a Combined Sections meeting, Tampa, Florida, 12–16 February. Available at: http://www.uwsp.edu/education/facets/links_resources/4413.pdf. Date accessed: 12 March 2011.

Gonsalves, A. 2006. Gen Y taking technology to new level. Information Week.Availableat:http://www.informationweek.com/news/global-cio/showArticle.jhtml?articleID=191600959. Date accessed: 14 November 2010.

Graham, CR. 2004. Blended learning systems: definition, current trends, and future directions, in *Handbook of blended learning: global perspectives, local designs*, edited by Bonk, CJ & Graham, CR. San Francisco, Calif: Pfeiffer Publishing.

Greenhaus, JH, Callanan, GA & Godshalk, VM. 2010. *Career management.* 4th edition. Thousand Oaks, Calif: Sage

Grobler, PA, Wärnich, S, Carrell, MR, Elbert, NF & Hatfield, RD. 2002. *Human resource management in South Africa.* 2nd edition. Cornwall: Thompson.

Guay, F, Ratelle, CF, Senécal, C, Larose, S & Deschênes, A. 2006. Distinguishing developmental from chronic career indecision: self-efficacy, autonomy, and social support. *Journal of Career Assessment* 14(2):235–251.

Guild, PB. 2001. Diversity, learning style and culture: new horizons for learning. Available at: <u>http://www.newhorizons.org/strategies/styles/guild.htm</u>. Date accessed: 1 September 2010.

Gulati, S. 2008. Compulsory participation in online discussions: is this constructivism or normalisation of learning? *Innovations in Education and Teaching International* 45(2):183–192.

Gunderman, RB, Williamson, KB, Frank, M, Heitkamp, DE & Kipfer, HD. 2003. Learner-centred education. *Radiology* 227:15–17. Available at: <u>http://radiology.rsna.org/content/227/1/15.full.pdf+html</u>. Date accessed: 2 January 2011.

Guthridge, M Komm, AB & Lawson, E. 2006. The people problem in talent management. *McKinsey Quarterly* 00475394 (2):1–3.

Halse, ML & Mallinson, B.J. 2008. Investigating popular internet applications as supporting e-learning technologies for teaching and learning with Generation Y. Paper presented at the Emerge Conference, held at the University of Cape Town, 7–18 July.

Hanley, S. 1994. *On constructivism*. Available at: www: <u>http://www.inform.umd.edu/UMS+State/UMD-</u> projects/MCTP/Essays/Constructivism.txt. Date accessed: 7 December 2010. Hansen, RS. 2010. Perception vs reality: 10 truths about the Generation Y workforce. *Quintessential Careers*. Available at: <u>http://www.quintcareers.com/Gen-Y workforce.html</u>. Date accessed: 14 November 2010.

Hansen, RS & Hansen, K. 2010. What do employers *really* want? Top skills and values employers seek from job-seekers. *Quintessential Careers*. Available at: <u>http://www.quintcareers.com/job_skills_values.html</u>. Date accessed: 8 September 2010.

Hansson, M. 2009. *Being a manager in Sweden*. Going Global. Available at: http://old.goinglobal.com/hot_topics/sweden_hansson_manager.asp. Date accessed: 7 January 2011.

Harasim, L. 2000. Shift happens: online education as a new paradigm in learning. *Internet and Higher Education* 3:41–61.

Hase, S & Kenyon, C. 2001. *From andragogy to heutagogy*. ultiBASE Articles. Available at: <u>http://ultibase.rmit.edu.au/Articles/dec00/hase2.htm</u>. Date accessed: 10 January 2011.

Heathfield, SM. 2010a. Performance management checklist: step-by-step to a performance management system. *About.com: Human Resources*. Available at: <u>http://humanresources.about.com/od/performancemanagement/a/perfmgmt.htm</u>. Date accessed: 8 September 2010.

Heathfield, SM. 2010b. The new roles of the human resources professional. *About.com: Human Resources*. Available at: <u>http://humanresources.about.com/od/performancemanagement/a/perfmgmt.htm</u>. Date accessed: 8 September 2010.

Herselman, M & Britton, KG. 2002. Analysing the role of ICT in bridging the digital divide amongst learners. *South African Journal of Education* 22(4):270–274.

V=VI_List of research project topics and materials

Hewitt, E. 2003. E-Readiness and capacity building for ICT application. Paper presented at the United Nations Department of Economic and Social Affairs (UNDESA)/Caribbean Centre for Development Administration (CARICAD), 4th Caribbean Ministerial Consultation and High Level Workshop on Public Sector Management and Strategies for E-Government, Grenada, January 14–15. Available at: <u>http://unpan1.un.org/intradoc/groups/public/documents/caricad/unpan007692.pdf</u>. Date accessed: 6 January 2011.

Hicks, R & McCracken, J. 2009. Mentoring vs coaching: do you know the difference? *Physician Executive* July–August:71–73.

Higher Education Management and Information System (HEMIS). 2010. *Headcount enrolment in public higher education, by age*. Department of Higher Education and Training. Provided by Mr H de Beer, HEMIS.

Hoffman, E. 2010. Technology-enabled learning. *The National Aeronautics and Space Administration (NASA) Ask Magazine* 39:4–5. Available at: <u>http://search.nasa.gov/search/search.jsp?nasaInclude=technology+enabled+learnin</u> g. Date accessed: 16 November 2010.

Honey, P & Mumford, A. 1982. *The manual of learning styles*. Maidenhead: Peter Honey.

Hootstein, E. 2002. Wearing four pairs of shoes: the roles of e-learning facilitators, in *Proceedings of World Conference on E-Learning in Corporate, Government, Healthcare, and Higher Education,* edited by Driscoll, M & Reeves, T. Chesapeake, VA: AACE:457–462

Hrastinski, S. 2008. Asynchronous and synchronous e-learning. EDUCAUSEQuarterly31(4)(October–December).Availableat:http://www.educause.edu/EDUCAUSE+Quarterly/EDUCAUSEQuarterlyMagazineVolum/AsynchronousandSynchronousELea/163445.Date accessed: 6January 2011.
Huang, H. 2002. Toward constructivism for adult learners in online learning environments. *British Journal of Educational Technology* 33(1):27–37.

Hung, LNQ. 2006. Using SWOT analysis to understand the institutional environments: a guide for Can Tho university. *Essays in Education* 18 (Fall 2006):1–11. Available at: <u>http://www.usca.edu/essays/vol18fall2006.html</u>. Date accessed: 23 November 2010.

Hunsaker, PL. 2001. Training in management skills. New Jersey: Prentice Hall.

IBM Corporation. 2006. Improving the accessibility of government, education and healthcare. Report by the IBM Human Ability and Accessibility Centre. Available at: http://www-03.ibm.com/able/dwnlds/public_sector_accessible.pdf. Date accessed: 4 January 2011.

Ifinedo, P. 2005. Measuring Africa's e-readiness in the global networked company: a nine-country data analysis. *International Journal of Education and Development using Information Communication Technology (IJEDICT)* 1(1):53–71.

Information Technologies Group. 2000. *Readiness for the networked world. A guide for developing countries*. Center for International Development (CID). Harvard University. Available online: <u>http://cyber.law.harvard.edu/readinessguide/readiness.html</u>. Date accessed: 14 August 2006.

Ingham, J. 2006. Closing the talent management gap. *Strategic HR Review* 20–23, Melcrum Publishing. Available at: <u>http://www.melcrum.com</u>. Date accessed: 10 July 2007.

International Labour Office (ILO). 2007. Regional skills and employability programme in Asia and the Pacific (SKILLS-AP) Report: Developing skills and employability for young people in Asia and the Pacific. Available at: <u>http://www.ilo.org/public/english/region/asro/bangkok/skills-ap/docs/young.pdf</u>. Date accessed: 29 July 2010. Internet World Stats. 2009. Usage and population statistics. Available at: <u>http://www.internetworldstats.com/af/za.htm</u>. Date accessed: 29 December 2010.

Iskold, A. 2007. *Rethinking "Crossing The Chasm"*. Available at: <u>http://www.readwriteweb.com/archives/rethinking_crossing_the_chasm.php</u>. Date accessed: 13 March 2009.

Ivancevich, JM. 2004. *Human resource management*. 9th edition. New York: McGraw Hill.

Jackson, B. 2007. Focusing on the positive. *Tactics* August:25.

Jackson, CJ, Furnham, AF, Forde, LD & Cotter, T. 2000. The dimensional structure of the Eysenck Personality profiler. *British Journal of Psychology* 91:223–239.

Jackson, C & Lawty-Jones, M. 1996. Explaining the overlap between personality and learning style. *Personality individual differences* 20(3):293–300.

Jacobs, RL & Washington, C. 2003. Employee development and organizational performance: a review of literature and directions for future research. *Human Resource Development International* 6(3):343–354.

Johannes, H. 2007. In search of the latent structure of an e-learning practitioner construct. PhD thesis. University of Pretoria, Pretoria.

Joint Information Systems Committee (JISC). 2010. *Developing students' digital literacy to give them the best chance of success*. Available at: <u>http://www.jisc.ac.uk/supportingyourinstitution/cribsheets/digilit.aspx#</u>. Date accessed: 6 January 2011.

Joint Information Systems Committee (JISC). 2009. *Effective practice in a digital age*. Available at: http://www.jisc.ac.uk/media/documents/publications/effectivepracticedigitalage.pdf. Date accessed: 31 December 2010.

Jonassen, DH. 2006. A constructivist's perspective on functional contextualism. *Educational Technology, Research and Development* 54(1):43–47.

Jonassen, DH. 1991. Objectivism vs constructivism. *Educational Technology Research and Development* 39(3):5–14.

Jordan, C. 2006. Building motivation into online education. Distance Education and E-learning White Paper. ITEC Department of Instructional Technologies at San Francisco State University (SFSU). Available at: <u>http://itec.sfsu.edu/wp/860wp/F06_860_jordan_motivation.pdf?q=5-education-</u> <u>building-people</u>. Date accessed: 1 September 2010.

Judhi, N, Pa'Wan, F, Othman, NA & Moksin, H. 2010. Factors influencing internal and external employability of employees. *Business and Economics Journal* (BEJ-11):1–10. Available at: <u>http://astonjournals.com/manuscripts/Vol2010/BEJ-11_Vol2010.pdf</u>. Date accessed: 15 September 2010.

Juutinen, S & Saariluoma, P. 2009. Emotional obstacles for e-learning: a user psychological analysis. *European Journal of Open, Distance and E-learning.* Available at: <u>http://www.eurodl.org/?article=402</u>. Date accessed: 11 September 2010.

Kahn, B. 2005. *Managing E-learning strategies: design, delivery, implementation and evaluation*. London: Information Science Publishing.

Kaner, S, Lind, L & Toldi, C. 2007. *Facilitator's guide to participatory decisionmaking*. San Francisco: Jossey-Bass.

Kaniki, AM. 2009. Doing an information search, in *Research in practice: applied methods for the social sciences,* edited by Terre Blanche, M, Durrheim, K & Painter, D. Cape Town: University of Cape Town Press.

Kanjee, A. 2009. Assessment research, in *Research in practice: applied methods for the social sciences,* edited by Terre Blanche, M, Durrheim, K & Painter, D. Cape Town: University of Cape Town Press.

Kanuka, H. 2006. Instructional design and e-learning: a discussion of pedagogical content knowledge as a missing construct. *e-Journal of Instructional Science and Technology* 9(2):1–17. Available at: <u>http://www.ascilite.org.au/ajet/e-jist/docs/vol9_no2/papers/full_papers/kanuka.pdf</u>. Date accessed: 31 October 2010.

Keller,H.(Sa).HelenKellerquotes.Availableat:http://www.brainyquote.com/quotes/authors/h/helen_keller_3.html. Date accessed:11 December 2010.

Kelly, K. 2009. From an encounter to text: collecting data in qualitative research, in *Research in practice: applied methods for the social sciences,* edited by Terre Blanche, M, Durrheim, K & Painter, D. Cape Town: University of Cape Town Press.

Kezi Communications. 2009. Generation Y still keeps marketers guessing. *Bizcommunity.com: Daily Media, Marketing and Advertising News*, 18 June. Available at: <u>http://www.bizcommunity.com/Article/196/347/37031.html</u>. Date accessed: 21 August 2009.

Khosrow-Pour, M. 2006. *Emerging trends and challenges in information technology management*. Hershey, PA: Idea Group Publishing.

Kim, K & Bonk, CJ. 2006. The future of online teaching and learning in higher education. *Educause Quarterly* 4:22–30.

Kinchin, IM. 2004. Investigating students' beliefs about their preferred role as learners. *Educational Research* 46(3):301–312.

Kleiman, LS. 2011. Human Resource Management. Reference for Business, *Encyclopaedia for business*. 2nd edition. Available at:

http://www.referenceforbusiness.com/management/Gr-Int/Human-Resource-Management.html. Date accessed: 3 January 2011.

Knowles, MS, Holton III, EF & Swanson, RA. 2005. *The adult learner: the definitive classic in adult education and human resource development*. 6th edition. San Diego, Calif: Elsevier.

Kosak, L, Manning, D, *D*obson, E, Rogerson, L, Cotnam, S, Colaric, S & McFadden, C. 2004. Prepared to teach online? Perspectives of Faculty in the University of North Carolina System. *Online Journal of Distance Learning Administration* 7(3) Fall.

Kottolli, A. 2008. Barriers to e-learning. *Blog: thoughts on marketing, innovation and leadership*. Available at: <u>http://arunkottolli.blogspot.com/2008/09/barriers-to-e-learning.html</u>. Date accessed: 23 August 2009.

Krueger, RA & Casey, MA. 2009. *Focus groups: A practical guide for applied research*. Thousand Oaks, CA: Sage Publications.

Kundu, SC, Malhan, D & Kumar, P. 2007. Human resource management practices in shipping companies: a study. *Delhi Business Review* 8(1):75–88.

Kuvaas, B. 2009. A test of hypotheses derived from self-determination theory among public sector employees. *Employee Relations* 31(1):39–56.

Kuvaas, B. 2006. Performance appraisal satisfaction and employee outcomes: mediating and moderating roles of work motivation. *International Journal of Human Resource Management* 17(3):504–522.

Kwache, PZ. 2007. The imperatives of information and communication technology for teachers in Nigeria higher education. *Journal of Online Teaching and Learning* 3(4):395–399.

Lam, CF & Gurland, ST. 2008. Self-determined work motivation predicts job outcomes, but what predicts self-determined work motivation? *Journal of Research in Personality* 42:1109–1115.

Landes, L. 2006. Getting the best out of people in the workplace. *The Journal for Quality and Participation* Winter:27–29.

Latham, GP. 2007. *Work motivation: history, theory, research and practice*. Thousand Oaks: Sage.

Latham, GP & Locke, EA. 1979. Goal setting: a motivational technique that works. *Organizational dynamics* Autumn:68–80.

Latham, GP & Locke, EA. 2006. Enhancing the benefits and overcoming the pitfalls of goal setting. *Organizational Dynamics* 35(4):332–340.

Lee, B. 2007. Compensation incentives that attract and retain employees. *Canadian Consulting Engineer Magazine*. Available at: http://www.docstoc.com/docs/24692275/Compensation-Incentives-That-Attract-and-Retain-Employees. Date accessed: 16 December 2010.

Lee, CH & Bruvold, NT. 2003. Creating value for employees: investment in employee development. *International Journal of Human Resource Management* 14(6):981–1000.

Lee-Post, A. 2009. E-learning success model: an information systems perspective. *Electronic Journal for e-Learning* 7 (1, May):61–70. Available at: <u>http://www.ejel.org/Volume-7/v7-i1/Lee-Post.pdf</u>. Date accessed: 26 November 2009.

Le Roux, K. 2009. Section Head: Information Technology Support, North-West University, Potchefstroom Campus. Interview, 1 September, Potchefstroom.

Leask, C. 2005. Coaching and mentoring: bridging the gap to company profitability.MediaReleaseGIMT,October.<u>Available</u>at:http://66.102.9.104/search?q=cache:IW6cfwhtU80J:www.gimt.co.za/articles/017-newcoach.pdf+mentoring+and+coaching&hl=en&gl=za&ct=clnk&cd=16.Dateaccessed: 23 October 2006.

Levine, SJ. 2001. The challenge of helping adults learn: characteristics of adult learners and implications for teaching technical information. Paper presented at a Learner Associates workshop. Available at: <u>http://www.learnerassociates.net/workshop/adltlrn1.pdf</u>. Date accessed: 30 October 2010.

Lewis, RE & Heckman, RJ. 2006. Talent management: a critical review. *Human Resource Management Review* 16(2):139-154.

Li, A & Butler, AB. 2004. The effects of participation in goal setting and goal rationales on goal commitment: an exploration of justice mediators. *Journal of Business and Psychology* 19(1):37–51.

Lincecum, L. 2000. The effects of software disruption on goal commitment, task selfefficacy, computer self-efficacy, and test performance in a computer-based instructional task. PhD thesis. Texas Tech University.

Liu, A. 2010a. Managing behavioral styles. *Lab Manager Magazine*. Available at: <u>http://www.labmanager.com/mtips.asp?ID=122</u>. Date accessed: 4 January 2011.

Liu, Y. 2010b. Social media tools as a learning resource. *Journal of Educational Technology Development and Exchange* 3(1):101–114.

Liu, Y. 2007. A comparative study of learning styles between online and traditional students. *Journal of Educational Computing Research* 37(1):41–63.

Liu, CL, Wang, CK, Tan, OS, Koh, C & Ee, J. 2009. A self-determination approach to understanding students' motivation in project work. *Learning and Individual Differences* 19:139–145.

Llorens, S, Salanova, M, & Grau, R. 2002. Training to technological change. *Journal of Research on Technology in Education* 35(2):206–213.

Locke, EA & Latham, GP. 2002. Building a practically useful theory of goal setting and task motivation: a 35-year odyssey. *American Psychologist* 57(9):705–717.

Locke, EA. & Latham, GP. 1990. Work motivation and satisfaction: light at the end of the tunnel. *American Psychological Society* 1(4):240–246.

Lock, EA, Latham, GP & Erez, M. 1988. The determinants of goal commitment. *Academy of Management Review* 13(1):23–39.

Locke, EA & Latham, GP. 1984. *Goal setting: a motivational technique that works*. New Jersey: Prentice Hall.

Luthans, F & Kreitner, R. 1975. *Organisational behaviour modification*. Glenview, III: Scott Foresman.

Luthans, F & Kreitner, R. 1985. *Organizational behavior modification and beyond: an operant and social learning approach*. Glenview, Illinois: Scott Foresman.

Lycette, B & Herniman, J. 2008. New goal-setting theory. *Industrial Management* September:25–30.

Lynch, M. 2010. Basic needs and well-being: a self-determination theory view. Paper as part of an annual VISTAS project, sponsored by the American Counselling Association. Available at: <u>http://counselingoutfitters.com/vistas/VISTAS_Home.htm</u>. Date accessed: 13 January 2011.

Machado, C. 2007. Developing an e-readiness model for HEIs: results of a focus group study. *British Journal of Educational Technology* 38(1):72–82.

Madden, LE. 1997. Motivating students to learn better through own goal-setting. *Education* 117(3):411–414.

Magolda, MB & Terenzini, PT. 2010. Learning and teaching in the 21st century: trends and implications for practice. *Ser Professor Universitário*. Available at: <u>http://www.serprofessoruniversitario.pro.br/ler.php?modulo=10&texto=1525</u>. Date accessed: 14 November 2010.

Magoulas, GD & Chen, S. 2006. Editorial: human factors in personalised systems and services. *Interacting with computers* 18:327–330.

Malhotra, Y & Galetta, DF. 1999. Extending the technology acceptance model to account for social influence: theoretical bases and empirical validation. Proceedings of the 32nd Hawaii International Conference on System Sciences, 5–8 January, Maui, Hawaii.

Markkula, M. 2006. Creating favourable conditions for knowledge society through knowledge management, e-Governance and e-Learning. Paper presented as a keynote presentation at the FIG Workshop on e-Governance, Knowledge Management and e-Learning in Budapest, Hungary, 27–29 April. Available at: <u>http://www.fig.net/pub/monthly_articles/june_2006/june_2006_markkula.pdf</u>. Date accessed: 22 November 2010.

Markland, D, Ryan, RM, Tobun, VJ & Rollnick, S. 2005. Motivational interviewing and self-determination theory. *Journal of Social and Clinical Psychology* 24(6):811–831.

Marshall, M & Harrison, S. 2005. It's about more than money: financial incentives and internal motivation. *Quality Safety and Health Care* 14:4–5.

Maslow, AH. 1946. *Motivation and personality*. 3rd edition. New York: Harper & Row. List of research project topics and materials 360 Mason, W & Watts, DJ. 2009. Financial incentives and the "performance of crowds". Paper presented at the Association for Computing Machinery (ACM) Special Interest Group on Knowledge Discovery and Data Mining (SIGKDD) Human Computation Workshop, 28 June, Paris, France. Available at: <u>http://portal.acm.org/citation.cfm?id=1600150&picked=prox&CFID=2329892&CFTO</u> KEN=70288777. Date accessed: 16 December 2010.

Matei, GS, Bernau, M, Heyworth, F, Pohl, U & Wright, T. 2007. *First steps in teacher training: a practical guide*. European Centre for Modern Languages, Council of Europe Publishing. Available at: http://archive.ecml.at/mtp2/publications/C7_TrainED_E_internet.pdf. Date accessed: 10 January 2011.

Matthias, H, Schneckenberg, D & Wildt, J. 2001. Online pedagogy: innovative teaching and learning strategies in ICT environments. Background paper of the Collaborative European Virtual University (CEVU) workgroup: Online pedagogy. Available at:

http://www.europace.org/articles%20and%20reports/WP1_WG7_8_BP.pdf. Date accessed: 7 January 2011.

Maugis, V, Choucri, N, Madnick, SE, Siegel, MD, Gillett, SE, Haghseta, F, Zhu, H & Best, ML. 2005. Global e-readiness: for what? Readiness for e-banking. *Information Technology for Development* 11(4):313–342.

Maurer, TJ, Pierce, HR & Shore, LM. 2002. Perceived beneficiary of employeedevelopment activity: a three-dimensional social exchange model. The Academy ofManagementReview27(3):432–444.

Maurer, T & Weeks, K. 2010. Interior design in practice: case studies of successful business models. New Jersey: John Wiley & Sons.

May, T. 2006. *Social research: issues, methods and process*. New York: Open University Press.

Mayo, A. 2000. The role of employee development in the growth of intellectual capital. *Personnel Review* 29(4): 521–533.

McCauley, C & Wakefield, M. 2006. Talent management in the 21st century: help your company find, develop and keep its strongest workers. *The Journal for Quality and Participation* Winter:4–7.

McConnell International. 2001. Ready? Net. Go! Partnerships leading the global economy. E-readiness Report. Available at: <u>http://www.witsa.org/papers/e-readiness2.pdf</u>. Date accessed: 12 May 2007.

McCrimmon, M. 2007. *Employee development: using coaching skills to develop employees*. Available at: <u>http://www.suite101.com/content/employee-development-a33574</u>. Date accessed: 16 October 2010.

McDonald, R, Harrison, S, Checkland, K, Campbell, SM & Roland, M. 2007. Impact of financial incentives on clinical autonomy and internal motivation in primary care: ethnographic study. *British Medical Journal*:334–1357. Available at: <u>http://www.bmj.com/content/334/7608/1357.full?rss=1</u>. Date accessed: 16 December 2010.

McLeish, A. 2002. Employability skills for Australian small and medium sized enterprises. Commonwealth Department of Science and Training. Available at: <u>http://www.dest.gov.au/NR/rdonlyres/18847ED7-5812-40F5-8F38-</u> 3BAE473B7C88/2216/SME research.pdf. Date accessed: 12 September 2010.

McNamara, C. 2010. Organizational change and development: includes field of organization development. Free Management Library. Available at: <u>http://managementhelp.org/org_chng/org_chng.htm</u>. Date accessed: 27 October 2010.

McNamara, C. (Sa). *Organizational performance management*. Free Management Library. Available at: <u>http://managementhelp.org/org_perf/org_perf.htm</u>. Date accessed: 5 January 2011.

McNulty, DM. 2009. The theory of planned induced apathy. *Ezine Articles*. Available at: <u>http://ezinearticles.com/?The-Theory-of-Planned-Induced-Apathy&id=2726992</u>. Date accessed: 28 November 2010.

Meek, VL & Davies, D. 2009. Policy dynamics in higher education and research: concepts and observations, in *2009 Report on UNESCO Forum on Higher Education Research and Knowledge*. Paris: UNESCO. <u>http://unesdoc.unesco.org/images/0018/001831/183168e.pdf</u>. Date accessed: 21 November 2010.

Mercer Consulting Outsourcing Investments. 2010. Organizations focus on employee engagement to attract and retain top talent: emphasis is back on pay programs. Available at: <u>http://www.mercer.com/press-releases/1383780</u>. Date accessed: 8 September 2010.

Miller, K. 2005. Charisma of an online learning facilitator: do you have it? *California Virtual Campus*. Available at: <u>http://www.cvc.edu/2005/05/charisma-of-an-online-facilitator-do-you-have-it/</u>. Date accessed: 27 November 2009.

Milliken, J & Barnes, LP. 2002. Teaching and technology in higher education: student perceptions and personal reflections. *Computers & Education* 39:223–235.

Mills, J. 2011. *DISC profile*. Available at: <u>http://www.discprofile.com/</u>. Date accessed: 4 January 2011.

Mind Tools. 2010. Schein's career anchors. Available at: http://www.mindtools.com/pages/article/newCDV_99.htm. Date accessed: 31 July 2010.

Mitra, S. 2010. Strategy roundtable: online education startups. *ReadWrite Start*. Available at: <u>http://www.readwriteweb.com/start/2010/06/strategy-roundtable-online-education-startups.php</u>. Date accessed: 11 September 2010.

Moller, AC, Ryan, RM & Deci, L. 2006. Self-determination theory and public policy: improving the quality of consumer decisions without using coercion. *Journal of Public Policy and Marketing* 25(1):104–116.

Momberg, C. 2004. The relationship between personality traits and vocational interests in a South African context. MA dissertation, University of Pretoria, Pretoria.

Moolman, HB. 2007. e-Readiness of warehouse workers: an explorative study. PhD thesis in Education (submitted in partial fulfilment). Pretoria, University of Pretoria.

Moolman, HB & Blignaut, S. 2008. Get set! e-ready, ... e-learn! The e-readiness of warehouse workers. *Educational Technology and Society* 11(1):168–182.

Moore, GA. 2002. Crossing the chasm. New York: HarperCollins.

Moore, AH, Moore, JF & Fowler, S. 2011. *Faculty development for the Net Generation*. Available at: http://www.educause.edu/Resources/EducatingtheNetGeneration/FacultyDevelopme http://www.educause.edu/Resources/EducatingtheNetGeneration/FacultyDevelopme http://www.educause.edu/Resources/EducatingtheNetGeneration/FacultyDevelopme http://www.educause.edu/Resources/EducatingtheNetGeneration/FacultyDevelopme http://www.educause.edu/Resources/EducatingtheNetGeneration/FacultyDevelopme http://www.educause.edu/Resources/EducatingtheNetGeneration/FacultyDevelopme

Morrison, M. 2007. Learning styles – questionnaires and instruments. Article of Rapid business improvement (RapidBI), Available at: <u>http://rapidbi.com/created/learningstyles/</u>. Date accessed: 19 May 2008.

Motah, M. 2007. Learning, types of learning, traditional learning styles and the impact of e-learning on the performance of secondary school students: the perception of teachers. Paper presented at the 2007 Computer Science and IT Education Conference, held in Mauritius, 16–18 November. Available at: <u>http://csited.org/2007/70MotaCSITEd.pdf</u>. Date accessed: 11 September 2010.

Mote, D. 2011. Performance appraisal and standards. *Encyclopaedia of Business*. 2nd edition. Available at: <u>http://www.referenceforbusiness.com/encyclopedia/Per-</u><u>Pro/Performance-Appraisal-and-Standards.html</u>. Date accessed: 7 January 2011.

Msimang, Z. 2008. South Africa: Generation Y must ask "why"? (opinion). *Mail & Guardian*, 8 May.

Mucha, RT. 2004. The art and science of talent management. *Organization Development Journal* 22(4):96–100.

Murphy, E. (Sa). *Constructivism: from personal beliefs to theoretical principles*. Available at: <u>http://www.mun.ca/educ/faculty/mwatch/fall02/Murphy.htm</u>. Date accessed: 7 December 2010.

Murphy, TH & Margulies, J. 2004. Performance appraisals. Paper presented at the ABA Labour and Employment Law Section Equal Employment Opportunity Committee, Mid-Winter Meeting, 24–27 March.

Mutula, SM. 2006. An assessment of e-readiness of small and medium sized enterprises in the ICT sector in Botswana, with special reference to information access. PhD thesis, Johannesburg: University of Johannesburg.

Mutula, SM. 2003. Assessment of Africa's telematics, policy and regulatory infrastructure: potential for e-learning. Paper presented at the conference of the National Tertiary Education Union, held at the University of KwaZulu-Natal, Durban, 9–11 July.

Naidoo, N. 2005. *Managers watch out for "Generation Y*". Available at: <u>http://www.busrep.co.za/index.php?fSectionId=970&fArticleId=2630972</u>. Date accessed: 21 August 2009.

National Aeronautics and Space Administration (NASA). 2007. Mentoring guide.NASA's Foundations of Influence, Relationships, Success and Teamwork (FIRST)Programme.Availableat:

http://www.usfirst.org/uploadedFiles/Community/FRC/Team_Resources/Mentoring% 20Guide.pdf. Date accessed: 12 January 2011.

National Center for Education Statistics (NCES). 2008. Projections of education statistics to 2017. Washington, DC: US Department of Education. Available at: <u>http://nces.ed.gov/pubs2008/2008078.pdf</u>. Date accessed: 6 November 2010.

National Center for Education Statistics (NCES). 2003. National assessment of adult literacy. Washington, DC: US Department of Education. Available at: <u>http://nces.ed.gov/pubs2003/2003495rev.pdf</u>. Date accessed: 6 November 2010.

Navran, FJ. 2010. *A customer service paradox*. Navran Associates Organizational Ethics Compliance and Leadership. Available at: <u>http://www.navran.com/article-customer-service.html</u>. Date accessed: 12 January 2011.

Nel, C. 2010. Head: Student Learning and Reading Development: North-West University (NWU), Potchefstroom Campus. Personal Interview, 14 November 2010, Potchefstroom.

New Zealand Council for Educational Research (NZCER). 2004. Critical success factors and effective pedagogy for e-learning in tertiary education. Background paper for ITP New Zealand. Available at: http://www.minedu.govt.nz/NZEducation/EducationPolicies/TertiaryEducation/PublicationSectors/TertiaryEducation/NZ CERFinalReport.pdf. Date accessed: 15 March 2007.

Nickols, F. 2007. Performance appraisal: weighed and found wanting in the balance. *The Journal for Quality and Participation* Spring:13–16.

Niemiec, CP, Lynch, MF, Vansteenkiste, M, Bernstein, J, Deci, EL & Ryan, RM. 2006. The antecedants and consequences of autonomous self-regulation for college: a self-determination theory perspective on socialization. *Journal of Adolescence* 29:761–775.

Njenga, JK & Fourie, LCH. 2010. The myths about e-learning in higher education. *British Journal of Educational Technology* 41(2):199–212.

North-West University (NWU). 2011. *Progress Report 2011/1*. Teaching and Learning with Technology Task Group. Potchefstroom Campus. 15 January 2011.

North-West University (NWU). 2010a. *Management development programme*. Leadership Academy, Institutional Office, NWU.

North-West University (NWU). 2010b. *Student enrolments for 2010*. Management Information Database: Institutional Office, North-West University.

North-West University (NWU). 2010c. *Memorandum: Teaching and Learning with Technology*. Potchefstroom Campus.

North-West University (NWU). 2010d. NWU@ a glance. Available at: <u>http://www.nwu.ac.za/nwu/glance.html</u>. Date accessed: 4 November 2010.

North-West University (NWU). 2010e. *Student use of information communications technology (ICT) at the Potchefstroom Campus of the North-West University.* Preliminary Report on a survey conducted by Academic Support Services, Potchefstroom Campus, NWU.

North-West University (NWU). 2010f. *Vision, mission and values*. Available at: <u>http://www.nwu.ac.za/export/sites/default/nwu/nwu/2.1.1-vision_e.pdf</u>. Date accessed: 14 September 2010.

North-West University (NWU). 2009a. *Teaching and learning framework*. Available at: <u>https://intranet.nwu.ac.za/en/fileplan/teaching/index.html</u>. Date accessed: 5 February 2010.

North-West University (NWU). 2009b. *Institutional plan, 2010–2012*. Available at: <u>http://www.nwu.ac.za/export/sites/default/nwu/nwu/2.1.2.1-IP-2010-2012.pdf</u>. Date accessed: 12 November 2010.

North-West University (NWU). 2007a. *Teaching and learning policy*. Available at: <u>http://www.nwu.ac.za/export/sites/default/nwu/gov_man/policy/8P-TL_e.pdf</u>. Date accessed: 14 November 2008.

North-West University (NWU). 2007b. *Quality policy*. Available at: <u>http://www.nwu.ac.za/export/sites/default/nwu/gov_man/policy/2P-2.3-quality e.pdf</u>. Date accessed: 16 February 2008.

North-West University (NWU). 2005a. *Performance management policy*. Available at: <u>https://intranet.nwu.ac.za/en/fileplan/human-resources/index.html#6</u>. Date accessed: 14 March 2009.

North-West University (NWU). 2005b. *Staff development policy*. Available at: <u>http://www.nwu.ac.za/export/sites/default/nwu/gov_man/policy/3P-3.4-</u> <u>staff_development_e.pdf</u>. Date accessed: 20 October 2010.

North-West University (NWU). 2005c. *Academic staff promotion policy*. Available at: <u>https://intranet.nwu.ac.za/opencms/export/intranet/html/en/in-im-</u> hro/handleidingdeel4.html. Date accessed: 21 September 2010.

Ntoumanis, N. 2005. A prospective study of participation in optional school physical education using a self-determination theory framework. *Journal of Educational Psychology* 97(3):444–453.

Nykodym, N. 1996. Public sector performance appraisal and effectiveness: a casestudy.PublicPersonnelManagement.Availableat:http://www.allbusiness.com/human-resources/employee-benefits/567661-1.html.Date accessed: 14 September 2008.

Ogrean, C, Herciu, M, & Belascu, L. 2009. Competency-based management and global competencies: challenges for firm strategic management. *International Review of Business Research Papers* 5 (4):114–122.

Oliver, C. 2001. Taking policy governance to the mainstream. *Board Leadership* Jan–Feb:6–8.

O'Neill, M. 2010. Generational preferences. *Knoll Workplace Research*. Available at: <u>http://www.knoll.com/research/downloads/WP_GenerationalDifferences.pdf</u>. Date accessed: 14 November 2010.

Organisation for Economic Co-operation and Development (OECD). 2004. Science, technology and innovation for the 21st century. Meeting of the OECD Committee for Scientific and Technological Policy at Ministerial Level, 29–30 January. Final Communique. Available at: http://www.oecd.org/document/0,2340,en_2649_34487_25998799_1_1_1_0.html Date accessed: 2 November 2010.

Organisation for Economic Co-operation and Development (OECD).1996.knowledge-basedeconomy.Availablehttp://www.oecd.org/dataoecd/51/8/1913021.pdfDate accessed: 8 November 2010.

Oyedele, O. 2010. How managers can motivate their employees. *SciTopics: research summaries by experts*. Available at: <u>http://www.scitopics.com/How Managers can Motivate their Employees.html</u>. Date accessed: 1 September 2010.

Oyedemi, TD. 2009. Social inequalities and the South African ICT access policy agendas. *International Journal of Communication* 3:151–168.

Palade, V, Howlett, RJ & Jain, LC. 2003. Knowledge-based Intelligent Information and Engineering Systems International Conference, Oxford, UK, September. Proceedings, Part II. Available at: <u>http://www.pubzone.org/pages/publications/showVenue.do?venueId=7048</u>. Date accessed: 14 January 2011.

Palloff, RM & Pratt, K. 2000. Making the transition: helping teachers to teach online. Paper presented at the 2000 EDUCAUSE Conference held in Nashville, 10–13 October. Available at: http://net.educause.edu/ir/library/pdf/EDU0006.pdf. Date accessed: 11 September 2010.

Pandor, N. 2007. Address by the Minister of Education at the colloquium on Information and Communication Technology (ICT) education and training, Birchwood Conference Centre. 19 March. Available at: http://www.info.gov.za/speeches/2007/07031914451001.htm. Date accessed: 3 January 2011.

Papasolomou-Doukakis, I. 2002. The role of employee development in customer relations: the case of UK retail banks, Corporate Communications: An International Journal 7(1):62-76.

Park, O & Lee, J. 2001. Adaptive Instructional Systems, in The handbook of research for educational communications and technology, edited by Jonassen, DH. A publication of the Association for Educational Communications and Technology. Available at: http://www.aect.org/edtech/ed1/firstedition.asp. Date accessed: 14 January 2011.

Parliamentary Monitoring Group. 2010. University and college student enrolment, plans to improve throughput rate, registration challenges and interventions, National Student Financial Aid Scheme (NSFAS) and pre-registration fee, discussed at a meeting of the Parliamentary Monitoring Committee, 2 February. Available at: http://www.pmg.org.za/report/20100202-university-college-student-enrolment-plansimprove-throughput-rate-re. Date accessed: 2 November 2010.

Pearsall, J. 2001. New Oxford dictionary of English. Oxford: Oxford University Press.

Pebble, T, Hargraves, H, Leach, L, Naidoo, K, Suddaby, G & Zepke, N. 2005. Impact of student support services and development programmes on student outcomes in undergraduate tertiary study. Report to the Ministry of Education, Wellington, New Zealand. Available at: http://www.educationcounts.govt.nz/ data/assets/pdf file/0013/7321/ugradstudento utcomes.pdf. Date accessed: 19 October 2010. List of research project topics and materials

Perry, JL, Mesch, D & Paarlberg, L. 2006. Motivating employees in a new governance era: the performance paradigm revisited. *Public Administration Review* July/August:505–514.

Perryer, C & Jordan, C. 2005. The influence of leader behaviors on organizational commitment: a study in the Australian public sector. *International Journal of Public Administration* 28:379–396.

Pienaar, S. 2011. Interview with the Information Technology Laboratory manager, North-West University, Potchefstroom campus, 17 January, Potchefstroom.

Powell, S. 2008. The management and consumption of organisational creativity. *Journal of Consumer Marketing* 25(3):158–166.

Prenhall. 2007. *Succession planning*. Web definitions. Available at: <u>wps.prenhall.com/wps/media/objects/213/218150/glossary.html</u>. Date accessed: 10 July 2007.

Price, C. 2009. Why don't my students think I'm groovy? The new "r"s for engaging millennial learners. Paper presented at the Georgia Health Sciences University's Tech Fair, Augusta, Georgia, 30 October.

Princeton University. 2009. *WordNet: a lexical database for English*. Available at: wordnet.princeton.edu/perl/webwn. Date accessed: 3 March 2009.

Proctor, T & Doukakis, I. 2003. Change management: the role of internal communication and employee development. *Corporate Communications* 8(4):268–277.

Public Administration Leadership and Management Academy (PALAMA). 2010. *Strategic human resource management*. Manual for the Executive Development Programme (EDP). PALAMA: Pretoria.

Public Service Commission (PSC). 2010. Assessment of the state of human resource management in the public service. Available at: http://www.psc.gov.za/docs/reports/2010/PSC_March_2010_Review.pdf. Date accessed: 21 March 2010.

Public Service Commission (PSC). 2007. *Toolkit for the management of poor performance in the public service*. Available at: <u>http://www.psc.gov.za/guidelines</u>. Date accessed: 18 March 2010.

Public Service Commission (PSC). 2000. *Report on career management in the public service*. Available at: <u>http://www.psc.gov.za/docs/reports/2000/career</u>. Date accessed: 10 March 2009.

Pugno, M. 2008. Economics and the self: a formalisation of self-determination theory. *The Journal of Socio-Economics* 37:1328–1346.

Pusser, B. 2002. *The knowledge economy and postsecondary education: report of a workshop*. Centre for Education, National Research Council. Washington DC: National Academy Press.

Quantec Research. 2010. *EIU digital economic rankings, 2010*. Available at: <u>http://www.quantec.co.za/aboutus/news/2010071</u>. Date accessed: 29 December 2010.

Rabak, L & Cleveland-Innes, M. 2006. Acceptance and resistance to corporate elearning: a case from the retail sector. *Journal of Distance Education* 21(2):115–134.

Rademan, DJ & De Vos, HD. 2001. Performance appraisals in the public sector: are they accurate and fair? *Journal of Industrial Psychology* 27(1):54–60.

Ramsden, P. 2008. The future of higher education teaching and the studentexperience.Availablewww.heacademy.ac.uk/.../paulramsden_teaching_and_student_experience.doc.Date accessed: 2 November 2010.

Ramsden, P. 2003. *Learning to teach in higher education*. London: RoutledgeFalmer.

Rautenbach, L. 2007. An electronic-learning (e-learning) readiness model for distance education in the workplace. PhD thesis. North-West University, Potchefstroom.

Redman, T & Wilkinson, A. 2001. *Contemporary human resource management: text and cases*. Harlow: Prentice Hall.

Rees, WD & Porter, C. 2003. Appraisal pitfalls and the training implications: part 1. *Industrial and Commercial Training* 35(7):280–284.

Reeve, J. 2009. Reinforcement. Article for *Education.com*. Available at: <u>http://www.education.com/reference/article/reinforcement/</u>. Date accessed: 7 January 2011.

Reeves, TC. 1997. Evaluating what really matters in computer-based education. Available at: <u>http://www.educationau.edu.au/archives/CP/reeves.htm</u>. Date accessed: 24 November 2009.

Reid, S. 2007. Underprepared college students: meeting the challenge in a postsecondary reform environment. *ProQuest Dissertations and Theses*. Available at: <u>http://gradworks.umi.com/32/93/3293562.html</u>. Date accessed: 14 November 2010.

Reid, MA, Barrington, H & Brown, M. 2004. *Human resource development: beyond training interventions*. London: Chartered Institute of Personnel and Development.

Republic of South Africa, vide South Africa (SA).

Research and Markets. 2009. Internet access in South Africa 2010. Brochure ofmarketresearchreport.Availableat:

http://www.researchandmarkets.com/reportinfo.asp?report_id=1280829. Date accessed: 29 December 2010.

Resnick, HS. 2007. Employee performance is leader's responsibility. *Work Systems*. Available <u>http://www.worksystems.com/freeResources/leadershipPrinciples/employee_perform</u> ance.html. Date accessed: 5 January 2011.

Robbins, SP. 2009. *Organisational behaviour: global and South African perspectives*. Cape Town: Pearson Education.

Robbins, S.P. 2003. *Organizational behaviour*. 10th edition. New Jersey: Prentice Hall.

Robertson, I. 2008. Sustainable e-learning, activity theory and professional development. Paper presented at the Australian Society for Computers in Learning in Tertiary Education (Ascilite) Conference, Melbourne, Australia, 30 November to 3 December 2008. Available at: http://www.ascilite.org.au/conferences/melbourne08/procs/robertson.pdf. Date accessed: 24 September 2010.

Roca, JC & Gagné, M. 2008. Understanding e-learning continuance intention in the workplace: a self-determination theory perspective. *Computers in Human Behaviour* 24:1585–1604.

Rockler-Gladen, N. 2006. Generation Y college students. *Suite101*. Available at: <u>http://www.suite101.com/content/generation-y-a7763</u>. Date accessed: 14 November 2010.

Rogers, EM. 2003. Diffusion of innovations. 5th edition. New York: Free Press.

Rogers, EM. 1995. Diffusion of innovations. 4th edition. New York: Free Press.

Rogers, EM. 1962. Diffusion of innovations. 1st edition. Glencoe: Free Press.

Roper, A. R. 2007. How students develop online learning skills. EDUCAUSEQuarterlyMagazine30(1).Availableat:http://www.educause.edu/EDUCAUSE+Quarterly/EDUCAUSEQuarterlyMagazineVolum/HowStudentsDevelopOnlineLearni/157435.Date accessed: 1 September 2010.

Rossiter, DE. 2006. Embedding e-learning in universities: analysis and conceptualisation of change processes. PhD thesis, Queensland University of Technology, Australia, May.

Roy, K. 2006. The impact of learning styles on interactivity in asynchronous e-learning. *Performance Improvement* 45(10):22.

Ryan, RM, & Deci, EL. 2000. Intrinsic and extrinsic motivations: classic definitions and new directions. *Contemporary Educational Psychology* 25(1):54–67.

Rycus, JS & Hughes, RC. 2000. What is competency-based in-service training? Paper presented at the Training, Resource, Advocacy, and Information Network (TRAINet) Forum, Glen Arbor, Michigan, Pennsylvania. Available at: www.narccw.org/TRAINet/Resource%20Paper%201.pdf. Date accessed: 28 October 2010.

Saari, LM & Judge, TA. 2004. Employee attitudes and job satisfaction. *Human Resource Management* 43(4):395–407.

Sakai. 2009. *2.3 Sakai tools*. Available at: <u>https://confluence.sakaiproject.org/display/DOC/2.3+Sakai+Tools</u>. Date accessed: 4 January 2011.

Salmon, G. 2003. *e-Moderating: the key to teaching and learning online*. London: Taylor & Francis Books.

Salmon, G. 2002. *Etivities: The key to active online learning*. Sterling, Va: Stylus.

Sax, LJ, Gilmartin, SK & Bryant, AN. 2003. Assessing response rates and non-response bias in web and paper surveys. *Research in Higher Education* 44(4):409–432.

Schang, SL. 2007. Crossing the technology adoption chasm in the presence of network externalities: implications for DoD. PhD thesis, Naval Posgraduate School, Monterey, California, USA.

Schien, E. 2007. Career anchors revisited: implications for career development in the 21st century. *NHRD Journal* 1(4):27-33.

Schreurs, J, Sammour, G & Ehlers, U. 2008. ERA – E-learning readiness analysis: a e-health case study of e-learning readiness, WSKS 2008, *Communications in Computer and Information Science* 19:267–275.

SchoolNet Africa. 2007. e-Readiness. *African Educational Knowledge Warehouse*. Available at: <u>http://www.schoolnetafrica.org/english/policy_centre/e-readiness.html</u>. Date accessed: 17 April 2007.

SchoolNet Africa. 2008. *e-Readiness as a tool for ICT development*. Available at: <u>http://www.schoolnetafrica.org/fileadmin/resources/E-readiness as a tool.pdf</u>. Date of access: 16 April 2008.

Schuler, RS. 1981. Personnel and human resource management. St Paul: West.

Selden, SC & Brewer, GA. 2000. Work motivation in the senior executive service: testing the high performance cycle theory. *Journal of Public Administration Research and Theory* 10(3):531–550.

Senge, P. 1990. *The fifth discipline: the art and practice of the learning organization*. London: Random House.

Shadare, OA & Hammed, TA. 2009. Influence of work motivation, leadership effectiveness and time management on employees' performance in some selected

industries in Ibadan, Oyo State, Nigeria. *European Journal of Economics, Finance and Administrative Sciences* 16:7–16.

Shah, N & Irani, Z. 2010. Examining employee attitudes and behaviours towards organisational change using supervisor and peer relations. Paper presented at the European, Mediterranean and Middle Eastern Conference on Information Systems, April, Abu Dhabi, United Arab Emirates.

Shamir, B. 1996. Meaning, self and motivation in organizations, in *Motivation and leadership at work.* 6th edition, edited by Steers, RM, Porter, LW & Bigley, GA. New York: McGraw-Hill:149–165.

Sharpe, R, Benfield, G & Francis, R. 2006. Implementing a university e-learning strategy: levers for change within academic schools. *ALT-J Research in learning technology* 14(2):135–151.

Shellabear, S. 2002. Competency profiling: definition and implementation. TrainingJournal–August2002.Availableat:www.dancinglion.com/mediabank/assets/.../Competency profiling.pdf.Dateaccessed: 16 October 2010.

Shelton, CD, McKenna, MK & Darling, JR. 2002. Leading in the age of paradox: optimizing behavioral style, job fit and cultural cohesion. *Leadership and Organization Development Journal* 23(7):372–379.

Shen, D, Laffey, J, Lin, Y & Huang, X. 2006. Social influence for perceived usefulness and ease-of-use of course delivery systems. *Journal of Interactive Online Learning* 5(3):270–282.

Sieber, L. 2008. The employability paradox. MBA dissertation, Pretoria, University of Pretoria. Available at: <u>http://upetd.up.ac.za/thesis/available/etd-03162010-160056/unrestricted/dissertation.pdf</u>. Date accessed: 10 January 2011.

Simons, PR. 2008. Dimensions of on-the-job learning styles. *Applied Psychology: An International Review* 57(3):417–440.

Šiugždinienė, J. 2008. Line manager involvement in human resource development. *Viešoji Politika Ir Administravimas* 25:32–37. Available at: <u>http://www.ktu.lt/lt/mokslas/zurnalai/vpa/vpa25/VPA_Nr.25_J.Siugzdiniene_p.32-</u> 37.pdf. Date accessed: 27 October 2010.

Sivo, SA, Saunders, C, Chang, Q & Jiang, JJ. 2006. Research methodology: literature review on research methodology for social and design science. Blog with excerpt from article: how low should you go? Rates and the validity of inference in IS questionnaire research. *Journal of the Association for Information Systems* 7(6). Available at: <u>http://doresearch.wordpress.com/2008/08/12/%E2%80%9Chow-low-should-you-go-low-response-rates-and-the-validity-of-inference-in-is-questionnaire-research%E2%80%9D/</u>. Date accessed: 15 May 2011.

Smith, GP. 2010. *Management and human resources: transform your organization from high turnover to high retention*. Business know-how. Available at: https://www.businessknowhow.com/manage/retain-employees.htm. Date accessed: 8 September 2010.

Smith, MK. 2003. Learning theory. *Ideas, thinkers, practice (Infed)*. Available at: <u>http://www.infed.org/biblio/b-learn.htm</u>. Date accessed: 2 November 2010.

Smith, MK. 2001. Peter Senge and the learning organization. *Infed Articles*. Available at: <u>http://www.infed.org/thinkers/senge.htm</u>. Date accessed: 10 January 2011.

Smith, MK. 2000. Curriculum theory and practice. *The encyclopaedia of informal education*. Available at: <u>http://www.infed.org/biblio/b-curric.htm</u>. Date accessed: 6 January 2011.

Smyre, R. 2006. On searching for new genes: a 21st century DNA for highereducation.NewHorizonsforLearning.Availableat:

http://www.marthalakecov.org/~building/future/smyre4.htm. Date accessed: 6 January 2011.

Snieder, R & Larner, K. 2009. *The art of being a scientist: a guide for graduate students and their mentors.* New York: Cambridge University Press.

Song, L, Singleton, ES, Hill, JR & Koh, MH. 2004. Improving online learning: student perceptions of useful and challenging characteristics. *Internet and Higher Education* 7:59–70.

South Africa (SA). 2004a. *National education information policy*, 7 September (Notice 1950 of 2004). Pretoria: Government Printers.

South Africa (SA). 2004b. *Draft White Paper on e-Education*, 26 August (Notice 1869 of 2004). Cape Town: Formeset Printers.

South Africa (SA). Skills Development Act 97 of 1998.

South Africa (SA). 1997. *Human Resource Management in the Public Service White Paper*, 3 December (Notice 18594 of 1997). Pretoria: Government Printer.

South Africa (SA). Higher Education Act 101 of 1997.

South Africa (SA). Labour Relations Act 66 of 1995.

South African Government Information. 2010. *Education*. Available at: <u>http://www.info.gov.za/aboutsa/education.htm</u>. Date accessed: 6 January 2011.

Stajkovic, AD & Luthans, F. 1997. A meta-analysis of the effects of organisational behaviour modification on task performance. *Academy of Management Journal* 40:1122–1149.

Steyn, M, Badenhorst, J & Kamper, G. 2010. Our voice counts: adolescents' view on their future in South Africa. *South African Journal of Education* 30:169–188.

Steyn, GM & Van Niekerk, EJ. 2002. *Human resource management in education*. Paarl: Paarl Print.

Stockley, D. 2003. What is e-learning? *e-Learning Solutions*. Available at: <u>http://derekstockley.com.au/elearning-definition.html</u>. Date accessed: 24 August 2009.

Stolovitch, HD, Clark, RE & Condly, SJ. 2002. *Incentives, motivation and workplace performance: research and best practices*. Research conducted for the International Society for Performance Improvement (ISPI) and the Incentive Research Foundation (IRF). Available at: <u>http://www.incentivecentral.org/pdf/Workplace-IRFfinal.pdf</u>. Date accessed: 11 January 2011.

Stone, NJ. 2008. Human factors and education: evolution and contributions. *The Journal of Human Factors and Ergonomic Society* (50):534–539. Available at: <u>http://hfs.sagepub.com/content/50/3/534.full.pdf+html</u>. Date accessed: 4 January 2011.

Sturges, J, Conway, N, Guest, D & Liefooghe, A. 2005. Managing the career deal: the psychological contract as a framework for understanding career management, organizational commitment and work behavior. *Journal of Organizational Behavior* 26(7):821–838.

Sun, H & Zhang, P. 2006. Causal relationships between perceived enjoyment and perceived ease of use: an alternative approach. *Journal of the Association for Information Systems* 7(9):618–645.

Sutton, C. 2004. Online learning: does it pass the taste test? Australian FlexibleLearningCommunity.Availableat:http://community.flexiblelearning.net.au/TeachingTrainingLearners/content/article_6893.htm.Date accessed: 1 April 2009.



Swinton, L. 2006. Kolb's learning style inventory and Kolb's learning cycle explained: no fluff, no filler, just facts. Available at: <u>http://www.mftrou.com/kolb-learning-style-inventory.html</u>. Date accessed: 12 October 2006.

Switser, P. 2010. Managing behavioral styles. *The Bookkeepers Zone*. Available at: <u>http://www.thebookkeepers.co.za/articles/managing-behavioural-styles/</u>. Date accessed: 4 January 2011.

Tadmore, Z. 2006. The Triad Research University or a post 20th century research university model. *Higher Education Policy* 19:287–298.

Takalani, T. 2008. Barriers to e-learning amongst postgraduate black students in higher education in South Africa. MEd dissertation, Stellenbosch, University of Stellenbosch.

TalentAlign. 2007. *Talent management processes in organisations*. Available at: http://www.talentalign.com/talent_management_processes_in_organization.htm. Date accessed: 9 July 2007.

Taljard, M. 2010. Head: Writing Laboratory. Centre for Academic and Professional Language Practice, North-West University (NWU), Potchefstroom campus. Personal Interview, 4 November, Potchefstroom.

Tatum, M. 2010. What is job competency? *Wise Geek*. Available at: <u>http://www.wisegeek.com/what-is-job-competency.htm</u>. Date accessed: 16 October 2010.

Terre Blanche, M & Durrheim, K. 2009. Histories of the present: social science research in context, in *Research in practice: applied methods for the social sciences,* edited by Terre Blanche, M, Durrheim, K & Painter, D. Cape Town: University of Cape Town Press.

Terre Blanche, M, Kelly, K & Durrheim, K. 2009. Why qualitative research? in *Research in practice: applied methods for the social sciences,* edited by Terre Blanche, M, Durrheim, K & Painter, D. Cape Town: University of Cape Town Press.

Thansingam, S & Soong, SKA. 2007. Interaction patterns and knowledge construction using synchronous discussion forums and video to develop oral skills. Paper presented at the Australian Society for Computers In Learning In Tertiary Education (ASCILITE), 2–5 December, Singapore. Available at: http://www.ascilite.org.au/conferences/singapore07/procs/thanasingam.pdf. Date accessed: 6 January 2011.

The Computer Language Company Inc. 2010. *Technology adoption life cycle definition*. The CBS Interactive Business Network. Available at: <u>http://www.bnet.com/topics/technology+adoption+life+cycle</u>. Date accessed: 5 January 2011.

The online encyclopedia. (Sa). Questionnaire. Available at: http://www.encyclo.co.uk/define/questionnaire. Date accessed: 9 December 2010.

Thomas, JA & Pederson, JE. 2003. Reforming elementary science teacher preparation: what about extant teaching beliefs? *School Science and Mathematics* 103(7):319–330.

Thomas, KM, Willis, LA & Davis, J. 2007. Mentoring minority graduate students: issues and strategies for institutions, faculty and students. *Equal Opportunities International* 26(3):179–192.

Thomas International. (Sa). Available at: http://www.thomasint.cn/Theory.asp and http://www.thomasint.cn/ent02a.asp. Date accessed: 3 September 2004; 11 June 2005.

Thomas International. 2005. *DISC profiling instruments*. Resources provided by Thomas International Registered Analyst.

Thomson, R & Mabey, C. 2001. *Developing human resources*. Oxford: Butterworth-Heinemann.

Thompson, AA & Strickland, AJ. 2003. *Strategic management: concept and cases*. 13th edition. New York: McGraw-Hill.

Trinka, J. 2009. What's a manager to do? *GovLeaders.Org.* Available at: <u>http://www.govleaders.org/whats_a_manager_to_do.htm</u>. Date accessed: 9 March 2009.

Tu, Y, Shih, M & Tsai, C. 2008. Eighth graders' web searching strategies and outcomes: The role of task types, web experiences and epistemological beliefs. *Computers and Education* 51(2008):1142-1153.

Turk, W. 2011. Coach your team. Defense Acquisition, Technology & Logistics(AT&L) Magazine, January–February:39–41. A publication of the DefenceAcquisitionUniversity, Belvoir, Virginia. Available at:http://www.dau.mil/pubscats/ATL%20Docs/Jan-Feb11/Turk jan-feb.pdf.Dateaccessed: 12 January 2011.

Twigg, CA. 2004. A little knowledge is a dangerous thing. *The Learning MarketSpace*. Available at: <u>http://www.thencat.org/Newsletters/Jul04.htm#1</u>. Date accessed: 12 January 2011.

United Kingdom National College for School Leadership (UKNCSL). 2006. Jargon Buster. *E-learning Facilitation Toolkit.* Available at: <u>http://www.ncsl.org.uk/mediastore/image2/facilitatorstoolkit/jargon_buster.htm</u>. Date accessed: 23 August 2006.

United Nations (UN). 2010. General Assembly of the United Nations: Speakers in second committee, discussing Information and Communications Technology for development, stress importance of universal access to broadband connectivity. Department of Public Information, News and Media Division, New York. Available at:

http://www.un.org/News/Press/docs/2010/gaef3290.doc.htm. Date accessed: 4 January 2011.

United Nations (UN). 2006. *A background. Economic Commission for Africa (ECA), Science and Technology Network of the United Nations (ESTNET)*. Available at: <u>http://www.uneca.org/estnet/about/aboutus.htm</u>. Date accessed: 4 January 2011.

United Nations Development Programme (UNDP). 2006. Incentive systems: incentives, motivation and development performance. A UNDP capacity development resource. Paper presented at the Learning Network on Capacity Development (LENCD) International Forum, 3–5 October, Nairobi, Kenya. Available at: <u>http://lencd.com/data/docs/233-Concept%20Note_Incentive%20Systems.pdf</u>. Date accessed: 8 January 2011.

United Nations Economic Commission for Europe (UNECE). 2010. *Draft policy brief on the topic "Age-friendly employment: Policies and practices"*. Working Group on Ageing. Available at: <u>http://www.unece.org/pau/_docs/age/2010/3.a_Age-friendly%20employment%20policies%20and%20practices.pdf</u>. Date accessed: 10 January 2011.

United Nations Economic and Social Commission for Asia and the Pacific (UNESCAP). 2009. Technical paper: information and communications technology and disaster risk reduction division. *Competency-based Training Guidelines for ICT Initiatives at the Community Level.* Available at: http://www.unescap.org/idd/papers.asp. Date accessed: 16 October 2010.

United Nations Educational, Scientific and Cultural Organization (UNESCO). 2008. ICT competency standards for teachers. Policy framework. Available at: <u>http://cst.unesco-ci.org/sites/projects/cst/The%20Standards/ICT-CST-Policy%20Framework.pdf</u>. Date accessed: 13 March 2011.

United States Agency of International Development (USAID). 2005. *Africa education initiative: South Africa case study*. Monitoring and Evaluation for the Africa Bureau

Education Division. Available at: <u>http://pdf.usaid.gov/pdf_docs/PNADG232.pdf</u>. Date accessed: 6 January 2011.

University of Calgary. (Sa). Blended approach to teaching and learning: a position paper. Available at: <u>http://commons.ucalgary.ca/documents/Blended%20Learning%20Position%20Paper</u> .pdf. Date accessed: 7 January 2011.

University of Minnesota, Office of Human Resources. (Sa). Available at: http://www1.umn.edu/ohr/toolkit/development/index.html. Date accessed: 9 March 2009.

Van Aswegen, K. 2010. Subject specialist: Computer Science and Information Systems, North-West University (NWU), Potchefstroom Campus. Personal Interview, 4 November, Potchefstroom.

Van der Riet, M & Durrheim, K. 2009. Putting design into practice: writing and evaluating research proposals, in *Research in practice: applied methods for the social sciences,* edited by Terre Blanche, M, Durrheim, K & Painter, D. Cape Town: University of Cape Town Press.

Van der Watt, MA. 2008. The integration of critical reflection as a learning strategy in the training of health science practitioners. PhD thesis. Pretoria, University of Pretoria. Available at: <u>http://upetd.up.ac.za/thesis/available/etd-10222008-115359/</u>. Date accessed: 14 September 2009.

Van Ryneveld, L. 2005. Surviving the game: interaction in an adult online learning community. PhD thesis. Pretoria, University of Pretoria. Available at: http://upetd.up.ac.za/thesis/available/etd-03082005-220804. Date accessed: 16 February 2007.

Vansteenkiste, M, Neyrinck, B, Niemiec, P, Soenens, B, De Witte, H & Van den Broeck, A. 2007. On the relations among work value orientations, psychological

need satisfaction and job outcomes: a self-determination theory approach. *Journal of Occupational and Organizational Psychology* 80:251–277.

Vathanopas, V & Thai-ngam, J. 2007. Competency requirements for effective job performance in the Thai public sector. *Contemporary Management Research* 3(1, March):45–70.

Verbruggen, M, Sels, L & Forrier, A. 2007. Organizational versus external career support: unravelling the relationship between organizational career management and the need for external career counselling. Report for the *Steunpunt Werk en Sociale Economie* of the Research Centre of Organisation Studies, Katholieke Universiteit Leuven, Belgium. Available at: <u>www.steunpuntwse.be/download/nl/1994199/pdf</u>. Date accessed: 12 January 2011.

Vermeulen, L. 2010. Effective career management of your employees to ensure the correct and proper career path development. Paper presented at the Global Prospectus Talent Management Conference, Midrand, Johannesburg, 3–4 August.

Vermeulen, L. 2008. A quest for the best: talent management as strategic priority. Conference proceedings *Journal of Public Administration* November:409–418.

Vermeulen, L. 2007. Talent management as strategic tool in retaining human capital. Conference proceedings. *Journal of Public Administration* November:272–284.

Vermeulen, L. 2005. The application and perceived usefulness of WebCT tools in the training of future public service officials. Paper presented at the annual conference of the International Association of Schools and Institutes of Administration (IASIA), July, Como, Italy.

Vermeulen, L & Johannes, H. 2006. Bridging the e-readiness chasm: elusive human factors. Conference proceedings. *Journal of Public Administration* September:361–376.

Vroom, VH. 1964. Work and motivation. New York: Wiley.

Warlick, D. 2010. *Are they students or are they learners?* Teaching and learning in the new information landscape. Blog, 8 October. Available at: http://davidwarlick.com/2cents/?p=2762. Date accessed: 3 January 2011.

Waterhouse, S. 2008. How to attract and hire the best ... every time. *Eyes on sales: people with a passion for selling*. Available at: <u>http://www.eyesonsales.com/content/article/how_to_attract_and_hire_the_bestevery_time/</u>. Date accessed: 16 October 2010.

Webster, M. 2011. Merriam Webster online dictionary. Available at: <u>http://www.merriam-webster.com/dictionary/technology</u>. Date accessed: 7 May 2011.

Wegge, J & Haslam, SA. 2005. Improving work motivating and performance in brainstorming groups: the effects of three group goal-setting strategies. *European Journal of Work and Organizational Psychology* 14(4):400–430.

Weiner, IB, Graham, JR, Schinka, JA & Naglieri, JA. 2003. *Handbook of psychology: assessment psychology*. New Jersey: John Wiley & Sons.

Weldon, C. 2010. Characteristics of the adult learner. Available at: <u>http://www.suite101.com/content/characteristics-of-the-adult-learner-a210116</u>. Date accessed: 30 October 2010.

Wellins, RS, Smith, AB & Erker, S. 2009. *Nine best practices for effective talent management*. Development Dimensions International (DDI). Available at: www.ddiworld.com/.../ddi/ininebestpracticetalentmanagement/wp.pdf. Date accessed 30 July 2010.

Wenschlag, RE. 2006. The role of goal setting and performance feedback in achieving peak performance. *Vital learning for business and life*. Available at: <u>www.vital-learning.com/documents/AffiliateLeadership.pdf</u>. Date accessed: 18 September 2007.
Westburn dictionary of markering. 2002. Semi-structured questionnaire. Available at: http://www.westburnpublishers.com/marketing-dictionary/s/semi-structured-questionnaire.aspx. Date accessed: 6 December 2010.

Westerman, J. 2007. When motivating generation Y in the classroom. Essays on teaching excellence. Available at: <u>http://teaching.uchicago.edu/ete/06-07/Westerman.html</u>. Date accessed: 14 October 2010.

Witt Communications. (Sa). Available at: <u>http://www.wittcom.com/what_is_disc.htm</u>. Date of access: 24 April 2005; 14 July 2005.

Whitmore, P. 2000. The maieutics of goal articulation: Motivating the choices of highest import. PhD thesis, Stanford University, USA.

Wolf, PD. 2006. Best practices in the training of faculty to teach online. *Journal of Computing in Higher Education* 17(2):47–78.

Wolfe, RL. 1996. *Crisp: systematic succession planning: building leadership from within*. United States of America: Crisp Publications.

Wong, A & Fitzsimmons, J. 2008. Factors affecting professor facilitator and course evaluation in an online graduate program. *Proceedings ascilite*, Melbourne.

World Bank. 2003. Lifelong learning in the global knowledge economy: challengesfordevelopingcountries.Availableat:http://siteresources.worldbank.org/INTLL/Resources/Lifelong-Learning-in-the-Global-Knowledge-Economy/lifelonglearningGKE.pdf.Date accessed: 31 October 2010.

Worthen, CB & Jefferson, JF. 2009. The non-traditional student: myths and misconceptions – what really works. Paper presented at the 29th conference of the Adult Higher Education Alliance, held in Chicago, Illinois, 6–9 October.

Wright, PM, Dunford, BB & Snell, SA. 2001. Human resources and the resource based view of the firm. *Journal of Management* 27:701–721.

Xu, Y & Tuttle, BM. 2004. Understanding work styles. *Business and Economic Review* October–December:22–24.

Yates, BL. 2004. Applying diffusion theory: adoption of media literacy programs in schools. *Studies in Media and Information Literacy Education* 4(2):1–12.

Yates, BL. 2001. Applying diffusion theory. Paper presented at the Instructional and Developmental Communication Division International Communication Association Conference, Washington, DC, USA, May 24–28.

Zemsky, R & Massy, WF. 2004. *Thwarted innovation: what happened to e-learning and why*. Available at: http://www.irhe.upenn.edu/Docs/Jun2004/ThwartedInnovation.pdf. Date accessed: 14 October 2005.

Zhu, Y. 2007. What drives differences in reward allocation principles across countries and organizations? *Academy of Management Perspectives* August:90–92.

Zuckerman, M, Porac, J, Lathin, D, Smith, R, & Deci, EL. 1978. On the importance of self-determination for intrinsically motivated behavior. *Personality and Social Psychology Bulletin* 4:443–446.

Dear Prof XXXXX

I am currently busy with my doctoral studies. The title of my thesis is: *Key human factors in the e-readiness of academic employees at higher education institutions.*

I am hereby kindly requesting your permission to distribute my questionnaire amongst employees in your School. I am interested in feedback from both employees that make use of eFundi, as well as those who do not make use of eFundi. Input from both these groups will be valuable to my study.

It will be highly appreciated if you can grant my request.

Kind regards Luni Vermeulen



Dear Colleague

I am busy with my D Litt et Phil Degree, titled: *Key human factors in the e-readiness of academic employees at higher education institutions.* I am currently in the data collection phase.

Prof XXXXX granted me permission to distribute my questionnaire in your School. If you are willing to participate, it will be highly appreciated if you can complete the attached questionnaire. I am interested in feedback from both employees that make use of eFundi, as well as those who do not make use of eFundi. Input from both these groups will be valuable to my study.

If you prefer to complete a hard copy of the questionnaire, please let me know and I will gladly forward it to you. You can return the questionnaire via e-mail to <u>Luni.Vermeulen@nwu.ac.za</u> or send it via internal mail to box 322, before 21 October 2009.

Kindly note that completion of the questionnaire is anonymous and voluntary and that your contribution to this discussion will be treated as part of collective feedback; you will therefore not be identified as an individual.

Kind regards Luni Vermeulen

ANNEXURE C





QUESTIONNAIRE FOR DOCTORAL STUDY: LUNI VERMEULEN

Dear Participant

The questionnaire forms part of the data collection for a doctoral thesis titled *Key human factors in the e-readiness of academic employees at higher education institutions*. The North-West University (NWU), Potchefstroom campus, serves as case study for the research. As an academic employee at the NWU your input is valuable and significant to the empirical research component of the study. Your answers and comments will be highly appreciated.

Generally e-learning refers to learning that can be accessed by any electronic means (Alessi & Trollip, 2001:377). For the purpose of this study, the e-learning focus is particularly on web-based teaching and learning. Whereas e-learning makes use of a variety of ICT tools, including the internet, web-based teaching and learning refers to learning specifically through the use of the internet/web-based applications (*White Paper on e-Education*, 2004:16).

The questionnaire comprises of two parts: **Section A**, which deals with questions pertaining to biographical information; and **Sections B to E**, which deals with questions pertaining to work style profiles, motivation, e-learning and employee training and development respectively. Sections B to E consist of both structured and open-ended questions.

Instructions:

• At the *structured questions:* kindly <u>choose only one option</u>, unless otherwise indicated

At the *open-ended questions*: you are requested to give your personal opinion/experience/perception as academic employee.
 <u>Please note that the space provided for your answer () is unlimited. Therefore you do not need to limit your answer</u>.

It will take approximately 15-20 minutes to complete the questionnaire.

Thanking you in advance

Luni Vermeulen

Lecturer: Public Management and Governance

School for Social and Development Studies

Luni.Vermeulen@nwu.ac.za / 082 373 1589

SECTION A: BIOGRAPHICAL INFORMATION

Designation:

School & Faculty:

Years of teaching experience:

Highest qualification:

Age:

SECTION B: WORK STYLE PREFERENCES

Please answer all questions by *ticking* only <u>one option</u>, unless indicated otherwise.

1. When I am confronted with a new technology/innovation in the work environment, I will:

- A 🗌 Immediately embrace and welcome the new technology/innovation
- B Adapt to the new technology/innovation relatively fast
- C Immediately see the risks involved, but will eventually see advantages of the technology/innovation over a period of time when the usability and success of the technology/innovation is proven.
- D Dislike the disruptions of new technology/innovations and will be very slow to adapt
- E Never adapt to a new technology/innovation

2. My preferred approach and attitude towards learning a new skill such as teaching online will be:

- A 🗌 I adapt easily and prefer to deal with new challenges and experiences
- B I want to understand the practical value and use of what I am being taught first
- C 🗌 I want to see good structure and a logical step-by-step approach
- D I want to have ample time to think intensively about the activities and concepts provided to me

3. The following statement describes me best:

A 🗌 I am concerned with results, competitive, focused on achieving goals, like to solve problems, and love challenges

B I like people and want to be liked in return, am optimistic and outgoing

C 🗌 I am sympathetic, friendly, a good listener, a great team player, work hard and create a stable environment

D I have high standards, am a perfectionist, and prefer systems, processes and procedures and consistency

SECTION C: MOTIVATION

4. When new challenging goals are assigned to me, I:

- A 🗌 resist it, because I already carry a heavy workload
- B 🗌 resist it because it is given in brief abrupt instructions
- C 🗌 resist it because a rationale is not provided
- D 🗌 do it because it is expected of me
- E _____ embrace the opportunity as it motivates me to develop my skills and demonstrate competence

5. I prefer to receive:

- A Difficult and specific goals
- B Difficult and general goals
- C
 Easy and specific goals
- D
 Easy and general goals

6. I am best motivated by (*more than one option can be chosen at this question*):

A A financial incentive

- B
 Promotion
- C 🗌 Recognition from my School Director/Dean
- D 🗌 Flexi time or the opportunity to work from home
- E Support with resources (eg. provision of a laptop; cell phone)

If you have chosen more than one option, kindly indicate which one is most important to you:

SECTION D: USE OF AND PERCEPTIONS ON e-LEARNING PLATFORM

7. I use eFundi to the following extent for my modules:

- A All my modules (undergraduate, postgraduate, full time and distance learning)
- B All undergraduate students
- C 🗌 All postgraduate students
- D All distance learning students
- E Some of my modules

Please specify (eg, two undergraduate, one postgraduate):

F 🗌 None of my modules

Please provide a reason:

- 8. I use the following tools of eFundi for my modules (*more than one tool can be chosen at this question*):
- A Announcements
- C
 Schedule
- D Assignments
- E 🗌 Gradebook
- F 🗌 Forums
- G 🗌 Messages
- H Dodcasting
- I 🗌 Wikis
- J e-Testing & Quizzes
- K 🗌 Bloggers
- 9. The following statement describes my level of comfortableness with teaching online best:
- A I am not comfortable with teaching online and therefore do not use the e-learning platform
- B I am not comfortable with teaching online, but use the e-learning platform because it is expected of me by my School Director/Dean
- C I was initially not comfortable with teaching online, but became more comfortable in time

- D 🗌 I am comfortable with teaching online as long as I can stick to the basics (providing information and resources; assessing assignments; posting details and due dates on the schedule)
- E 🗌 I am comfortable with teaching online and am using more and more of the online tools
- F I am completely comfortable with teaching online (and use tools such as podcasting and bloggers)
- 10. I do not use the e-learning platform (or initially did not use it) because (*more than one option can be chosen at this question*):
- A 🗌 It is out of my comfort zone
- B I am not sufficiently computer literate
- C 🗌 I did not receive training
- D 🗌 It is an additional workload
- E I prefer "talk-and chalk"
- F 🗌 I am not interested in technology
- 11. I am ready and willing to include cutting edge technology such as podcasting and online games in my modules on eFundi:
- A Definitely
- B 🗌 On a limited basis
- C 🗌 Maybe
- D Only if it is expected of me

F 🗌 No	
G 🗌 I already include it	t

12. I use the following social network/s (*more than one option can be chosen at this question*):

- A Facebook B Twitter C Badoo D Yedda E MXit
- F
 - Please specify:
- G 🗌 None

<u>Please note that the space provided for your answers (</u>) is unlimited. Therefore you do not need to limit your <u>answers</u>.

- 13. Do you *currently use* your module/s on eFundi interactively? How?
- 14. Do you plan to use your module/s on eFundi interactively? How?

15. What is your general perception of e-learning?

SECTION E: TRAINING AND DEVELOPMENT

- 16. I view training for the use of an e-learning platform as online learning facilitator as:
- A 🗌 Imperative
- B Important, but not necessary for myself
- C
 Not necessary, these days everybody knows how to find their way online
- 17. The following should be included in training for the use of an e-learning platform (*more than one option can be chosen at this question*):
- A Paper-based notes/user manual
- B An instructor that gives verbal instructions
- C An instructor which demonstrates
- D Visual aids (eg. video's; simulations)

If you have chosen more than one option, kindly indicate which one you perceive to be most important:

18. When being taught how to teach online for the first time, I:

- A 🗌 loved it and could not wait to begin teaching online
- B looked forward to start teaching online, once I have mastered the new skill
- C [] felt more comfortable to teach online once I have obtained the skill, although e-learning will never be my first choice
- D Still did not want to teach online

19. I want the following to be included in e-learning training (more than one option can be chosen at this question):

- A How to use the tools of the e-platform (eg. posting of messages and information, assessment of assignments)
- B How to customise a module site (add photos, images, banners; change colours and fonts; add voice clips, podcasts, simulations)
- C 🗌 How to facilitate discussions to ensure optimal participation
- 20. Do you think training on becoming an online learning facilitator should be included in your personal career development plan?
- 21. What type of assistance do you expect from your School Director/Dean to support you for online teaching and learning?

22. Do you think e-readiness should be assessed during a performance appraisal and subsequently be included in the development plan of an academic employee that is not e-ready.?Please motivate your answer.

Thank you for your time and your participation in this study!

FOCUS GROUP DISCUSSION

Researcher: Luni Vermeulen

Research: D Litt et Phill degree in Public Administration (UNISA)

Place: North-West University, Potchefstroom Campus, Building F13, Room 106 Time: 10:30 – 12:30 Date: 10 December 2010

I hereby confirm that my participation in this focus group is voluntary and that I may quit participating in the discussion at any time. I give my consent to participate in this focus group discussion, provided that my contribution to this discussion will be treated as part of collective feedback and that I will not be identified as an individual.

Signature: Participant

Signature: Researcher

FOCUS GROUP DISCUSSION

10 December 2010 North-West University (Potchefstroom campus)

RESEARCHER: LUNI VERMEULEN

D Litt et Phil degree in Public Administration (UNISA)

CONSENT

- Voluntary
- Can quit participation at any time
- Confidential no individual will be identified



INTRODUCTION

• Title of thesis:

• KEY HUMAN FACTORS IN THE E-READINESS OF ACADEMIC EMPLOYEES AT HIGHER EDUCATION INSTITUTIONS

- Performance appraisal
- Employee Development; Career Development
- Key human factors:
 - Personal profile patterns (DISC factors Thomas International)
 - Learning style
 - \cdot Pace and style of technology adoption
- Motivators



E-LEARNING

 e-Learning refers to learning that can be accessed by any electronic means

(Alessi & Trollip, 2001:377)

 e-Learning involves the use of a computer or electronic device (e.g. a mobile phone) in some way (to a bigger or lesser extent) to teach or provide learning material

(Stockley, 2003)



E-READINESS

 E-readiness implies that the circumstances or features of being ready for e-learning are in place

(Rautenbach, 2007:iv)

 E-readiness is not limited to <u>physical readiness</u> (infrastructure, hardware, software), but also consists of the <u>non-physical</u> <u>readiness</u> of an organisation such as *inter alia* psychological readiness

(Rautenbach, 2007:iv)

- · Comfortability
- Willingness
- Knowledge and adaptability of teaching and learning strategies

E-READINESS ASSESSMENT

- > Assessment tools were primarily used to assess e-readiness of:
 - countries (especially developing countries),
 - governments,
 - companies (e.g. banking; property)

• to a limited extent HEIs

 Current e-readiness assessment tools -facilities, connectivity, accessibility, hardware, software, the digital divide, technological status, policies and regulations (Machado, 2007:73-75)

• No e-readiness assessment tool for academic employees



CONTEXT

- The technology revolution of the 21st century presents <u>HEIs</u> with one of the biggest adaptation <u>challenges</u> in its history.
- HEIs should either adjust to this innovative reality or they will be in danger of losing its standing as a principal educational institution

(Amirault & Visser, 2009: 62)

GENERATION Y

- Born between 1980-1990; SA from 1990
- fun and relaxed
- > absorbed with technology
- a traditional approach does not appeal
- stimulus junkies" who gets bored easily
- skilful at multi-tasking, think at a high speed and are fervently broad-minded in terms of diversity
- an astonishingly innovative generation
- will seek reinforcement and constant feedback on a regular basis (Business Report, 2005:1)
- high level of technical literacy
- some of the most used technologies of Gen Y live virtual classrooms, podcasts, blogs, social networks and collaborative editing

(Halse & Mallinson, 2008:1)

GENERATION Y (Cont.)

HEMIS

Age	2008 Enrolment		2009 Enrolment	
	Total	%	Total	%
18	82 742	10.3	93 614	11.2
19	76 665	9.6	80 935	9.7
20	78 780	9.8	80 424	9.6
21	72 855	9.1	72 480	8.6
22	60 339	7.5	60 097	7.5
Total:	799 490	46.3	837 779	46.6



GENERATION Y (Cont.)

NWU

Year of birth	2010 Undergraduate contact learners (Potchefstroom)		2010 Undergraduate contact learners (All campuses)	
	Total	%	Total	%
1992	89	1.2	555	4.4
1991	2 774	37.8	4 033	32.2
1990	876	11.9	1 668	13.3
1989	298	4.1	730	5.8
1988	605	8.2	919	7.3
1987	383	5.2	623	5
Total	5 025	68.4	8 528	68
Total undergraduate registrations	7 342			





DISCUSSION QUESTIONS

- 1. How do you perceive the <u>current role of e-learning</u> in the higher education environment?
- 2. How do you perceive the <u>future role of e-learning</u> in the higher education environment?
- 3. <u>What should be done</u> to ensure that e-learning is able to fulfil this role?
- 4. What are your perceptions pertaining to the <u>expectations of</u> <u>learners</u> in terms of e-learning?
- 5. How should the <u>role of academic employees</u> change to accommodate learners' expectations?





FOCUS GROUP DISCUSSION

Researcher: Luni Vermeulen Research: D Litt et Phill degree in Public Administration (UNISA)

Place: North-West University, Potchefstroom Campus, Building F13, Room 106 Time: 10:30 – 12:30 Date: 10 December 2010

Discussion questions

- 1. How do you perceive the current role of e-learning in the higher education environment?
- 2. How do you perceive the future role of e-learning in the higher education environment?
- 3. What should be done to ensure that e-learning is able to fulfil this role?
- 4. What are your perceptions pertaining to the expectations of students in terms of e-learning?
- 5. How should the role of academic employees change to accommodate students' expectations?

HR/P 03



Personal development plan for academic staff

Title, initials and surname:

University number:

Faculty/school/section:

1 Which of the following fields of competence require further development in terms of the strategies and priorities of the division, thus making it essential for the position?

Areas of comp	etence	According to judgment of staff member	According to judgment of manager
Teaching-learning			
Research and post graduate training			
Organisational and management tasks			
Academic and professional qualificati	ons and registrations		
Contribution to foundation, value syst	em and science practice		
Professional behaviour			

2 Identify four (4) areas for development of skills and do planning for the achievement thereof

	Development priorities	Act	ion plan
	(focus on the four most important development priorities)	Internal/external training opportunity, in-service training and/or formal study	Supply full details
Priority 1:			
Priority 2:			
Priority 3:			
Priority 4:			

Signature of staff member

Signature of manager

Date



NORTH-WEST UNIVERSITY YUNIBESITI YA BOKONE-BOPHIRIMA NOORDWES-UNIVERSITEIT INSTITUTIONAL OFFICE

HR/P 01

PERFORMANCE AGREEMENT FORM FOR ACADEMIC STAFF

THE OBJECTIVE OF THIS PERFORMANCE AGREEMENT FORM IS:

- To use joint planning and dialogue between the dean/director(s) and the academic staff member to reach an agreement about the participation and goals of the staff member regarding his/her teaching, research and other functions, in order to achieve the goals and targets of the faculty, school and research entity.
- To serve as a basis for dialogue to identify objectives for staff development.
- To serve as a basis for self-appraisal by the staff member and for evaluation of work performance of the staff member by the dean/director(s).

GUIDELINES FOR USING THE FORM:

- The form is completed annually by the staff member him/herself, and it is then finalised after dialogue with his/her director(s).
- Only categories relevant to the staff member in question are completed.

Name of staff member	School and subject group/programme	Year

1. TEACHING

Time = %

1.1 Under- and postgraduate (including honours and master's course units, but excluding theses)

Course unit	Section of course unit presented by the staff member	Estimated number of students

- 1.2 Way in which teaching will be implemented and developed (e.g. via development/revision of study guides, study material, teaching methods, utilisation of information technology) and the expected outcomes:
- 1.3 Marketing of expertise with regard to teaching, in order to generate a third stream of income (e.g. further training, short courses):

- 1.4 Particular tasks and related objectives regarding teaching (e.g. subject chair, programme co-ordinator, writing of SAQA unit standards):
- 1.5 Personal development objectives regarding teaching (e.g. obtaining particular knowledge and skills) and methods to achieve them (e.g. attending courses, improving qualifications, undertaking study trips):

2. 2.1 2.1.1(a)	RESEARCH AND POSTGRADUATE TRAINING Participation in research: Research focus area/unit in which you do your research:	Time =	%
2.1.1(b)	If your research currently does not resort under a focus area, which you plan to link it (where applicable).	indicate the focus a	area with

2.1.2 Research objectives for the year of this task agreement:

2.1.3 Main points of research objectives for the following three years:

2.1.4 Planned research outputs upon which are agreed, e.g. articles, books or contributions in books, conference presentations (including participation in workshops and symposiums), research reports or project reports, patents:

2.1.5 Objectives for the year regarding managerial inputs in terms of research (only for members of a managerial structure in a focus area/unit and/or a leader of a sub-programme):

2.2 Supervision of M- and D-students regarding dissertations and theses (kindly attach a detailed form if this space is insufficient):

Name of student	Degree	On which level has the title been approved (i.e., faculty board or senate)?	Capacity, i.e. supervisor, co- supervisor, assistant supervisor, promoter, co- promoter, assistant promoter	Expected date of submission

2.3 Marketing of research expertise (e.g. projects to generate a third stream of income):

2.4	Development objectives as researcher (obtaining knowledge and skills) and methods to
	implement them (courses, improving qualifications, study trips):

2.5	Participation in activities of the academic and research community at large (e.g. subject associations, editorial boards, professional boards):
i.	OTHER IMPORTANT FUNCTIONS/TASKS FOCUSED ON THE STRATEGIC OBJECTIVES OF THE FACULTY/SCHOOL/ Time = % RESEARCH ENTITY
3:	The total agreed estimated time expenditure for all three categories must add up to 100%
natu	ure of staff member

Name and signature of dean* / school director

Name and signature of director of research entity (if applicable)

Date

* The performance agreement of directors regarding their academic task is made with the dean.

Original details: p:\engels\task agreement form for academic staff.doc

ANNEXURE I



NORTH-WEST UNIVERSITY YUNIBESITI YA BOKONE-BOPHIRIMA NOORDWES-UNIVERSITEIT INSTITUTIONAL OFFICE

HR/P 02

SELF-EVALUATION FOR ACADEMIC STAFF

NAME	FACULTY/SCHOOL

Use this form to evaluate your job performance for the past year. Please refer to the performance agreement which you signed with your dean/director(s). After completion you must arrange a meeting with your dean/director(s) in order to complete the personal development plan and plan your performance agreement for the following year.

1. TEACHING

1.1 Self evaluation scale for teaching

You evaluate your own teaching on the 7-point scale.

If your performance in a certain aspect is excellent/superior/magnificent/100%, you will award yourself 7. If you think that you did not perform at all in a certain aspect, you will award yourself 1. The ratings between 1 and 7 on the scale reflect positions between these extremes. Mark the appropriate block with an X.

		1	2	3	4	5	6	7
1.	. How students experience my teaching according to the							
	teaching-evaluation scale							1
2.	Quality and availability of study guides							
3.	Appropriateness of my teaching methods							
4.	Appropriateness of my prescribed study material							
5.	My contribution to course development							
6.	The extent to which I stimulate independent study							
7.	Quality and frequency of evaluation (tests, examinations and							
	assignments)							
8.	Effective use of appropriate teaching media							

			1	2	3	4	5	6	7
9.	The	extent to which I integrate the following education principles							
	during my teaching (award a rating for each)								
	\diamond	Purposefulness							
	\diamond	Planning							
	\diamond	Motivation							
	\diamond	Individualising							
	\diamond	Independent working by students							
	\diamond	Experience/concretising							
	\diamond	Mastering							
	\diamond	Communication							
	\diamond	Evaluation							

1.2 Briefly state what you consider to be your strengths in undergraduate teaching during the past year.



1.3 Write down what you consider to be your weaknesses in undergraduate teaching for the past year.

·····

1.4 Read the performance agreement you signed a year ago with your director/dean. Indicate to which extent you were successful in the undergraduate teaching you agreed on. (Use a separate sheet of paper if necessary).

.....

1.5 Write down the obstacles/limitations you experienced in the execution of your teaching duties as agreed upon.

2. RESEARCH AND POSTGRADUATE TEACHING

2.1 Read the performance agreement for the past year. To which extent do you think you were successful?

2.2 Which obstacles and/or limitations did you experience in this aspect of your task?

······

2.3 SUPERVISION OF M- AND D-STUDENTS

Read 2.2 of the performance agreement for the past year. Write down the number of students who completed their M- and D-studies under your supervision in the past year, as well as the capacity in which you were involved (supervisor, co-supervisor, assistant supervisor, promoter, co-promoter, assistant promoter).

2.4 RESEARCH

2.4.1 Write down your research outputs in the past year for the performance areas below:

	NUMBER
research articles which are recognised for subsidy	
other research articles	
	NUMBER
books	
contributions in books	

List of research project topics and materials

	approved research reports and/or project reports				
	patents				
	presentations about research and development work				
	research grants for research and development work: institutions and amounts				
	any other research performance you would like to mention.				
2.4.2	If you did not perform in research as agreed upon, briefly supply reasons.				
2.4.3	Which obstacles/limitations (if any) did you experience in the execution of your research task?				
отн	ER IMPORTANT FUNCTIONS/TASKS FOCUSED ON THE STRATEGIC GOALS OF				
THE	FACULTY/SCHOOL/RESEARCH ENTITY				
3.1	Write down how you reached your goals that were planned in the performance agreement.				
	~ 				

.....

3.

421

	3.2	If you did not perform as agreed upon, briefly supply reasons.			
	3.3	Which obstacles/limitations did you experience regarding this aspect?			
4.	SUB	JECT-SPECIFIC, PROFESSIONAL AND PERSONAL DEVELOPMENT			
	4.1	Briefly describe your main activities and results thereof in terms of knowledge/skills with			
		regard to subject-specific, professional and personal development during the past year?			
	4.2				
		Write down any obstacles/limitations that you experienced			

.....

SIGNATURE STAFF MEMBER

.....

DATE

.....

 $\textbf{Original details:} p:\engels\hr-p 02 self-evaluation form for academic staff.doc 2007-09-05$



NORTH-WEST UNIVERSITY YUNIBESITI YA BOKONE-BOPHIRIMA NOORDWES-UNIVERSITEIT INSTITUTIONAL OFFICE

Academic Development and Support

FEEDBACK FORM - MINI-CONTACT SESSION

Criteria for the evaluation of a mini-contact session presented by a newlyappointed colleague during CNL

Instructions

- 1. Please use this form to give feedback to your colleague on his/her presentation of a mini-contact session.
- 2. Complete the feedback form directly after the mini-contact session and give it to the colleague you assessed

Name of presenter of mini-contact session:

.....

Use the following scale to assess the various aspects of facilitation

1	2	3	
Not competent	Competent	Excellent	
		("more than competent")	

You are requested to use the *"column for commment"* for more particular feedback to your colleague.
	ASPECT OF FACILITATION	GRADE	COMMENT					
	ORGANISATION OF THE CONTACT SESSION							
1.	Have the outcomes for the mini- contact session been stated clearly?							
2.	Have the outcomes been attained by the end of the session?							
3.	Has the session run its course within the given 15 minutes?							
	FACILITA	TION AND AS	SESSMENT SKILLS					
4.	Ability to involve learners actively during the contact session (demonstration of group work techniques)							
5.	Ability to integrate formative assessment during the contact session							
	DEMON	ISTRATION OF	MEDIA USAGE					
6.	Applicability of media selection							
7.	Skills in using media							
то	TAL	/ 21						
	General Comments							
Str	Strong point							
Are	Area to be developed							

Original information: BASAEB p:\dokument nuut\8. onderrigleer\knd\mini contact feedback.doc 2007-10-22



Institutional Teaching Excellence Award (ITEA) Guidelines for panel evaluation of contact sessions

ITEA: Assessment of teaching and learning during the contact situation with students

The NWU is aware that teaching and learning differ between small and large class groups, as well as between faculties and different subject disciplines. These differences should as far as possible be taken into account during the assessment of submissions. It is proposed that subject experts serve on the assessment panel, so that the uniqueness of each subject may be adequately dealt with.

It is also proposed that teaching and learning in the class situation and other teaching opportunities should be evaluated according to the guidelines below. For each category the bottom and top ends of success are indicated to serve as a guideline for the awarding of a score on a five-point scale.

1. Course of teaching

1.1 Introduction

- The material was not linked to students' prior knowledge; there was no problem description / introduction to the subject; no clear objective and learning outcomes were identified. The lecturer plunged into the new material.
- There was efficient linking with previous contact periods by means of the testing of prior knowledge; students were motivated and interest was stimulated; the problem description / introduction was clearly formulated; the objective and learning outcomes of the contact session were logically explained and well formulated.

1.2 Development

- There was no logical structure / pattern; less relevant contents were not distinguished; core concepts and essences were not provided; no familiar examples were used to which new contents could be linked. Learning activities were not referred to at all in the study guide.
- During the presentation of new learning content clear structure and relationships were identified; core concepts were repeatedly highlighted and essences were emphasised; perspective was provided; sufficient and very appropriate examples (pegs) were used to establish links with new contents. There was continuous feedback regarding learning activities in the study guide, as well as additional applications. Students see the specialist in action.

1.3 Conclusion

- No summary and application of the discussed subject content were provided; learning gains were not assessed; no follow-up activities were given. Contents were left hanging in the air.
- Discussed contents were summarised and applied; achievement of identified learning outcomes (learning gains) was assessed; thoroughly planned follow-up activities (assignments) were given.

2. Integration of education principles

2.1 Purposefulness

- Vague; nobody knows what the lecturer is trying to achieve.
- Clear learning outcomes were identified and all learning activities are directed at achieving the learning outcomes.

2.2 Planning

- Presentation was incoherent and disordered.
- The entire presentation reflected thorough planning the content was presented in a logically ordered manner. Learning activities during the contact session logically linked up with information as explained in the study guide.

2.3 Level of presentation

- Development was slow, level of content low and simple, outcomes incorrectly pitched and students were uninvolved and disinterested.
- An outstandingly high level of content presentation with the outcomes correctly pitched. Students literally had no choice but to be interested and to pay attention.

2.4 Individualisation

- The class was treated as a homogenous entity and students were not involved as individuals.
- The lecturer also worked with and put questions to individuals. Recognition was clearly given to different learning styles and abilities. The lecturer knows the students in smaller classes by name.

2.5 Learner self-activity

- Students were totally passive. There is no guarantee that learning took place.
- The lecturer constantly involved students. Ample opportunity was created to experience theoretical knowledge in practical form.

2.6 Comprehensive view

- Only the essential information was provided, without making any connections whatsoever.
- The lecturer made a real attempt to explain the relevance of every piece of information in the context of the more comprehensive structure.

2.7 Experience

- The content was restricted to the theoretical level. Facts, descriptions and information were simply communicated verbally.
- The content was explained by means of examples, demonstrations, illustrations etc. The students experienced the content.

2.8 Mastering

- The lecturer does not accept that mastering of the subject material is his responsibility as well. He/she took no steps to ensure that students had mastered the work.
- The lecturer went to considerable trouble to ensure that mastering took place, by asking questions, incorporating self-assessment opportunities, summarising, revising, etc.

2.9 Socialisation and group work

- There was a rigid, formal atmosphere in which no or little interpersonal reactions took place.
- The lecturer created much opportunity for interpersonal reaction and exchange of ideas in groups. Two-way communication was encouraged.

2.10 Assessment

- No assessment was done. It was simply taken for granted that the students learned something.
- The lecturer constantly assessed the mastering of the identified learning outcomes. After the contact session summative assessment was done.

3. Appropriateness and variety of teaching-learning methods

- The lecturer is lecture-oriented. The teaching-learning methods used are in general not appropriate for achieving the learning outcomes.
- The lecturer used a variety of strategies, such as group work, video teaching, case studies, lecturing, discussion, questions-and-answers, visual representation etc. The methods used by the lecturer promoted learning and the mastering of learning outcomes.

4. Preparation and use of teaching media

- A logical scheme did not unfold on the writing board. Transparencies are mostly handwritten. Information could just as well have been written on the board.
- The lecturer used the most appropriate teaching medium for the presentation and discussion of learning contents. Transparencies and other media contributed to the promotion of learning, the achieving of outcomes and the clear transmission of the message. During the presentation and discussion a logical scheme unfolded on the writing board.

5. Teaching communication

5.1 Verbal communication

- There were defects in the lecturer's speech, such as talking too loudly, too softly and/or too fast or swallowing words. Strange terms and expressions were used to the point of irritation, such as "good", "right", "okay" etc.
- The lecturer could be heard clearly. Language use was exquisite and marked by good expression.

5.2 Non-verbal communication

- Certain irritating habits which distract students' attention were displayed. No contact with students could be observed.
- Appropriate facial expressions and body movements were used. The lecturer made eye contact and achieved constant contact with students.

Approved: IM - 26 April 2006

Approved by Institutional Senate on 10 May 2006

Huidige gegewens: HOSTNAME I:\users\basascvr\dokumente\annatjie\masters-meesters\itou-itea\itea appendix g - assessment of contact session.doc 2011-10-05

ITEA: Assessment of contact opportunity at NWU

CANDIDATE:		EVALUATOR:	Date:							
FACULTY:	L	SUBJECT GROUP/DEPARTMENT								
			1	SCOP	RE					
CATEGORY	FACTORS	Poor	Below average	Average	Good	Excellent	Total			
		1	2	3	4	5				
	Introduction									
Course of teaching-learning	Development									
	Conclusion									
	Purposefulness									
	Planning									
	Level of presentation									
	Individualisation									
Integration of education	Learner self-activity									
Principles	Experience									
	Mastering									
	• Socialisation and group work									
	Comprehensive view									

	Assessment			
Appropriateness and variety of	Appropriateness			
teaching-learning methods	Variety			
	Planning			
Preparation and use of teaching media	• Applicability and effectiveness			
	Readability			
	Verbal			
Teaching communication	Non-verbal			
TOTAL				

ITEA: Remarks

CANDIDATE:		EVALUATOR:	Date:
	Introduction		
Course of teaching-learning	Development		
	Conclusion		
	Purposefulness		
	Planning		
	Level of presentation		
	Individualisation		
Integration of education	Learner self-activity		
Principles	Experience		
	Mastering		
	Socialisation and group work		
	Comprehensive view		
	Assessment		
Appropriateness and variety of	Appropriateness		
teaching-learning methods	Variety		
	Planning		
Preparation and use of teaching media	Applicability/effectiveness		
	Readability		
Tooching communication	• Verbal		
reaching communication	Non-verbal		



NORTH-WEST UNIVERSITY YUNIBESITI YA BOKONE-BOPHIRIMA NOORDWES-UNIVERSITEIT INSTITUTIONAL OFFICE

ITEA: Guidelines for the ITEA portfolio to be submitted as part of the ITEA evaluation process

1. Introduction

The award for *outstanding teaching and learning* is made to a lecturer with consideration of *the entire spectrum of activities pertaining to teaching and learning*. A portfolio is compiled during the course of the year and submitted for assessment by the end of October. The portfolio must show evidence of sound content and teaching strategies, and provide details on assessment policies and practices.

2. Elements required in the portfolio

The following elements must be included in the portfolio:

2.1 Teaching Philosophy

Each candidate is to submit a statement of no more than two pages on his/her teaching philosophy. The statement should provide information on

- personal innovation in the development of a well designed learning environment and how such innovative ideas have contributed to student learning.
- choice of teaching methods that suit the outcomes, subject matter, the learners' needs and learning styles,
- consistency between outcomes, teaching strategies and forms of assessment,
- enthusiasm for a subject and its impact on learners,
- the use of learning activities and how the success of these are measured,
- how learners are encouraged to study,
- how learners are encouraged to develop critical thinking strategies,
- feedback to learners to enhance their learning,
- teaching and learning approach and the outcomes as well as reasons for failure and success and what changes are made to enhance learning,
- professional development and leadership. (Candidates can outline how he/she keeps their knowledge and understanding of the subject up to date, and describe strategies that are used to reflect on teaching practice and develop new skills and continually improve teaching and learning.),
- effective participation in curriculum design and development.

2.2 Study guides

- Study guides for all modules the candidate presents during the relevant year, should be submitted to the evaluation panel.
- Each study guide must contain the following:
 - o administrative matters;
 - o clear learning outcomes at different levels (amongst others, critical outcomes);
 - study guidance;

- learning activities;
- o assignments to stimulate independent study; and
- self-assessment exercises.

If a candidate did not develop a study guide him/herself, he/she may explain how and why he/she plans certain changes.

2.3 Assessment

- For every module that is presented two of the most recent examination papers, semester tests and class tests must be submitted. In this category the following aspects are assessed:
 - The quality and frequency of tests and assignments.
 - The variation of question types in question papers (questions at different levels of thinking according to Bloom's taxonomy).
 - A written report from the internal/external moderator on examination papers, memorandums and marks allocated. The moderator must be a senior subject expert.
 - The marking scheme or memorandum of each exam paper and test.
 - The relation between assignments and questions in exam papers and tests with the identified learning outcomes.
 - The methods to control independent study and to assess the written work of students.
 - The quality of feedback given to students regarding their work, e.g. tests, assignments and homework.
 - The opportunities for self-assessment.
 - The pass rate that should be an indication of the extent to which students were successful in achieving the identified outcomes.

2.4 How students experience the lecturer's teaching/facilitation

- The questionnaire, section A, with 15 questions, as proposed by the Institutional Office, must be used. (See appendix F)
- Lecturers must be assessed by students in all the modules they present during the course of the relevant year. A member of the faculty's teaching committee, the programme manager/leader or the school director must conduct these assessments.
- Results must be processed and included in the portfolio together with written remarks from the students.
- As students are the clients or target group of the entire teaching and learning programme, an award can only be made if the student feedback indicates that the students also experience the lecturer as outstanding (at least 75% on the teaching feedback scale).

2.5 Peer assessment

- At least 3 contact sessions must be assessed by peers (i.e. the evaluation panel). One of these assessments is conducted by prior arrangement.
- The proposed guidelines will be used as criteria. (See appendix G)
- The extent to which teaching principles are applied during class sessions or other teaching opportunities will be taken into consideration.
- In order to obtain the award, the applicant must be assessed by his/her peers as outstanding, and must obtain a score of at least 75%.
- The teaching advisor from the campus ADS-office will provide the final marks and feedback for this category when the portfolio is assessed.

2.6 For the final assessment, the different categories carry the following weights:

Category	Weight
Student assessment of teaching	20
Peer assessment	20
Study material evaluation	15
Evaluation of assessment (questionnaires, memoranda, etc.)	15
Appropriate choice and use of teaching media	5
Suitable variety of teaching-and-learning strategies	10
Innovative contribution to programme/module development; exceptional contribution to development of a learning environment; proof of attendance of teaching development activities (e.g. attendance of free campus workshops or other relevant activities aimed at teaching development sponsored by the faculty)	15
TOTAL	100

Approved: IM: 26 April 2006 Approved by Institutional Senate on 10 May 2008

Huidige gegewens: HOSTNAME I:\users\basascvr\dokumente\annatjie\masters-meesters\itou-itea\itea appendix c - guidelines portfolio proces.doc 2011-10-05

ANNEXURE N



Score card for evaluation of ITEA teaching portfolio at the NWU (2010)

Name of participant:	
School / Department:	
Faculty:	
Campus:	

Category	Weight		Mark	Comments
Student evaluation of teaching	20	Average achieved on questionnaire – see Appendix F		Comments from evaluation panel will be added at appropriate time
Peer evaluation	20	The average mark awarded by evaluation panel (3 members) will be added at appropriate time		Comments from evaluation panel will be added at appropriate time
Study guides and study material	15	 Module information: outcomes, credits, assessment, study material, module plan Teaching and learning 		

Category	Weight		Mark	Comments
		design: support, exercises, monitoring, self evaluation, exam questions		
		3. Relevance and variation of materials		

Category	Weight		Mark	Comments
Assessment (exam papers, memorandums, marks etc.)	15	 Formative assessment. Assessment techniques. Relation between outcomes, assignments, exam questions. Variation of questions on different levels; Bloom's taxonomy. Allocation of marks. Feedback. Memorandums. Students' performance 		
Suitable choice and use of teaching media	5	 Examples of students work Quality of media Is learning promoted. Effectiveness in helping students to master outcomes . Variation: tranparancies, video material, slides, powerpoint slides, computer programmes, simulations, etc. 		
Variation in effective teaching and learning strategies	10	 Is active learning promoted. Is mastering of outcomes promoted. Is collaborative learning promoted. E-learning 		

Category	Weight		Mark	Comments
Other innovative contributions in the teaching and learning environment	15	 Innovation in designing of the learning environment Design / development of a new/ existing program/module Acquisition of critical and specific outcomes 		
TOTAL	100			

Approved: IM – 26 April 2006 Approved by Institutional Senate on 10 May 2006 Huidige gegewens: PUKGEBRUIKER I:\users\basascvr\dokumente\annatjie\masters-meesters\itou-itea\itea appendix d - score card itea portfolio.doc 2011-10-05

Oorspronklike gegewens: p:\wwdata\verka\verkapaneelevaluering.doc 2002-12-03